

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Eberl, Jakob Korbinian

Research Report The Collateral Framework of the Eurosystem and Its Fiscal Implications

ifo Beiträge zur Wirtschaftsforschung, No. 69

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Eberl, Jakob Korbinian (2016) : The Collateral Framework of the Eurosystem and Its Fiscal Implications, ifo Beiträge zur Wirtschaftsforschung, No. 69, ISBN 978-3-95942-014-3, ifo Institut - Leibniz-Institut für Wirtschaftsforschung an der Universität München, München

This Version is available at: https://hdl.handle.net/10419/167366

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



ifo Beiträge69zur Wirtschaftsforschung

The Collateral Framework of the Eurosystem and Its Fiscal Implications

Jakob Korbinian Eberl



Herausgeber der Reihe: Clemens Fuest Schriftleitung: Chang Woon Nam

ifo Beiträge69zur Wirtschaftsforschung

The Collateral Framework of the Eurosystem and Its Fiscal Implications

Jakob Korbinian Eberl



Bibliografische Information der Deutschen Nationalbibliothek

Die Deutsche Nationalbibliothek verzeichnet diese Publikation in der Deutschen Nationalbibliografie; detaillierte bibliografische Daten sind im Internet über http://dnb.d-nb.de abrufbar

ISBN-13: 978-3-95942-014-3

Alle Rechte, insbesondere das der Übersetzung in fremde Sprachen, vorbehalten. Ohne ausdrückliche Genehmigung des Verlags ist es auch nicht gestattet, dieses Buch oder Teile daraus auf photomechanischem Wege (Photokopie, Mikrokopie) oder auf andere Art zu vervielfältigen. © ifo Institut, München 2016

Druck: ifo Institut, München

ifo Institut im Internet: http://www.cesifo-group.de

Preface

This thesis was written by Jakob Eberl while having been a research assistant at the Center for Economic Studies (CES) at the University of Munich. It was completed in December 2015 and accepted as a doctoral thesis by the Department of Economics at the University of Munich in May 2016. The thesis tells the story of the Eurosystem's collateral framework and its fiscal implications in seven chapters (neglecting the introduction and the conclusion), which can be conceptually assigned to two parts.

The first part comprises Chapters 2 to 5 and elaborates on the Eurosystem's collateral framework. Chapter 2 provides an introductory overview of the development of Eurosystem's monetary policy throughout the financial and sovereign debt crisis. It tracks the crisis mitigation of the Eurosystem and identifies collateral criteria as being crucial to this crisis mitigation, given that it facilitated increased liquidity provision. Chapter 3 provides an in-depth analysis of the Eurosystem collateral framework. It compiles a comprehensive narrative database of general principles and amendments to the collateral framework. Moreover, the effects of amendments to the collateral framework on eligible marketable assets ("collateral pool") are elaborated along several dimensions. Chapter 4 addresses the Eurosystem's risk control of collateral assets and how this was adjusted in response to factors that were exogenous and endogenous to the Eurosystem. Moreover, shortcomings of risk control are identified. Chapter 5 sheds light on the effects of the Eurosystem's collateral framework as well as risk control by tracking the development of credit quality of collateral pledged with the Eurosystem. The model elaborated in this chapter argues that it is rationale for counterparties to pledge collateral of relatively low quality with the Eurosystem while relatively good collateral is pledged on the market ("adverse selection of collateral").

The fiscal implications of the collateral framework are investigated in the second part, which contains Chapters 6 to 8. Chapter 6 formulates the framework for analyzing the fiscal implications, which is based on the analysis of fiscal sustainability. The common framework of fiscal sustainability under certainty is extended to uncertainty such that it facilitates deriving an intuitive indicator for assessing fiscal implications. Chapter 7 elaborates on the fiscal implication of collateral criteria that evolves from the close link between collateral criteria, central bank finances and the government budget. It derives the optimal level of collateral criteria from a fiscal perspective and shows that uniform collateral criteria in a monetary union of heterogeneous countries give rise to cross-country fiscal implications. Chapter 8 addresses the fiscal implication of collateral criteria that originates from the close link between collateral criteria and explicit government guarantees to the financial sector. It shows that the Eurosystem's collateral framework involves government discretion to free up collateral by giving explicit guarantees. This discretion is vital to government incentives to grant guarantees.

Finally, *Chapter 9* provides a summary and rethinks the Eurosystem's collateral framework based on the findings of the thesis.

JEL classification: E51, E52, E58, E62, E63, H12, H63, H77, H81.

Keywords: central banking, monetary policy, fiscal policy, Eurosystem, collateral, collateral framework, haircut, fiscal sustainability, central bank recapitalization, government guarantee.

Danksagung

Die vorliegende Arbeit entstand während meiner Tätigkeit als wissenschaftlicher Mitarbeiter am Center for Economic Studies und dem Lehrstuhl für Nationalökonomie und Finanzwissenschaft der Ludwig-Maximilians-Universität München. Sie wäre ohne die Unterstützung vieler Personen in meinem beruflichen und privaten Umfeld in dieser Form nicht zustande gekommen. Daher will ich nachfolgend denjenigen danken, die mich im Laufe meiner Promotion besonders unterstützt haben.

An erster Stelle möchte ich meinen Doktorvater Hans-Werner Sinn nennen. Für sein Vertrauen und seine Bereitschaft mir in meinen Forschungsinteressen freien Lauf zu lassen sowie für seine zahlreichen wertvollen Anregungen und Hilfestellungen möchte ich ihm herzlich danken. Sein Einfluss hat nicht nur diese Arbeit, sondern auch mein ökonomisches Verständnis nachhaltig geprägt. Darüber hinaus danke ich Clemens Fuest für seine Bereitschaft, die Rolle des Zweitgutachters zu übernehmen. Christoph Trebesch danke ich für die Unterstützung im Verlauf der vergangenen Jahre und seine Bereitschaft als mündlicher Prüfer zu fungieren. Teile dieser Arbeit entstanden während meines Forschungsaufenthalts als Visiting PhD Student am Massachusetts Institute of Technology. Unterstützt wurde ich dabei von Jim Poterba, wofür ihm mein besonderer Dank gilt.

In den vergangenen Jahren habe ich besonders von der Zusammenarbeit und dem intensiven Austausch mit Christopher Weber profitiert. Erkenntnisse dieser Zusammenarbeit finden sich in dieser Dissertation wieder. Christoph Winter danke ich insbesondere für seine zahlreichen Anmerkungen zur vorliegenden Arbeit. Danken möchte ich auch Darko Jus, mit dem ich in den ersten Jahren meiner Promotionszeit eng zusammen arbeiten durfte. Besonderer Dank für die gute Zusammenarbeit gilt zudem meinen (teils ehemaligen) Lehrstuhlkollegen Andreas Blöchl, Florian Buck, Susanne Wildgruber, Michael Stimmelmayr, Christian Beermann und Nadjeschda Arnold. Große Unterstützung auf administrativer Ebene erhielt ich von Ursula Baumann, Ulrike Gasser, Martina Grass, Renate Meitner und Susanne Wormslev.

Nicht zuletzt danke ich Sissi und meinen Eltern für ihre fortwährende und bedingungslose Unterstützung – dafür, dass ihr immer für mich da wart/seid.

Jakob Berl

Vünchen, im Dezember 2015

The Collateral Framework of the Eurosystem and its Fiscal Implications

Inaugural-Dissertation

zur Erlangung des Grades Doctor oeconomiae publicae (Dr. oec. publ.)

eingereicht an der Ludwig-Maximilians-Universität München 2015

vorgelegt von Jakob Korbinian Eberl

Referent: Professor Dr. Dr. h.c. mult. Hans-Werner Sinn Korreferent: Professor Dr. Clemens Fuest Promotionsabschlussberatung: 11. Mai 2016

Contents

Li	st of	Abbre	eviations/Acronyms	iv
\mathbf{Li}	st of	\mathbf{Symb}	ols	\mathbf{vi}
\mathbf{Li}	st of	Figure	es	viii
Li	st of	Tables	5	xi
1	Intr	oducti	ion	1
	1.1	Motiva	ation	1
	1.2	Main	Findings, Related Literature and Contributions of the Thesis	5
	1.3	Struct	ure of the Thesis	11
2	The	• Mone	etary Policy of the Eurosystem and the Growing Importance of the)
	Col	lateral	Framework	13
	2.1	Overv	iew of the Monetary Policy of the Eurosystem	14
	2.2	Asset	Purchases	16
		2.2.1	Covered Bond Purchase Programmes	18
		2.2.2	Securities Market Programme	19
		2.2.3	Asset-Backed Securities Purchasing Programme	21
	2.3	Loans		22
	2.4	The G	rowing Importance of the Eurosystem Collateral Framework	27
3	The	colla	teral Framework of the Eurosystem	29
	3.1	Collat	eral Criteria of the Eurosystem	30
		3.1.1	The Initial Collateral Framework: Tier 1 and Tier 2 Eligibility	30
		3.1.2	A Comprehensive Collateral Framework: the Single List	31
		3.1.3	Two Parallel Collateral Frameworks	31
		3.1.4	General Principles of the Collateral Framework	32
	3.2	Develo	opment of Collateral Criteria of the Eurosystem	36
		3.2.1	Changes to General Collateral Criteria	36
		3.2.2	Changes to the Collateral Criteria of Government-Linked Securities	41
		3.2.3	Changes to the Collateral Criteria of Securities Traded on Non-Regulated	
			Markets	42
		3.2.4	Changes to the Collateral Criteria of Bank Bonds	43
		3.2.5	Changes to the Collateral Criteria of Asset-Backed Securities	48

		3.2.6	Changes to the Collateral Criteria of Corporate Bonds	50
		3.2.7	Preliminary Summary	51
	3.3	Develo	opment of the Pool of Eligible Marketable Assets (Collateral Pool)	52
		3.3.1	Refinement of the Collateral Pool Development	54
		3.3.2	Newly Eligible and Ineligible Marketable Assets	54
		3.3.3	Eligible Marketable Assets in Selected Countries	56
		3.3.4	Credit Quality of Eligible Marketable Assets	60
		3.3.5	Denomination of Eligible Marketable Assets	65
		3.3.6	Non-Regulated Markets and Eligible Marketable Assets	65
		3.3.7	Government Guarantees and Eligible Marketable Assets	67
		3.3.8	Residual Maturity of Eligible Marketable Assets	67
		3.3.9	Lessons from Analyzing the Collateral Pool Development	68
4	\mathbf{Ris}	k Cont	crol of the Eurosystem	70
	4.1	The S	et of Risk Control Measures	71
	4.2	Haircu	its as the Key Risk Control Measure	72
		4.2.1	General Principles of the Haircut Specification	74
		4.2.2	Development of Applied Haircuts	78
		4.2.3	Descriptive Analysis of Applied Haircuts	82
	4.3	Evalua	ation and Shortcomings of the Haircut Specification	88
		4.3.1	Evaluation of the Haircut Specification	89
		4.3.2	Shortcomings of the Haircut Specification	90
	4.4	Poten	tial Effects of the Simplified Haircut Specification	95
5	Col	lateral	Pledged with the Eurosystem	98
	5.1	Develo	opment of Collateral Pledged with the Eurosystem	99
	5.2	The R	tepo Market as the Eurosystem's Competitor for Collateral $\ldots \ldots \ldots$	102
		5.2.1	General Aspects of Repurchase Agreements	102
		5.2.2	Collateral and Haircuts in the European Repo Market \hdots	107
	5.3	A Mo	del of Adverse Selection of Collateral	111
		5.3.1	Model with Private and Public Liquidity Provision	112
		5.3.2	The Eurosystem as Provider of Public Liquidity	120
	5.4	Credit	Quality of Collateral Pledged with the Eurosystem Revisited	133
6	Fra	mewor	k for Analyzing Fiscal Implications of Collateral Criteria	137
	6.1	The C	Concept of Fiscal Sustainability	138
	6.2	Fiscal	Sustainability under Certainty	141
	6.3	The C	Concept of Fiscal Risk	146
	6.4	Fiscal	Sustainability under Uncertainty	150
	6.5	Overv	iew of Fiscal Implications of Eurosystem Collateral Criteria	155
		6.5.1	Collateral Criteria and Government Debt	155
		6.5.2	Collateral Criteria and Central Bank Finances	157
		6.5.3	Collateral Criteria and Government Guarantees	158

7	Fisc	cal Implication from Collateral Criteria and Central Bank Finances	160
	7.1	Relevance of Central Bank Finances	161
	7.2	Stylized Facts on Central Bank Finances in the Eurosystem	163
	7.3	Collateral Criteria and Central Bank Finances	166
		7.3.1 Collateral Criteria and the Income from Money Creation	167
		7.3.2 Collateral Criteria and Emergency Liquidity Assistance	185
		7.3.3 Collateral Criteria and Intra-Eurosystem Claims/Liabilities	187
	7.4	Central Bank Finances and Central Bank Recapitalization	190
		7.4.1 Internal Recapitalization	190
		7.4.2 External Recapitalization	192
		7.4.3 Circumvented Recapitalization	192
	7.5	Optimal Level of Collateral Criteria from a Fiscal Perspective	193
		7.5.1 Fiscally Desired Level of Collateral Criteria with One Country and One	
		Central Bank	195
		7.5.2 Fiscally Desired Levels of Collateral Criteria with Two Countries and One	
		Central Bank	196
8	Fisc	cal Implication from Collateral Criteria and Government Guarantees	203
	8.1	Government Discretion to Create Collateral	204
	8.2	Stylized Facts on Government Guarantees in the Eurozone	206
	8.3	Common Rationale for Government Guarantees	209
	8.4	Novel Rationale: Importance of Collateral Criteria for Government Guarantees .	210
		8.4.1 Expected Costs of a Government Guarantee	211
		8.4.2 Benefits of a Government Guarantee	213
		8.4.3 The Optimal Level of Government Guarantees	215
9	Cor	nclusion	220
	9.1	Summary: the Eurosystem Collateral Framework and its Fiscal Implications	220
	9.2	An Improved Collateral Framework from a Fiscal Perspective	221
		9.2.1 Structure of the Proposed Collateral Framework	221
		9.2.2 Principles of the Proposed Collateral Framework	223
Aj	ppen	ndix to Chapter 5	226
Aj	ppen	ndix to Chapter 7	229
Re	efere	ences	231

List of Abbreviations/Acronyms

ABSs	Asset-Backed Securities		
ABSPP	Asset-Backed Securities Purchase Programme		
ACC	additional credit claim		
ANFA	agreement on net-financial assets		
AP	asset purchases		
BSLI	balance sheet leanness indicator		
CAP	cash provider		
CBPP	Covered Bond Purchase Programme		
CCP	central (clearing) counterparty		
CDS	credit default swap		
CE	certainty equivalent		
CF	cash flow		
CMBS	commercial mortgage-backed security		
COP	collateral provider		
CQS	credit quality step		
DBRS	Dominion Bond Rating Service		
\mathbf{EC}	European Commission		
ECAF	European Credit Assessment Framework		
ECAI	External Credit Assessment Institution		
ECB	European Central Bank		
EEA	European Economic Area		
EEC	European Economic Community		
ELA	Emergency Liquidity Assistance		
EMIR	European Market Infrastructure Regulation		
EMU	European Monetary Union		
ESCB	European System of Central Banks		
ESM	European Stability Mechanism		
EU	European Union		
FCC	fonds communs de créances		
FTPL	fiscal theory of the price level		
GC	general collateral		
GDP	Gross Domestic Product		
GLOC	Gresham's Law of Collateral		
GNP	Gross National Product		
HRL	high-risk lending		
ICMA	International Capital Market Association		

IFAC	International Federation of Accountants	
IMF	International Monetary Fund	
ISIN	International Securities Identification Number	
LC	liquidity category	
LOLR	lender of last resort	
LRL	low-risk lending	
LTRO	longer-term refinancing operation	
MRO	main refinancing operation	
MRR	main refinancing rate	
NAMA	National Asset Management Agency	
NCB	National Central Bank	
NPG	no-Ponzi game condition	
OMO	open market operation	
OMT	outright monetary transactions	
PD	probability of default	
PVBC	present-value budget constraint	
repo	repurchase agreement	
RMBD	retail mortgage-backed debt instrument	
RMBS	residential mortgage-backed security	
SLTRO	supplementary longer-term refinancing operation	
SMEs	small and medium enterprises	
SMP	Securities Market Programme	
SPV	special purpose vehicle	
STEP	Short-Term European Paper	
Target	Trans-European Automated Real-Time Gross Settlement Express Transfer System	
TBTF	too big to fail	
TFEU	Treaty on the Functioning of the European Union	
TLTRO	targeted longer-term refinancing operation	
TVC	transversality condition	
UCITS	Undertakings for Collective Investment in Transferable Securities	
UN	United Nations	

List of Symbols

A	securities purchased by the central bank
b_k	fraction of subsidized bonds in CQS $k \in \{1, 3\}$
c^e	value of contingent explicit liability; cost of government guarantee to eligible assets
\bar{c}^e	cost of government guarantee to ineligible asset
c^i	value of contingent implicit liability
D	government debt
d	government debt relative to GDP
G	government spending
g	growth rate of GDP
h	haircut
h_a	add-on haircut
h_{CB}	haircut applied by the central bank
h_e	effective haircut
h_L	haircut applied by a private lender
h_n	nominal haircut
i	market rate of interest; interest rate on government debt
i_L	repo rate; interest rate in collateralized loan
k	capital key of an NCB
L	refinancing loans given by the central bank
M	monetary base
m	initial margin
pb	primary balance relative to GDP
P	price level
p^e_λ	probability of occurrence of contingent explicit liability
p^i_λ	probability of occurrence of contingent implicit liability
p	counterparty success probability
p_A	success probability of security purchased by the central bank
p_L	success probability of central bank refinancing loan
q	collateral success probability
\hat{q}	minimum credit rating threshold
R	aggregate surplus in the repo market
R_B	rent of borrowers
R_{CB}	rent of the central bank (Eurosystem)
R_L	rent of lenders
r	real interest rate
s	signal on bond quality

\hat{s}_k	minimum rating in CQS $k \in \{1, 3\}$	
T	tax revenue	
V	present value	
v	valuation markdown	
Y	GDP	
α	return to borrower from outside option	
β	discount rate	
Γ	transfer from the central bank to the government	
Γ_{int}	transfer from the central bank to the government in case of income retention	
γ	transfer from the central bank to the government relative to GDP	
δ	central bank income from money creation (seignorage)	
δ_{as}	seignorage measured by asset-side approach	
δ_{fisc}	fiscal seignorage	
δ_{mon}	monetary seignorage	
δ_{opp}	opportunity cost seignorage	
ϵ	misvaluation of bond quality; rating distortion	
ζ	lenders' demand for bonds	
Θ	set of marketing borrowers	
θ	bond quality	
$ heta_{lpha}$	bond quality threshold for borrowers turning to market	
$ heta_{lpha_k}$	bond against which public repo conditions are pooled within CQS k	
$\hat{ heta}_k$	minimum bond quality in CQS k	
κ	liquidity of collateral; recovery rate of loan	
λ	parameter of Poisson process	
μ	lenders' belief on average quality of traded bonds	
ξ	residual maturity of collateral asset	
π	inflation rate	
ρ	bond price	
σ	depositor impatience/panic	
au	haircut subsidy provided by the central bank	
u	income retained by the central bank	
$ u_{req}$	retained income required to restore central bank finances	
Φ	collateral value of asset/bond; value of explicit government guarantee	
Φ_{CB}	collateral value assigned by central bank	
Φ_L	collateral value assigned by lender	
φ	error in theoretical valuation	
χ	(conditional) collateral success probability in case of close links	
Ψ	theoretical value assigned to asset/bond	
Ω	market value of asset/bond	

List of Figures

1.1	Estimated central bank collateral scarcity in selected Eurozone countries	2
1.2	Evolution in liquidity provision of the Eurosystem	4
1.3	Stylized fiscal implications of relaxed Eurosystem collateral criteria	5
2.1	Development of the implementation of loans and asset purchases	16
2.2	Volume of loans and asset purchases	17
2.3	Composition of asset purchases	19
2.4	How NCBs perform reverse transactions	24
2.5	Composition of refinancing loans according to maturity	25
2.6	Main refinancing rate and number of refinancing operations	26
3.1	Changes to general collateral criteria	37
3.2	Long-term government credit ratings for selected countries	40
3.3	Changes to the collateral criteria of government-linked securities $\ldots \ldots \ldots \ldots$	41
3.4	Changes to the collateral criteria of securities traded on non-regulated markets \ldots	42
3.5	The size of the Short-Term European Paper (STEP) market	44
3.6	Changes to the collateral criteria of bank bonds	45
3.7	Changes to the collateral criteria of asset-backed securities (ABSs) $\ldots \ldots \ldots$	48
3.8	Temporal distribution of classified amendments to collateral criteria $\ldots \ldots \ldots$	52
3.9	Published information on eligible marketable assets (Eurosystem collateral pool) $\ .$.	53
3.10	Refined information on the collateral pool by country and by asset type \ldots .	55
3.11	Newly eligible and ineligible marketable assets	56
3.12	Newly eligible marketable assets by country and by asset type	57
3.13	Newly eligible marketable assets and national collateral pools by asset type $\ . \ . \ .$	58
3.14	Newly eligible marketable assets and the collateral pool by credit quality	62
3.15	Newly eligible marketable assets and national collateral pools by credit quality \ldots	63
3.16	Newly eligible marketable assets and the collateral pool by denomination $\ldots \ldots \ldots$	65
3.17	Newly eligible marketable assets and the collateral pool by type of market	66
3.18	Newly eligible marketable assets and the collateral pool with respect to guarantees .	67
3.19	Average residual maturity of eligible marketable assets	68
3.20	Broadening of the Eurosystem collateral pool and national collateral pools	69
4.1	Changes to Eurosystem haircuts	79
4.2	Quantitative development of nominal valuation haircuts	84
4.3	Clustering of asset properties in haircut specification	92
4.4	Distribution of assets within the residual maturity clusters	94

4.5	Clustering of asset credit quality (exemplified for the rating scale of S&P) $\ldots \ldots$	95
4.6	Distribution of eligible marketable assets within CQSs	95
4.7	Average haircut to eligible marketable and pledged assets	96
5.1	Eurosystem refinancing loans by country	99
5.2	Pledged collateral by asset type (published yearly data)	100
5.3	Inferred pledged collateral by country and by credit quality (estim. monthly data) .	101
5.4	Inferred collateral pledged at the country level by asset type	103
5.5	Inferred collateral pledged at the country level by credit quality	105
5.6	Collateral pledged in the European repo market by country	108
5.7	Collateral pledged in tri-party repos by asset type and by credit quality \ldots .	109
5.8	Stylized course of action for a repo \hdots	111
5.9	Equilibrium with symmetric information	114
5.10	Equilibrium with asymmetric information	115
5.11	Equilibrium with positively distorted signal	117
5.12	Outside option and no market unraveling	119
5.13	Outside option and partial market unraveling	120
5.14	Outside option and complete market unraveling	121
5.15	Pooling of public repo conditions I: subsidization of collateral $\ldots \ldots \ldots \ldots \ldots$	123
5.16	Pooling of public repo conditions II: application of supplementary haircuts	125
5.17	Pooling of public repo conditions III: valuation errors	126
5.18	Perfectly-pooled public repo conditions (within CQSs) and correct ratings	128
5.19	Perfectly-pooled outside option (within CQSs) and positively distorted ratings $\ . \ .$	131
5.20	Refined credit quality of inferred pledged collateral I $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	134
5.21	Refined credit quality of inferred pledged collateral II	135
5.22	Average default probability of eligible marketable and pledged assets	136
6.1	Fiscal sustainability and debt dynamics under certainty $\ldots \ldots \ldots \ldots \ldots \ldots$	145
6.2	Probability distributions according to Poisson process	153
6.3	Fiscal sustainability and debt dynamics under uncertainty	154
7.1	Annual profits of selected NCBs and the ECB as well as transfers to governments	164
7.2	Balance sheet leanness indicator for selected NCBs	165
7.3	Risk buffer of selected NCBs and the ECB	167
7.4	Importance of interest income from central bank loans and asset purchases	168
7.5	Stylized central bank balance sheet	172
7.6	Infinite chain of central bank loans under certainty	173
7.7	Infinite chain of uncollateralized central bank loans under uncertainty	175
7.8	Infinite static chain of collateralized central bank loans under uncertainty	176
7.9	Infinite static chain of collat. central bank loans under uncertainty; $\kappa \in [0, 1]$	178
7.10	Infinite dynamic chain of collat. central bank loans under uncertainty; $\kappa \in [0, 1]$	179
7.11	Course of events for collateralized loan under uncertainty	181
7.12	Default risk of Eurosystem counterparties	183

7.13	Liquidity risk of eligible marketable assets	184
7.14	Approximated provision of Emergency Liquidity Assistance by NCBs	187
7.15	Intra-Eurosystem claims and liabilities	189
7.16	Stylized central bank balance sheet with intra-Eurosystem claim/liability	190
7.17	Fiscally desired level of collateral criteria: one country and one central bank	195
7.18	Fiscally desired levels of collateral criteria: two countries and one central bank \ldots	197
7.19	Uniform level of collateral criteria and extent of risk-sharing	198
7.20	Indication on the fiscally preferred level of collateral criteria in selected countries	200
7.21	Sovereign debt crisis and fiscally desired levels of collateral criteria	201
8.1	New government guarantees to eligible marketable assets in selected countries	205
8.2	Government guarantees to eligible marketable assets	207
8.3	Structure of new gov. guarantees: guarantees to already- vs. newly eligible assets	208
8.4	Structure of new gov. guarantees: guarantees by asset type and by country \ldots .	209
8.5	Course of action for bond pledged in <i>private</i> repo	212
8.6	Course of action for bond pledged in <i>public</i> repo	213
8.7	Optimal levels of explicit government guarantees by asset type	217
8.8	Increase in depositor impatience and optimal levels of expl. gov. guarantees	217
8.9	Deterioration in the fiscal position and optimal levels of expl. gov. guarantees	218
8.10	Relaxation of collateral criteria and optimal level of expl. gov. guarantees	219
9.1	Structure of the proposed two-tier collateral framework	222

List of Tables

2.1	Kinds of open market operations and available instruments $\ldots \ldots \ldots \ldots \ldots$	15
2.2	Overview of asset purchase programs	18
2.3	Comparison of operational criteria for the Covered Bond Purchase Programmes \ldots	20
2.4	Details on government bond holdings under the Securities Market Programme	21
3.1	Changes to the collateral criteria of securities with close links	34
3.2	Harmonized rating scale of the Eurosystem	36
3.3	Summary and classification of major amendments to collateral criteria	51
4.1	Eurosystem risk control measures	72
4.2	Differentiation of Eurosystem haircuts at the end of 2014	81
4.3	Nominal valuation haircuts during the pre-crisis phase	83
4.4	Changes to nominal valuation haircuts during the debt crisis phase $\ldots \ldots \ldots \ldots$	87
4.5	Changes to nominal valuation haircuts during the remittent debt crisis phase	88
4.6	Overview of supplementary haircuts applied by the Eurosystem	89
5.1	Combinations of clearing and collateral management in private repos	106
5.2	Haircuts applied to selected government bonds by CCPs and the Eurosystem	110
6.1	Definitions of fiscal sustainability	140
6.2	Government liability matrix	148
7.1	Fiscal implications of different levels of \hat{q}	199
8.1	Monthly average value of new government guarantees in selected countries	206
8.2	Probability of government guarantee to be called	212
8.3	Marginal costs and benefits of explicit government guarantees by asset type	216

1

Introduction

1.1 Motivation

Five counterparties defaulted on refinancing loans from the Eurosystem to the total amount of EUR 10.3 bn when the financial crisis hit Europe in the fall of 2008.¹ According to the European Central Bank (ECB), loans under default were mostly collateralized by assets of limited liquidity that had to be restructured to allow for recovery. Claims unsettled by the restructured collateral assets were deprived from governments that are entitled to National Central Banks' (NCBs) profit. Write-off losses from weakly collateralized refinancing loans that adversely affect governments' budget may not remain an isolated case as the Eurosystem has substantially relaxed its collateral framework during recent years.

The collateral framework is the guiding principle for Eurosystem liquidity provision in terms of both lending and asset purchases. Only eligible marketable assets as specified by the collateral framework qualify for purchases by the Eurosystem (ECB 2011d). Refinancing loans are required to be adequately collateralized (ECB 2012e). Adequacy of collateral implies "[...] first, that the Eurosystem is protected from incurring losses in its credit operations and, second, that sufficient collateral should be available to a wide set of counterparties [...]" (ECB 2007a, pp. 101-2). This manifests the trade-off inherent in the Eurosystem's interpretation of collateral adequacy as it makes two opposing demands on collateral quality: while collateral quality should be sufficiently high to protect the Eurosystem in case of counterparty default (*risk protection*), it should also be sufficiently low to ensure the availability of collateral to counterparties (*collateral availability*).² The Eurosystem solves this trade-off by specifying collateral criteria as the "efficient outcome of a cost-benefit analysis" (ECB 2015c, p. 16).

¹ See ECB, "Eurosystem Monetary Policy Operations in 2008," Press Release, 5 March 2009, SIBERT 2009 as well as BELKE and POLLEIT 2010.

² Stipulation for collateral availability is inter alia related to how central banks implement monetary policy. Central banks predominantly relying on lending should ensure a larger availability than central banks focusing on asset purchases (CHEUN et al. 2009).

The Eurosystem always deemed eligible a wide range of collateral in its endeavor to ensure collateral availability (TABAKIS and WELLER 2009, ECB 2015c). However, this endeavor was challenged throughout recent crises, which threatened collateral availability and raised concerns about collateral scarcity constraining banks' funding. Figure 1.1 depicts collateral scarcity in selected countries from 2007 to 2013 measured in terms of the amount of pledged collateral relative to marketable assets eligible in a country.³ While collateral scarcity was absent prior to the financial crisis in all selected countries, it became a pressing issue (interpreted as values larger than 50%, see dashed line) in Greece, Ireland and Portugal and it substantially increased in Italy and Spain. In accordance with collateral availability as the second demand of collateral adequacy, the Eurosystem substantially and successively relaxed its collateral framework to counter country-specific collateral scarcity.⁴

Figure 1.1: Estimated central bank collateral scarcity in selected Eurozone countries

The figure elaborates on the development of central bank collateral scarcity in selected Eurozone countries from 2007 to 2013. Collateral scarcity is measured in terms of the amount of pledged collateral relative to marketable assets eligible in a country. While collateral scarcity was absent prior to the financial crisis in all selected countries, it became a pressing issue (interpreted as values larger than 50%, see dashed line) in Greece, Ireland and Portugal and it substantially increased in Italy and Spain.



Greece: gaps in the time series owing to ineligibility of government-related collateral; see Chapter 3. Source: author's calculation; European Central Bank, *Eligible Assets Database*.

However, the relaxation of collateral criteria implies that the quality of pledged collateral deteriorates such that the risk exposure of the Eurosystem would increase if the additional collateral risk was not sufficiently hedged. Two broad types of liquidity provision with different risk implications prevail for the Eurosystem, relating to its handling of collateral. First, the Eurosystem provides liquidity by purchasing collateral assets. This gives rise to direct risk since the Eurosystem would have to write off liquidity in case of asset default ("single default"). Second, the Eurosystem provides liquidity based on collateralized loans. This implies indirect risk as liquidity would have to be written off only if both the counterparty and the collateral defaulted ("dou-

 $^{^{3}}$ See also Section 5.1 for further details. Values larger than 100% are possible as the analysis cannot factor in eligible non-marketable assets, given that no information is available.

⁴ See e.g. COEURÉ 2012, LEVELS and CAPEL 2012, SINGH 2013 and ECB 2014a.

ble default"). The relaxation of collateral criteria implies that direct risk from asset purchases increases as the quality of acquired assets declines. Likewise, indirect risk from collateralized loans increases as the quality of risk protection in terms of collateral declines. Prior to the financial crisis, the Eurosystem provided liquidity exclusively by lending against collateral complying with its initially high quality requirements. This is referred to as low-risk lending (LRL). Owing to relaxations of its collateral framework, the Eurosystem altered its liquidity provision towards loans collateralized by low-quality collateral, i.e. high-risk lending (HRL), and asset purchases (AP).

Figure 1.2 mirrors the evolution of Eurosystem liquidity provision from 2007 to 2013. It details the development of the volume of refinancing loans provided against two types of collateral, i.e. (i) collateral estimated to having complied with the pre-crisis credit rating threshold (LRL) and (ii) collateral of lower credit quality (HRL).⁵ Moreover, it illustrates the development of the volume of asset purchases (AP).⁶ Volumes are given relative to those in January 2007 to reflect the evolution in liquidity provision by classifying extended liquidity provision in terms of LRL, HRL and AP. The figure indicates that the Eurosystem extended liquidity by LRL in late 2007 when the first tensions in interbank markets occurred. It readopted this type of liquidity provision from the fall of 2008 to the intensification of the sovereign debt crisis in mid-2010. It turned to providing additional liquidity by HRL as well as AP in the summer of 2010 when it likely started to accommodate refinancing loans against collateral of lower credit quality and to purchase assets. In late 2011, it further performed LRL again such that it provided additional liquidity in 2012 and 2013 by AP as well as LRL and HRL.

Relaxations of collateral criteria bear fiscal implications. The Eurosystem successively increased its risk exposure owing to the evolution of liquidity provision, which is closely related to relaxations of collateral criteria. Risk protection deteriorated from LRL against high-quality collateral to HRL against lower-quality collateral (with insufficient risk hedging) to purchases of collateral. Lower risk protection facilitates the likelihood of a central bank bailout, which constitutes a fiscal risk in terms of a contingent government liability that is not contractually agreed. Hence, the monetary policy of the Eurosystem can have effects on the fiscal policy of Eurozone governments when it would have to be adjusted in response to budgetary spillovers from monetary policy. Furthermore, relaxed collateral criteria bear a further fiscal implication when relaxations affect the eligibility of government-guaranteed assets. Amendments to collateral criteria can acknowledge governments' discretion to free up eligible assets via explicit guarantees to otherwise ineligible assets. In this case, amendments shape incentives to initiate new or extend existing explicit government guarantees, i.e. to assume fiscal risk in terms of contingent government liabilities that are contractually agreed.

Figure 1.3 elaborates on the hypothetical fiscal implications of the Eurosystem collateral framework by indicating the development of the fiscal position of two aggregates of selected Eurozone

⁵ The figure is based on the quality of pledged collateral estimated by the author in Section 5.1. It depicts the development of volume from January 2007 to December 2013 as estimates on collateral credit quality are only available for this period.

⁶ NCBs' asset purchases under the so-called agreement on net-financial assets (ANFA) are neglected here as reliable information is scarce and they officially are no monetary policy issue, rather serving investment purposes. See e.g. ANDERSON and STALLINGS 2013 as well as HOFFMANN 2015 on asset purchases under the ANFA.



Figure 1.2: Evolution in liquidity provision of the Eurosystem

The figure elaborates on the evolution in liquidity provision of the Eurosystem from 2007 to 2013. It details volumes of loans against collateral estimated to having complied with the pre-crisis minimum credit rating (LRL) and collateral of lower credit quality (HRL) together with the development of the volume of asset purchases (AP). Volumes are stated relative to the respective volumes in January 2007 to reflect the evolution in liquidity provision of the Eurosystem.

countries (Germany, Netherlands, Finland and Luxembourg as more financially solid countries and Greece, Ireland, Portugal, Spain, Italy and Cyprus as crisis-stricken countries) in terms of the unweighted average government debt ratios from 2006 to 2014.⁷ The presumption that amendments to collateral criteria bear fiscal implications implies that this development would have been different if the Eurosystem had not relaxed collateral criteria during recent years. The depicted areas are the result of a counterfactual exercise, providing a very stylized indication of the counterfactual development of debt ratios. Accordingly, they reflect the hypothetical development of government debt in the absence of amendments to collateral criteria. The debt ratio would have been smaller for the aggregate of the more financially sound countries as it is claimed that these countries did not benefit from relaxed collateral criteria. By contrast, the aggregate of crisis-stricken countries is considered as a beneficiary of relaxed collateral criteria such that their debt ratio would have been larger in the absence of relaxed collateral criteria. Moreover, the figure provides the unweighted average of explicit government guarantees in addition to government debt ratios for both country-aggregates (dashed lines) in due consideration of the effect of relaxed collateral criteria on government incentives to give explicit guarantees. It suggests that this more comprehensive measure of government indebtedness departed from

⁷ Fiscal sustainability indicators are the obvious alternative for assessing the development of fiscal positions, as pioneered by BLANCHARD 1990 and applied by e.g. the European Commission (EC). See BALASSONE and FRANCO 2000 and CHALK and HEMMING 2000 for discussions. The indicators measure the necessary adjustment of fiscal policy to ensure fiscal sustainability over a given time horizon. Despite their intuitive appeal, quantitative assessments based on forward-looking indicators are flawed as indicators rely on estimates of future developments of variables over an arbitrary time horizon. Fiscal sustainability indicators of the EC are inadequate for assessing the development of fiscal positions as they lack availability and comparability as specific parameters such as the time horizon were adjusted.

debt ratios for both country-aggregates over time as governments extended explicit guarantees inter alia owing to relaxed collateral criteria.

Figure 1.3: Stylized fiscal implications of relaxed Eurosystem collateral criteria

The figure hypothesizes the fiscal implications of relaxed Eurosystem collateral criteria, distinguishing between two aggregates of Eurozone countries. Green refers to the aggregate of financially sound countries while red depicts financially stricken countries. The areas provide a highly stylized indication of the counterfactual development of government debt (in terms of the unweighted average government debt ratio), i.e. the hypothetical development absent of amendments to the collateral framework. It is assumed that government debt would have been lower for the aggregate of financially sound countries and higher for financially stricken countries. Moreover, the figure provides the unweighted average of explicit government guarantees to eligible marketable assets together with government debt for the two aggregates (dashed lines). It suggests that this more comprehensive measure of government indebtedness departed from the progress of debt ratios for both country-aggregates over time as governments extended explicit guarantees inter alia owing to relaxed collateral criteria.



1.2 Main Findings, Related Literature and Contributions of the Thesis

The dissertation derives five main findings that evolve from the analysis of the Eurosystem collateral framework and its fiscal implications. The first three findings are related to the collateral framework, while the further two findings seize upon the fiscal implications of amendments to it.

Main Finding 1: The Eurosystem has intensified the frequency of amendments to its collateral framework over recent years. This has broadened the collateral pool both horizontally (quantitatively) and vertically (qualitatively), as well as affecting its composition. A narrative database of amendments to the collateral framework is compiled, revealing that the Eurosystem has intensified the frequency of amendments in response to the financial and the sovereign debt crisis. Amendments were predominantly directed at relaxing collateral criteria. While they were mostly general as a first response to the financial crisis, the Eurosystem has refined activity towards asset type-specific criteria over time. Moreover, the analysis shows that the Eurosystem has broadened the pool of eligible marketable assets ("collateral pool") both horizontally and vertically. Horizontally, it extended the collateral pool quantitatively, i.e. it deemed eligible additional asset types of initial credit quality. Vertically, the Eurosystem expanded the pool qualitatively by lowering the minimum credit rating threshold for initial types of eligible assets. The investigation of amendments to the collateral framework suggests that the collateral pool has changed both in size and composition. Indeed, an analysis of the collateral pool confirms this suggestion, revealing that amendments have affected the pool with respect to geographical composition, composition by asset type, credit quality of eligible marketable assets, denomination of eligible marketable assets and the average residual maturity of eligible marketable assets.

Main Finding 2: The Eurosystem applies valuation haircuts as the major risk mitigation measure in a simplified fashion based on a reduced set of available information. This implies a subsidization of eligible assets relative to the market and imperfect mitigation of collateral risk. The Eurosystem applies a set of measures to mitigate collateral risk that is specified in the collateral framework. Risk control measures and particularly valuation haircuts as the most important measure were adjusted in response to major exogenous events such as the financial crisis as well as endogenous events such as amendments to the collateral framework. Despite being the major risk mitigation tool, adjustments to haircuts were infrequent and not always specifically related or delayed to observable events. Moreover, the Eurosystem haircut application is associated with three flaws, giving rise to simplified haircuts that neglect available information on asset properties. The negligence of available information stems from (i) the application of the first-best rule when the pivotal rating is identified, (ii)the sequential application of valuation markdowns and haircuts and *(iii)* the clustering of asset properties. The application of simplified haircuts implies pooling of refinancing conditions, i.e. the alignment of collateral values assigned to assets with different properties. The pooling of refinancing conditions has two implications: first, a fraction of assets is subsidized relative to the market; and second, risk control is imperfect as collateral risk is not fully hedged. This increases the Eurosystem's risk exposure, in particular when collateral criteria are relaxed.

Main Finding 3: The application of simplified haircuts makes the Eurosystem prone to the adverse selection of collateral. This reinforces the increase of the risk exposure and contributes to understanding empirical evidence that the default probabilities of eligible and pledged collateral have developed asymmetrically. The Eurosystem likely attracted collateral assets of relatively low quality owing to the application of simplified haircuts and the pooling of refinancing conditions. This phenomenon of adverse selection of collateral is referred to as Gresham's Law of Collateral (GLOC), according to which high-quality collateral is used on the market while low-quality collateral is pledged with the central bank. The attraction of low-quality collateral would be intensified and beyond the control of the central bank if credit ratings were positively distorted. The adverse selection of collateral facilitates the understanding of empirical evidence that the default probabilities of eligible and pledged collateral have developed asymmetrically in the Eurozone over recent years. While the default probability of eligible marketable assets developed relatively stable from 2007 to 2013, the default probability of pledged collateral more than trebled over the same period. This indicates that eligible assets were adversely selected to the Eurosystem in line with GLOC, which reinforced the increase in the Eurosystem's risk exposure over recent years.

Main Finding 4: Amendments to uniform collateral criteria in a monetary union bear cross-country fiscal implications, giving rise to risk-sharing and a transfer of fiscal sustainability across heterogeneous countries. The optimal level of collateral criteria from a fiscal perspective is the solution to the trade-off between expected costs and benefits that accrue to the Treasury from amendments to collateral criteria. Relaxed collateral criteria are beneficial to the Treasury as they enhance liquidity in the financial sector, which amplifies demand for government bonds and lowers government borrowing costs. However, they are also costly to the Treasury in terms of a higher probability of central bank bailout, which constitutes an implicit fiscal risk, i.e. a contingent government liability that is not contractually agreed. Specifically, the relaxation of Eurosystem collateral criteria boosted the probability of a central bank bailout by facilitating central bank risk exposure from higher collateral risk, Emergency Liquidity Assistance (ELA) and intra-Eurosystem imbalances. As Eurozone governments bear the cost of potential central bank bailout unequally and shocks to fiscal positions were asymmetric during the sovereign debt crisis, fiscally desired levels of collateral criteria diverged. Divergence was particularly strong as several countries experiencing the largest deterioration in the fiscal position would have only had to contribute little to the central bank bailout. If the central bank relaxed collateral criteria from an initially strict level to the level preferred by crisis-stricken countries, amendments to collateral criteria would affect fiscal positions. Accordingly, they would increase risk-sharing and transfer fiscal sustainability from financially solid countries to crisis-stricken countries.

Main Finding 5: Amendments to collateral criteria bear fiscal implications as changes to the eligibility of government-guaranteed assets shape government incentives to give explicit guarantees to eligible marketable assets. Amendments to the collateral framework have further fiscal implications when they affect the eligibility of governmentguaranteed assets. Upon first glance, collateral criteria appear to be a monetary policy issue while government guarantees are a fiscal policy one. However, following further consideration, amendments to the collateral criteria of government-guaranteed assets shape government incentives to initiate new or extend existing explicit guarantees. Explicit guarantees constitute an explicit fiscal risk, i.e. a contingent government liability that is contractually agreed. Collateral criteria are vital for government incentives to grant explicit guarantees when they acknowledge government discretion to create eligible collateral via guarantees. Therefore, a novel rationale characterizes the optimal level of explicit government guarantees from a fiscal perspective by trading off the costs and benefits of giving explicit guarantees. Explicit guarantees to eligible assets bear (i) lower expected costs as the government would pay a fraction of the guarantee back to itself via profit distribution in the Eurosystem if the guarantee was called, as well as (ii) higher benefits as guarantees facilitate liquidity in the financial sector, which lowers government borrowing cost via the effect of enhanced liquidity on the demand for government bonds (see Main Finding 4). The narrative database of amendments to the collateral framework revealed that the Eurosystem acknowledged government discretion to create collateral via explicit guarantees to e.g. own-use uncovered bank bonds. The analysis of explicit government guarantees given to eligible marketable assets from 2007 to 2013 suggests that governments contributed to the provision of additional collateral throughout the financial and sovereign debt crisis. Hence, the Eurosystem collateral framework had a bearing on the accumulation of explicit fiscal risk and contributed to intensify the nexus between governments and the financial sector.

The main findings point to the following strands of the literature, which the dissertation complements.

First, the dissertation adds to the body of literature that analyzes and compares *collateral* frameworks of central banks. This body has recently expanded as several central banks have extensively amended collateral criteria as a monetary policy tool since the onset of the financial crisis. CHAILLOUX et al. 2008a review early responses of major central banks to the financial crisis. CHAILLOUX et al. 2008b and CHEUN et al. 2009 provide surveys of the principles shaping the collateral framework of central banks and track their adaptions during the early stage of the financial crisis. TABAKIS and WELLER 2009 provide a comprehensive overview of the collateral framework of the Federal Reserve, the Eurosystem and the Bank of Japan. More recent comparative studies are provided by ECB 2013a, BIS 2013b and ECB 2014a. A general discussion of central bank collateral frameworks is provided by BINDSEIL and PAPADIA 2009, RULE 2012 and BINDSEIL 2014. GROS et al. 2012 outline the major reason why changes to collateral criteria differed between the Federal Reserve and the Eurosystem. The authors argue that while the early stage of the financial crisis (2007-09) was similar on both sides of the Atlantic, the subsequent period was unique to the Eurozone. Accordingly, the Eurosystem collateral framework was amended the most frequently. First overviews of the Eurosystem collateral framework and its amendments are detailed by HOFMANN 2011, ECB 2013c, SINN 2012, 2014b and ECB 2015c. In-depth surveys and analyses of the amendments were more recently published by EBERL and WEBER 2014a, BELKE 2015 and NYBORG 2015. The dissertation complements and extends recent in-depth investigations by compiling a narrative database comprising all relevant amendments to the Eurosystem collateral framework from 2001 to 2014. It provides detailed information on the chronological sequence of amendments, in general and differentiated by asset type. Moreover, new data provided by the Deutsche Bundesbank is investigated to track the effects of amendments to collateral criteria on the development of the Eurosystem collateral pool as well as national collateral pools. Specifically, the analysis of this data provides innovative insights into the credit quality of eligible assets that were hitherto not available. It also facilitates the distinction between quantitative and qualitative broadening of the collateral pool. Hence, the dissertation records the dimension of quantitative and qualitative broadening of the Eurosystem and national collateral pools, as well as further effects of amendments to the Eurosystem collateral framework on collateral pools.

Second, the dissertation contributes to the literature on *risk management of the Eurosystem*, which has received increased attention with the Eurosystem assuming a riskier monetary policy throughout the financial and sovereign debt crisis. Several important studies are collected in BINDSEIL et al. 2009. BINDSEIL 2009a examines central bank activity during the financial crisis from a risk management perspective. BINDSEIL 2009b provides an overview of the different

types of risk to central banks, particularly from central bank policy operations. BINDSEIL and PAPADIA 2009 argue that the risk of eligible assets can be harmonized and mitigated to the level consistent with central bank risk tolerance by means of appropriate risk mitigation. Asset eligibility should subsequently base on a cost-benefit analysis. GONZÁLEZ and MOLITOR 2009 explore in depth with respect to methodological issues of risk mitigation and credit risk assessment in central bank policy operations. Furthermore, the authors motivate the application of different risk mitigation measures and discuss how they are used by the Eurosystem. TABAKIS and WELLER 2009 compare risk mitigation measures and credit risk assessment in central bank policy operations for the Federal Reserve, the Eurosystem and the Bank of Japan. HEINLE and KOIVU 2009 present an approach to estimate tail risk for a portfolio of collateralized central bank loans. More recent analyses of Eurosystem risk control include BINDSEIL 2014, EBERL and WEBER 2014b, ECB 2015c and NYBORG 2015, among others. The dissertation complements this body of literature by providing a comprehensive survey of Eurosystem risk control measures, their application and development from 2001 to 2014. Valuation haircuts are identified as the principal risk control measure and their quantitative development is tracked from 2001 to 2014 for all eligible asset types. It is shown how the Eurosystem adjusted valuation haircuts in response to major exogenous factors such as the financial and the sovereign debt crisis as well as endogenous factors such as its amendments to the collateral framework. Moreover, shortcomings in the application of valuation haircuts are identified.

Third, the dissertation adds to the literature on *adverse selection of collateral*. This phenomenon is referred to as Gresham's Law of Collateral (GLOC), adapted from "Gresham's law" describing the tendency of bad money to drive out good money (e.g. MACLEOD 1856 and VON HAYEK 1976). The analogy between Gresham's law for money and collateral is narratively established by CHAILLOUX et al. 2008a,b, while the adverse selection of collateral is discussed in SINGH 2013. BINDSEIL 2014 seizes on the phenomenon of GLOC and discusses four measures that central banks could in principle consider to counteract the concentration of low-quality collateral owing to adverse selection. The adverse selection of collateral was first analyzed within a model by EWERHART et al. 2006. The authors derive a preference of borrowers to pledge illiquid assets with the central bank taking into account different pricing of liquidity risk by the market and the central bank. NYBORG et al. 2002 argue that the adverse selection of collateral would be the natural outcome of banks using collateral efficiently owing to heterogeneity in the opportunity cost of collateral pledge. EWERHART and TAPKING 2008 confirm this finding that the least liquid and most risky assets will be deposited with the central bank. The adverse selection of collateral is observed for the Eurosystem by BINDSEIL and PAPADIA 2006 from 1999 to 2005 and by CHAILLOUX et al. 2008b from 2004 to 2007. The latter study also cites the adverse selection of collateral for the Federal Reserve in the outset of the financial crisis. EBERL and WEBER 2013 elaborate on and confirm the adverse selection of collateral for the Eurosystem within a theoretical model. Most recently, FECHT et al. 2015 document adverse selection (referred to as "systemic arbitrage") for German banks' borrowing from the Eurosystem from 2006 to 2010. In this context, systemic arbitrage is also addressed by NYBORG 2015. The dissertation extends the literature on the adverse selection of collateral as follows. First, it addresses GLOC specifically for the Eurosystem within a theoretical model that takes up certain peculiarities

10

of the Eurosystem collateral framework and risk control. Second, it emphasizes the role of rating agencies for the extent of adverse selection. Third, it adduces descriptive evidence for the occurrence of GLOC in the Eurozone from 2007 to 2013. Finally, the finding of adverse selection of collateral is used to estimate the credit quality of collateral pledged with the Eurosystem. It thereby provides novel insights since the Eurosystem publishes information on the quantity but not the quality of pledged collateral.

Fourth, the dissertation complements and extends the currently small body of literature on *fiscal* implications of the Eurosystem collateral framework that elaborates on spillovers from Eurosystem collateral criteria to fiscal policy.⁸ A broad approach is usually taken, analyzing the link between monetary and fiscal policy (and vice versa). Important contributions to this approach include but are not limited to SARGENT and WALLACE 1981, AIYAGARI and GERTLER 1985, KING and PLOSSER 1985, LEEPER 1991, COCHRANE 2001, AFONSO 2008, IZE and OULIDI 2009 as well as WALSH 2010. The link originates from the assumption of different yet inherently linked tasks by the central bank and the government, whereby the central bank performs monetary policy and ensures price stability while the government implements fiscal policy in terms of spending and taxation. Both monetary and fiscal policy can contribute to satisfying the government budget constraint: while fiscal policy can be adjusted in terms of spending and revenue, monetary policy can alter money creation, which affects the transfer of financial means from the central bank to the government. Accordingly, printing money can contribute to alleviate distressed public finances at the cost of inflation. A situation in which fiscal policy is expected to adjust such that the government budget constraint holds while monetary policy is set freely is called a *Ricardian* regime with *monetary dominance*. By contrast, a *non-Ricardian* regime with *fiscal dominance* assumes that monetary policy ensures that the government budget constraint is met. Limited attention has been devoted to the relation between collateral policy and fiscal policy, despite the former growing in importance as it facilitated non-standard monetary policy. Existing analyses are usually narrative and focused on the risk from lending against low-quality collateral and assets purchases (e.g. SIBERT 2009, BUITER and RAHBARI 2012b as well as SINN 2012, 2014b). While this risk can culminate in the need for central bank bailout, this was often negated by the assertion that central banks do not care about their finances as they would always be able to recapitalize themselves through money creation (e.g. DE GRAUWE and JI 2013). The dissertation challenges this assertion and collects evidence suggesting that central banks de facto take their financial position into account. Moreover, it argues that the Eurosystem became particularly prone to central bank bailout owing to the relaxation of collateral criteria. It is unrolled that the Eurosystem put at risk its income from money creation, implicitly lowered the collateral criteria for ELA and facilitated the accumulation of huge intra-Eurosystem balances. Thereby, the Eurosystem increased the expected cost of central bank bailout to governments. Taking into account these costs in the government budget constraint to link collateral and fiscal policy, the dissertation extends the literature by elaborating on the

⁸ Accordingly, it only touches the financial and real effects of collateral policy such as the overproduction of lowquality assets and tilts in the composition of bank portfolios. See e.g. BUITER and SIBERT 2005, CHAILLOUX et al. 2008b, CHAPMAN et al. 2010, ASHCRAFT et al. 2011, MILES and SCHANZ 2014, NYBORG and ÖSTBERG 2014, NYBORG 2015, FECHT et al. 2015 and VAN BEKKUM et al. 2015.

optimal level of collateral criteria from a fiscal perspective.⁹ It reveals how uniform collateral criteria in a monetary union of heterogeneous countries give rise to risk-sharing and the transfer of fiscal sustainability countries.

Finally, the dissertation contributes to the literature, which addresses the importance of collateral criteria for government incentives to grant explicit guarantees. A novel rationale for government guarantees is presented that differs from the common rationale. DIAMOND and DY-BVIG 1983 claim that the government should provide support to illiquid financial institutions, facing a potential run on their deposits owing to panic among depositors, based on the selffulfilling belief of deposit withdrawal. According to ibid., government guarantees would then constitute a costless and fully effective means to prevent panic-driven bank runs. However, the recent financial crisis suggests that bank runs may not be driven by irrational panic rather caused by deterioration of fundamentals such as economic conditions and the value of bank assets, which gives rise to insolvencies of banks. Moreover, governments experienced hard times in providing sufficient resources for necessary guarantees. In this case, government guarantees cannot fully foreclose the possibility of bank runs; rather, they can entail substantial costs, thus rendering them a potential threat to the government budget. Accordingly, the novel rationale elaborated in the dissertation characterizes the optimal level of government guarantees from a fiscal perspective by trading off the costs and benefits of giving explicit guarantees that are inter alia determined by collateral criteria.

1.3 Structure of the Thesis

The dissertation tells the story of the Eurosystem collateral framework and its fiscal implications in seven chapters, which can be conceptually assigned to two parts. The first part comprises Chapters 2, 3, 4 as well as 5 and elaborates on the Eurosystem collateral framework. The fiscal implications of the collateral framework are investigated in the second part, which contains Chapters 6, 7 and 8. Chapter 2 provides an introductory overview of the development of Eurosystem monetary policy throughout the financial and sovereign debt crisis. It tracks the crisis mitigation of the Eurosystem and identifies collateral criteria as being crucial to this crisis mitigation, given that it facilitated increased liquidity provision. The Eurosystem mitigates risk from liquidity provision in two steps: first, it requires credit operations to be based on collateral; and second, collateral assets are subject to risk control. Stepwise risk mitigation is reflected in the structure of the dissertation. Chapter 3 provides an in-depth analysis of the Eurosystem collateral framework. It compiles a comprehensive narrative database of general principles and amendments to the collateral framework. Moreover, the effects of amendments to the collateral framework on eligible marketable assets (collateral pool) are elaborated along several dimensions. Chapter 4 addresses the Eurosystem risk control of collateral assets and how this was adjusted in response to factors that were exogenous and endogenous to the Eurosystem. Moreover, shortcomings of risk control are identified. Chapter 5 sheds light on the effects

⁹ Naturally, the fiscal perspective is simply one of several, given that collateral criteria can also be considered in terms of monetary policy transmission and financial stability (BINDSEIL 2014). However, the fiscal perspective must not be neglected as central bank collateral first and foremost serves the purpose of risk mitigation to the central bank, the finances of which are closely intertwined with the government budget.

of the Eurosystem collateral framework as well as risk control by tracking the development of credit quality of collateral pledged with the Eurosystem. The model elaborated in this chapter shows that it is rationale for counterparties to pledge collateral of relatively low quality with the Eurosystem while relatively good collateral is pledged on the market (adverse selection of collateral). This suggests that the relaxation of collateral criteria had particularly strong adverse effects. Chapter θ formulates the framework for analyzing the fiscal implications of the collateral framework, which is based on the analysis of fiscal sustainability. The common framework of fiscal sustainability under certainty is extended to uncertainty such that it facilitates deriving an intuitive indicator for assessing fiscal implications. Chapter 7 elaborates on the fiscal implication of collateral criteria that evolves from the close link between collateral criteria, central bank finances and the government budget. It derives the optimal level of collateral criteria from a fiscal perspective and shows that uniform collateral criteria in a monetary union of heterogeneous countries give rise to cross-country fiscal implications. Chapter 8 addresses the fiscal implication of collateral criteria that originates from the close link between collateral criteria and explicit government guarantees to the financial sector. It shows that the Eurosystem collateral framework involves government discretion to free up collateral by giving explicit guarantees. This discretion is vital to government incentives to grant guarantees. Finally, Chapter 9 provides a summary and rethinks the Eurosystem collateral framework based on the findings of the dissertation.

The Monetary Policy of the Eurosystem and the Growing Importance of the Collateral Framework

This chapter provides an overview of Eurosystem monetary policy, its general principles and its development following the outset of the financial crisis and throughout the European sovereign debt crisis. It also emphasizes the central role assigned to the collateral framework in the crisis mitigation of the Eurosystem. The monetary policy of the Eurosystem prior to the crisis can be stylized as follows: it allotted refinancing credit to counterparties at a variable-rate tender procedure for either one week or three months at a relatively high interest rate against collateral of high quality. After the outset of the financial crisis, the Eurosystem mitigated crisis by fully allotting refinancing credit at a fixed-rate tender procedure for various periods of up to three years at a historically low interest rate against a myriad of collateral of ever-declining quality. In addition, the Eurosystem launched several programs to purchase certain types of assets in the primary and secondary market that complied with its relaxed collateral framework. Therefore, the collateral framework was key, albeit it was less prominent to Eurosystem monetary policy and crisis mitigation as collateral availability became the de facto only constraint to liquidity provision. Therefore, relaxations of collateral criteria facilitated the new style of lending in terms of the full allotment of refinancing credit as well as asset purchases. The chapter is structured as follows. Section 2.1 provides an overview of Eurosystem monetary policy and its crisis measures. Sections 2.2 and 2.3 examine the two types of open market operations (OMOs), *i.e.* "outright transactions" (asset purchases) and "reverse transactions" (loans), as well as their application. Finally, Section 2.4 carves out that amendments to the Eurosystem collateral framework were crucial as they were fundamental to liquidity provision in terms of loans and asset purchases.

2.1 Overview of the Monetary Policy of the Eurosystem¹⁰

Article 18.1 of the Statute of the European System of Central Banks (ESCB) and of the ECB allows the Eurosystem to operate in financial markets "to achieve [its] [...] objectives and carry out its tasks" (ECB 2012e, p. 238). For this purpose, the monetary policy kit of the Eurosystem contains three tools: *(i)* OMOs, *(ii)* the minimum reserve system and *(iii)* standing facilities.¹¹ It applies these tools to steer short-term money market rates and ensure its functioning through the provision of liquidity to commercial banks.¹² Unlike other central banks, the Eurosystem pursues a wide and decentralized implementation of its monetary policy, i.e. interaction between NCB and a multitude of eligible counterparties. In order to be eligible, counterparties have to obey the minimum reserve system and be supervised by an European Economic Area (EEA) national authority, as well as being considered financially sound (see ECB 2015c).¹³ The focus is placed upon OMOs in the remainder, as the major monetary policy tool for crisis mitigation.

The Eurosystem performs OMOs via five instruments. The most important is the reverse transaction in which it lends liquidity to counterparties against collateral assets. Second, the Eurosystem can also purchase or sell assets on the market, which it calls outright transactions.¹⁴ Finally, it can issue ECB debt certificates, make foreign exchange swaps and collect fixed-term deposits. These five instruments can be assigned to three kinds of OMOs that differ in their objectives: *(i)* refinancing operations, *(ii)* fine-tuning operations and *(iii)* structural operations. Refinancing operations are performed on a regular basis to facilitate liquidity in financial markets. Finetuning operations are performed on an ad-hoc basis with the aim of managing liquidity in the market and steering interest rates, particularly to smooth the interest rate effects of unexpected liquidity fluctuations in the market. Structural operations are applied to adjust market liquidity over the longer term such that the Eurosystem can adapt its structural position towards the financial sector. Table 2.1 depicts the instruments with which the three kinds of OMOs are performed. The focus is placed upon reverse and outright transactions, which have been the dominant instruments of the Eurosystem to mitigate crisis via market intervention.¹⁵

Reverse transactions are the main instrument as they are used to perform all three kinds of OMO. They play the most important role in the Eurosystem's refinancing operations and are applied based upon either a repurchase agreement (repo) or a collateralized loan. In a repo,

¹⁰ For further details on the response of the Eurosystem to the financial crisis, see among many others e.g. ECB 2010c,d, TRICHET 2010 and SINN 2012, 2014b. For its response to the European sovereign debt crisis, see e.g. COUR-THIMANN and WINKLER 2013, CLAEYS 2014 and SINN 2014b. Moreover, the annex of the monthly bulletins of the ECB provides a comprehensive record and chronology of monetary policy measures of the Eurosystem.

¹¹ For a comprehensive explanation of the Eurosystem's monetary policy framework, see ECB 2015b, commonly referred to as "General Documentation".

¹² More specifically, OMOs serve the purpose of steering interest rates, managing market liquidity and signaling the monetary policy stance. The minimum reserve system aims at stabilizing money market interest rates and creating/enlarging structural liquidity shortage. Standing facilities serve the purpose of providing or absorbing overnight liquidity, signaling the monetary policy stance and bounding overnight interest rates.

¹³ Exceptions are possible, e.g. for branches of non-EEA banks within the Eurozone. Furthermore, operational criteria by the respective NCB have to be met.

¹⁴ There is no general restriction on the market in which assets are purchased, i.e. purchases in both the primary and secondary market are permitted. However, according to ECB 2012e, Article 21.1), the Eurosystem is restricted to purchasing government bonds only in the secondary market.

¹⁵ For details on foreign exchange swaps, fixed-term deposits and the issuance of ECB debt certificates as the other instruments to perform OMOs, see ECB 2011d.

Table 2.1: Kinds of open market operations and available instruments

The table shows the instruments by which the Eurosystem can perform the three kinds of OMOs. Reverse and outright transactions have been the dominant instruments to mitigate crisis. While reverse transactions can be used for all three kinds of OMOs, outright transactions can exclusively be performed as a structural operation.

KIND OF OMO	Instrument	Purpose
refinancing operation	reverse transaction	steer short-term interest rates, man- age liquidity, signal monetary policy stance, provide long-term refinancing
fine-tuning operation	reverse transaction, foreign exchange swap, fixed-term deposit	manage liquidity, smooth effects on interest rates of unexpected liquidity fluctuations
structural operation	reverse transaction, out- right transaction, issuance of ECB debt certificate	adjust structural position of the Eu- osystem towards financial sector

Source: author's compilation; European Central Bank.

ownership of an asset is transferred from the borrower to the lender together with an agreement on reversing the transaction through a retransfer of the asset at a future point in time. By contrast, in a collateralized loan, ownership of the asset is retained by the borrower and the lender receives an enforceable security interest over the asset.¹⁶ In the remainder of this chapter, reverse transactions are referred to as *loans* for simplicity.

The other instrument to perform OMOs that is investigated is outright transactions. They are exclusively performed for structural operations, usually by NCBs, although the Governing Council of the ECB can decree the ECB to perform outright transactions when deemed necessary. Akin to reverse transactions in terms of repos, outright transactions involve a transfer of ownership, albeit with no envisaged reverse transfer of ownership. In the remainder of this chapter, outright transactions are labeled as *asset purchases* for simplicity.

The onset of the financial crisis in the fall of 2008 substantially altered the execution of OMOs, particularly with respect to loans (in terms of reverse transactions) and asset purchases (in terms of outright transactions). This is illustrated in Figure 2.1 for the period from 2008 to 2014.¹⁷ While amendments above the timeline apply to loans, actions below are related to asset purchases. The Eurosystem initiated five asset purchase programs between 2008 and 2014, i.e. the first Covered Bond Purchase Programme (CBPP), the Securities Market Programme (SMP), the second and third CBPP as well as the Asset-Backed Securities Purchase Programme (ABSPP). Moreover, and as an immediate response to market turnoil in October 2008, the Eurosystem performed numerous amendments to its lending operations, especially with respect to the maturity of loans and assets deemed eligible as collateral. Hence, maturities were successively extended, e.g. in October 2008, when the maturity of longer-term refinancing operations (LTROs) was prolonged from three to six months. Furthermore, collateral criteria were extensively amended, most prominently in October 2008 when the minimum credit rating threshold for eligible assets

¹⁶ See Chapter 5 for further distinction between repurchase agreements and collateralized loans.

¹⁷ Changes to the interest rate applied to loans are also a major amendment. For the sake of clarity, interest rate changes are not reflected in Figure 2.1 but are addressed in Section 2.3.

was lowered for the first time. Figure 2.1 provides a *qualitative* indication for the assessment that Eurosystem crisis management between 2008 and 2014 predominantly involved accommodating the conditions for loans complemented by asset purchases.

Figure 2.1: Development of the implementation of loans and asset purchases

The figure depicts the major amendments to the implementation of OMOs by means of loans and asset purchases from 2008 to 2014. Amendments to loans are shown above the timeline while major events with respect to asset purchases are given below the timeline. The figure indicates *qualitatively* that the Eurosystem mitigated crisis predominantly by adjusting conditions for loans complemented by asset purchases.



A quantitative indication for this assessment is provided by Figure 2.2, which shows the development of the volume of Eurosystem's loans and asset purchases from 1999 to 2014.¹⁸ The volume of loans substantially increased following the onset of the financial crisis and subsequently remained at a high level, before returning to its pre-crisis level in mid-2011. Thereafter, the reintensification of the European sovereign debt crisis led to an even larger increase in the volume of loans culminating in mid-2012. By contrast, asset purchases did not take place earlier than July 2009, when the Eurosystem set up its first asset purchase program directed at covered bonds. The volume of asset purchases increased during the onset of the European sovereign debt crisis in mid-2010. Thereafter, four programs directed at different asset types followed and the purchase volume increased accordingly, particularly after the sovereign debt crisis reintensified. The following two sections take a closer look at the development of asset purchases and loans, i.e. the blue and red areas in Figure 2.2.

2.2 Asset Purchases¹⁹

Article 18.1 of the Statute of the ESCB and of the ECB allows the Eurosystem to operate in financial markets inter alia by performing outright transactions, i.e. by purchasing and selling assets (ECB 2012e). Hence, asset purchases are a standard instrument to perform OMOs, although their application is intended within structural operations only such that their frequency is not standardized. From a legal perspective, asset purchases imply a full transfer of owner-

¹⁸ Loans comprise liquidity provided by the Eurosystem via main refinancing operations (MROs), LTROs, finetuning reverse operations, structural reverse operations, marginal lending facility and credits related to margin calls. See also Figure 2.5.

¹⁹ NCBs' asset purchases under the so-called ANFA are neglected here as they were discovered very recently and officially are no monetary policy issue, rather serving investment purposes. See e.g. ANDERSON and STALLINGS 2013 as well as HOFFMANN 2015 on asset purchases under the ANFA.



The figure illustrates the volume of loans and asset purchases of the Eurosystem from 1999 to 2014. Having moderately risen prior to the financial crisis, the volume of loans substantially increased after (i) its onset (October 2008) and (ii) the reintensification of the European sovereign debt crisis (mid-2011). The Eurosystem purchased assets as of July 2009. The figure *quantitatively* confirms the qualitative indication (Figure 2.1) that the Eurosystem mitigated crisis predominantly via loans complemented by asset purchases.



Source: author's illustration; Euruopean Central Bank.

ship from the selling counterparty to the Eurosystem at a price in accordance with best market practice.²⁰ A priori, the Eurosystem does not impose any restriction on the range of counterparties from which it may acquire assets. In general, the Eurosystem purchases assets based upon bilateral transactions, meaning that it applies no tender within such transactions. The transactions are usually carried out by the NCBs on instruction from the ECB, although the ECB can purchase the assets itself under exceptional circumstances. Only eligible marketable assets as defined by the Eurosystem's collateral framework are to be purchased. Since the Eurosystem considers asset purchases as part of its common monetary policy, purchased assets are held to maturity. They are initially valued at acquisition cost and later on at amortized cost, i.e. at book value. The Eurosystem purchases assets on the market as any other investor and hence it does not inherently receive preferable treatment in the event of asset default.²¹

Table 2.2 seizes on the five asset purchase programs launched by the end of 2014 (Figure 2.1), summarizes the programs in chronological order and gives the respective targeted asset type, the start and end dates as well as the peak volume. Overall, the Eurosystem purchased assets to the amount of EUR 328.7 bn by the end of 2014. Among these assets, 32.8% (EUR 107.7 bn) were covered bonds acquired under CBPPs 1-3, only 0.5% were Asset-Backed Securities (ABSs) purchased under the ABSPP and 66.7% were government bonds acquired under the SMP.

While Table 2.2 provides only peak volumes, i.e. the highest volume of assets purchased under the respective program, the left panel of Figure 2.3 reveals the development of purchase volumes

 $^{^{20}}$ For this and the following general principles of asset purchases, see ECB 2011d.

²¹ Nevertheless, the ECB received preferable treatment in the Greek debt restructuring of 2012, cf. e.g. GULATI et al. 2013, SINN 2014b as well as TREBESCH and ZETTELMEYER 2014.

Program	Asset Type	Start Date	End Date	Peak Volume ^a
CBPP1	covered bonds	02/07/2009	30/06/2010	EUR 61.7 bn
SMP	government bonds	10/05/2010	06/09/2012	EUR 219.3 bn
CBPP2	covered bonds	03/11/2011	31/10/2012	EUR 16.4 bn
CBPP3	covered bonds	20/10/2014	$ongoing^{a}$	EUR 29.6 bn
ABSPP	ABSs	21/11/2014	$\operatorname{ongoing}^{\mathrm{a}}$	EUR 1.7 bn

Table 2.2: Overview of asset purchase programs

The table lists the asset purchase programs of the Eurosystem in chronological order. Three programs were targeted at covered bonds, of which only the last one remains ongoing. The latest purchase program was launched at the end of 2014 and directed at ABSs. The most important program was the SMP, which was targeted at government bonds. It was started in May 2010 and terminated in September 2012 with a peak volume of EUR 219.3 bn.

^a By the end of 2014.

Source: author's compilation; European Central Bank.

under the respective programs over time. The right panel shows the composition of purchases by asset types. It is apparent that after the SMP was launched, government bonds accounted for the lion's share of acquired assets. The remainder of this section investigates the purchase programs for the three types of assets separately, focusing on the explanation of the developments given in Figure 2.3.

2.2.1 Covered Bond Purchase Programmes (CBPPs)

By the end of 2014, the Eurosystem had launched three purchase programs directed at covered bonds.²² Although the names of the programs differ only in the digit indicating their order, technical features have substantially differed, particularly between the first two and the last CBPP.²³ Table 2.3 provides an overview of the features and highlights the differences of the third CBPP. All three programs were targeted at euro-denominated covered bonds issued by financial institutions located in the Eurozone. These bonds had to have a first-best rating of "triple B" to be eligible for purchase.²⁴ In contrast to the first two CBPPs, CBPP3 was set up broadly.²⁵ This manifests for instance in the duration of *at least* 24 months of CBPP3 (in contrast to 12 months for CBPP1 and 2) and the absence of a target volume. Moreover, while CBPP1 and 2 were directed towards covered bonds of certain maturities and minimum values, neither a targeted maturity nor a minimum value was specified for CBPP3. Finally, whereas all three programs have been targeted at covered bonds in both the primary and secondary market,

²² For the role of covered bonds in the Eurozone, see ECB 2008a; for the CBPPs and market impacts, see BEIRNE et al. 2011 and SZCZERBOWICZ 2014.

 $^{^{23}\,}$ For the technicalities of CBPP1 to CBPP3, cf. ECB 2009, ECB 2011b and ECB 2014b.

²⁴ This corresponds to credit ratings equivalent to BBB-/Baa3/BBBL; for the harmonized rating scale of the Eurosystem that matches ratings over rating agencies, see Section 3.1.4.3. For CBPP1, a first-best rating of AA/Aa2 was laid down "as a rule" (ECB 2009, p. 19). However, the rating could not be lower than BBB-/Baa3/BBBL, which hence was the actual minimum rating.

²⁵ The only restriction relative to the previous programs was determined in terms of an issue share limit of 70% for all covered bonds acquired under the CBPP3. For covered bonds being issued in Greece or Cyprus, this limit was 30%. This means that the Eurosystem does not purchase more than 70% or 30%, respectively, of any given bond issuance/International Securities Identification Number (ISIN).
Figure 2.3: Composition of asset purchases

The left panel shows the development of the nominal value of assets purchased under the five programs from 2009 to 2014. It reveals that covered bonds acquired under the CBPP1 were accumulated until July 2010, after which their nominal value continuously decreased. The value of covered bonds acquired under the CBPP2 and 3 and of ABSs purchased under the ABSPP was rather small until the end of 2014. Two phases can be identified with respect to government bond purchases under the SMP. In the first phase (05/2010 to 08/2011), the Eurosystem bought bonds issued by the governments of Greece, Ireland and Portugal. The purchase of Italian and Spanish government bonds initiated the second phase (as of 08/2011), which was more important in quantitative terms.



CBPP3 also allows for the purchase of retained covered bonds directly from issuers.²⁶ The relaxations of eligibility requirements may explain the observation that the Eurosystem bought substantially more bonds in the first month under the CBPP3 (EUR 12.7 bn) than under the previous CBPPs (EUR 4.2 bn and EUR 1.6 bn, respectively).

2.2.2 Securities Market Programme (SMP)

The SMP was the second asset purchase program and the first directed at government bonds.²⁷ First government bond purchases were undertaken in May 2010 in the secondary market, prompting a full discussion on whether the Eurosystem would be legally allowed to perform such purchases.²⁸ The SMP was terminated by the Eurosystem's announcement of outright monetary transactions (OMT) in September 2012. By the terminal date, NCBs acquired government bonds in strict proportion to their shares in ECB capital in the total amount of more than EUR 200 bn in two phases (left panel of Figure 2.3). In the first phase, lasting from May 2010 to August 2011, the Eurosystem bought bonds issued by the governments of Greece, Ireland and Portugal. The decision to also buy Italian and Spanish government bonds initiated the second phase as of August 2011. Government bond holdings decreased after a peak volume of almost EUR 220 bn in February 2012, amounting to EUR 144 bn by the end of 2014. The holdings can

²⁶ According to ECB data, the fraction of covered bonds purchased under the CBPP2 in the primary market increased from 15% in November 2011 to 37% in October 2012. For CBPP3, the share of purchases in the primary market during the first month of the program amounted to 24%.

²⁷ For details on the SMP, see ECB 2010a. For studies of the SMP and its impacts, see e.g. MANGANELLI 2012, ESER and SCHWAB 2013, GHYSELS et al. 2014 as well as TREBESCH and ZETTELMEYER 2014. See e.g. MEANING and ZHU 2011 as well as GLICK and LEDUC 2012 for market impacts of central bank asset purchases.

²⁸ These discussions were mainly concerned with the question of whether purchases of government bonds would be compatible with Article 123 of the Treaty on the Functioning of the European Union (TFEU) (cf. EU 2012), which prohibits direct purchases of government bonds by the Europystem. See e.g. SINN 2014a as well as BUITER and RAHBARI 2012b for opposing views. See also FUEST 2013.

Table 2.3: Comparison of operational criteria for the Covered Bond Purchase Programmes

The table compares the operational criteria for the three purchase programs targeted at covered bonds. Provisions made for CBPP3 differing from those for previous programs are highlighted. Most importantly, CBPP3 has a longer duration and no target volume was specified. Moreover, the Eurosystem acquires retained covered bonds under the CBPP3 and neither a targeted maturity nor a minimum volume of covered bonds was specified.

	CBPP1	CBPP2	CBPP3	
DURATION	12 months	12 months	at least 24 months	
TARGET VOLUME	EUR 60 bn	EUR 40 bn	none	
Peak Volume	EUR 61.7 bn	EUR 16.4 bn	n/a	
Allocation of Pur-	"across the Eurogene"	"across the Furgene"	"carried out progressively by the ECB	
CHASES	across the Eurozone	across the Eurozone	and the NCBs"	
Market	primary and secondary	primary and secondary	primary and secondary	
PURCHASE OF RE-	no	no	Noc	
TAINED BONDS	110	110	усэ	
Guiding Eligibility	colletoral framowork	collatoral framowork	collateral framework	
Principle	conateral framework	conateral framework		
CURRENCY	EUR	EUR	EUR	
MATURITY	between 3 to 10 years	up to 10.5 years	none	
MINIMUM DATING	first-best rating of $AA/Aa2$ as a rule;	first host rating of "triple R"	first-best rating of "triple B"	
	in any case not below "triple B"	inst-best fatting of triple b		
MINIMUM VALUE	EUR 500 mn as a rule; in any case	EUB 300 mp	nono	
WIINIMUM VALUE	not below EUR 100 mn $$	Een 500 mil	none	
PURCHASES PER BOND	no limit	no limit	70% per ISIN; $30%$ in case of	
ISSUANCE	110 111111		Greek and Cypriot covered bonds	

Source: author's compilation; European Central Bank.

be expected to continuously decrease over the coming years as the Eurosystem intends to hold purchased government bonds to maturity.

Table 2.4 indicates the fractions according to which the Eurosystem bought government bonds from the five countries based on end-of-year government bond holdings.²⁹ Accordingly, Italian government bonds accounted for almost half of the bonds acquired under the SMP, followed by Spanish (about 20%), Greek (about 15%), Portuguese (about 10%) and Irish government bonds (about 6%). Moreover, Table 2.4 details the average maturity of purchased government bonds. The average maturity of total government bond holdings amounted to 4.3 years by the end of 2012 and slightly decreased to 3.9 and 3.7 years by the end of 2013 and 2014, respectively. The Eurosystem sterilized government bond purchases to avoid inflationary pressure (via the effect of bond purchases on the monetary base). Accordingly, the Eurosystem offset acquisitions by weekly fine-tuning operations equal to the volume of government bonds held, hence leaving the monetary base unaffected.³⁰ The Eurosystem suspended these fine-tuning operations as of 10 June 2014. The SMP was terminated with the decision for OMT on 6 September 2012. By the end of 2014, government bond purchases under the OMT had not yet taken place.

Table 2.4: Details on government bond holdings under the Securities Market Programme

The table presents information published by the ECB on the holdings of government bonds from Ireland, Greece, Spain, Italy and Portugal. The largest fraction are Italian government bonds, followed by Spanish, Greek, Portuguese and Irish bonds. Overall, the Eurosystem held EUR 218 bn of government bonds at the end of 2012, which subsequently decreased to EUR 185.7 bn (end of 2013) and EUR 149.4 bn (end of 2014). The average residual maturity of government bond holdings amounted to 3.7 years at the end of 2014.

	Nominal Value ^a			Ø Re	Ø Residual Maturity ^b		
	end-2012	end-2013	end-2014	end-2012	end-2013	end-2014	
IRELAND	14.2~(6.5%)	9.7~(5.2%)	9.7~(6.5%)	4.6	5.3	4.3	
GREECE	33.9~(15.6%)	27.7 (14.9%)	19.8~(13.3%)	3.6	3.4	3.5	
Spain	44.3~(20.3%)	38.8~(20.9%)	28.9~(19.3%)	4.1	3.6	3.8	
ITALY	102.8~(47.2%)	89.7~(48.3%)	76.2~(51%)	4.5	4.1	3.8	
Portugal	22.8~(10.5%)	19.8~(10.7%)	14.9 (10%)	3.9	3.4	3.3	
Total	218	185.7	149.4	4.3	3.9	3.7	

^a in EUR bn; relative values in brackets (deviations possible owing to rounding).

^b in years.

Source: author's calculation; European Central Bank.

2.2.3 Asset-Backed Securities Purchase Programme (ABSPP)

The last asset purchase program initiated by the Eurosystem by the end of 2014 was directed at ABSs (see ECB 2015a). In the initial phase, the ECB intended to acquire ABSs on its own,

²⁹ The ECB does not publish detailed data on purchased government bonds. See BARCLAYS CAPITAL 2012 and SINN 2012, 2014b for inferred asset purchases by country over time. There, it is assumed that the Eurosystem bought government bonds broadly in proportion to the size of the underlying bond markets. This implies that the Eurosystem purchased about 50%, 25% and 25%, respectively, of Greek, Irish and Portuguese government bonds during the first phase, as well as 66% and 33%, respectively, of Italian and Spanish government bonds (and a very small amount of Portuguese and Irish bonds) during the second phase. See also TREBESCH and ZETTELMEYER 2014.

 $^{^{30}\,}$ These weekly fine-tuning operations were performed as fixed-term deposits, see Table 2.1.

whereas later on purchases will be undertaken in a decentralized fashion by the NCBs. ABSs shall be purchased in the primary and secondary market from counterparties eligible for regular monetary policy operations and counterparties used by the Eurosystem for the investment of their euro-denominated investment portfolios, as well as entities deemed eligible by ECB's Governing Council on a case-by-case basis. The Eurosystem intends to purchase euro-denominated ABSs that are issued in the Eurozone and comply with its collateral framework. ABSs must have a second-best rating of "triple B" to be eligible for purchase. However, this rating requirement is waived for ABSs issued in Greece and Cyprus in compliance with the suspension of the minimum rating requirement for these two countries in the collateral framework (see Section 3.2.2). Akin to asset purchases under the CBPP3, there is neither a specified target volume nor a minimum issuance volume. The Eurosystem applies issue share limits of 70% in general and 30% to those ABSs issued in Greece and Cyprus. ABSs holdings purchased under the ABSPP amounted to EUR 1.7 bn by the end of 2014 (see EUROPEAN CENTRAL BANK 2015a).

2.3 Loans

The Eurosystem regularly conducts OMOs as refinancing operations to steer short-term interest rates and market liquidity (cf. Table 2.1).³¹ Refinancing operations are performed as MROs and LTROs, for which the same collateral assets are eligible. MROs are the regular liquidity-providing transactions, performed weekly with a usual period of one week. LTROs are performed monthly and usually have a period of three months. For both cases, the Eurosystem announces whether it wants to distribute liquidity through a fixed or variable-rate tender. In the former, banks bid the amount of money they want to borrow at a predefined interest rate, while in the latter, they bid the amount and the interest rates at which they want to borrow. Refinancing operations are performed in terms of reverse transactions (cf. Table 2.1). NCBs have the option to conduct reverse transactions as repos or collateralized loans (ECB 2015b).

Within a (liquidity-providing) repo, the NCB and the counterparty agree on two transfers at two distinct points in time. Initially, the NCB buys an asset from the counterparty and at some predefined date in the future the transaction is reversed as the counterparty repurchases the asset from the NCB at a specified price.³² Hence, legal ownership of the asset is transferred to the NCB for a limited period of time. The counterparty effectively acts as the borrower and the NCB as the lender. The interest rate applied to the repo is implied by the difference between the sale and repurchase price.

The second possible way to perform a reverse transaction is a collateralized loan, i.e. the NCB makes a loan to a counterparty that is secured by assets. Collateral assets are retained by the counterparty and ownership is only transferred to the NCB in case of counterparty default. In practice, assets pledged as collateral have to be stored at a safe custody account of the responsible

³¹ Besides refinancing operations, the Eurosystem provides liquidity via fine-tuning reverse operations, structural reverse operations, marginal lending facility and credits related to margin calls. However, these lending practices play only a minor role in the Eurosystem (see Figure 2.5). Moreover, NCBs can provide ELA to solvent yet illiquid counterparties that lack eligible collateral. The Eurosystem considers ELA as not being part of the single monetary policy. See Section 7.3.2 for a further discussion.

³² For further details on repos, see e.g. DUFFIE 1996 and GARBADE 2006. Repos and particularly their use and importance in the European interbank market is further investigated in Chapter 5 and esp. Section 5.2.

NCB at a clearing and depositary institution. To collateralize a loan, the counterparty can choose to use either the earmarking system in which each pledged asset is earmarked for the specific transaction or a pooling system in which pledged assets are consolidated. Collateral assets are valued in terms of an adjusted market value that needs to exceed provided liquidity over the entire period of the loan.³³ Adjustment of the market value is based on the application of a haircut, which serves as risk control to the Eurosystem as it reflects inter alia credit risk, liquidity and maturity of the security.³⁴ The NCB credits the adjusted market value to the collateral account of the counterparty, which can subsequently borrow to this extent. In case of counterparty default, the NCB can liquidate the collateral to settle its claims in terms of principal and interest as well as administrative cost. In case of correct risk mitigation, the liquidation value is adequate to settle the claims. Any higher proceeds belong to the counterparty while any residual remains outstanding if only part of the claim can be met.

Repos and collateralized loans are economically similar.³⁵ Repos de facto represent collateralized loans since the NCB temporarily receives collateral in return for provided liquidity. The main difference is that ownership of the collateral is transferred in case of repos, whereas the borrower retains ownership in collateralized loans. Each NCB decides individually whether reverse transactions are performed as repos or collateralized loans.³⁶ Figure 2.4 reveals how NCBs performed reverse transactions at the end of 2014. It shows that only five out of 19 NCBs accept repos with counterparties. The NCBs of Belgium, Italy, Luxembourg and Spain permit both repos and collateralized loans, while the NCB of Estonia uses repos exclusively. 18 NCBs perform refinancing operations as collateralized loans with 14 NCBs using exclusively this type of transaction. Hence, the vast majority of NCBs perform refinancing operations as collateralized loans, exposing the Eurosystem to potential tedious legal issues as collateral assets remain with the counterparty and have to be transferred in case of counterparty default.

The red area in Figure 2.2 indicates that the Eurosystem substantially extended liquidity provision via reverse transactions (in terms of repos and collateralized loans) after the onset of the financial crisis and throughout the European sovereign debt crisis. The substantial extension was rendered possible by changes to the following six determinants of credit terms, which are key to any loan or repo and are examined in the following:³⁷

1. maturity;	2. interest rate;	3. counterparty;		
4. amount;	5. collateral;	6. haircut.		

The Eurosystem already extended *maturities* prior to the onset of the financial crisis in the fall of 2008 (cf. Figure 2.1). In April 2008, the Eurosystem introduced supplementary longer-term refinancing operations (SLTROs) with a maturity of six months, i.e. doubling the regular

 ³³ Depending on whether earmarking or pooling is used, this has to be ensured for each individual transaction or a counterparty in total.
³⁴ See Chanter 4 for an extension discussion of side control of the Encounterpart in successful the condition of the second seco

³⁴ See Chapter 4 for an extensive discussion of risk control of the Eurosystem in general and the application of haircuts in specific.

³⁵ See RUCHIN 2011 for a comparison of repos and collateralized loans from a legal perspective. Additional differences between repos and collateralized loans are addressed in Section 5.2 when repos in the interbank market are compared to collateralized loans from the central bank.

³⁶ The decision usually depends on the market practice of the particular NCB and peculiarities of its national legislation.

 $^{^{37}\,}$ See also Adrian et al. 2013 for the determinants and Figure 2.1 for the amendments.



The figure reveals how NCBs perform reverse transactions. The decision lies with the NCBs to perform them as repurchase agreements and/or collateralized loans (ECB 2015b). The vast majority of NCBs apply collateralized loans only, whereas only four NCBs allow for both types and only Estonia favors repurchase agreements.



Source: author's compilation based on information provided to the author by National Central Banks.

maturity of LTROs.³⁸ Thereafter, the Eurosystem extended maturities successively, introducing 12- and 36-month terms for LTROs. The first 12-month LTRO tender was settled in June 2009, before two 36-month LTROs followed in December 2011 and February 2012. Moreover, the Eurosystem introduced additional targeted longer-term refinancing operations (TLTROs) with a maturity of up to four years and an early repayment option, the first of which was settled in September 2014. The aim of TLTROs is to support bank lending such that liquidity provision is conditional on counterparties' outstanding loans and net lending. The extension of maturities had two major effects, which are illustrated in Figure 2.5.³⁹ The figure illustrates the development of lending volumes to eligible counterparties with respect to maturity. First, the left panel reveals a shift from MRO- to LTRO-based lending, i.e. counterparties replaced short-term funding by longer-term funding.⁴⁰ Liquidity was predominantly granted via MROs prior to the financial crisis with about three quarters. However, the tide turned with the onset of the financial crisis. By the end of 2014, about 80% of lending took place via LTROs and in May 2011 almost exclusively via LTROs (about 96%). Second, the right panel zooms into the crises times, also revealing a shift within longer-term funding. The successive extension of maturities induced counterparties to draw liquidity via LTROs with the respective longest maturity. By the end of 2014, liquidity was almost solely provided via the three-year LTROs.

 $^{^{38}}$ The second SLTRO tender with a maturity of six months was settled in July 2008 and various followed in 2009.

³⁹ Note that three-year LTROs comprise TLTROs in Figure 2.5.

⁴⁰ "Other" comprises fine-tuning reverse operations, structural reverse operations, marginal lending facility and credits related to margin calls.

Hence, the extension of maturities had two effects: first, counterparties switched from MRO- to LTRO-based funding; and second, they drew liquidity from LTROs with the longest maturity.⁴¹



The figure illustrates the development and composition of loans and how they were affected by the extension of maturities. The left panel indicates that the extension induced a shift in funding from MROs to LTROs. The right panel reveals that this shift was intensified between 2007 and 2014 as counterparties drew liquidity via LTROs with the longest maturity.



Together with the extension of maturities, the Eurosystem substantially lowered the *interest rate* applied in lending, i.e. the main refinancing rate (MRR). Figure 2.6 details the development of the MRR from 1999 to 2014. There was large variation in the MRR, which was lowered successively after the financial crisis hit. While the end-of-month average MRR amounted to 3.1% between January 1999 and September 2008, it plummeted to 0.94% between October 2008 and December 2014. At the end of 2014, the interest rate reached a historical low of 0.05%. This development spurred the substantial increase in Eurosystem lending, as depicted in Figure 2.2.

The Eurosystem accounts for a wide range of *counterparties* eligible as the third determinant of credit terms. As explained above, eligible counterparties have to obey the minimum reserve requirement, be supervised by an EEA national authority and be considered financially sound. By the end of 2014, the ECB listed 8,296 monetary financial institutions in the European Union (EU), 5,555 (67%) of which are subject to and not exempt from the Eurosystem's minimum reserve requirement and hence are deemed eligible counterparties.

The *amount* was limited by variable-rate tenders without full allotment prior to the financial crisis. Since October 2008, the Eurosystem has performed liquidity provision via fixed-rate tenders with full allotment.⁴² The switch to fixed-rate tenders with full allotment was crucial for the substantial expansion of liquidity provision as it implies that the Eurosystem determines the interest rate and eligible counterparties can borrow without limit at the given rate of interest. Moreover, the Eurosystem markedly increased the number of refinancing operations since 2008,

⁴¹ Longer maturities of provided liquidity endorsed banks to alleviate maturity mismatch between the investment side and the funding side of their balance sheets (ECB 2010d).

⁴² The Europystem applied fixed-rate tenders (without full allotment) at the very beginning of the third stage of the European Monetary Union (EMU) (January 1999 to June 2000). In relation to Figure 2.6, this implies that the depicted interest rate corresponds to the fixed MRR from January 1999 to June 2000 and from October 2008 to December 2014, while it is the marginal MRR from June 2000 to October 2008. See also VERGOTE et al. 2010.

The figure depicts the development of the MRR between 1999 and 2014 on the right scale. The MRR was successively lowered after the onset of the financial crisis, amounting to a historical low of 0.05% at the end of 2014. Moreover, the figure details the number of monthly refinancing operations (left scale). While the Eurosystem conducted 5.5 operations on average per month during pre-crisis times, the average increased to 8.9 between October 2008 and December 2014.



as shown in Figure 2.6. While the Eurosystem performed 5.5 refinancing operations per month on average during pre-crisis times, the average increased to 8.9 operations between October 2008 and December 2014. In these operations, the Eurosystem granted refinancing credit to the amount of EUR 87.3 tn (pre-crisis) and EUR 50.7 tn (October 2008 to December 2014). This implies an increase in the average monthly provision of liquidity from EUR 746.5 bn to EUR 889.6 bn.⁴³ These developments facilitated the availability of liquidity and contributed to its substantial expansion, as is evident in Figure 2.2. With refinancing credit being fully allotted at a higher frequency, liquidity is de facto limited by eligible collateral only (see e.g. NYBORG and STREBULAEV 2001).

The Eurosystem took a multitude of actions to prevent any potential limitation of eligible collateral, i.e. to ensure *collateral* availability in the Eurozone and more particularly in certain crisis-stricken countries. The collateral framework and its development is described and analyzed in depth in Chapter 3. Moreover, the chapter investigates the development of eligible marketable assets (collateral pool) and how it was affected by amendments to collateral criteria. The use of eligible collateral is analyzed in Chapter 5. Amendments to collateral criteria kept sufficient eligible marketable assets available, as Figure 1.1 indicates.

The *haircut* is the sixth determinant of credit terms. Haircuts are closely connected to collateral in the Eurozone for two reasons. On the one hand, both collateral criteria and haircuts are laid down in the collateral framework of the Eurosystem. On the other hand, collateral and haircuts constitute risk mitigation in liquidity provision, which takes place in two steps: first, collateral

⁴³ However, the average number of bidders per transaction decreased from 375 to 154. The information is provided on the homepage of the ECB.

is taken as protection against the risk of counterparty default; and second, haircuts are applied to discount the value of collateral assets as hedge against collateral-related risk. Haircuts are identified as the key risk control measure of the Eurosystem in Chapter 4, where the general principles of haircut determination and the development of Eurosystem haircuts are elaborated in depth.

2.4 The Growing Importance of the Eurosystem Collateral Framework

Central bank collateral frameworks in general and the framework of the Eurosystem in specific has gained increasing importance over the last decade. Their basic function is to define the set of assets eligible for liquidity provision by central banks (in terms of both loans and asset purchases). Furthermore, collateral frameworks determine the quantity of liquidity that counterparties can draw from central banks in exchange for collateral assets by laying down the rules for asset valuation. According to NYBORG 2015, this places central bank collateral frameworks at the core of the monetary and financial system, and it renders them "one of the most complex and economically significant elements of monetary policy implementation" (BINDSEIL 2013, p. 5).

The observable growing *general importance* of central bank collateral frameworks is based on the two opposing effects of amendments to collateral criteria (see Chapter 1). Collateral is the primary means of risk mitigation in central bank liquidity provision. Hence, relaxation of collateral criteria implies a trade-off between ensuring liquidity access of counterparties, which facilitates the implementation of monetary policy and the assumption of additional credit risk (CHAILLOUX et al. 2008b). Central banks oppose potential collateral scarcity by relaxing collateral criteria to reshape the market perception of high-quality assets (LEVELS and CAPEL 2012, BIS 2013a). The importance of collateral criteria for market perception is reinforced by SINGH 2013, who defines the post-Lehman period as a "new collateral space" in which activity of additional market participants such as central banks influenced collateral availability. Collateral availability is particularly important when interest rates approach the zero-lower bound and central banks strive to extend liquidity provision. In this case, collateral availability is the de facto constraint to liquidity provision. However, extended liquidity provision is associated with credit risk, which is best mitigated by lending only against "good collateral".⁴⁴ Relaxing collateral criteria to free up additional yet lower-quality collateral amplifies credit risk. Furthermore, NYBORG 2015 emphasizes the growing importance of central bank collateral criteria as they shape conditions in the market for liquidity, which spill over to financial markets.

The *specific importance* of the Eurosystem collateral framework is revealed by its key role in the extension of liquidity provision in the Eurozone.⁴⁵ The General Documentation of the Eurosystem stipulates that any liquidity-providing OMO should be based on assets that are compliant with the collateral framework (ECB 2015b). Accordingly, it prescribes that only eligible marketable assets as specified by the collateral framework qualify for being purchased by

⁴⁴ "Good collateral" in terms of BAGEHOT 1873. See GOODHART 1999 and TUCKER 2009. For the role of collateral in central bank risk protection, see also STONE et al. 2011.

⁴⁵ NYBORG 2015 attributes an integral part of the Eurosystem's fight to save the euro to the collateral framework.

the Eurosystem, thus rendering the collateral framework the guiding principle for the numerous asset purchase programs.⁴⁶ The importance of the collateral framework for lending is grounded in the Statute of the ESCB and of the ECB, with Article 18.1 requiring that Eurosystem credit operations are based on adequate collateral (ECB 2012e and Chapter 1). Hence, the relaxation of collateral criteria has been crucial to the Eurosystem in terms of performing its standard and non-standard measures of monetary policy in recent years.⁴⁷ It served as a major means of crisis mitigation as freed-up assets were considered eligible for both asset purchases and collateralized loans. Moreover, it was key to the accumulation of intra-Eurosystem imbalances (see Section 7.3.3) and ELA, which is provided to illiquid yet solvent banks against collateral not complying with the uniform criteria (see Section 7.3.2).

Therefore, the importance of the Eurosystem collateral framework has been fostered over recent years when the Eurosystem relaxed it to substantially extend liquidity provision. The growing importance was initially disregarded,⁴⁸ although it has been more recently widely acknowledged and emphasized by e.g. SINN 2012, COUR-THIMANN and WINKLER 2013, EBERL and WEBER 2013, ECB 2013c, CASSOLA and KOULISCHER 2014, EBERL and WEBER 2014a,b, CLAEYS 2014, CLAEYS et al. 2014, DRECHSLER et al. 2015, WOLFF 2014, SINN 2014a,b, BELKE 2015, EBERL and WEBER 2015, ECB 2015c, FECHT et al. 2015 and SINN 2015b. The Eurosystem collateral framework is comprehensively addressed in the next chapter, as well as reflecting the central topic of the remainder of this dissertation.

⁴⁶ Note that this likely does not hold for government bond purchases under the ANFA for which purchase criteria thus far are confidential.

⁴⁷ For the evaluation of Eurosystem monetary policy with respect to standard and non-standard measures, see LENZA et al. 2010, ECB 2011e, COUR-THIMANN and WINKLER 2013 as well as CLAEYS 2014.

⁴⁸ Early pointers concerning the importance of the collateral framework can be found in e.g. ECB 2010d and SINN 2010b.

The Collateral Framework of the Eurosystem⁴⁹

This chapter provides an in-depth analysis of the Eurosystem collateral framework and its effects on the pool of eligible marketable assets ("collateral pool"). It compiles a narrative database of general principles of the collateral framework as well as changes to collateral criteria from 2001 to 2014 in general and by asset type, eliciting information from all relevant official documents released by the Eurosystem.⁵⁰ The chapter carves out two effects of the frequent amendments of the collateral framework. On the one hand, they made the collateral framework opaque, which contradicts the Eurosystem's claim for a "simple and transparent" collateral framework (cf. ECB 2007b, p. 87). On the other hand, they facilitated broadening the collateral pool both horizontally (i.e. the Eurosystem deemed eligible additional asset types of given credit quality) and vertically (i.e. the Eurosystem deemed eligible additional assets of lower credit quality). The chapter is structured as follows. Section 3.1 describes the Eurosystem collateral framework in terms of its emergence and general principles. Section 3.2 analyzes the development of collateral criteria concerning the evolution of general collateral criteria and criteria specific to asset types, *i.e.* debt instruments issued or guaranteed by governments, debt instruments traded on nonregulated markets, covered and uncovered bank bonds, ABSs as well as corporate bonds. Finally, Section 3.3 analyzes the effects of amendments to collateral criteria on the collateral pool along different dimensions, i.e. its geographical composition, its composition by asset type, credit quality, denomination and residual maturity. Moreover, the effects of extended eligibility of government quarantees and securities traded on non-regulated markets for the development of the collateral pool are scrutinized.

⁴⁹ Sections 3.1 and 3.2 base on EBERL and WEBER 2014a and Section 3.3 on EBERL and WEBER 2015.

⁵⁰ The majority of information originates from official documents of the Eurosystem. In particular, this includes official documents of the General and Temporary Framework as well as press releases. Additional information was provided to the author by the Eligible Assets Team in bilateral communication.

3.1 Collateral Criteria of the Eurosystem⁵¹

As discussed in Section 2.4, Article 18.1 of the Statute of the ESCB and the ECB allows the Eurosystem to operate in financial markets based on "adequate collateral". The Eurosystem ensures adequacy of collateral through its collateral framework. Accordingly, this section examines the framework in terms of its initial design, the "Single List" of eligible assets, the distinction between general and temporary collateral criteria and its general principles.

3.1.1 The Initial Collateral Framework: Tier 1 and Tier 2 Eligibility

The Eurosystem initially (i.e. in 2001) distinguished between two classes of eligible assets, namely tier 1 and tier 2 assets. Various types of marketable and non-marketable assets were initially deemed and made eligible over time within the two classes. Marketable assets are assets for which active markets exist and which may be liquidated in a short time, such as government bonds, bank bonds and corporate bonds.⁵² Hence, these assets typically feature a market value. By contrast, non-marketable assets are not traded on a regular market; rather, they are dealt in private transactions or held by the owner to maturity.⁵³ Consequently, they generally lack a market value. In particular, the type of asset, the type and residence of the issuer/debtor/guarantor, the place of issuance, the credit quality of the asset and its denomination are crucial elements for eligibility and thus they were subject to alteration.

The initial distinction between tier 1 and tier 2 assets was considered necessary as economic integration across the Eurozone was in its early stages and differences in financial structures remained substantial. Therefore, NCBs were given the right to consider eligible for tier 2 certain assets regarded as particularly important for national financial markets and banking systems. Uniform criteria were only applied to tier 1 assets, permitting two types of assets: debt certificates issued by the ECB or NCBs prior to the adoption of the euro and debt instruments issued or guaranteed by entities established in the EEA that were admitted to trading on regulated and non-regulated markets (see also Section 3.2.3).⁵⁴ Moreover, tier 1 assets had to meet "high credit standards" to ensure the financial soundness of the pledging counterparty (ECB 2000). These standards were not further specified at the beginning. Furthermore, debt instruments had to be denominated in euro and NCBs were prohibited to accept as collateral "own-use debt instruments", i.e. debt instruments pledged by the issuing counterparty itself. Collateral criteria for tier 2 assets were not uniform but rather established idiosyncratically by NCBs and subject

⁵¹ See e.g. ECB 2011f and DEUTSCHE BUNDESBANK 2015 for further details on the role of collateral in monetary policy of the Eurosystem.

⁵² At the end of 2014, the set of eligible marketable assets comprised ECB debt certificates, central government debt instruments, debt instruments issued by central banks, local and regional government debt instruments, supranational debt instruments, covered bank bonds, credit institutions debt instruments, debt instruments issued by corporate and other issuers and ABSs. The ECB publishes a daily list of eligible marketable assets on ISIN-basis.

⁵³ At the end of 2014, the set of eligible non-marketable assets comprised credit claims/bank loans, retail mortgage-backed debt instruments (RMBDs) and fixed-term deposits. Credit claims are defined as the debt obligations of a debtor vis-à-vis a Eurosystem counterparty. RMBDs are debt instruments (promissory notes or bills of exchange) that are secured by a pool of retail mortgages but fall short of full securitization. Only Irish mortgage-backed promissory notes have been labeled as RMBDs. Due to non-marketability, the Eurosystem does not provide information on the amount of eligible non-marketable assets.

⁵⁴ These entities included Eurosystem, public sector and private sector entities from EEA countries as well as international and supranational institutions.

to approval by the ECB Governing Council. Tier 2 assets had to be either debt instruments (marketable and non-marketable) or equities traded on regulated markets.⁵⁵ Tier 2 assets also had to be denominated in euro and were ineligible if they were own-use.

3.1.2 A Comprehensive Collateral Framework: the Single List

As economic and financial integration proceeded, the ECB repealed the possibility of the idiosyncratic acceptance of assets by NCBs and introduced the Single List as a comprehensive framework for eligible assets in January 2007.⁵⁶ The two-tier system was phased out such that tier 2 assets not qualifying for the Single List remained eligible until 31 May 2007.⁵⁷ The Single List has drawn a distinction between marketable and non-marketable assets. While marketable assets comprised former tier 1 assets, two types of assets were initially summarized as nonmarketables: credit claims, i.e. bank loans, and RMBDs. The ECB thereby harmonized the former idiosyncratic criteria for tier 2 assets, with the exception of credit claims for which certain idiosyncratic acceptance criteria remained in place.⁵⁸

3.1.3 Two Parallel Collateral Frameworks

At the end of 2014, collateral criteria were specified in two parallel frameworks established by the Eurosystem: (i) the General Framework and (ii) the Temporary Framework. Whenever assets comply with the eligibility criteria, NCBs are in principal obliged to accept assets as collateral. General collateral criteria in terms of the General Framework are defined in the General Documentation (see ECB 2015b). Furthermore, the Temporary Framework complements the General Framework and lays down additional temporary collateral criteria that were deemed necessary when financial markets entered turmoil and the full allotment of refinancing credit via fixed-rate tenders called for additional collateral. This Temporary Framework can be adapted to local needs upon the condition that certain minimum risk control requirements are met (see ECB 2015c). Hence, the Eurosystem brought the Temporary Framework into being in October 2008, which was initially intended to expire at the end of 2009. However, as liquidity demand

⁵⁵ Moreover, the ECB could authorize NCBs to include other assets, e.g. debt instruments issued by credit institutions not complying with Article 22.4 of Directive 88/220/EEC amending Directive 85/611/EEC (here-after referred to as "Undertakings for Collective Investment in Transferable Securities (UCITS) Directive"). According to the ECB, a credit institution is either (i) an undertaking whose business is to receive deposits or other repayable funds from the public and grant credit for its own account or (ii) an undertaking or any other legal person, other than those under (i), which issues means of payment in the form of electronic money. For the sake of simplicity, credit institutions are hereafter referred to as banks. Article 22.4 of the UCITS Directive specifies the following requirements: (i) the issuer of the bond must be a credit institution; (ii) the issuance has to be governed by a special legal framework; (iii) the issuing institutions must be subject to special prudential public supervision; (iv) the set of eligible assets to cover the bond must be defined by law; (v) the cover asset pool must provide sufficient collateral to cover bondholder claims throughout the whole lifetime of the covered bond; and (vi) bondholders must have priority claim on the cover asset pool in case of the issuer's default. Covered bonds complying with the requirements are considered particularly safe and will be denoted as "UCITS-compliant covered bank bonds" or simply as "covered bank bonds" in the following.

 $^{^{56}\,}$ See e.g. ECB 2006c for a description of the Single List.

⁵⁷ Units of French fonds communs de créancess (FCCs; French securization funds) that were previously eligible as tier 1 remained eligible until 31 December 2008.

⁵⁸ The Single List allowed for idiosyncratic criteria regarding the minimum threshold for the size of credit claims and charging a handling fee until December 2011. Thereafter, NCBs were allowed to idiosyncratically accept credit claims within the additional credit claims (ACCs) framework, see Section 3.2.1.2.

further grew, rather than returning to normal, it was maintained to complement the General Framework.

As a result, having initially been designed as a "crisis framework", established for a limited period and intended to provide additional collateral to facilitate the temporary extension of liquidity provision, the Temporary Framework turned into a long-lasting framework that remained in force at the end of 2014. Important collateral criteria that were initially envisaged to be temporary were assumed in the General Framework or in force without an expiration date.

3.1.4 General Principles of the Collateral Framework

Three general principles of the Eurosystem collateral framework are discussed hereafter: (i) close links between counterparties; (ii) principles for assessing the value of eligible assets; and (iii) the European Credit Assessment Framework (ECAF), according to which the credit quality of eligible assets is appraised.

3.1.4.1 Close Links

The ineligibility of assets with close links was already part of the initial General Framework in 2001. Accordingly, assets were deemed ineligible if issued by the counterparty itself or any other closely related entity. The most extreme case of close links is the own use of assets, whereby the asset is issued and pledged by the same party. Close links were initially defined according to Directive 2000/12/EC of the EC, focusing on links in participation or control (see EC 2000).⁵⁹ Four exceptions were made with respect to (*i*) close links between the pledging counterparty and public authorities of EEA countries; (*ii*) close links in trade bills, i.e. trade bills for which at least one entity (other than a credit institution) was liable in addition to the pledging counterparty; (*iii*) close links in UCITS-compliant covered bank bonds; and (*iv*) cases in which debt instruments were protected by specific legal safeguards comparable to (*iii*), which were not further specified.

The definition of close links and collateral criteria applied to assets were successively altered, as shown in Table $3.1.^{60}$ In May 2005, the Eurosystem deviated from the EC's general definition of close links, matching the definition to its collateral framework.⁶¹ The definition was slightly

⁵⁹ According to Directive 2000/12/EC, the Eurosystem defined close links as situations in which two or more entities were linked by (i) participation, which meant "the ownership, direct or by way of control, of 20% or more of the voting rights or capital of an undertaking", or (ii) control, which meant "the relationship between a parent undertaking and a subsidiary, in all the cases referred to in Article 1.1 and 1.2 of Directive 83/349/EEC, or a similar relationship between any natural or legal person and an undertaking; any subsidiary undertaking of a subsidiary undertaking shall also be considered a subsidiary of the parent undertaking which is at the head of those undertakings." Furthermore, a situation in which two or more entities were "permanently linked to one and the same person by a control relationship shall also be regarded as constituting a close link between such persons," see EC 2000 in conjunction with EEC 1983. The EC emphasizes that this definition only laid down minimum criteria for the provision of close links.

⁶⁰ The table also includes developments specific to asset types. For the description of these specific applications of close links, see the respective sections below.

⁶¹ As of May 2005, the Eurosystem defines close links as situations in which (i) "the counterparty owns 20% or more of the capital of the issuer, or one or more undertakings in which the counterparty owns the majority of the capital own 20% or more of the capital of the issuer, or the counterparty and one or more undertakings in which the counterparty owns the majority of the capital together own 20% or more of the capital of the issuer;" or (ii) "the issuer owns 20% or more of the capital of the counterparty, or one or more undertakings in

changed in January 2007 when the close links provision became not only relevant to issuers but also to debtors and guarantors. At the same time, trade bills with close links were deemed ineligible. The ECB made a significant change in February 2009, which is also addressed in Section 3.2.2 and Chapter 8. As government guarantees gained importance with the onset of the financial crisis, the Eurosystem deemed government-guaranteed debt instruments with close links eligible. Hence, marketable and non-marketable debt instruments with close links were deemed eligible (including own-use), provided that they were guaranteed by a government of an EEA country and complied with the general collateral criteria. Furthermore, RMBDs with close links likewise became eligible.

3.1.4.2 Valuation Principles

Valuation principles lay down rules concerning how to assess the value of assets pledged as collateral. These principles hold crucial importance since the Eurosystem applies its risk control measures based on this valuation.⁶² Valuation principles were already broadly specified in the initial collateral framework, before being successively altered over time. At the end of 2014, the Eurosystem assessed the value of a marketable asset based on either its market price or a theoretical value. First, the market price on the business day preceding the valuation date is considered. If more than one price is quoted, the lowest of these prices is used. In the absence of a price on the preceding business day, the last trading price is adduced. If the last available price is older than (or has not moved for at least) five days, the Eurosystem assigns a theoretical value to the asset. Valuation markdowns are applied when the value is not supported by a market price for covered and uncovered bank bonds as well as ABSs.⁶³ For non-marketable assets, a theoretical value or the outstanding amount is used.

Hence, the Eurosystem implemented cautious principles regarding the valuation of eligible assets, although potential valuation errors cannot be fully ruled out. Indeed, this holds for both marketable and non-marketable assets and is particularly relevant for own-use assets that are not traded. The absence of a market price requires theoretical valuation, which is prone to valuation errors. In case of overvalued assets, the liquidity provided would be too high. Furthermore, risk control measures (e.g. the haircut) applied to an overvalued asset would not fully hedge the underlying risk and in case of default the pledged security might be insufficient to cover outstanding claims.^{64,65}

which the issuer owns the majority of the capital own 20% or more of the capital of the counterparty, or the issuer and one or more undertakings in which the issuer owns the majority of the capital together own 20% or more of the capital of the counterparty;" or *(iii)* "a third party owns both the majority of the capital of the counterparty and the majority of the capital of the issuer, either directly or indirectly, through one or more undertakings in which that third party owns the majority of the capital," (see ECB 2005, p. 35).

⁶² See Chapter 4 for Eurosystem risk control.

⁶³ Valuation markdowns are deductions directly applied to the theoretical value prior to the application of further risk control measures, see Section 4.1.

 $^{^{64}}$ The haircut reflects the difference between the value that the Eurosystem assigns to an eligible asset and the amount of the loan, see Section 4.1.

⁶⁵ Imagine that the true value of an asset that a bank pledges with the Eurosystem is 80 but unknown to the Eurosystem. Moreover, the Eurosystem cannot consult the market price, e.g. because the asset is not traded. Due to the lack of information, the Eurosystem may overvalue the asset and assign a theoretical value of e.g. 100. For every haircut less than 20%, the Eurosystem would grant refinancing credit that is not entirely collateralized. For a haircut of e.g. 10%, the Eurosystem would be left with a loss of 10 in case of bank default.

Table 3.1: Changes to the collateral criteria of securities with close links

The table depicts the evolution of collateral criteria of securities with close links from 2001 to 2014. The most extreme case of close links is the own use of assets, whereby the asset is issued and pledged by the same party. Initial criteria were broadly defined and seldom changed prior to the onset of the financial crisis. Thereafter, amendments became more frequent and related to specific asset types (especially bank bonds).

Date	Action	Asset Type
01/01/2001	Ineligibility of debt instruments with close links (with exceptions); definition of close links according to Directive $2000/12/EC$ (ECB/2000/7)	all
30/05/2005	Modification of the definition of close links beyond Directive $2000/12/EC$ (ECB/2005/2)	all
01/01/2007	Ineligibility of trade bills with close links $(ECB/2006/12)$	trade bills
01/02/2009	Eligibility of government-guaranteed debt instruments and of RMBDs with close links; ineligibility of ABSs with close links within a currency hedge (ECB/2008/13)	all, RMBDs, ABSs
10/10/2010	Eligibility of residential real estate loan-backed structured covered bank bonds with close links $(ECB/2010/13)$	covered bank bonds (ABSs)
01/02/2011	Eligibility of commercial mortgage loan-backed structured covered bank bonds with close links $(ECB/2010/30)$	covered bank bonds (ABSs)
19/12/2011	Ineligibility of ABSs with close links within an interest rate hedge (ECB/2011/25, ECB/2012/11, ECB/2012/17, ECB/2012/18, ECB/2013/4)	ABSs
03/07/2012 то 02/05/2013	Limitation of use of government-guaranteed bank bonds with close links (ECB/2012/12, ECB/2011/25, ECB/2012/17, ECB/2012/18)	bank bonds
03/01/2013	Eligibility of further non-UCITS-compliant covered bank bonds with close links $(ECB/2012/25)$	covered bank bonds
03/05/2013 то 28/02/2015	Release of NCBs obligation to accept eligible government-guaranteed uncovered bank bonds with close links where the guarantor is a country under an EU/IMF program and whose credit assessment does not meet high credit standards; limitation of the use of government-guaranteed uncovered bank bonds with close links (ECB/2013/4)	uncovered bank bonds
01/03/2015	Ineligibility of government-guaranteed uncovered bank bonds with close links $(ECB/2013/6)$	uncovered bank bonds

Source: author's compilation.

3.1.4.3 The European Credit Assessment Framework (ECAF)

Since January 2007, the Eurosystem has pursued the objective that eligible assets comply with uniform credit rating standards by establishing the ECAF.⁶⁶ The ECAF was introduced to determine the creditworthiness of assets based on information provided by different credit assessment sources.⁶⁷ The Eurosystem stipulated in its initial General Framework that at least one credit assessment from an eligible ECAI for the issued security *(issue)* had to comply with "high credit standards". The ECB specified in January 2007 that in the absence of a rating for the issue, the creditworthiness of the *issuer* would be decisive. In case of a guaranteed issue, the creditworthiness of the *quarantor* was considered in third place. Thereby, the Eurosystem established a pecking order of credit ratings that takes into account the (1) issue, (2) issuer and (3)guarantor in determining eligibility. The respective next credit rating was only used when the precedent rating was not available. The Eurosystem defined down the pecking order in October 2013 when it decided that program ratings could be considered equal to issue ratings.⁶⁸ Moreover, credit ratings of the issuer or guarantor have been taken equally into account according to a first-best rule in the absence of an issue or program rating. Hence, the Eurosystem extended the first step and equated the second and third steps of the pecking order. A first-best rule has already been applied since January 2007, when issue ratings from eligible ECAIs differ.⁶⁹

The Eurosystem undertook a major step towards making ratings from different ECAIs comparable in January 2011 when it introduced its "harmonized rating scale". At the end of 2014, the Eurosystem took into account credit ratings from four rating agencies: Standard & Poor's (S&P), Moody's, Fitch and Dominion Bond Rating Service (DBRS). As ECAIs assess credit risk according to different rating scales, the harmonized rating scale aimed at standardizing the divergent scales.⁷⁰ The Eurosystem defined three credit quality steps (CQSs) for short- and long-term credit assessments, as illustrated in Table 3.2.⁷¹ Since October 2013, the Eurosystem has distinguished short- (i.e. assets with an original maturity of up to 390 days) from long-term assets (i.e. assets with an original maturity of more than 390 days). For the former, both short-

⁶⁶ Uniform credit rating standards are fundamental as risk control is performed based on the credit rating assigned to assets. See Chapter 4 as well as BRENDEL et al. 2015 and ECB 2015c.

⁶⁷ More specifically, the ECAF relies on four credit assessment sources: (i) External Credit Assessment Institutions (ECAIs), (ii) NCBs' in-house credit assessment systems, (iii) counterparties' internal ratings-based systems and (iv) third-party providers' rating tools. As ECAIs cover all eligible issuers/debtors/guarantors from EEA or non-EEA G10 countries, they play the most important role, particularly for the assessment of marketable assets (see ibid.). (ii) and (iv) cover only country-specific non-financial corporations, while the use of (iii) has to be approved by NCBs and is subject to performance monitoring. Hence, credit rating requirements are generally defined in terms of ECAIs ratings.

⁶⁸ See BRENDEL et al. 2015 for further information and observed obscurities in the implementation of the redefined pecking order by NCBs.

⁶⁹ For ABSs, the Eurosystem set up the second-best rule in October 2010, according to which the second-best available credit rating also had to comply with the minimum rating threshold for ABSs.

⁷⁰ The S&P long-term rating scale comprises 22 credit rating notches from AAA to D, while the scale of Moody's comprises 21 (Aaa to C), the scale of Fitch 20 (AAA to D) and DBRS differentiates between 26 notches (AAA to D).

⁷¹ Table 3.2 shows that the three CQSs are defined as follows: CQS 1 ranges from a long-term rating of "triple A" (Eurosystem's notation) equal to "Aaa" from Moody's and "AAA" from Fitch, S&P and DBRS to a rating of "Aa3" from Moody's and "AA-" from Fitch and S&P as well as "AAL" from DBRS. CQS 2 equals at least "single A" (Eurosystem), which means a rating of "A-" by Fitch or S&P, "A3" by Moody's or "AL" from DBRS. This is deemed equivalent to a probability of default (PD) over a one-year horizon of 0.1%. Finally, CQS 3 corresponds to a credit rating of at least "triple B" (Eurosystem), which is equal to "BBB-" from Fitch or S&P, "Baa3" from Moody's and "BBBL" from DBRS. CQS 3 is considered equivalent to a PD of 0.4% over a one-year horizon. For convenience, the notation of the Eurosystem is used in the following whenever possible.

and long-term ratings are considered on a first-best rule basis. For the latter, only long-term ratings are used.

Table 3.2: Harmonized rating scale of the Eurosystem

The table displays the harmonized rating scale according to which credit ratings of the four agencies are matched. Three ratings of DBRS are highlighted as they were belatedly shifted (Brendel et al. 2015). The long-term rating BBBL was introduced in CQS 3 in April 2014. The short-term rating R-1L was moved from CQS 1/2 to 3 and R-2L was introduced to CQS 3 in 2014. Hence, the Eurosystem relaxed collateral criteria by deeming eligible assets with a first-best rating of BBBL or R-2L from DBRS.

		Credit Quality Step (CQS)			
		1	2	3	
	DBRS		R-1H, R-1M	R-1L, R-2H, R-2M, R-2L	
Short-	Fitch		F1+, F1	F2	
Term	Moody's		P-1	P-2	
	S&P's		A-1+, A-1	A-2	
	DBRS	AAA to AAL	AH to AL	BBBH to BBBL	
Long-	Fitch	AAA to AA-	A+ to A-	BBB+ to BBB-	
TERM	Moody's	Aaa to Aa3	A1 to A3	Baa1 to Baa3	
	S&P's	AAA to AA-	A+ to A-	BBB+ to BBB-	

Source: author's illustration; European Central Bank; see also BRENDEL et al. 2015.

3.2 Development of Collateral Criteria of the Eurosystem

Having described the initial collateral framework and its general principles, this section elaborates on the development of the collateral framework. It starts by discussing general changes affecting the eligibility of marketable and non-marketable assets. This discussion is followed by a description of the amendments to collateral criteria of bonds issued or guaranteed by governments and debt instruments traded on non-regulated markets. Finally, changes to the collateral criteria of three asset types are investigated, namely bank bonds, ABSs and corporate bonds.

3.2.1 Changes to General Collateral Criteria

Modifications to the collateral criteria applied to marketable and non-marketable assets were manifold and are arranged in Figure 3.1. The timeline shows that the Eurosystem's activity in modifying general collateral criteria has amplified over time. In the first years after setting up its collateral framework (2001-2007), the Eurosystem made only a few modifications aimed at standardizing and harmonizing the framework. Idiosyncratic collateral criteria were scrapped and no amendments were made in response to market developments. Nonetheless, since the onset of the financial crisis in the fall of 2008, the Eurosystem has intensified amendments to the collateral framework to alleviate turmoil in financial markets.

Figure 3.1: Changes to general collateral criteria

The timeline depicts the amendments to general collateral criteria from 2001 to 2014. The Eurosystem made only few modifications aimed at standardizing and harmonizing the framework during the first years (2001-2007). During that time, idiosyncratic collateral criteria were abolished and no amendments were made in response to market developments. The Eurosystem intensified amendments since the onset of the financial crisis in the fall of 2008 in response to financial market turmoil.



3.2.1.1 Standardization and Harmonization of the Collateral Framework (2001-2007)

The Eurosystem made first amendments to the collateral framework in March 2004 when marketable assets were divided into four categories of decreasing liquidity, i.e. liquidity categories (LCs).⁷² Moreover, the nature of a guarantee for eligible tier 1 and tier 2 assets was further specified.⁷³

Euro-denominated debt instruments issued by entities established in countries that were part of the G10 but not the EEA, i.e. the US, Canada, Japan and Switzerland, became eligible as of May 2005. In contrast to this relaxation of collateral criteria, equities were deemed ineligible. Furthermore, the Eurosystem made first efforts towards the Single List (see Section 3.1.2) by announcing that it would phase out the two-tier system and replace it with a uniform framework of marketable and non-marketable assets. The Single List was subsequently introduced in January 2007 as the most substantial step towards standardizing and harmonizing the collateral framework.

3.2.1.2 Modification of the Collateral Framework in Response to the Crises (2008-2014)

The endeavor of the Eurosystem to create a coherent collateral framework was interrupted by the collapse of Lehman Brothers in September 2008. As a prompt response to turmoil in financial markets, the Eurosystem made one of the most fundamental changes to its collateral framework in October 2008 by lowering the minimum credit rating threshold for eligible assets (except for ABSs) from "single A" to "triple B", i.e. right above junk status. In order to capture additional risk involved in the acceptance of low-quality assets, the Eurosystem applied a uniform add-on haircut to assets with a rating lower than "single A", i.e. to assets later summarized in CQS 3. Whereas the lowering was initially intended to be temporary, it was prolonged by several legal acts until January 2011. Accordingly, it was adopted in the General Framework and hence it remains in force without an expiration date.⁷⁴

The Eurosystem initially collected fixed-term deposits for fine-tuning purposes only, i.e. to steer market liquidity. This became particularly relevant after the Eurosystem started asset purchases in July 2009 (see Section 2.2). Nevertheless, the Eurosystem already allowed counterparties to pledge fixed-term deposits as collateral as of October 2008. As eligibility was only temporarily valid, it had to be prolonged twice (in December 2008 and January 2010). Since January 2011, fixed-term deposits have been included in the General Framework as a third type of eligible non-marketable assets.

⁷² Central government debt instruments and debt instruments issued by central banks were subsumed in LC 1. Local and regional government debt instruments as well as jumbo Pfandbrief-style debt instruments, agency debt instruments and supranational debt instruments were summarized as LC 2. All traditional Pfandbrief-style debt instruments, credit institution debt instruments and debt instruments issued by corporate and other issuers were marked as LC 3, while all eligible ABSs were classified as LC 4.

⁷³ From then onwards, a guarantee has been accepted if the guarantor had unconditionally and irrevocably guaranteed the obligation with respect to the payment of principal, interest and any other amounts. The guarantee was to be payable on first demand but not necessarily in the case in which the guarantor was a government. Moreover, the obligations of the guarantor had to rank at least equally with unsecured obligations of the guarantor.

⁷⁴ The initial legal act came into force on 25 October 2008 and expired on 30 November 2008. The first prolongation lasted from 1 December 2008 until 31 December 2009, while the follow-up prolongation was valid between 1 January 2010 and 31 December 2010.

As another immediate response to the onset of the financial crisis, the Eurosystem deemed temporarily eligible guaranteed subordinated assets together with debt instruments denominated in pounds sterling, yen or US dollars as of November 2008. This was undertaken under the provision that the foreign currency debt instruments were issued and settled in the Eurozone and that the issuer was established in the EEA. Temporary acceptance was again repeatedly prolonged until December 2010.⁷⁵ The eligibility of debt instruments denominated in pounds sterling, yen or US dollars was reintroduced in November 2012 and remained in place at the end of 2014.

In February 2009, the Eurosystem undertook an amendment that appears minor upon first glance yet actually holds great significance in practice, whereby the list of accepted ECAIs was expanded to include DBRS as the fourth agency.⁷⁶ The inclusion of another ECAI should be advantageous as additional information can be taken into account in the assessment of collateral quality. However, as the first-best rule is applied to eligible assets other than ABSs in case of divergent credit ratings (see Section 3.1.4.3), each ECAI has the power to be pivotal, i.e. it exerts influence on whether assets are (i) eligible or (ii) subject to lower haircuts applied in CQS 1/2 or to higher haircuts in CQS 3. With each additional ECAI, the competition among rating agencies to provide the pivotal rating increases. Figure 3.2 depicts the development of long-term credit ratings of the governments of Ireland, Portugal, Italy and Spain, as assessed by the four ECAIs. DBRS temporarily provided the best credit rating to these countries and hence was pivotal.⁷⁷ For Italy (March 2013 to December 2014) and Spain (August 2012 to December 2014), DBRS was the only ECAI rating the governments' creditworthiness in CQS 1/2. From August 2011 to December 2014, DBRS was the only ECAI rating the government of Ireland in CQS 3. The minimum credit rating threshold was waived for bonds issued or guaranteed by the Portuguese government from July 2011 to August 2014 (see following section). Portugal intended to leave the EU/IMF program in early-2014 under the prerequisite that it was able to refinance in the market. The eligibility of Portuguese government bonds as collateral with the Eurosystem was key to this prerequisite but it was not ensured without suspension of the minimum rating threshold because DBRS provided the best rating of BBBL, although this did not comply with CQS 3 prior to April 2014 (cf. Table 3.2). After introducing BBBL to CQS 3 in April 2014, bonds issued or guaranteed by the Portuguese government remained eligible despite revoking the suspension of the minimum rating threshold in August 2014. The effects of these amendments to collateral criteria may have been particularly strong as for government-guaranteed bonds the rating of the guarantor was put to the second step of the pecking order (October 2013), according to which the decisive credit rating was determined (cf. Section 3.1.4.3).⁷⁸

 $^{^{75}}$ Prolongations were made in December 2008 and January 2010 until December 2010.

⁷⁶ DBRS itself states that it has been an accepted ECAI since 1 January 2008. However, it first appeared as accepted ECAI in a guideline released in October 2008, which entered into force on 1 February 2009 (cf. ECB 2008b). DBRS is a small Canadian agency compared to S&P, Moody's and Fitch, which account for a combined market share of about 95%. It had to close down all its European branches in the aftermath of the financial crisis.

⁷⁷ NYBORG 2015 confirms the important role of DBRS, documenting that the first-best rating of DBRS was worth around EUR 200 to 300 bn in increased collateral value. Banks likewise profited from generous credit ratings of DBRS, as described in BRENDEL et al. 2015.

⁷⁸ Moreover, the Eurosystem deemed eligible own-use government-guaranteed debt instruments since February 2009, which coincides with the acceptance of DBRS as ECAI.



The figure details the long-term local currency credit ratings of the governments of Ireland, Portugal, Italy and Spain, as assessed by the four ECAIs. DBRS temporarily provided the best credit rating to these countries. Accordingly, it was pivotal for eligibility of Portuguese government bonds and the assessment of Irish, Spanish and Italian government bonds in CQS 1/2.



Source: author's compilation; European Central Bank; S&P; Moody's; Fitch; DBRS.

The Eurosystem gave NCBs room to idiosyncratically adapt collateral criteria for credit claims to local needs under the so-called ACCs framework in December 2011.⁷⁹ NCBs were conferred the right to set up own collateral criteria and risk control measures for credit claims, which subsequently had to be approved by the ECB Council. However, potential losses on pledged credit claims under the ACCs framework were not mutualized but rather borne by the respective NCB.⁸⁰ Seven NCBs (Central Bank of Ireland, Banco de España, Banque de France, Banca d'Italia, Central Bank of Cyprus, Oesterreichische Nationalbank and Banco de Portugal) put forward such national collateral criteria. Moreover, as of January 2014, NCBs have been allowed to accept credit claims that are included in a pool of other credit claims or backed by real estate assets in case of exceptional circumstances. Further room for the idiosyncratic acceptance of collateral was conceded to NCBs in July 2014 when they were allowed to deem eligible short-

⁷⁹ See TAMURA and TABAKIS 2013 for a comparison of collateral criteria for credit claims applied by NCBs, as well as further information on the role of credit claims as collateral in recent years.

⁸⁰ See NAGEL 2012. According to Guideline ECB/2012/18, NCBs were only allowed to provide assistance to each other when bilaterally agreed and approved by the ECB Council.

term debt instruments issued by non-financial corporations that do not comply with general collateral criteria.

3.2.2 Changes to the Collateral Criteria of Government-Linked Securities

This section examines the amendments to collateral criteria of debt instruments issued or guaranteed by public entities (with the right to levy taxes, hereafter referred to as governments) depicted in Figure 3.3.⁸¹

Figure 3.3: Changes to the collateral criteria of government-linked securities

The figure chronologically arranges the amendments to collateral criteria of securities issued or guaranteed by governments. Most importantly, the Eurosystem suspended the minimum credit rating threshold for several crisis-stricken countries, i.e. for Greece (May 2010), Ireland (April 2011 to August 2014), Portugal (July 2011 to August 2014) and Cyprus (May 2013).



Although the minimum rating threshold was already reduced to "triple B" for governmentrelated assets in October 2008 (as for all assets except for ABSs), several countries continued to struggle to meet this threshold. In order to keep debt instruments issued or guaranteed by these countries eligible, the Eurosystem successively suspended the application of the minimum rating threshold to marketable debt instruments issued or guaranteed by these governments, i.e. for Greece (May 2010),⁸² Ireland (April 2011 to August 2014), Portugal (July 2011 to August 2014) and Cyprus (May 2013).⁸³ The Eurosystem extended the eligibility of own-use collateral to government-guaranteed assets in February 2009. In principle, this enabled banks to securitize assets into bonds, ask the government for guarantee and pledge them owing to the government guarantee.

⁸¹ See also Chapter 8 for the importance and fiscal implications of amendments to collateral criteria of governmentguaranteed bonds.

⁸² The suspension for Greece was repealed in February 2012, before being suspended again from March 2012 to July 2012. Reintroduced suspension was conditional upon the Greek government's ability to provide collateral enhancement in the form of a buy-back scheme. In December 2012, the minimum rating threshold was waived again without further condition.

⁸³ For the Cypriot government, the suspension was repealed in June 2013 and reintroduced in July 2013.

3.2.3 Changes to the Collateral Criteria of Securities Traded on Non-Regulated Markets

The requirement for marketable assets to be traded on accepted regulated and non-regulated markets was already contained in the initial collateral framework.⁸⁴ The Eurosystem successively modified its requirements, thereby increasing the number of accepted non-regulated markets over time.⁸⁵ Figure 3.4 illustrates the development of collateral criteria of assets traded on accepted non-regulated markets in due consideration of the market for Short-Term European Papers (STEPs, see below) as a controversial example for a non-regulated market.

Figure 3.4: Changes to the collateral criteria of securities traded on non-regulated markets

The figure illustrates the development of the collateral criteria of assets traded on non-regulated markets from 2001 to 2014. It reveals that the Eurosystem initially refined requirements for accepted non-regulated markets to standardize the collateral framework but deviated from this approach during the financial crisis. Most importantly, the Eurosystem deemed eligible uncovered bank bonds traded on non-regulated markets, e.g. the Short-Term European Paper (STEP) market.



The Eurosystem refined the requirements with which accepted non-regulated markets had to comply in May 2005 for the purpose of creating a comprehensive collateral framework. The Eurosystem referred to three principles according to which non-regulated markets were assessed, i.e. *(i)* safety, *(ii)* transparency and *(iii)* accessibility.⁸⁶ Moreover, the Eurosystem pointed out that it would not aim at assessing the intrinsic quality of each market.

The Eurosystem brought into force an amendment in November 2007 to expand the eligibility of assets traded on non-regulated markets when the first indication of distress in interbank markets appeared. It continued accepting marketable assets other than uncovered bank bonds that had been issued prior to May 2007 in certain non-regulated markets not fully complying with the three principles.⁸⁷ Specifically, these were non-regulated markets that fulfilled the requirements for safety and accessibility, although not for transparency.

As distress in interbank markets aggravated in the fall of 2008, bank bonds issued on accepted non-regulated markets were deemed eligible in October 2008, subject to an add-on haircut.⁸⁸

⁸⁴ For a market to be considered as regulated, it has to comply with criteria as defined according to the Investment Services Directive (93/22/EEC), while accepted non-regulated markets have to comply with certain principles defined by the Eurosystem (see Footnote 86).

⁸⁵ In January 2007, the list of accepted non-regulated markets comprised 18 markets from 11 countries. MTS Slovenia was accepted in February 2009 and the Irish Global Exchange Market in January 2013.

⁸⁶ According to Guideline ECB/2005/2, the principle of safety meant certainty regarding transactions, particularly concerning the validity and enforceability of transactions. Transparency was interpreted as unimpeded access to information on the market's rules of procedures and operation, the financial features of the assets, the price formation mechanism and the relevant prices and quantities. Finally, accessibility referred to a market's rules of procedures and operation, which allowed the Eurosystem to obtain information and conduct transactions when needed for these purposes.

 $^{^{87}\,}$ The amendment was prolonged in October 2009 and remained valid until December 2010.

⁸⁸ For an extensive discussion of the evolution of collateral criteria applied to bank bonds, see Section 3.2.4.

Eligibility was prolonged twice in December 2008 and January 2010, until December 2010. It was adopted in the General Framework in January 2012 and the add-on haircut was scrapped.

Whenever acceptance requirements for non-regulated markets are sufficiently strict, risk to the Eurosystem stemming from the eligibility of assets traded on such markets should not significantly differ from the risk of assets traded on regulated markets. However, two arguments can be adduced to challenge this view. First, the Eurosystem itself declared that it would not aim at assessing the intrinsic quality of each non-regulated market. Second, the application of the three principles laid down for acceptance of non-regulated markets may be questioned. The principle of *transparency* should guarantee "unimpeded access to information on market rules of procedures and operations, the financial features of the assets, the price formation mechanism, and the relevant prices and quantities" (ECB 2005, p. 35). This principle was not only repeatedly suspended (see above) but also laxly applied, as the following paragraph will discuss.

The STEP Market as Accepted Non-Regulated Market

A notable example of an accepted non-regulated market is the STEP market, which the Eurosystem included into its list of eligible non-regulated markets in April 2007.⁸⁹ For a STEP to be eligible, it had to comply with the collateral criteria for the specific asset type. Depending on the issuer, STEPs can be "uncovered bank bonds", "corporate bonds" or "other marketable assets". Although STEPs were principally eligible as collateral, STEPs issued by banks were excluded. However, the Eurosystem revoked this derogation in October 2008, as discussed above.

Figure 3.5 illustrates the development of the size of the STEP market. It reveals that a trebling of the nominal value of STEPs outstanding accompanied the announcement that the Eurosystem would accept the STEP market as a non-regulated market in September 2006. Another increase can be observed in October 2008, at the same time when STEPs issued by banks were deemed eligible. Subsequently, the value of STEPs outstanding persisted at a high level. Without claiming causal effects, the figure suggests that the concurrence of the market development and eligibility of STEPs increased the amount of eligible collateral.⁹⁰

Moreover, the STEP market indicates why the acceptance of non-regulated markets for collateral purposes could pose substantial risk to the Eurosystem (see also SINN 2012, 2014b). The STEP market is managed by a sub-organization of the European Banking Federation and supervised by the Banque de France. Detailed data on traded STEPs is provided by Euroclear France to the Banque de France, although it is not fully reported to the ECB. This lack of disclosure of information lies in contrast to the Eurosystem's principle of transparency.

3.2.4 Changes to the Collateral Criteria of Bank Bonds

STEPs issued by banks are just one specific type of bank bonds that the Eurosystem deems eligible as collateral. Generally, bank bonds are marketable assets and hence the general collateral

⁸⁹ See ECB, "First Publication of STEP Yield Statistics," Press Release, 2 April 2007, and ECB, "Assessment of STEP for Collateral Purposes in Eurosystem Credit Operations," Press Release, 15 September 2006.

⁹⁰ The importance of the acceptance of the STEP market for providing additional collateral is further addressed in Section 3.3.6.

Figure 3.5: The size of the Short-Term European Paper (STEP) market

The figure depicts the development of the size of the STEP market in terms of the nominal value of STEPs outstanding from 2006 to 2014. It indicates that both the announcement of the STEP market as an accepted non-regulated market and the eligibility of STEPs issued by banks coincided with increases in the nominal value of outstanding STEPs.



criteria as described in Section 3.2.1 apply. In addition, specific eligibility criteria were established for bank bonds, the development of which is discussed in the following and summarized in Figure 3.6. One can differentiate between covered and uncovered bank bonds. For covered bank bonds, an important distinction is made regarding the compliance of the bonds with the UCITS Directive, see Footnote 55. Moreover, both covered and uncovered bank bonds can imply close links or even be own-use.

3.2.4.1 Initial Collateral Criteria of Bank Bonds

Initially, covered bank bonds had to comply with the UCITS Directive to be eligible within tier 1, where they were classified into either LC 2 or 3. While "Jumbo Pfandbrief-style debt instruments" were included in LC 2,⁹¹ "traditional Pfandbrief-style debt instruments" and all other eligible covered bank bonds were assorted in LC 3. Non-UCITS-compliant covered as well as uncovered bank bonds were only eligible provided that an NCB included them in its tier 2 list. Owing to a similar structure, covered bank bonds were initially considered equivalent to ABSs, but not vice versa.⁹²

⁹¹ "Jumbo Pfandbrief-style instruments" were defined as debt instruments with an issuing volume of at least EUR 500 mn, for which at least two market makers provided regular bid and ask prices. All other Pfandbrief-style debt instruments were labeled as "traditional".

⁹² The major difference between covered bonds and ABSs is that covered bonds usually remain on the issuer's balance sheet and are typically more regulated, while ABSs are usually off-balance sheet and generally tranched.

OF

THE

Figure 3.6: Changes to the collateral criteria of bank bonds

The figure illustrates the development of the collateral criteria of covered and uncovered bank bonds from 2001 to 2014. It reveals that there were numerous changes to the eligibility of bank bonds, particularly following the onset of the financial crisis. Accordingly, bank bonds were the asset type subject to the most amendments to collateral criteria.



3.2.4.2 Amendments to Collateral Criteria of Bank Bonds

The Eurosystem deviated from the equivalent treatment of covered bank bonds and ABSs as of May 2006 by clarifying additional specific eligibility criteria were to be met for ABSs (see Section 3.2.5).⁹³ Moreover, UCITS-compliant covered bank bonds did not have to meet the minimum credit rating threshold from January 2007 to December 2007. Uncovered bank bonds ("debt instruments other than covered bank bonds that are issued by credit institutions", ECB 2006a) were deemed eligible with the introduction of the Single List in January 2007, provided that they were admitted to trading on a regulated market.

Amendments to the liquidity classifications of bank bonds were performed in February 2009 and January 2011.⁹⁴ In February 2009, "jumbo covered bank bonds" were marked as LC 2,⁹⁵ "traditional covered bank bonds" as LC 3, and all uncovered bank bonds ("credit institution debt instruments, unsecured") were reclassified from LC 3 into LC 4, just like "unsecured debt instruments issued by financial corporations other than banks" in January 2011. Furthermore, all non-UCITS-compliant covered bank bonds were explicitly included into LC 3.

The Eurosystem had to cope with the potential perils stemming from the eligibility of bank bonds guaranteed by crisis-stricken countries irrespective of credit rating (see Section 3.2.2) when the fiscal positions of European governments deteriorated. As of March 2012, NCBs were thus exempted from their obligation to accept eligible *bank bonds* guaranteed by countries under an EU/IMF program *or* by countries whose credit assessment did not comply with the minimum credit rating.⁹⁶ This attempt to mitigate risk was alleviated in May 2013 when the exemption was narrowed to *uncovered* bank bonds *with close links* guaranteed by countries under an EU/IMF program *and* whose credit assessments did not meet the minimum credit rating.⁹⁷

In the same line of risk precaution, the pledge of uncovered bank bonds has been limited as of March 2009. Since then, uncovered bank bonds of the same issuer group (i.e. identical issuer or closely linked issuers) could only be pledged to the extent that the haircut-adjusted value of these bonds would not exceed 10% of the total value of the collateral submitted by the counterparty. The limitation was further tightened to 5% in January 2012 and it was generalized to all unsecured debt instruments (i.e. issued by credit institutions and also by any other institution) in October 2010. Hence, excessive exposure to one counterparty and its affiliates was counteracted and the potential mutual provision of collateral between two banks

⁹³ However, the Eurosystem did not explicitly distinguish between covered bonds and ABSs before January 2013, when it emphasized that "for the purpose of the Eurosystem legal framework related to monetary policy, covered bonds are not considered ABSs" (ECB 2012d, p. 32).

⁹⁴ These amendments are important as the Eurosystem performs risk control based on liquidity classifications, i.e. it applies haircuts according to liquidity classifications.

⁹⁵ The requirements for a debt instrument to be classified as "Jumbo covered bank bond" (previously: "Jumbo Pfandbrief-style debt instrument") were modified whereby the issuing volume was increased from EUR 500 mn to at least EUR 1 bn and at least three market makers (instead of two) had to provide regular bid and ask quotes, see also Footnote 91.

⁹⁶ For instance, the Deutsche Bundesbank decided to refuse such uncovered bank bonds from Greece, Portugal and Ireland as of May 2012, according to NAGEL 2012.

 $^{^{97}}$ As of March 2015, uncovered bank bonds with close links are ineligible such that these provisions lapse.

was limited. However, government-guaranteed uncovered bank bonds were excluded from any limitation. 98

The Eurosystem restricted the pledge of covered bank bonds in March 2013 by introducing additional collateral criteria. Since then, the Eurosystem has prohibited the pool of assets underlying covered bank bonds to contain ABSs. Exemptions were made with respect to several specific types of ABSs.⁹⁹ Moreover, the restrictions did not apply to covered bank bonds that had been on the list of eligible ABSs by November 2012. These bonds remained eligible until November 2014.

3.2.4.3 Eligibility of Bank Bonds with Close Links

The general eligibility of marketable assets with close links has already been addressed in Section 3.1.4.1. The general collateral criteria were also applied to bank bonds with close links. Moreover, the Eurosystem applied specific criteria to bank bonds with close links, the development of which is captured in Table 3.1 and discussed hereafter.¹⁰⁰

Initially, only UCITS-compliant covered bank bonds with close links were deemed eligible. In October 2010, eligibility was extended to certain non-UCITS-compliant covered bank bonds, i.e. residential real estate loan-backed structured covered bank bonds. The eligibility of non-UCITS-compliant covered bank bonds was extended to those backed by commercial mortgage loans in February 2011. A more general rule for the eligibility of bank bonds with close links was established in January 2013, referring to any "covered bank bonds for which all criteria set out in Part 1, Points 68 to 70 of Annex VI to Directive 2006/48/EC are complied with, except for the limits on guaranteed loans in the cover pool" (ECB 2012d, p. 34).

Government-guaranteed bank bonds with close links became eligible in February 2009 (see Section 3.2.2). Such bonds could be substantially pledged as no limitation was in place prior to July 2012.¹⁰¹ In July 2012, the Eurosystem limited the pledge of such bonds to the nominal value of bonds submitted by 3 July 2012 (i.e. the day when the relevant guideline entered into force). However, the Eurosystem defined this restrictive measure down by making deviations from the limitation possible subject to Council approval. Moreover, the Eurosystem narrowed the limitation to government-guaranteed *uncovered* bank bonds with close links in May 2013. As of March 2015, the Eurosystem no longer accepts as collateral government-guaranteed uncovered bank bonds with close links as well as covered bonds with such bank bonds in the underlying pool.

⁹⁸ Moreover, the limitation was waived for uncovered bank bonds with a haircut-adjusted total value of less than EUR 50 mn and for bonds already submitted as collateral prior to 20 January 2009. The latter exception was valid until 1 March 2010.

⁹⁹ Specifically, the following ABSs remained allowed in the underlying pool of covered bank bonds: (i) ABSs that complied with the requirements laid down in Directives 2006/48/EC and 2006/49/EC; (ii) ABSs that were originated by a member of the same consolidated group of which the issuer of the covered bonds is also a member; or (iii) ABSs that are used as a technical tool to transfer mortgages or guaranteed real estate loans from the originating entity into the cover pool. See ECB/2012/25.

¹⁰⁰ As the Eurosystem did not explicitly distinguish between covered bank bonds and ABSs for collateral purposes prior to January 2013, changes to the application of close links to covered bank bonds are also indicated as changes to ABSs (but not vice versa) until that date.

 $^{^{101}\,\}mathrm{See}$ Chapter 8 and SINN 2014b.

3.2.5 Changes to the Collateral Criteria of Asset-Backed Securities

Like covered bank bonds, ABSs are a possible way for banks to free up the balance sheet, although banks have preferred the former over the latter for two reasons: first, most covered bank bonds can be used as collateral even with close links; and second, lower haircuts were applied to covered bank bonds than to ABSs (see Section 4.2.2). It was discussed in the previous section that the Eurosystem did not explicitly differentiate between ABSs and covered bank bonds prior to January 2013.¹⁰² Therefore, it is presumed that collateral criteria specified for covered bank bonds were also applied to ABSs prior to May 2006, when the Eurosystem first set out ABSs-specific collateral criteria. These ABSs-specific criteria are discussed in the following and summarized in Figure 3.7.

Figure 3.7: Changes to the collateral criteria of asset-backed securities (ABSs)

The figure illustrates the development of collateral criteria of ABSs from 2001 to 2014, marking a story of both tightening and relaxation. During a first period between 2006 and 2011, the Eurosystem limited the eligibility of ABSs but relaxed requirements thereafter (2011 to 2012). Since January 2013, a tight accompanying monitoring of underlying assets has been introduced.



The development of ABSs-specific criteria marks a story of both tightening and relaxation. During a first period between 2006 and 2011, the Eurosystem limited the eligibility of ABSs but relaxed criteria thereafter (2011 to 2012). Since January 2013, it has started to introduce a tight accompanying monitoring of assets underlying ABSs.

3.2.5.1 Tightening Collateral Criteria of ABSs (2006-2011)

ABSs were excluded from the requirement for eligible assets to have a fixed and unconditional principal amount introduced in May 2006. However, specific requirements were laid down for assets underlying ABSs, i.e. they had to be legally acquired in accordance with the laws of an EU member state and be a "true sale" that was enforceable against any third party and beyond

¹⁰² ABSs were not explicitly included in the initial collateral framework. They were first mentioned in March 2004 and classified into LC 4, i.e. the lowest category possible at that time. The classification was irrespective of the issuer or credit rating.

the reach of the originator and its creditors.¹⁰³ Furthermore, two types of ABSs were deemed ineligible: (i) ABSs comprising credit-linked notes; and (ii) ABSs issued by entities established in non-EEA G10 countries, i.e. the US, Canada, Japan and Switzerland. ABSs not complying with these criteria remained eligible until October 2006.

The Eurosystem explicitly excluded ABSs from the lowering of the minimum credit rating threshold for marketable and non-marketable assets in October 2008. Soon afterwards, the minimum credit rating requirement for ABSs was even tightened in March 2009. The Eurosystem defined its high credit standards for ABSs as "single A" over the *lifetime* of the ABSs and the highest possible credit rating of "triple A" had to be obtained at *issuance*. At the same time, the Eurosystem further restricted the criteria for underlying assets. Multiple securitization was foreclosed by the prevention that underlying assets could comprise tranches of other ABSs. ABSs that did not meet the restrictions but were issued prior to 1 March 2009 remained eligible for another year. ABSs were shifted to (the lowest) LC 5 when uncovered bank bonds were introduced into LC 4 in February 2009. As of October 2010, the Eurosystem required at least two ECAIs-credit ratings to meet the minimum credit rating thresholds for ABSs. In case of distinct ratings, the "second-best rule" was applied.¹⁰⁴ The application of this second-best rule lies in contrast to the treatment of all other assets for which the less strict first-best rule was applied, see Section 3.1.4.3. With respect to underlying assets, the Eurosystem also demanded that ABSs should not comprise swaps, other derivatives or synthetic securities.¹⁰⁵

3.2.5.2 Relaxing Collateral Criteria of ABSs (2011-2013)

The Eurosystem changed course in December 2011, abandoning its restrictive collateral policy with respect to ABSs. It lowered the minimum credit rating threshold for specific ABSs with only one sort of underlying assets, i.e. homogeneous ABSs.¹⁰⁶ For ABSs backed by loans to small and medium enterprises (SMEs) or residential mortgages only, the minimum rating threshold at issuance was lowered from a second-best rating of "triple A" to "single A".¹⁰⁷ Such ABSs were considered eligible provided that they fulfilled additional criteria.¹⁰⁸

The lowering of the rating threshold was intended to be sustained until September 2012. Nevertheless, it was further lowered in June 2012 to a second-best rating of "triple B" both at issuance

¹⁰³ "True sale" means that the underlying assets are transferred by the seller to a special purpose vehicle (SPV), which becomes entitled to the cash flows generated by the underlying assets (including those resulting from a sale of the assets).

¹⁰⁴ Depending on the date of issuance, the Eurosystem implemented the following regimes. For ABSs issued prior to 1 March 2010 with only one credit assessment, an additional assessment had to be obtained before 1 March 2011. For ABSs issued before 1 March 2009, both credit assessments had to meet a rating of at least "single A" over lifetime. For ABSs issued between 1 March 2009 and 28 February 2010, the first-best credit assessment had to comply with "triple A" at issuance and "single A" over lifetime and the second-best assessment with "single A" both at issuance and over lifetime.

¹⁰⁵ This requirement did not prevail for swaps used in ABSs transactions strictly for the purpose of hedging. ABSs that did not comply with this requirement but had been eligible before October 2010 were deemed eligible for another year.

¹⁰⁶ The Eurosystem thereby made a move towards accepting only ABSs considered as "plain vanilla", i.e. ABSs made from a single pool of underlying assets.

 $^{^{107}\,\}mathrm{The}$ minimum rating requirement over lifetime was kept at "single A".

¹⁰⁸ First, the counterparty pledging the ABSs or any third party with close links was not allowed to provide an interest rate hedge. Second, the underlying assets were not allowed to contain loans that were non-performing at issuance nor structured, syndicated or leveraged at any time.

and over lifetime. Moreover, this lowering was not made exclusive for ABSs backed by loans to SMEs or residential mortgages only, but it also applied to homogeneous ABSs from commercial mortgages, auto loans as well as leasing and consumer finance and heterogeneous ABSs comprising loans to SMEs and residential mortgages.¹⁰⁹ Since July 2014, credit card receivables have been added to the list of assets underlying homogeneous eligible ABSs.

3.2.5.3 Tightening the Monitoring of Eligible ABSs (as of January 2013)

Since January 2013, ABSs have occupied an exceptional role within marketable assets as an explicit differentiation has been made between "common eligibility requirements" and "additional eligibility criteria applicable to ABSs". The "loan-level reporting initiative" was gradually introduced, aimed at making ABSs in general and the underlying assets in particular more transparent. Accordingly, the Eurosystem demanded comprehensive and standardized loan-level data on the pool of underlying assets.¹¹⁰ ABSs must be backed by homogeneous assets; otherwise, the underlying assets cannot be reported in accordance with the loan-level reporting requirement.¹¹¹ Therefore, heterogeneous ABSs have been deemed ineligible as of January 2013 and any other ABSs not complying with the loan-level reporting requirement as of April 2014.

3.2.5.4 Collateral Criteria of ABSs with Close Links

For ABSs with close links, asset-specific collateral criteria beyond general criteria discussed in Section 3.1.4.1 were put into force (see also Table 3.1). As the Eurosystem did not explicitly differentiate between covered bank bonds and ABSs prior to January 2013, it is presumed that the development of the eligibility of covered bank bonds with close links (Section 3.2.4.3) also touches upon the eligibility of ABSs with close links.

3.2.6 Changes to the Collateral Criteria of Corporate Bonds

Corporate bonds were labeled as "debt instruments issued by corporate and other issuers" (ECB 2003),¹¹² and they have always been eligible for collateral purposes provided that they comply with the general collateral criteria for marketable assets. Thus, corporate bonds were also subject to the general changes of collateral criteria applied to marketable assets extensively discussed in Section 3.2.1, without specific provisions having been made.

¹⁰⁹ At first, this lowering was temporarily valid until September 2012, before being prolonged in September 2012 until March 2013. In March 2013, it was included in the General Framework.

¹¹⁰ Loan-level reporting requirements were introduced for residential mortgage-backed securities (RMBSs) as well as ABSs backed by loans to SMEs on 1 January 2013. Since 1 March 2013, the requirements have also been imposed to commercial mortgage-backed securities (CMBSs). For consumer finance ABSs, leasing ABSs and auto loan ABSs, the requirements became obligatory as of 1 January 2014 and for credit card ABSs by 1 April 2014.

¹¹¹ The underlying assets were to be reported using a single template for the specific asset type to meet the loan-level requirement, which could not be fulfilled for heterogeneous ABSs.

¹¹² The label was changed to "debt instruments issued by non-financial corporations and other issuers" (ECB 2010b) in January 2011.

3.2.7 Preliminary Summary

Thus far, this chapter has aimed to provide an in-depth analysis of the Eurosystem collateral framework and its development in terms of a narrative database. Table 3.3 summarizes the most important amendments to the collateral criteria performed by the Eurosystem between 2001 and 2014, indicating the amendments as tightening or relaxing the collateral criteria accordingly.

Table 3.3: Summary and classification of major amendments to collateral criteria

The table consolidates the narrative database of the Eurosystem collateral framework by pointing out and classifying the most significant amendments. The table conveys the overall impression that (i) the Eurosystem intensified amendments to the collateral framework throughout recent years of crisis, and that (ii) activity was predominantly directed at relaxing collateral criteria.

DATE	AMENDMENT	CLASSIFICATION	
DAIE	AMENDMENT	tightening	relaxing
30/05/2005	ineligibility of equities	\checkmark	
01/01/2007	abolition of idiosyncratic collateral criteria (introduction of Single List)	\checkmark	
15/09/2008	collapse of Lehman Brothers		
25/10/2008	lowering of minimum credit rating for all assets except for ABSs from "single A" to "triple B";		\checkmark
	eligibility of bank bonds traded on STEP market		\checkmark
14/11/2008	eligibility of marketable debt instruments issued in pounds sterling, yen or US dollars		\checkmark
01/02/2009	eligibility of own-use government-guaranteed debt instru- ments;		\checkmark
	DBRS accepted as fourth ECAI		\checkmark
01/03/2009	increase of minimum credit rating for ABSs from "single A" to "triple A" at issuance	\checkmark	
05/2010	intensification of European sovereign debt crisis		
$\begin{array}{r} \hline 06/05/2010,\\ 01/04/2011,\\ 07/07/2011,\\ 03/05/2013,\\ 09/05/2013 \end{array}$	suspension of minimum credit rating for debt instruments issued or guaranteed by the governments of Greece, Ireland, Portugal; later by governments under an EU/IMF program and Cyprus		\checkmark
	idiosyncratic acceptance of credit claims by NCBs;		\checkmark
19/12/2011	lowering of minimum credit rating for specific ABSs from "triple A" to "single A" at issuance		\checkmark
29/07/2012	lowering of minimum credit rating for all ABSs from "sin- gle A" to "triple B" at issuance and over lifetime		\checkmark
03/01/2013	ineligibility of heterogeneous ABSs	✓	
01/04/2014	ineligibility of ABSs not complying with the loan-level reporting requirement	\checkmark	

Source: author's compilation.

The table conveys the overall impression that (i) the Eurosystem intensified amendments to the collateral framework in response to the financial crisis, and that (ii) activity was predominantly directed at relaxing collateral criteria. This impression arises from a qualitative consideration and is reinforced by the quantitative dimension reflected in Figure 3.8.¹¹³ The figure reveals how amendments to collateral criteria scatter over time. While changes were scarce prior to the financial crisis, the Eurosystem ramped up activity as of 2007, whereby the number of changes

¹¹³ Moreover, the Eurosystem released only six official documents affecting collateral criteria prior to the financial crisis. From the onset of the crisis in 2008 to the end of 2014, this number increased to 44.

increased from 25 prior to 74 after the Lehman event. While amendments were rather general as a first response to the financial crisis, the Eurosystem refined changes over the years towards asset type-specific criteria. In Figure 3.8, the changes are classified as either tightening or relaxing measures. Changes that cannot be classified accordingly are neglected. The bars sum up changes in either direction differentiated by asset type. Relaxing measures prevail in response to the financial crisis (particularly in 2008) and throughout the European sovereign debt crisis. Overall, 48 relaxing and 27 tightening amendments can be identified since the introduction of the collateral framework in 2001.

Figure 3.8: Temporal distribution of classified amendments to collateral criteria

The figure reveals the chronological distribution of changes to collateral criteria from 2001 to 2014 according to the author's compilation. The left panel differentiates between changes applying to all assets ("general") and changes to specific asset types ("asset type-specific"). The Eurosystem ramped up activity as of 2007 and refined activity over the years towards amending asset type-specific criteria. The right panel shows the author's classification of the changes as either tightening or relaxing. The panel indicates that relaxing changes prevail during both the financial crisis and the European sovereign debt crisis.



The narrative database of changes to the Eurosystem collateral framework presented in this chapter thus far indicates that the Eurosystem broadened the collateral pool both *horizontally* and *vertically*.¹¹⁴ Horizontally, it broadened the collateral pool *quantitatively*, i.e. it deemed eligible additional asset types of given credit quality. Vertically, the Eurosystem broadened the pool *qualitatively* by lowering the minimum credit rating threshold for given types of eligible asset types. The next section elaborates on the broadening along both dimensions by analyzing the development of the size and composition of the collateral pool.

3.3 Development of the Pool of Eligible Marketable Assets (Collateral Pool)

Information published by the ECB on the development of the collateral pool is depicted in Figure 3.9. The ECB only provides historical data in aggregated terms, i.e. data on eligible marketable assets by asset type, only yearly from 2004 to 2011 and quarterly since 2012.¹¹⁵ No record is available for eligible non-marketable assets as these are usually traded over the

¹¹⁴ CHAILLOUX et al. 2008b make an akin differentiation but refer to the horizontal dimension as the currencies that are accepted.

¹¹⁵ For the sake of comparability, Figure 3.9 uses yearly data only.

counter.¹¹⁶ Therefore, Figure 3.9 shows the total nominal value of eligible marketable assets by asset type and reveals that the collateral pool changed in two ways: the pool substantially grew from 2004 to 2014 (by 82.5%), while its composition varied. The only year in which the pool contracted relative to the previous year was 2011, which can be attributed to the expiration of two major relaxations of the collateral framework: (i) the eligibility of bank bonds traded on non-regulated markets; and (ii) the eligibility of assets denominated in certain foreign currencies. The two amendments were reintroduced throughout 2012 such that the collateral pool recovered thereafter.

Figure 3.9: Published information on eligible marketable assets (Eurosystem collateral pool)

The figures shows the development of the Eurosystem collateral pool by asset type from 2004 to 2014, based on data published by the ECB. It reveals that the collateral pool substantially increased during that time. Owing to the availability of highly-aggregated data only, the lessons from the figure are limited.



Source: author's illustration; European Central Bank.

The lessons from published data given in Figure 3.9 are limited for two reasons: (i) data is aggregated for asset types such that asset-specific information is not available; and (ii) data is averaged over end-of-month stocks at the end of each year/quarter such that time-specific information is not available. The remainder of this section aims to reveal new insights into the effects of amendments to collateral criteria on the development of the collateral pool by investigating both asset- and time-specific information. Therefore, it refines information provided in Figure 3.9 for the major crises period from May 2007 to December 2013. This refinement is based on historical data from the Eligible Assets Database provided to the author by the Deutsche Bundesbank. The Eligible Assets Database is the list of all eligible marketable assets on ISIN-basis and hence it may comprise up to 50,000 unique marketable assets.¹¹⁷ The ECB has published the list daily since April 2010. As the list is only published for the day, no historical record is available from the ECB. Based on end-of-month lists of eligible marketable

 $^{^{116}}$ Hence, the following analysis is limited to the extent that only eligible marketable assets are investigated.

¹¹⁷ It comprised 25,348 marketable assets with a nominal value of EUR 9.9 tn in May 2007, 51,374 assets (EUR 11.6 tn) in November 2008 and 36,930 assets (EUR 13.9 tn) in December 2013.

assets from May 2007 to December 2013, Figure 3.9 is refined and extended in the following, providing new insights into the development of the collateral pool along different dimensions.¹¹⁸ The analysis of the monthly stock of eligible marketable assets (collateral pool) is complemented by the investigation of the monthly flow of eligible marketable assets (newly eligible marketable assets). While the analysis of the collateral pool contributes to obtain holistic insights into the development of eligible marketable assets along various dimensions, the investigation of newly eligible assets better fits for tracking the effects of amendments to the collateral framework on eligible assets.

3.3.1 Refinement of the Collateral Pool Development

Figure 3.10 refines Figure 3.9 by revealing the development of the collateral pool by country (left panel) and asset type (right panel) based on monthly data from May 2007 to December 2013.¹¹⁹ The right panel confirms the observations from Figure 3.9 that the collateral pool substantially increased from 2007 to 2013 and that its composition also changed. This can be attributed to the numerous changes to general collateral criteria that affected all asset types. However, the value of eligible ABSs decreased over time—an observation that can be referred to both market developments and the stricter collateral criteria of ABSs. Furthermore, the eligible amount of uncovered bank bonds increased, particularly after the onset of the financial criteris, which triggered changes to collateral criteria, e.g. with respect to bank bonds traded on accepted non-regulated markets. The left panel illustrates that the geographical distribution of eligible assets changed over time, e.g. while the nominal value of eligible assets in Germany remained relatively constant, it increased in France and Italy. Although amendments to the collateral framework were broadly defined and affected all asset types, their effects obviously differed across countries, which led to asymmetrical geographical developments.

3.3.2 Newly Eligible and Ineligible Marketable Assets

The analysis of the monthly stock of eligible marketable assets (collateral pool) in the previous section is complemented by the investigation of the monthly flow of eligible marketable assets (newly eligible marketable assets) as amendments to collateral criteria determine assets that enter the collateral pool ("newly eligible assets") or leave it ("newly ineligible assets").¹²⁰ Figure 3.11 gives the monthly nominal value of newly eligible and ineligible marketable assets from May 2007 to December 2013. As relaxations of collateral criteria dominated after the onset of the financial crisis, the nominal value of newly eligible collateral is more volatile than the nominal value of newly ineligible. Moreover, the nominal value of newly eligible assets runs on

¹¹⁸ The Eligible Assets Database differentiates between (i) bonds, (ii) medium-term notes, (iii) (treasury) bills/commercial papers/certificates of deposit, (iv) Jumbo covered bonds, (v) traditional covered bonds, (vi) abs, (vii) multi-cédulas, and (viii) structured covered bonds. For the sake of comparability, assets based on ISINs were matched to the asset types recognized in Figure 3.9, based on the issuer and asset specification given in the database.

¹¹⁹ The distribution across countries is calculated in terms of the residence of the collateral issuer.

¹²⁰ Newly eligible assets are defined as assets that appear in the collateral pool in one month but not in the previous one. Newly ineligible assets are assets that are contained in the collateral pool in one month but not in the subsequent one, corrected for assets that became ineligible owing to maturity.
Figure 3.10: Refined information on the collateral pool by country and by asset type

The figure refines Figure 3.9 by showing the development of the collateral pool by country (left panel) and asset type (right panel) from May 2007 to December 2013 based on monthly data provided to the author by the Deutsche Bundesbank. The right panel confirms the previous observation that the collateral pool substantially increased, as well as revealing that its composition by asset type changed. Moreover, the left panel indicates that the composition by country changed over time, implying asymmetrical developments of eligible marketable assets in selected countries.



a higher level than newly ineligible assets as the collateral pool grew over time. In December 2010, the value of newly ineligible assets jumps as in January 2011 both bank bonds traded on non-regulated markets and marketable assets denominated in pounds sterling, yen and US dollars became ineligible. During 2012, several jumps in the value of newly ineligible assets can be attributed to changes to the collateral criteria of bonds issued or guaranteed by the Greek government, e.g. the withdrawal of the suspension of the minimum credit rating for such bonds in February and July 2012. For newly eligible assets, the development is marked by three jumps. The first jump occurred immediately after the onset of the financial crisis in October and November 2008, when the Eurosystem successively introduced the eligibility of lower-rated assets, bank bonds traded on non-regulated markets and assets denominated in pounds sterling, yen and US dollars. The figure shows that these amendments triggered a substantial increase in the nominal value of newly eligible collateral. The second jump lasted from December 2011 to March 2012, with two consecutive substantial increases in the value of newly eligible marketable assets. The jumps can be attributed to the reintroduced eligibility of bank bonds traded on non-regulated markets (January 2012) together with the two three-year LTROs allotted during that time ("big bazooka"). The Eurosystem substantially increased refinancing credit via the two LTROs (see Section 2.3) and this additional credit had to be collateralized, largely by government-guaranteed assets (see Chapter 8). The last jump in November 2012 reflects the reintroduction of the eligibility of assets denominated in pounds sterling, yen and US dollars.

Figure 3.12 investigates the area under the green line in Figure 3.11, i.e. the development of the nominal value of newly eligible assets, by country (left panel) and asset type (right panel). The left panel indicates that newly eligible assets were freed up across all countries and particularly in France and Italy, thus confirming the findings of Figure 3.10. Moreover, other Eurozone countries as well as Great Britain and the US (depicted as "others") gained in importance, most



The figure shows the development of the nominal value of newly eligible and ineligible marketable assets from 2007 to 2013. As collateral criteria were predominantly relaxed rather than tightened during that time, the nominal value of newly eligible assets is more volatile than the nominal value of newly ineligible assets. As the collateral pool was growing over time, the nominal value of newly eligible collateral runs on a higher level than newly ineligible assets. Jumps in nominal values highlighted by the shaded areas can be attributed to changes in collateral criteria.



Source: author's calculation; European Central Bank, Eligible Assets Database

importantly owing to the eligibility of foreign currency-denominated assets.¹²¹ The right panel reveals that predominantly government bonds and uncovered bank bonds became newly eligible.

3.3.3 Eligible Marketable Assets in Selected Countries

This section merges the analyses for the collateral pool and newly eligible marketable assets and scrutinizes both newly eligible marketable assets and collateral pools in selected countries by asset type. This is supported by Figure 3.13, which considers newly eligible marketable assets in the left panels and national collateral pools (i.e. the pool of eligible marketable assets in each country) in the right panels. The comparison of the panels for each country provides a better understanding of how amendments to collateral criteria affected the development of newly eligible collateral and the composition of national collateral pools.

While the German collateral pool grew only slightly from May 2007 to December 2013 (8%), collateral pools considerably increased in Spain (56.4%), France (90.2%) and Italy (52.3%). The Irish collateral pool substantially increased from the onset of the financial crisis in the fall of 2008 to the end of 2010 (53.5%), while the Portuguese pool steadily increased (63%), although with a dint from May 2012 to April 2013. The Greek collateral pool grew until January 2012 (64.6%) but subsequently dropped to a level lower than in May 2007. In all countries, government bonds accounted for the largest fraction of eligible marketable assets.

¹²¹ As of May 2005, collateral issued by entities from the US, Canada, Japan and Switzerland has been deemed eligible (see Section 3.2.1).

Figure 3.12: Newly eligible marketable assets by country and by asset type

The figure elaborates on the area under the green line in Figure 3.11 and details the development of newly eligible marketable asset by country (left panel) and asset type (right panel) from 2007 to 2013. Although newly eligible collateral was spread across countries, a substantial fraction was issued in France. The right panel confirms that predominantly government and uncovered bank bonds became newly eligible.



The composition of the German collateral pool remained relatively stable with covered bank bonds slightly losing importance, being outweighed by uncovered bank bonds and other marketable assets. Moreover, the value of newly eligible assets considerably increased after the relaxation of collateral criteria in October and November 2008 and it also remained on a high level throughout 2012.

The Spanish collateral pool increased mainly owing to the increase in government bonds. While ABSs played an important role in Spain until February 2009, their value dropped in March 2009 owing to two changes of the collateral framework (see Section 3.2.5), i.e. the tightening of the minimum credit rating for ABSs (to "triple A" at issuance and "single A" over lifetime) and the ineligibility of ABSs comprising tranches of other ABSs. This development is also reflected in the value of newly eligible ABSs, which was high prior to 2009 but covered and uncovered bank bonds as well as corporate bonds became newly eligible assets thereafter. Hence, the drop in ABSs was compensated by increases in bank and corporate bonds.

In France, the collateral pool increased over time, mainly attributed to increases in the amount of government bonds and uncovered bank bonds. The increase in uncovered bank bonds was largely due to the eligibility of bank bonds traded on non-regulated markets from October 2008 to December 2010 and as of January 2012. This is also reflected in the development of newly eligible assets, which largely comprised uncovered bank bonds.

The Greek collateral pool almost exclusively comprised government bonds until mid-2010. Since the outbreak of the sovereign debt crisis in May 2010, the value of eligible government bonds stabilized and uncovered bank bonds gained in importance (albeit largely guaranteed by the Greek government, cf. Chapter 8). The Greek collateral pool diminished in 2012 when the Eurosystem revoked the suspension of the minimum credit rating threshold for bonds issued or guaranteed by the Greek government in February. In March 2012, the suspension was reintroduced, albeit conditional upon the provision of a buy-back scheme (cf. Section 3.2.2). When Greek government bonds were deemed ineligible even if a buy-back scheme was provided as of Figure 3.13: Newly eligible marketable assets & national collateral pools by asset type

The figure analyzes newly eligible marketable assets by asset type in selected countries (left panels) together with their effect on national collateral pools (right panels) from 2007 to 2013. It shows that national collateral pools grew, largely owing to increases in newly eligible government bonds and uncovered bank bonds. ABSs lost in importance in most countries and covered bank bonds played only a minor role, aside from in Spain.



Source: author's calculation; European Central Bank, Eligible Assets Database.





Source: author's calculation; European Central Bank, Eligible Assets Database.

July 2012, the Greek collateral pool fully drained. Ineligibility lasted until December 2012 when the minimum credit rating threshold was again waived. Changes to eligibility are also reflected in the development of newly eligible assets, indicating two spikes in newly eligible government bonds. The first spike coincides with the introduction of the conditional suspension of the minimum rating in March 2012 and the second with the reintroduced unconditional suspension in December 2012. Interestingly, the Eligible Assets Database indicates that the Greek collateral pool was empty between July and November 2012 although only bonds issued or guaranteed by the Greek government were subject to changes in collateral criteria.

The Italian collateral pool increased over time as inter alia the value of newly eligible uncovered bank bonds rose, although this is largely attributed to newly eligible government bonds. The value of newly eligible government bonds remained high and occasional increases in the value of newly eligible uncovered bank bonds caused spikes in the value of newly eligible marketable assets.

Likewise, the collateral pool grew in Portugal from May 2007 to December 2013, albeit with a dint from May 2012 to April 2013 coinciding with the release of the NCBs' obligation to accept eligible government-guaranteed bank bonds provided that the guarantor is a country under an EU/IMF program or whose credit assessment does not meet high credit standards (cf. Section 3.2.4). However, the development of newly eligible collateral indicates that the dint was largely caused by fewer newly eligible uncovered bank bonds. While these bonds became newly eligible to a large extent after the onset of the financial crisis, the value of newly eligible uncovered bank bonds remained low from May 2012 to April 2013. This lack of uncovered bank bonds was compensated by newly eligible government bonds.

In Ireland, the collateral pool substantially expanded by the end of 2010, largely driven by newly eligible ABSs. Newly eligible ABSs thereafter almost came to a halt and the pool shrunk until May 2012. Interestingly, although the Eurosystem suspended the minimum credit rating for Irish government bonds as of April 2011, no such bond became newly eligible from April 2011 to January 2012. The lack of government bonds and ABSs was compensated by newly eligible corporate and uncovered bank bonds, with the latter being largely guaranteed by the Irish government (cf. Chapter 8).

In summary, Figure 3.13 provides new information on the development and composition by asset type of newly eligible marketable assets and national collateral pools, revealing two important findings. First, national collateral pools substantially grew from 2007 to 2013, whereby this growth differed across countries. Growth was largely due to increases in newly eligible government bonds and uncovered bank bonds. Accordingly, this implies, as a second finding, that the composition of newly eligible marketable assets and hence national collateral pools by asset type substantially differed across countries.

3.3.4 Credit Quality of Eligible Marketable Assets

The descriptive analysis thus far has made no distinction between the horizontal and vertical broadening of the collateral pool. However, Section 3.2.7 summarized that this distinction is

important given that the Eurosystem amended the collateral framework to broaden the collateral pool along both dimensions. Therefore, the following investigation extends the analysis and elaborates on the distinction between horizontal and vertical broadening of the collateral pool by analyzing the credit quality of newly eligible marketable assets and the collateral pool (both total and national). Hence, it resembles the previous analysis yet differentiates by the credit quality of eligible assets rather than asset types.

While the previous analysis refined (composition of the collateral pool by asset type) and extended available information (analysis at the country level), this section provides innovative insights as the Eligible Assets Database does not contain information on credit quality. Three procedures are possible to gain insights into the development of credit quality of eligible assets. First, the development of the average haircut applied to eligible assets can be used as a proxy for credit quality. However, Chapter 4 indicates that this proxy would be flawed as haircuts are subject to exogenous changes (irrespective of credit quality), which would affect the average haircut detached from credit quality. Second, credit assessments from rating agencies could be obtained, although this procedure proves impracticable as neither are all assets rated by rating agencies nor is there a historical record on credit ratings available. Therefore, a third approach is elaborated that combines information provided in the Eligible Assets Database and a self-compiled comprehensive database of haircuts applied by the Eurosystem from 2007 to 2013 (see also Chapter 4). The application of this inductive identification strategy provides the first outside-in analysis of credit quality of marketable assets eligible with the Eurosystem in refinancing operations between 2007 and 2013.¹²² Chapter 4 addresses that the Eurosystem essentially determines haircuts based on four factors: (i) liquidity, (ii) coupon, (iii) residual maturity and (iv) credit quality.¹²³ Taking advantage of information on the first three factors provided in the Eligible Assets Database, *credit quality* is inferred as the missing fourth factor from information on haircuts contained in the self-compiled database. Therefore, the analysis derives the Eurosystem's understanding of credit quality in terms of the credit quality of collateral clustered into CQSs, thereby providing new insights into the credit quality of eligible marketable assets.

The left panel of Figure 3.14 details the credit quality of newly eligible marketable assets, while the right panel shows the development of credit quality in the collateral pool. The green area represents assets in CQS 1/2, i.e. assets rated between "triple A" and "single A". As "single A" was the initial minimum credit rating threshold, the green area reflects the *horizontal* broadening of the collateral pool. By contrast, the red and blue areas represent the *vertical* broadening as the red area gives all assets within CQS 3 (i.e. lower than "single A" and at least "triple B") and the blue area shows assets rated lower than "triple B".¹²⁴ Therefore, the two areas jointly measure additional assets made available by lowering or waiving the minimum credit rating

 $^{^{122}}$ About 99% of eligible marketable assets could be unanimously matched over the entire time horizon by this technique. No more than 5% of assets could not be matched at each point in time.

¹²³ Moreover, supplementary haircuts on e.g. foreign currency assets are applied (cf. Section 4.2.3.2).

¹²⁴ Collateral rated lower than "triple B" was only deemed eligible in case of specific government-related assets for which the credit rating requirement was waived or within the ACCs framework (cf. Section 3.2.2). In case of ACCs, these assets are referred to as CQS 4 with a default probability of up to 1.0%, see e.g. MAHARAJ et al. 2012. As only eligible marketable assets are considered here, credit claims are not reflected in the figures.

threshold.¹²⁵ The figure reveals that collateral rated lower than the initial rating threshold steadily rose from October 2008 to December 2013 and particularly following the onset of the sovereign debt crisis in 2010. Hence, it suggests that the broadening of the collateral pool was achieved both horizontally and vertically.

Figure 3.14: Newly eligible marketable assets and the collateral pool by credit quality

The figure illustrates the development of credit quality of newly eligible marketable assets (left panel) and the Eurosystem collateral pool (right panel). The figure indicates the horizontal and vertical broadening of the collateral pool (Section 3.2.7). Credit quality deteriorated particularly following the onset of the sovereign debt crisis in 2010.



Source: author's calculation; European Central Bank, Eligible Assets Database.

The finding that the broadening was achieved both horizontally and vertically only partly holds for national collateral pools in Germany and France, which are inter alia depicted in Figure 3.15, illustrating the development of credit quality of newly eligible marketable assets and national The credit quality of eligible marketable assets remained high in Germany, collateral pools. where collateral was almost exclusively rated within CQS 1/2. A similar development is observable for France, although small fractions of newly eligible assets were rated within CQS 3 immediately after the onset of the financial crisis and throughout the sovereign debt crisis. National collateral pools were considerably broadened vertically in Ireland, Spain and Italy. While the credit quality of eligible marketable assets already dropped throughout the financial crisis in Ireland, deterioration was delayed in Spain and Italy, where credit quality worsened at the turn of 2011/2012. Credit quality declined the most in Greece and Portugal. In Portugal, deterioration commenced when the sovereign debt crisis set in and it further intensified such that the majority of collateral was rated within CQS 3 at the end of 2013. Likewise, collateral quality substantially dropped in Greece with the onset of the sovereign debt crisis and it did not recover thereafter. At the end of 2013, eligible marketable assets in Greece were exclusively of credit quality lower than "triple B" (CQS 4). Hence, no marketable assets would have been eligible in Greece at the end of 2013 if the Eurosystem had not waived the rating requirement for government-related assets.

¹²⁵ The yellow area represents the small fraction of assets that could not be inferred to a CQS.

Figure 3.15: Newly elig. marketable assets & national collateral pools by credit quality

The figure depicts the development of credit quality of newly eligible marketable assets (left panels) and national collateral pools (right panels). The credit quality of eligible marketable assets remained high in Germany and France where national collateral pools were hence predominantly broadened horizontally. By contrast, national collateral pools were broadened vertically to a considerable extent in Ireland, Spain and Italy. The credit quality of eligible marketable assets deteriorated the most in Greece and Portugal.



Source: author's calculation; European Central Bank, Eligible Assets Database.



Figure 3.15: Newly elig. marketable assets & national collateral pools by credit quality (cont.)

 $\label{eq:source:author's calculation; European Central Bank, \textit{Eligible Assets Database}.$

3.3.5 Denomination of Eligible Marketable Assets

An amendment to the collateral framework that is often considered important to the broadening of the collateral pool was the eligibility of marketable assets denominated in certain foreign currencies, i.e. pounds sterling, yen and US dollars. Such assets were made eligible in two phases (see Section 3.2.1.2): *(i)* from November 2008 to December 2010 and *(ii)* as of November 2012. These two phases are also apparent in Figure 3.16, which details the development of newly eligible marketable assets (left panel) and the collateral pool (right panel) by denomination. The figure reveals that the eligibility of foreign currency assets contributed to broadening the collateral pool, particularly following the onset of the financial crisis in the fall of 2008 and with renewed eligibility in November 2012.

Figure 3.16: Newly eligible marketable assets and the collateral pool by denomination

The figure elaborates on the development of newly eligible collateral (left panel) and the collateral pool (right panel) by denomination from 2007 to 2013. It differentiates between eligible marketable assets denominated in EUR and foreign currencies (GBP, JPY, USD). The panels reveal that the eligibility of foreign currency assets contributed to broadening the collateral pool, with the eligibility introduced in the fall of 2008 and after the renewed eligibility in November 2012.



Source: author's calculation; European Central Bank, Eligible Assets Database.

3.3.6 Non-Regulated Markets and Eligible Marketable Assets

The Eurosystem amended the collateral criteria of assets traded on non-regulated markets by (i) increasing the number of accepted non-regulated markets and (ii) deeming eligible additional asset types traded on accepted non-regulated markets (cf. Section 3.2.3). The upper panels in Figure 3.17 indicate that the Eurosystem succeeded in broadening the collateral pool, especially after the turmoil in financial markets in the fall of 2008. The upper left panel considers newly eligible marketable assets and differentiates between assets traded on non-regulated and regulated markets. It reveals that a substantial fraction of newly eligible assets were traded on accepted non-regulated markets. In particular, this fraction increased after bank bonds traded on non-regulated markets were deemed eligible as of October 2008.

The upper right panel illustrates that the increase in newly eligible assets traded on nonregulated markets affected the composition of the collateral pool as the fraction of assets traded on non-regulated markets increased over time. The lower panels seize upon the important role of the STEP market (see Section 3.2.3) and investigate the blue areas in the upper panels, i.e. they focus on newly (left panel) and total eligible marketable assets (right panel) traded on non-regulated markets. They differentiate between assets issued on the STEP market (by banks or by others) and other non-regulated markets (by banks or by others), confirming that the increase of eligible assets traded on non-regulated markets was largely due to the eligibility of bank bonds traded on these markets, especially on the STEP market. Bank bonds traded on non-regulated markets have been eligible during two phases (cf. Section 3.2.3), i.e. in a first phase from October 2008 to December 2010 and as of January 2012. By the end of 2013, almost half of eligible marketable assets traded on non-regulated markets were issued by banks.

Figure 3.17: Newly eligible marketable assets and the collateral pool by type of market

The figure elaborates on the development of the role of eligible assets traded on non-regulated markets in due consideration of the STEP market. The upper panels indicate that the Eurosystem succeeded in broadening the collateral pool by extending the acceptance of non-regulated markets. The lower panels refine the blue areas in the upper panels with due regard to the STEP market. They differentiate between assets issued in the STEP market (by banks or others) and other non-regulated markets (by banks or others). The figure emphasizes the importance of the eligibility of bank bonds traded on non-regulated markets such as the STEP market.



Source: author's calculation; European Central Bank, Eligible Assets Database

3.3.7 Government Guarantees and Eligible Marketable Assets

Section 3.2.2 inter alia discussed how the collateral criteria of government-guaranteed assets were amended. Figure 3.18 suggests that the collateral pool broadened owing to these amendments, particularly following the onset of the financial crisis. The left panel shows newly eligible marketable assets and the right panel illustrates the development of the collateral pool. The panels differentiate between assets that are (*i*) not guaranteed, (*ii*) guaranteed by governments and (*iii*) guaranteed by others (e.g. private sector entities). While the vast majority of eligible marketable assets was not guaranteed, government guarantees increased in importance. Government guarantees to eligible marketable assets are further addressed in Chapter 8, where the development observable in Figure 3.18 is investigated.¹²⁶

Figure 3.18: Newly eligible marketable assets and the collateral pool with respect to guarantees

The figure illustrates the development of newly eligible marketable assets (left panel) and the collateral pool (right panel) with respect to guarantees from 2007 to 2013. It shows that the amount of eligible guaranteed assets increased over time, particularly with government guarantees.



Source: author's calculation; European Central Bank, Eligible Assets Database.

3.3.8 Residual Maturity of Eligible Marketable Assets

Finally, the effects of changes to collateral criteria on the residual maturity of eligible marketable assets are scrutinized. Figure 3.19 illustrates the average residual maturity of newly eligible marketable assets as well as in the collateral pool (weighted by nominal value).¹²⁷ The figure highlights that the average maturity of newly eligible assets and accordingly in the collateral pool dropped owing to amendments to collateral criteria in October 2008, especially as STEPs issued by banks with per definition short maturity were deemed eligible. Thereafter, the average residual maturity slightly increased but jumped and dropped when STEPs issued by banks were temporarily deemed ineligible throughout 2011. The average residual maturity of newly eligible marketable assets experienced another jump in November 2012 when foreign denominated assets

¹²⁶ Recently, NYBORG 2015 pointed to the importance government guarantees in the Eurosystem collateral framework, emphasizing that it implies leeway for politics to influence collateral values.

¹²⁷ The collateral pool contained 57 government bonds, covered and uncovered bank bonds as well as corporate bonds with 9999 as the year of maturity between May 2007 and January 2013. These assets are neglected as outliers to avoid potential biases in the depiction of average residual maturities.

became eligible again. Hence, the figure reveals that the average residual maturity of newly eligible marketable assets and in the collateral pool fluctuated and decreased over time, inter alia reflecting amendments to collateral criteria. The average residual maturity in the collateral pool decreased from 5.5 years in May 2007 to 4.6 years in December 2013. Observable fluctuations in the average residual maturity of newly eligible marketable assets can be associated with amendments to collateral criteria, specifically (i) the eligibility of STEPs issued by banks and (ii) the eligibility of assets denominated in certain foreign currencies.

Figure 3.19: Average residual maturity of eligible marketable assets

The figure illustrates the average residual maturity of newly eligible marketable assets and in the Eurosystem collateral pool. It highlights that the average residual maturity of newly eligible assets considerably fluctuated, reflecting amendments to collateral criteria. The average maturity in the collateral pool decreased from 5.5 years in May 2007 to 4.6 years in December 2013.



3.3.9 Lessons from Analyzing the Collateral Pool Development

In conclusion, this section has revealed that the Eurosystem collateral pool has changed in both size and composition over recent years, reflecting amendments to collateral criteria. The Eurosystem broadened the collateral pool both horizontally and vertically by deeming more asset types eligible, as well as assets of lower credit quality. Moreover, it extended eligibility to assets denominated in certain foreign currencies, amplified the eligibility of assets traded on non-regulated markets and relaxed the collateral criteria of government-guaranteed assets.

The extent to which these measures broadened the Eurosystem collateral pool are reflected in the left panel of Figure 3.20. The size of the collateral pool in September 2008 is normalized to 100 to capture the effects of amendments to the collateral framework. The left panel reveals that the most important measures of the Eurosystem were to deem more assets of initial credit quality eligible ("horizontal"), relax the collateral criteria of assets traded on non-regulated markets and deem eligible more assets of lower credit quality ("vertical"). Moreover, the extended eligibility of government-guaranteed assets as well as assets denominated in foreign currency contributed to the broadening of the Eurosystem collateral pool. Figure 3.20: Broadening of the Eurosystem collateral pool and national collateral pools

The figure shows the broadening of the Eurosystem collateral pool (left panel) and national collateral pools (right panel) from 2007 to 2013. Pool size is normalized to 100 for pre-crisis sizes in September 2008. The left panel takes up broadening measures and suggests that quantitative was more important than qualitative broadening. Moreover, the extended eligibility of assets traded in non-regulated markets, government-guaranteed assets and assets denominated in foreign currency contributed to the broadening. The right panel seizes upon the broadening of national collateral pools and details total (solid line) and vertical broadening (dashed line). It shows that broadening was the strongest in France, Italy, Spain and Portugal but modest in Germany and even negative in Ireland. Vertical broadening was minor in Germany, France and Ireland but important in Italy and Spain. It was negligible in Portugal prior to 2011 but exploded thereafter.



Furthermore, the geographical composition of the collateral pool changed owing to asymmetric developments of national collateral pools, as shown in the right panel of Figure 3.20 for selected countries. The size of national collateral pools in September 2008 is again normalized to 100. Solid lines refer to the development of national collateral pools and dashed lines detail the extent of their vertical broadening. The panel confirms that national collateral pools grew the strongest in France, Italy, Spain and Portugal (in that order) from 2007 to 2013. The national collateral pool only slightly increased in Germany and it even shrunk in Ireland, although the Irish collateral pool exhibited the largest expansion during the financial crisis until the onset of the sovereign debt crisis. Vertical broadening of national collateral pools was modest in Germany, France and Ireland but stronger in Italy and Spain. It was minor in Portugal prior to 2011 but exploded thereafter, thus confirming the finding of Figure 3.15 that credit quality in the Portuguese collateral pool dropped in 2011. Hence, the Portuguese collateral pool was not only qualitatively broadened but also a fraction of the collateral pool was qualitatively replaced, i.e. assets within CQS 1/2 were replaced by assets within CQS 3.

4

Risk Control of the Eurosystem¹²⁸

The Eurosystem is exposed to two types of risk in its credit operations. Counterparties can default on their loans, which gives rise to (i) counterparty risk. Despite requiring counterparties to be supervised and deemed financially sound, the Eurosystem hedges counterparty risk by calling for collateral. Therefore, the Eurosystem can liquidate collateral to meet its claims in case of counterparty default. However, the assumption of collateral exposes the Eurosystem to (ii) collateral risk as the liquidation value of collateral is uncertain at the time when the loan is contracted. The Eurosystem applies a set of measures to mitigate collateral risk that are laid down in its collateral framework. Accordingly, this chapter explores the application and development of these risk control measures. It shows that the Eurosystem specifies risk control in a simplified fashion based on a reduced set of available information, which implies imperfect mitigation of collateral risk. The previous chapter revealed that the collateral pool has evolved along various dimensions over recent years, inter alia owing to amendments to the collateral framework. This chapter addresses how the Eurosystem accommodated risk control in response to its changing collateral pool. First, Section 4.1 provides an overview of risk control measures of the Eurosystem and identifies haircuts as the most important measure. Section 4.2 elaborates on haircuts. It derives general principles for their specification, compiles a narrative database of Eurosystem haircuts and provides a descriptive analysis of their evolution. Subsequently, Section 4.3 evaluates the specification of Eurosystem haircuts against the derived general principles and carves out its shortcomings. Finally, the potential effects of the shortcomings are addressed in Section 4.4.

 $^{^{128}\,\}rm This$ chapter bases on EBERL and WEBER 2014b.

The Eurosystem hedges risk from collateralized lending in two steps.¹²⁹ First, it requires counterparties to obey the minimum reserve requirement, be supervised and deemed financially sound (see ECB 2015c). Second, it calls for adequate collateralization of provided liquidity. Accordingly, the Eurosystem is exposed to collateral risk as the liquidation value of collateral is uncertain at the time when the loan is contracted. The Eurosystem applies a set of measures to mitigate collateral risk, which are summarized in Table 4.1 (see also ECB 2000, 2011d, 2015c).¹³⁰

Prior to the introduction of the Single List in January 2007, the Eurosystem established uniform risk control measures for tier 1 assets and NCBs were responsible for taking control for assets that they accepted in tier 2. The initial set of risk control measures comprised "valuation haircuts", "initial margins", "variation margins", "limits in relation to issuers/debtors or guarantors", "additional guarantees" and "exclusion". The Eurosystem restricted the set of disposable risk control measures to "valuation haircuts" and "variation margins" in March 2004. It announced to perform a biennial review of its risk control measures in 2008 in response to amendments to the collateral framework and developments in financial markets. Since then, valuation haircuts have been repeatedly updated (see next section) and partly adapted to collateral pool and market developments. Moreover, the Eurosystem revoked its restriction of the set of risk control measures in February 2009 such that all risk control measures could be applied "at any time if required to ensure adequate risk protection" (cf. ECB 2008b, p. 37). The Eurosystem successively introduced two additional means to address collateral risk in February and March 2009, i.e. "supplementary haircuts" and "limits in relation to the use of uncovered bank bonds".¹³¹ The latter measure has quantitatively limited the use of uncovered bank bonds with close links that were deemed eligible if they were guaranteed by a government (see Section 3.2.4). Hence, uncovered bank bonds of the same issuer group (i.e. identical issuer or closely linked issuers) could only be pledged to the extent that the haircut-adjusted value of these bonds would not exceed 10% (later 5%) of the total value of collateral submitted by a counterparty as of March 2009. This limitation was generalized to unsecured debt instruments with close links in October 2010 and explicitly added to the set of risk control measures. The implementation of the ACCs framework in December 2011 enabled NCBs to establish idiosyncratic risk control measures that could deviate from those applied within the collateral framework. The application of idiosyncratic risk control measures was made subject to approval of the Governing Council. This approval was considered unnecessary in exceptional circumstances as of May 2013 under the provision that the risk control measures would have been established by another NCB and approved by the Council.

Valuation haircuts and variation margins are the only risk control measures applied from January 2001 to December 2014. Valuation haircuts imply that the market or a theoretical value assigned to an eligible asset is deducted by a fraction ("haircut") to determine the collateral

¹²⁹ The analysis in this chapter elaborates on risk control in collateralized lending. While asset purchases and collateralized loans share many types of risk, there are important differences (ECB 2015c). Most importantly, the Eurosystem is exposed to potential losses from collateralized lending only in case of double default (i.e. counterparty and collateral default), although it might already suffer losses from asset purchases in case of

Table 4.1: Eurosystem risk control measures

The table provides an overview of the risk control measures applied by the Eurosystem. Valuation haircuts and variation margins are the most important measures and they were applied over the entire period.

MEASURE	DESCRIPTION	Application Period
Valuation Haircuts	The collateral value is calculated as some value of the asset less a certain percentage (haircut).	$01/2001$ - $\mathrm{present^a}$
Supplementary Haircuts	Haircuts that are applied beyond valuation hair- cuts if considered necessary.	$10/2008$ - $\rm present^{a,b}$
Initial Margins	Counterparties have to pledge collateral at least equal to liquidity plus the value of the initial mar- gin.	01/2001 - 03/2004; $02/2009 - \text{present}^{a}$
VARIATION MARGINS (MARKING TO MAR- KET)	The haircut-adjusted value of collateral has to be maintained over time and if it falls below a thresh- old, the Eurosystem calls for additional collateral.	01/2001 - present ^a
Limits in Relation to Use of Unsec. Debt Instruments	Limitation of the pledge of unsecured debt in- struments with close links as described in Section 3.2.4.	03/2009 - present ^{a,c}
Limits in Rela- tion to Issuer- s/Debtors/Guar- antors	Limitation of the exposure vis-à-vis issuer- s/debtors/guarantors in general or vis-à-vis spe- cific counterparties.	01/2001 - 03/2004; $02/2009 - \text{present}^{a}$
Additional Guarantees	Additional guarantees can be required from coun- terparties to accept certain assets as collateral.	01/2001 - 03/2004; $02/2009 - present^{a}$
Exclusion	Certain assets and/or counterparties can be excluded.	$\begin{array}{c} 01/2001 - 03/2004; \\ 02/2009 - \mathrm{present^a} \end{array}$

^a By the end of 2014.

^b The Eurosystem applied supplementary haircuts to e.g. assets within CQS 3 since October 2008. Supplementary haircuts for risk control were mentioned in February 2009 in the collateral framework. They were explicitly included as a risk control measure in October 2010.

^c The measure was first specified for uncovered bank bonds in March 2009 and generalized to unsecured debt instruments in October 2010.

Source: author's compilation; European Central Bank.

value. Valuation haircuts reflect expectations about the liquidation value of collateral in case of counterparty default. Variation margins subsequently ensure that this expected liquidation value is maintained over time. Amendments to the collateral framework making eligible more types of assets of different credit quality called for repeated adjustments to valuation haircuts, thus rendering them key to Eurosystem risk control (see also ECB 2015c).

4.2 Haircuts as the Key Risk Control Measure

The analysis of valuation haircuts as the key risk control measure of the Eurosystem is important for two reasons. First, haircuts are applied as a discount to the value of an eligible asset

single default (i.e. collateral default). Therefore, risk control differs between collateralized lending and asset purchases. See EUROPEAN CENTRAL BANK 2015c for an in-depth investigation.

¹³⁰See BINDSEIL et al. 2009 (in particular BINDSEIL 2009a, BINDSEIL and PAPADIA 2009, GONZÁLEZ and MOLITOR 2009, TABAKIS and WELLER 2009) as well as CHAILLOUX et al. 2008b for general discussions and contributions to central bank risk mitigation.

¹³¹ "Supplementary haircuts" were explicitly introduced as a risk control measure in October 2010, see Table 4.1.

such that they determine the amount of liquidity that the Eurosystem supplies for eligible assets. Therefore, the haircut is one of the six determinants of credit terms (see Section 2.3). Second, some central (clearing) counterparties (CCPs) adopt haircuts of the Eurosystem such that Eurosystem haircuts not only influence refinancing conditions with the Eurosystem but they also spill over to the interbank repo market (see Section 5.2.2 and NYBORG 2015).¹³²

There is relatively little work on haircut-setting within collateral frameworks of central banks. JOKIVUOLLE and PEURA 2000 provide an investigation of central bank haircuts on bank loans and ASHCRAFT et al. 2011 analyze the role of haircuts in collateralized central banks loans. ECB 2014a has recently extended ECB 2013a and presents insights into the role of haircuts within the Eurosystem collateral framework, comparing haircut ranges of the Eurosystem with the CCP-cleared segment of the repo market. Moreover, ECB 2015c emphasizes the role of haircuts in Eurosystem risk control and provides a sketchy overview of their development.

The value of a collateral asset is adjusted for risk control purposes by either (i) a haircut or (ii) an initial margin. For a haircut, a discount on some value of the collateral (e.g. the market value) is taken such that the asset's collateral value Φ is given by¹³³

$$\Phi \equiv (1-h)\Omega,\tag{4.1}$$

with h being the haircut and Ω the market value of the asset. The haircut can be expressed as

$$h = \frac{\Omega - \Phi}{\Omega}.\tag{4.2}$$

For an initial margin, the loan is over-collateralized by the amount of the initial margin m such that Φ reads as

$$\Phi \equiv \frac{\Omega}{1+m} \tag{4.3}$$

and the initial margin can be calculated as

$$m = \frac{\Omega - \Phi}{\Phi}.\tag{4.4}$$

The difference between a haircut and initial margin is only conceptual as both instruments have equivalent effects. Using Equations (4.1) and (4.3), the relation between a haircut and initial margin can be expressed as¹³⁴

$$h = \frac{m}{1+m}.\tag{4.5}$$

Owing to their importance in the risk control of the Eurosystem, the focus lies on haircuts in the following.

¹³² See e.g. MANCINI et al. 2015 for the relation between Eurosystem and Eurex haircuts. BIS 1999 provides an early description of the interbank repo market from the perspective of central banks. For further discussion on repo markets and their connection to the Eurosystem's refinancing operations, see Chapter 5. For the European interbank repo market, see DUNNE et al. 2011, ECB 2014d as well as MANCINI et al. 2015.

¹³³ For the general characterization of haircuts see also e.g. CHOUDHRY 2006, EUROCLEAR 2009, COMOTTO 2012, 2013.

 $^{^{134}\,{\}rm For}$ instance, a haircut of 5% would be approximated by an initial margin of 5.26%.

4.2.1 General Principles of the Haircut Specification

Lenders bear different risks in collateralized liquidity provision despite collateralization. They aim to minimize these risks and gain compensation for any remaining. This can be achieved by the application of two instruments, i.e. (i) the interest rate and (ii) the haircut.¹³⁵ The following discussion elaborates on the major risks in collateralized liquidity provision and emphasizes the relevant instrument to address these risks. Moreover, general principles of haircut specification are carved out.¹³⁶

Risk from collateralized liquidity provision only materializes in case of borrower default. In this case, the lender has to liquidate the collateral to meet outstanding claims (from principal and interest). As this *counterparty risk* is solely associated with the financial soundness of the borrower, it should generally be correlated with interest rates but not with haircuts (cf. COMOTTO 2012 as well as DANG et al. 2013a). FITCH RATINGS 2012 finds an indication for the absence of correlation based on empirical observations, while survey results by BIS 2010 suggest a correlation between counterparty risk and haircuts. Furthermore, DANG et al. 2013a find an indication for effects of counterparty risk on haircuts as they observe that borrowers pledging collateral of the same type and credit rating face different haircuts. However, ibid. attribute this finding to a reason other than counterparty risk, i.e. the risk that lenders face a better informed trader in case of collateral liquidation such that they are not compensated for their outstanding claims.¹³⁷ Hence, this risk should be addressed by the haircut.

This reflects the sequence of addressing risk in collateralized liquidity provision, i.e. counterparty and collateral risk. Counterparty risk is addressed by charging an interest rate together with demanding collateral. Collateral risk only materializes in case of counterparty default. If collateral risk was adequately addressed by the haircut, no counterparty risk would have to be mitigated via the haircut. Mitigating counterparty risk through the haircut would subsequently imply an inefficient haircut increase, given that it would curtail liquidity provision (given collateral scarcity).

Consider the simple participation constraint of a risk-neutral lender given by Equation (4.6). Let $(1-p) \in [0,1]$ represent the default probability of the counterparty and $(1-q) \in [0,1]$ the default probability of collateral. The interest rate premium in collateralized lending above the market rate of interest (repo rate) is denoted by $i_L \in [0,1]$, which is assumed to be zero for simplicity such that the interest rate premium equals the interest rate. $h \in [0,1]$ is the haircut on collateral that is assessed at its market value Ω . Collateral that the borrower has at disposal is assumed to be restricted to $\overline{\Omega}$. The lender can decide upon the utilization of available liquidity

¹³⁵ Lenders could also restrict the set of potential borrowers, which would correspond to a prohibitively high interest rate. Likewise, lenders could define assets that are deemed ineligible as collateral, which would be equivalent to a haircut of 100%.

¹³⁶ See also GONZÁLEZ and MOLITOR 2009 for a discussion of central bank haircut specification and NYBORG 2015 for general features of Eurosystem haircuts.

¹³⁷ This reasoning is based on an utility-based measure, called information acquisition sensitivity, cf. DANG et al. 2013a,b.

to the amount of $(1-h)\overline{\Omega}$ in either collateralized lending or alternative investment in the market at zero interest. The lender's participation constraint reads as

$$\underbrace{p(1+i_L)(1-h)\overline{\Omega} + (1-p)q\overline{\Omega}}_{\text{expected payoff from collateralized loan}} \stackrel{!}{=} \underbrace{(1-h)\overline{\Omega}}_{\text{payoff from alternative investment}} (4.6)$$

The left-hand side of Equation (4.6) gives the expected payoff to the lender from entering the collateralized loan. The lender receives principal and interest with probability p. With probability (1 - p), the counterparty defaults and the lender receives collateral to the amount of $\overline{\Omega}$, which can be liquidated with probability q.¹³⁸ ¹³⁹ The expected payoff from collateralized lending has to put the lender in the same financial position as alternative investment. Therefore, the expected payoff from the collateralized loan has to equal the liquidity of the lender as the market offers zero interest by assumption. Solving Equation (4.6) for i_L gives the following relationship between the interest rate i_L and the haircut h:

$$i_L = \left(1 - \frac{q}{1-h}\right) \left(\frac{1-p}{p}\right). \tag{4.7}$$

If counterparty risk is absent, i.e. p = 1, i_L is also equal to zero irrespective of collateral risk. However, in reality, lending is associated with counterparty risk, i.e. $p \in (0, 1)$. If both the interest rate and the haircut are available for risk mitigation, possible combinations of i_L and h that fulfill the lender's participation constraint are given by the negative and concave relationship between i_L and h implied by Equation (4.7).¹⁴⁰

Furthermore, consider the case in which only either the interest rate or the haircut is available for the mitigation of (counterparty and collateral) risk. If the haircut h is restricted to zero (h = 0), the interest rate depends on both counterparty and collateral risk such that $i_L = (1 - q) \left(\frac{1-p}{p}\right)$. Counterparty risk affects the interest rate to the extent that collateral risk is present. If the collateral was free of risk, i.e. q = 1, then the interest rate would also be equal to zero. By contrast, if the interest rate is restricted to zero $(i_L = 0)$, the haircut solely depends on the default probability of the collateral with h = 1 - q, i.e. the haircut would solely and fully reflect collateral risk. This gives a good approximation of the case of the Eurosystem. The Eurosystem sets a single interest rate as a matter of monetary policy and to eliminate pitfalls from having to differentiate pricing of operations based on counterparty creditworthiness. Therefore, the market interest rate is defined by the Eurosystem and liquidity is provided to a wide range of counterparties with different default probabilities at this interest rate. This calls for the implementation of risk control aimed at achieving the equivalence of collateral risk across eligible assets (see ECB 2015c).

¹³⁸ The lender will receive the entire liquidation value $q\overline{\Omega}$ in case of counterparty default only if this liquidation value falls short of principal and interest, i.e. $(1 + i_L)(1 - h)\overline{\Omega}$. Therefore, Equation (4.6) is subject to the constraint that $q\overline{\Omega} \leq (1 + i_L)(1 - h)\overline{\Omega}$.

¹³⁹ Note that without a loss of generality, collateral risk is analyzed in terms of credit risk only. Therefore, e.g. liquidity risk is neglected. See Section 7.3.1 for the liquidity risk of collateral.

¹⁴⁰ As $\partial i/\partial h < 0$ and $\partial^2 i/\partial h^2 < 0$, concavity suggests that as both a larger interest rate and a larger haircut are costly to the borrower, the optimal solution would be to mitigate risk via either the interest rate or the haircut. A larger interest rate is costly because it lowers the margin for the borrower to the extent that the loan does not pay off. A larger haircut decreases principal as Ω is restricted to $\overline{\Omega}$.

Collateral risk is in principle determined by seven types of risk that could materialize between loan contraction and collateral liquidation,¹⁴¹ i.e. (*i*) legal risk, (*ii*) operational risk, (*iii*) collateral quality (in terms of credit and liquidity risk), (*iv*) market risk (in terms of traded market and valuation risk), (*v*) interest-rate risk, (*vi*) wrong-way risk and (*vii*) exchange-rate risk. In the following, the effects of the seven properties on the haircut size are stylized and general principles of haircut specification are carved out.

First, the transfer of collateral ownership in case of counterparty default should be legally binding in terms of an easily enforceable written contract. In the absence of such a contract, the lender risks legal challenges and facing competition for collateral with other creditors, which would delay the settlement of outstanding claims. This *legal risk* should be addressed by the use of predefined contracts (Master Agreements), such as the Global Master Repurchase Agreement and the European Master Agreement. Only the lack of such a predefined contract should result in a larger haircut as the expectation of receiving the collateral quickly after counterparty default would be impaired.

Proposition 4.1 (Legal risk and haircut size). Legal risk arising from non-standard lending contracts that might be incomplete and difficult to enforce should c. p. increase the haircut size.

This legal risk may be accompanied by *operational risk*, which could arise from vague and complex processes both prior to and after counterparty default. Prior to default, a lack of margin regulation would lead to valuation risk (for initial and variation margins, see above). Even if margining is effective, there is a delay between the margin call and the actual adjustment. This operational risk should be reflected in the size of the haircut as it is conditional upon the volatility of the collateral value. The detection and formal declaration of counterparty default and legal processes after default may lead to further delays and potential complications, meaning that it should also be reflected in the haircut to the extent that they are collateral-specific.

Proposition 4.2 (Operational risk and haircut size). Operational risk arising from inefficient margining should c. p. augment the haircut size.

Once the counterparty has defaulted and the collateral is with the lender, it has to be liquidated to redeem outstanding claims. The outcome of this liquidation is affected by *collateral quality*, which is related to the creditworthiness of its issuer, i.e. the credit risk, as well as its liquidity. Hence, the more likely a default of the collateral issuer, the larger the haircut should be. Likewise, lower liquidity of collateral should imply a larger haircut.

Proposition 4.3 (Collateral quality and haircut size). Collateral quality is determined by collateral credit and liquidity risk.

- i) Higher credit risk as reflected by lower credit ratings should c. p. increase the haircut size.
- *ii)* Higher liquidity risk should c. p. result in a higher haircut size.

The liquidity of marketable assets is particularly determined by risk inherent in the market on which they are traded. This *market risk* manifests in adverse movements of the market value of

 $^{^{141}\,\}mathrm{See}$ e.g. European Repo Council 2012 and ECB 2015c for overviews of risk types.

collateral between collateral valuation and liquidation. In the event of a crisis and a shock to market confidence, liquidity deteriorates and sudden adverse price movements are likely. This downside volatility implies a greater risk that the lender is unable to settle its outstanding claims.¹⁴² Market risk is lower for official and active markets than for non-regulated markets since the liquidation of collateral is easy and price movements are more transparent and predictable. Nonetheless, the more complex and scarce an asset, the less it will be traded and the more difficult liquidation is, as well as the predictability of price movements. Furthermore, there might be endogenous price movements depending on the quantity of collateral to be liquidated. Moreover, the valuation of marketable assets gives rise to *valuation risk* as theoretical valuation is prone to errors.¹⁴³

Proposition 4.4 (Market/valuation risk and haircut size). Market risk manifests in the likelihood of adverse movements of the market value between collateral valuation and liquidation. Depending on the market on which assets are traded, valuation methodologies may vary and give rise to valuation risk.

- i) Higher market risk should c. p. increase the haircut size.
- ii) Valuation risk should c. p. result in a larger haircut size.

Furthermore, collateral assets typically carry interest in terms of a coupon payment, which affects market valuation and implies *interest-rate risk*, such that it should therefore be reflected in the haircut. Coupons can be variable or fixed-rate and are usually paid once or twice a year.¹⁴⁴ Consequently, market valuation increases between two coupon payments and drops on the day of coupon payment, because sellers in the market seek compensation for accrued interest between two coupon payments. The market price is called the "dirty price" and it only equals the "clean price" on days of coupon payments but is otherwise higher. Variation margins account for this erratic valuation as the collateral value is based on the dirty price. By contrast, zero-coupon bonds are traded at a discount as the interest payment is implied in the difference between the market price and nominal value. Therefore, the market price deviates from the nominal value for zero-coupon bonds because duration in the MACAULAY 1938 sense is equal to residual maturity. Residual maturity is another characteristic of collateral that drives up the duration and hence the interest-rate risk of the collateral. A longer residual maturity implies a higher degree of uncertainty about the development of the asset price. Hence, a longer residual maturity should result in larger haircuts. Moreover, zero-coupon bonds with a maturity exceeding one year bear more interest-rate risk and should receive a larger haircut.

Proposition 4.5 (Interest-rate risk and haircut size).

i) Interest-rate risk implied in zero-coupon assets (maturity > one year) should c. p. increase the haircut size.

¹⁴² Some of the volatility is absorbed by variation margins. However, at least the difference between the market value at time of counterparty default and the realized liquidation value has to be taken care of by the haircut as no margin calls can be made.

 $^{^{143}\}operatorname{Additionally},$ margining cannot mitigate risk when assets lack a market value.

¹⁴⁴ However, the variability of the coupon rate is irrelevant for the haircut as they continue to be paid to the borrower.

ii) Interest-rate risk owing to longer residual maturity should c. p. increase the haircut size.

Assets that are pledged as collateral may be issued by an entity with close links to the borrower (see Section 3.1.4.1). In case of close links between the borrower and the collateral issuer, counterparty risk and collateral risk are correlated. Such a correlation implies *wrong-way risk* and it is the highest for own-use collateral as collateral also defaults in case of counterparty default. The effectiveness of pledged assets in collateralized lending transaction is diminished with close links and should be deducted by a larger haircut.

Proposition 4.6 (Wrong-way risk and haircut size). Wrong-way risk arising from close links between the borrower and the collateral issuer should c. p. augment the haircut size.

Finally, potential exchange-rate risk has to be taken into account if the lender deems eligible collateral in foreign currency. Exchange rates can move between the dates of collateral valuation and collateral liquidation such that haircuts should be larger the higher the *exchange-rate risk*.

Proposition 4.7 (Exchange-rate risk and haircut size). Exchange-rate risk arising from the pledge of assets denominated in foreign currency should c. p. increase the haircut size.

4.2.2 Development of Applied Haircuts

This section compiles a narrative database on haircuts applied by the Eurosystem in determining the collateral value of assets. General principles of Eurosystem's haircut specification are carved out and a descriptive analysis of the development of haircuts is provided.

4.2.2.1 General Principles of Applied Haircuts

The Eurosystem lays down the haircuts that it applies to eligible assets within its collateral framework. Table 4.2 provides an overview of the dimensions according to which haircuts were differentiated at the end of 2014. Check marks indicate that haircuts were differentiated along that dimension while loops illustrate the opposite. Moreover, check marks in brackets indicate that differentiation was partial, i.e. it did not apply to all assets of the respective liquidity category (LC). Haircuts applied to marketable assets of LCs 1 to 4 were differentiated according to (*i*) CQS, (*ii*) residual maturity and (*iii*) coupon.¹⁴⁵ The haircut applied to LC 5 (which comprised ABSs only) was uniform across residual maturities and irrespective of coupon. Haircuts applied to credit claims were differentiated according to CQS as well as residual maturity in addition to the valuation method (theoretical or outstanding amount). By contrast, haircuts applied to RMBDs were uniform across residual maturities within CQS 1/2 and irrespective of the valuation method and coupon.¹⁴⁶

In addition to valuation haircuts, the Eurosystem applies supplementary haircuts. These can be applied in the form of (i) add-on haircuts, which are simply added to the valuation haircut,

¹⁴⁵ The classification of assets into LCs is based on the issuer and asset type and hence derives liquidity from both the asset type itself as well as the issuer. In October 2013, liquidity categories were renamed in "haircut categories" as their purpose is the assignment of haircuts. The initial name is used for comparability.

 $^{^{146}}$ Only RMBDs within CQS 1/2 were deemed eligible at the end of 2014 (cf. Section 3.1).

as well as *(ii)* valuation markdowns, which are directly applied at the level of valuation.¹⁴⁷ Whenever the two risk control measures are jointly applied, nominal haircuts (i.e. those that can be directly observed in the tables published by the Eurosystem) and effective haircuts (i.e. those that take into account add-on haircuts and valuation markdowns) differ. The effective haircut h_e is given by

$$h_e \equiv h_n + h_a + v - \underbrace{(h_n + h_a) \cdot v}_{\text{interaction term}},\tag{4.8}$$

with h_n being the nominal haircut, h_a the add-on haircut and v the valuation markdown. The effective haircut h_e comprises four components, i.e. (i) the (nominal) valuation haircut, (ii) the add-on haircut, (iii) the valuation markdown, and (iv) an interaction term between the three. This interaction term is further addressed in Section 4.3.

4.2.2.2 Development of Haircut Specification

The development of Eurosystem haircuts is illustrated in Figure 4.1 and explained hereafter (see also ECB 2015c). Haircuts were already an important part of the initial collateral framework in January 2001 when they were specified for tier 1 and tier 2 assets. Haircuts for tier 1 assets were differentiated according to the type of coupon (fixed-rate, floating-rate, inverse floating-rate) and residual maturity (four cluster: 0 to 1 year, 1 to 3 years, 3 to 7 years, and more than 7 years). The same clustering of residual maturity was applied to assets eligible in tier 2 and haircuts were specified for assets with limited liquidity and special features, as well as non-marketable assets.

Figure 4.1: Changes to Eurosystem haircuts

The figure illustrates the development of Eurosystem haircuts from 2001 to 2014. While haircuts became more differentiated prior to the financial crisis, they were only partly adapted to collateral pool as well as market developments during the financial and sovereign debt crisis. Moreover, the Eurosystem introduced supplementary haircuts to specifically mitigate risk from the extended eligibility of assets.



Source: author's compilation

This basic framework was first modified in March 2004, the features of which still prevailed at the end of 2014. Tier 1 assets were allocated into LCs according to the issuer and asset type. Two allocations were made irrespective of the issuer, namely ABSs were included in the lowest LC and

¹⁴⁷ Add-on haircuts were e.g. applied to eligible marketable assets denominated in pounds sterling, yen or US dollars from October 2008 to December 2010. When collateral denominated in these currencies was deemed eligible again in November 2012, the add-on haircut was replaced by a valuation markdown.

jumbo Pfandbrief-style debt instruments (later: jumbo covered bank bonds) were summarized in the second LC. Haircuts were differentiated more precisely according to residual maturity as the number of cluster increased from four to six (0 to 1 year, 1 to 3 years, 3 to 5 years, 5 to 7 years, 7 to 10 years, and more than 10 years). Moreover, haircuts were differentiated according to whether the coupon was fixed or zero. For assets with identical properties in terms of residual maturity and coupon, haircuts were decreasing in liquidity, i.e. haircuts increased from LC 1 to 4. Haircuts applied to tier 1 inverse floating-rate instruments were uniform over LCs, depending only on residual maturity. The haircut applied to debt instruments with variable-rate coupons was that applied to a debt instrument with 0 to 1 year residual maturity and fixed-rate coupon within the relevant LC. Haircuts applied to tier 2 assets were elaborated accordingly by including more cluster of residual maturity as well as differentiating between zero and fixed-rate coupon.

The two-tier system was phased out in May 2007 with the introduction of the Single List (cf. Section 3.1.2), which has drawn a distinction between marketable and non-marketable assets. Marketable assets comprised former tier 1 assets and the respective haircuts were applied. Haircuts imposed to credit claims with fixed-rate coupon were differentiated according to the residual maturity and valuation method, while haircuts applied to those with variable-rate coupon were irrespective of the valuation method.¹⁴⁸ A uniform haircut was imposed to RMBDs.

Amendments to haircuts in response to the financial crisis in October 2008 and November 2008 took the form of supplementary haircuts (cf. Section 3.2). First, a uniform add-on haircut was introduced to all eligible assets rated lower than "single A" as the minimum credit rating threshold was lowered from "single A" to "triple B". Second, a uniform add-on haircut was imposed to newly eligible bank bonds traded on accepted non-regulated markets. Third, a uniform add-on haircut was applied to subordinated assets that were deemed eligible with adequate guarantees. Fourth, a valuation markdown was implemented for theoretically valued subordinated assets. Fifth, the Eurosystem charged a uniform add-on haircut to assets denominated in pounds sterling, yen or US dollars that were deemed eligible for the first time from November 2008 to December 2010. Eligibility was reintroduced in November 2012 and the former uniform add-on haircut was replaced by a valuation markdown, which was larger for yen than for pounds sterling or US dollars.

In February 2009, LCs were rearranged with uncovered bank bonds allocated to the fourth and ABSs to the fifth LC (cf. Section 3.2.5). Haircuts on ABSs were harmonized, i.e. they were no longer differentiated with respect to residual maturity, while a uniform valuation markdown was introduced to theoretically valued ABSs. This valuation markdown was extended to theoretically valued covered and uncovered bank bonds in January 2011. At the same time, the Eurosystem lowered the minimum credit rating threshold to "triple B" in the General Framework and replaced the add-on haircut on such assets by more graduated haircuts. More specifically, credit quality was introduced as a fourth dimension of haircut differentiation (besides liquidity, residual maturity and coupon). However, differentiation according to credit quality was limited as only two cluster of credit quality were taken into account with CQSs 1/2 and 3. The limited differentiation of haircuts with respect to credit quality was also implemented for inverse

¹⁴⁸ This held for credit claims with interest payments with a resetting period shorter than one year. Otherwise, credit claims were considered to have a fixed-rate coupon and the respective haircut was applied.

Table 4.2: Differentiation of Eurosystem haircuts at the end of 2014

The table provides an overview of the dimensions according to which the Eurosystem differentiated haircuts at the end of 2014. Check marks indicate that haircuts were differentiated along the specific dimension, while loops illustrate the opposite. Moreover, check marks in brackets indicate that differentiation was partial, i.e. it did not apply to all assets of the respective LC.

	LC	CC	$\mathbf{\hat{S}}$		Residual Maturity			Coupon ^a V		VALU	ATION	Close Links ^c	Deno- mination ^d		
		1/2	3	0–1	1 - 3	3 - 5	5 - 7	7–10	>10	zero	fixed	$^{\mathrm{theo-}}$ retical $^{\mathrm{b}}$	outstanding amount	S	
E	1	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	0	0	0	\checkmark
ABI	2	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	(\checkmark)	0	(\checkmark)	\checkmark
ARKET ASSE1	3	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	(\checkmark)	0	(\checkmark)	\checkmark
	4	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	(\checkmark)	0	0	\checkmark
Μ	5	\checkmark	\checkmark	0	0	0	0	0	Ο	0	О	(\checkmark)	0	Ο	\checkmark
Non- Marketable Assets	Credit Claims	\checkmark	V	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	Ο	0	\checkmark	\checkmark	0	0
	RMBDs	\checkmark	0	0	0	0	0	0	0	0	Ο	0	0	0	0

^a Debt instruments with variable-rate coupons were treated as debt instruments with a fixed-rate coupon and 0-1 year residual maturity in the relevant LC. Specific haircuts are applied to inverse floating-rate debt instruments.

^b A supplementary haircut in terms of a valuation markdown was applied to theoretically valued covered and uncovered bank bonds as well as ABSs.

^c A supplementary haircut in terms of a valuation markdown was applied to own-use covered bank bonds, which differed across CQSs.

^d As of November 2012, a supplementary haircut in terms of a valuation markdown was applied to all marketable debt instruments denominated in foreign currency. It differed between yen and other foreign currencies, i.e. pounds sterling and US dollars. Source: author's compilation.

floating-rate debt instruments and credit claims with fixed interest payments. The lowering of the credit rating requirement for homogenous ABSs in June 2012 was accompanied with the introduction of larger haircuts for these newly eligible ABSs within CQS 3, which were differentiated with respect to underlying assets.¹⁴⁹ Furthermore, a valuation markdown was imposed to own-use covered bank bonds in October 2013, which was differentiated according to credit quality.

Section 3.2.2 revealed that the collateral criteria of bonds issued or guaranteed by governments were subject to substantial amendments and that country-specific criteria were introduced in May 2010 when the minimum credit rating was suspended for bonds issued or guaranteed by the government of Greece. This step was repeated for Ireland (April 2011), Portugal (July 2011) and Cyprus (May 2013). Specific and considerably larger haircuts were adopted for government bonds and government-guaranteed bank bonds as well as non-financial corporate bonds issued in Greece and Cyprus, but not for Ireland and Portugal. The specific haircuts were revised and lowered throughout 2014.

4.2.3 Descriptive Analysis of Applied Haircuts

Having carved out general principles and compiled a narrative database of the development of haircuts applied by the Eurosystem, this section elaborates on the quantitative development of haircuts. It aims to draw a comprehensive picture of the quantitative development of haircuts and takes into account the different components of effective haircuts of the Eurosystem: (i) the (nominal) valuation haircut h_n , and (ii) supplementary haircuts, i.e. add-on haircuts h_a and valuation markdowns v.

4.2.3.1 (Nominal) Valuation Haircuts

The following analysis of nominal haircuts suggests four broad phases of haircut development:

- PRE-CRISIS PHASE: April 2004 to September 2008;
- FINANCIAL CRISIS PHASE: October 2008 to December 2010;
- SOVEREIGN DEBT CRISIS PHASE: January 2011 to September 2013;
- REMITTENT SOVEREIGN DEBT CRISIS PHASE: October 2013 to December 2014.

The timing of the phases suggests that the Eurosystem responded to the onsets of the financial and the sovereign debt crisis with delays, whereby the response was faster for the former rather than the latter. Despite being the major risk mitigation tool, adjustments to haircuts were scarce and infrequent, as well as not always being specifically related to events during the phases (see also NYBORG 2015). For instance, haircut adjustments during the sovereign debt crisis phase were only secondarily addressed to government-related assets.

Figure 4.2 provides a comprehensive picture of the quantitative development of nominal haircuts applied by the Eurosystem from March 2004 to December 2014 to assets in LCs 1 to 5,

¹⁴⁹ For ABSs that did not have two ratings of at least "single A", the haircut was differentiated with respect to underlying assets, whereby those ABSs backed by commercial mortgages were subject to a larger valuation haircut than other eligible ABSs. The haircuts were lowered in October 2013.

inverse floating-rate debt instruments, credit claims as well as RMBDs, arranged by CQSs.¹⁵⁰ Based on the information illustrated in Figure 4.2, haircuts applied during the four phases are characterized according to the following two properties:¹⁵¹ first, the average haircut seizes on the level of haircuts; and second, the haircut range illustrates the spread of haircuts between the smallest and largest haircut. The range is defined as the difference between the haircut applied to assets with the shortest (0 to 1 year) and the longest (more than 10 years) residual maturity.

Pre-Crisis Phase The Eurosystem did not change haircuts within this first phase. Only assets of good quality, i.e. rated within CQS 1/2, were eligible and Table 4.3 indicates that initial average haircuts as well as haircut ranges were moderate and increased with diminishing liquidity of assets. Haircuts applied to ABSs were still differentiated according to residual maturity, while a uniform haircut of 20% was applied to all eligible RMBDs.

Table 4.3: Nominal valuation haircuts during the pre-crisis phase

The table characterizes haircuts applied prior to the financial crisis, indicating that initial average haircuts as well as haircut ranges were moderate and increased with decreasing liquidity.

	Average Haircut (in percentage points)	HAIRCUT RANGE (in percentage points)
LC 1	2.83	5
LC 2	4.08	6.5
LC 3	5	7.5
LC $5^{\rm a}$	6.25	10
INVERSE FLOATER	12.17	23
Credit Claims	11.5	10
RMBDs	20	0

^a ABSs made up for LC 4 at that time are indicated as LC 5 into which they were classified in February 2009.

Source: author's calculation.

Financial Crisis Phase Turmoil in financial markets predominantly impaired the quality of bank bonds, which were comprised in LCs 2 (Jumbo covered bonds) and 3 (traditional covered bonds and uncovered bonds) at the time. Moreover, the market for ABSs almost completely dried up. Higher credit risk as well as lower liquidity owing to the distressed market environment should—according to Proposition 4.3—have called for larger haircuts on bank bonds and ABSs.

While the Eurosystem promptly reacted to the turmoil by adapting its collateral framework in October 2008 and the subsequent months, amendments to haircuts were initially small. The lowering of the minimum credit rating threshold was accompanied by applying an add-on haircut of 5% to the newly eligible assets of lower quality within LCs 1 to 3. However, initial haircuts

¹⁵⁰ The figures illustrate the quantitative development of nominal valuation haircuts imposed on collateral with fixed-coupon payments. For credit claims, the average of the haircut applied to theoretically valued credit claims and those valued according to the outstanding amount is shown. The add-on haircut applied to eligible collateral in CQS 3 from October 2008 to December 2010 is treated as a nominal valuation haircut for the sake of comparability. All other supplementary haircuts are neglected.

¹⁵¹ Haircut properties are given for collateral with fixed-rate coupon and theoretical valuation in case of credit claims.





Source: author's compilation; European Central Bank.



Figure 4.2: Quantitative development of nominal valuation haircuts (cont.)

were not adjusted until February 2009 when the Eurosystem reacted to the changing market environment by reclassifying uncovered bank bonds in LC 4 and ABSs in LC 5. The average haircut applied to uncovered bank bonds was raised from 5% to 10%, while the range remained at 7.5%, i.e. uncovered bank bonds were effectively deducted by another add-on haircut of 5% irrespective of residual maturity as of February 2009. Haircuts applied to ABSs (LC 5), which were hitherto differentiated according to residual maturity, were harmonized to the haircut formerly applied to ABSs with the longest residual maturity (more than ten years). Therefore, the haircut increase was greater the shorter the residual maturity of ABSs, while the average haircut increased from 6.25% to 12%.

Sovereign Debt Crisis Phase While the European sovereign debt crisis intensified in the course of 2010, the Eurosystem did not amend haircuts until January 2011. The debt crisis mainly affected government bonds that were eligible as collateral within LC 1 as well as governmentguaranteed assets and bank bonds eligible in LCs 2 to 4. Larger haircuts should have been imposed on these assets according to Proposition 4.3 as credit quality and liquidity deteriorated. However, changes were not specifically directed at government-related assets but predominantly owed to the revision of the uniform add-on haircut of 5% applied to eligible assets of lower credit quality (CQS 3) in LCs 1 to 4. The uniform add-on haircut was replaced by more graduated haircuts, which were differentiated according to residual maturity and coupon (see Section 4.2.2.2). Table 4.4 provides an overview of the modifications that can be tracked in Figure 4.2. The comparison of Tables 4.3 and 4.4 indicates that haircuts applied to government bonds (incorporated in LC 1) were not subject to change. By contrast, modifications implied substantial increases in haircuts applied to assets within LCs 2 to 4 in CQS 3, which were larger in absolute terms for higher LCs (i.e. less liquid assets). Residual maturities within LCs were treated differently, as indicated by the pattern of change in Table 4.4 and tracked in Figure 4.2. The general impression is that absolute increases in haircuts mainly reflected an inverse u-shaped pattern, implying that increases were large for assets of medium (3-5 and 5-7 years) but small or zero for those with short (0-1 and 1-3 years) and long (7-10 and larger than 10 years) residual maturity. Average haircuts increased for all assets aside from those in LC 1 and high-quality assets (i.e. CQS 1/2) in LC 2. The credit quality and liquidity of bonds issued or guaranteed by several countries further impaired throughout 2011 with the aggravation of the sovereign debt crisis. The Eurosystem thus suspended (or extended the suspension) of the minimum credit rating for bonds issued or guaranteed by Greece, Ireland, Portugal and Cyprus. Declining credit quality and liquidity of government-related collateral in those countries was accompanied by specific modifications to haircuts applied to government-related assets from Greece and Cyprus. For Greece, the average haircut increased from 9.4% to 43.3% (government bonds) and from 22% to 53.3% (government-guaranteed bonds). For Cyprus, the average haircut increased from 9.4% to 37.5% (government bonds) and from 22% to 47.5% (government-guaranteed bonds).

Remittent Sovereign Debt Crisis Phase The Eurosystem reacted cautiously to the calming of financial markets, which suggested the remittance of the sovereign debt crisis as it did not modify haircuts until the end of 2014. The improved market environment in terms of improved

Table 4.4: Changes to nominal valuation haircuts during the debt crisis phase

The table provides an overview of the changes to nominal haircuts during the sovereign debt crisis phase, which can be tracked in Figure 4.2. Changes implied substantial increases in haircuts applied to assets within LCs 2 to 4 in CQS 3. Increases were larger in absolute terms for higher LCs, i.e. less liquid assets. Residual maturities within LCs were treated differently, as indicated by the pattern of change.

		HAIRCUTS I		HAIRCUTS IN CQS 3			
	Øa	Range ^a	Pattern	Øa	Range ^a	Pattern	
LC 1	2.8	5	no change	7.8	5	no change	
LC 2	4.1	6.5	no change	14.9	14	inverse u-shaped increase ^b	
LC 3	5.9	9.5	increase	23	21.5	inverse u–shaped increase	
LC 4	11.6	10.5	strict increase	32.7	24.5	inverse u–shaped increase	
LC 5	16	0	increase ^c	n/a	n/a	n/a	
Inverse Floater	17.5	20.5	inverse u–shaped increase ^d	56.3	48.5	uniform increase	
Credit Claims	15.1	12.5	inverse u–shaped increase	33.3	25	inverse u–shaped increase	
RMBDs	24	0	increase ^c	n/a	n/a	n/a	

^a In percentage points.

^b The inverse u-shaped pattern implies that the change is large for assets with medium (3-5, 5-7) but small or zero for assets with short (0-1, 1-3) and long (7-10, >10) residual maturity.

^c Irrespective of residual maturity.

^d Except for inverse floaters with residual maturity between 0 to 1 year. Source: author's compilation.

credit quality and liquidity of government-related assets should have resulted in a reduction of haircuts (Proposition 4.3). In fact, haircuts were modified for all LCs as well as credit claims and RMBDs in October 2013. However, modifications were small in magnitude relative to increases in haircuts performed during the previous (debt crisis) phase. Unlike previously, modifications were also directed at government bonds (or LC 1, respectively) such that haircuts were lowered for government bonds of high credit quality (CQS 1/2) but strictly increased for lower-quality assets (CQS 3). Exceptions were ABSs (LC 5) for which the uniform haircut was lowered by more than it was increased in the previous phase, as well as RMBDs, for which the haircut was further increased. The overall pattern of modifications was more divergent than previously, as Figure 4.2 indicates and Table 4.5 summarizes. Haircuts were predominately modified in a u-shaped pattern, i.e. changes were mainly directed to assets of shorter and longer residual maturity. As modifications were small in magnitude, average haircuts and ranges did not vary substantially relative to the previous phase. Moreover, the Eurosystem considerably lowered the specific haircuts applied to bonds issued or guaranteed by the government of Greece. The average haircut on Greek government bonds decreased to 21.8% (from 43.4%) and the haircut applied to bonds guaranteed by the Greek government fell to 30.3% (from 53.3%). Special haircuts for collateral related to the Cypriot government were not altered.

Table 4.5: Changes to nominal valuation haircuts during the remittent debt crisis phase

The table provides an overview of the changes to nominal haircuts during the remittent sovereign
debt crisis phase, which can be tracked in Figure 4.2. Modifications were small in magnitude
relative to haircut increases during the previous phase. Changes were also directed at government
bonds (or LC 1, respectively). Inverse floaters are neglected as no changes were made.

	Haircuts in CQS $1/2$			HA	Haircuts in CQS 3			
	Øa	Range ^a	Pattern	Øa	Range ^a	Pattern		
LC 1	2.2	4.5	u-shaped decrease ^b	9.4	7	increase		
LC 2	3.5	7	constant decrease ^c	14.9	15.5	u-shaped modification ^d		
LC 3	4.6	9.5	u-shaped decrease	19.5	19.5	decrease		
LC 4	11.6	10.5	no change	30.1	24.5	u-shaped decrease		
LC 5	10	0	decrease ^e	22	0	decrease ^e		
Credit Claims	15.1	12.5	u-shaped modification ^f	33.3	25.5	u-shaped increase		
RMBDs	39.5	0	increase ^e	n/a	n/a	n/a		

^a In percentage points.

^b The u-shaped pattern implies that the change in haircuts is small or zero for assets with medium (3-5, 5-7) but large for assets with short (0-1, 1-3) and long (7-10, >10) residual maturity.

 $^{\rm c}$ Except for assets with residual maturity of more than 10 years, for which the haircut was increased. $^{\rm d}$ Increase only for assets of the shortest (0-1) and longest (>10) residual maturity.

^e Irrespective of residual maturity.

^f Increase only for assets of with residual maturity 0-1, 7-10 and more than 10 years. Source: author's compilation.

4.2.3.2 Supplementary Haircuts

Section 4.2.2 has revealed that besides nominal haircuts, the Eurosystem also applies supplementary haircuts. Accordingly, this section briefly reviews and summarizes the quantitative development in the application of the two types of supplementary haircuts: (i) add-on haircuts, i.e. h_a , and (ii) valuation markdowns, i.e. v. Table 4.6 provides an overview of add-on haircuts and valuation markdowns applied by the Eurosystem from 2001 to 2014 together with the affected collateral, haircut size and the period of application. It reveals that the Eurosystem applied add-on haircuts in particular after the onset of the financial crisis and the following quantitative and qualitative broadening of eligible collateral. The Eurosystem later replaced former add-on haircuts and shifted towards the application of valuation markdowns.

4.3 Evaluation and Shortcomings of the Haircut Specification

Building on the previous analysis of haircuts, Eurosystem haircut application is evaluated against the general principles of haircut specification carved out in Section 4.2.1. Based on this evaluation, shortcomings of Eurosystem haircuts are subsequently pointed out.¹⁵²

 $^{^{152}\,\}mathrm{See}$ also NyBorg 2015 for an assessment of Eurosystem haircuts.

Table 4.6: Overview of supplementary haircuts applied by the Eurosystem

The table provides an overview of supplementary haircuts, i.e. add-on haircuts and valuation markdowns, applied by the Eurosystem from 2001 to 2014. It reveals that the Eurosystem applied add-on haircuts in particular during the financial crisis but later turned to valuation markdowns.

Affected Collateral	Type	Size	Application Period
eligible assets rated lower than "single A"	add-on haircut	5%	25/10/2008-31/12/2010
bank bonds traded on non- regulated markets	add-on haircut	5%	25/10/2008-31/12/2010
foreign currency collateral	add-on haircut	8%	14/11/2008 - 31/12/2010
subordinated assets	add-on haircut	10%	14/11/2008-31/12/2010
theoretically valued subordi- nated assets	valuation markdown	5%	14/11/2008-31/12/2010
theoretically valued ABSs	valuation markdown	5%	01/02/2009-today
theoretically valued covered and uncovered bank bonds	valuation markdown	5%	01/01/2011-today
foreign currency collateral	valuation markdown	$16\%/26\%^{\rm a}$	09/11/2012-today
own-use covered bank bonds	valuation markdown	$8\%/12\%^{\rm b}$	01/11/2013-today

 $^{\rm a}$ 16% for GBP and USD; 26% for JPY. $^{\rm b}$ 8% for bank bonds in CQS 1/2 and 26% for those in CQS 3. Source: author's compilation.

4.3.1 Evaluation of the Haircut Specification

Proposition 4.1 explored the effect of legal risk on haircut size and postulated that legal risk arising from non-standard lending contracts calls for larger haircuts as these contracts might be incomplete or difficult to enforce. The Eurosystem uses the European Master Agreement for its relevant transactions (ECB 2006b) and ensures that no legal risk arises from the contractual framework of a transaction. Hence, there is no necessity for larger haircuts owing to legal risk.

Proposition 4.2 addressed the link between operational risk and haircuts and claimed that operational risk from e.g. inefficient margining should imply larger haircuts. The Eurosystem applies variation margins according to which counterparties have to maintain the collateralization of principal over time (see Section 4.1). Recalculation is performed daily and includes accrued interest. Usually, a variation margin of 0.5% of principal is used as a trigger point for a margin call.

Proposition 4.3 elaborated on the bearing of collateral quality on haircut size and claimed that the collateral of lower credit quality and liquidity should have larger haircuts. The Eurosystem effectively distinguishes three CQSs in its collateral framework. Haircuts do not continuously increase with diminishing credit quality; instead, they jump when an eligible asset moves from CQS 1/2 to CQS 3, and drop vice versa. Hence, the haircut application of the Eurosystem only partially meets Proposition 4.3 with respect to the effect of credit quality on haircut size. Moreover, the Eurosystem takes collateral liquidity into account. Haircuts are differentiated according to five LCs and different haircuts are applied to eligible non-marketable assets. Assets are assigned to LCs depending on their type and issuer. Hence, the development of haircuts applied to marketable assets is only partially differentiated according to liquidity.

Proposition 4.4 considered the effect of market risk on haircut size and inter alia claimed that assets subject to valuation risk should bear larger haircuts. The Eurosystem applies a valuation markdown of 5% to theoretically valued covered and uncovered bank bonds as well as ABSs to address valuation risk. In addition, different haircuts are applied to credit claims depending on the valuation method. The Eurosystem recognized that bank bonds traded on non-regulated markets would bear higher market risk and thus it applied an add-on haircut to these assets from October 2008 to December 2010. However, the Eurosystem refrained from applying a supplementary haircut when bank bonds traded on non-regulated markets were again deemed eligible as of January 2012.

Proposition 4.5 asked for larger haircuts in case of higher interest-rate risk owing to e.g. the longer residual maturity of collateral. Eurosystem haircuts are differentiated with respect to six cluster of residual maturity. Generally, haircuts increase in residual maturity to recognize the higher interest-rate risk associated with longer duration. Analogous to haircut differentiation according to credit quality and liquidity, residual maturity does not imply a continuous effect on haircut size as haircuts only change when residual maturity moves from one cluster to the other. Hence, Proposition 4.5 is met with restrictions. The Eurosystem applies haircuts to zero-coupon assets that are the same or larger than to fixed-coupon assets. No difference is made in case of collateral maturing within one year as the duration does not significantly differ for these assets. Variable-rate coupons are treated like fixed-rate coupons and haircuts are applied according to residual maturity, which corresponds to the resetting period of the variable-rate coupon.

Proposition 4.6 dealt with the effect of wrong-way risk on haircut size. It called for larger haircuts the closer the connection between the counterparty and collateral issuer. In fact, the Eurosystem applies a supplementary haircut in terms of a valuation markdown of 8% to own-use covered bank bonds rated in CQS 1/2 and 12% in CQS 3. However, other assets eligible with close links (especially own-use government-guaranteed uncovered bank bonds) are not subject to a supplementary haircut.

Finally, Proposition 4.7 addressed exchange-rate risk and claimed larger haircuts when collateral is denominated in foreign currency. The Eurosystem deemed eligible collateral denominated in pounds sterling, yen or US dollar from November 2008 to December 2010 and again as of November 2012. It applied a uniform add-on haircut during the first period, which was replaced by a differentiated valuation markdown in November 2012.

4.3.2 Shortcomings of the Haircut Specification

The previous section revealed that the Eurosystem oversimplifies haircut specification as in particular Propositions 4.3 and 4.5 are only met to a limited extent. This section argues that the application of Eurosystem haircuts is associated with the following four flaws: (i) the application of the first-best rule; (ii) the remoteness from market conditions and infrequent revision; (iii)
the sequential application of valuation markdowns and haircuts; and (iv) the clustering of assets properties.

4.3.2.1 Application of First-Best Rule for Credit Assessment

Credit assessment is crucial for risk control as haircuts are applied based on the assessed credit quality. Section 3.1.4.3 showed that the Eurosystem assesses credit quality of eligible marketable assets (except for ABSs) based on a first-best rule. Accordingly, the Eurosystem only takes into account the best rating when credit ratings of rating agencies differ. Owing to the application of the first-best rule, the Eurosystem ignores available information on credit quality, which should be taken into account for risk control. Moreover, it incentivizes rating agencies to be pivotal, i.e. to exert influence on whether assets are eligible and/or on the applied haircut. This can give rise to an artificial "race to the top" of ratings irrespective of the actual credit quality and facilitate "ratings shopping" by counterparties as it was generally observed in the preface of the financial crisis (e.g. SKRETA and VELDKAMP 2009, BECKER and MILBOURN 2011, BOLTON et al. 2012). One facile way to alleviate this problem would be to apply a second-best or even last-best rule in case of differing rating assessments, which would imply a requirement for a certain minimum number of assessments. The second-best rule together with the requirement of at least two credit ratings was already applied to ABSs as of October 2010.

4.3.2.2 Remoteness from Market Conditions and Infrequent Revision

The Eurosystem differentiates haircuts along various dimensions such that they reflect assetspecific properties such as credit quality and liquidity (albeit to a limited extent). However, haircuts fail to reflect the latest developments in market conditions owing to their low level of market input and infrequent revision. Section 4.2.3 revealed that haircuts were revised infrequently and often lagged in response to developments in financial markets and amendments to collateral criteria. NYBORG 2015 documents an average time between broad haircut revision of more than three years. Indeed, this impression is confirmed by Figure 4.2.

4.3.2.3 Sequential Application of Valuation Markdowns and Haircuts

Another shortcoming is the sequential application of risk control measures. Nominal and add-on haircuts are applied after valuation markdowns are imposed, which are applied directly when collateral valuation takes place as they aim to alleviate uncertainty from asset valuation. Therefore, it makes sense to apply valuation markdowns as a precautionary means to theoretically valued assets. However, Table 4.6 indicates that valuation markdowns replaced former add-on haircuts and were also applied to foreign currency collateral irrespective of the valuation method. This is precarious as the sequential application gives rise to interaction between the valuation markdown and haircuts (see also Section 4.2.2.1). This interaction term is reflected by

$$\Phi = (1 - h_n - h_a)(1 - v)\Omega = (1 - h_n - h_a - v + \underbrace{(h_n + h_a) \cdot v}_{\text{interaction term}})\Omega.$$
(4.9)

Interaction effectively augments the collateral value of an asset as it alleviates the applied haircut. This alleviating effect is greater the larger the haircut applied to the assets, i.e. the lower the collateral quality. The interaction term would vanish if valuation markdowns and haircuts were applied simultaneously, with larger haircuts resulting accordingly.

4.3.2.4 Clustering of Asset Properties

The Eurosystem recognized four determinants of nominal haircuts applied to eligible marketable assets at the end of 2014: (i) coupon, (ii) liquidity, (iii) credit quality and (iv) residual maturity. The haircut is differentiated according to whether the coupon is fixed, zero or variable. Therefore, all potential coupon structures are considered and all available information affects the haircut size. However, this does not hold for the other three determinants, for which information is summarized into five (liquidity), two (credit quality) and six (residual maturity) cluster. This reflects the Eurosystem's endeavor to specify haircuts that take account "of the main [...] risk differences, in as few dimensions as possible" (ECB 2015c, p. 29). Neglecting coupon structure for simplicity, the nominal haircut h_n in Equation (4.8) can be formulated as $h_n(q,\kappa,\xi)$, i.e. it depends on credit quality q, liquidity κ and residual maturity ξ . The effect of q, κ and ξ on h_n as assumed by the Eurosystem in haircut specification is stylized in Figure 4.3. The obvious shortcoming is that q and κ are discrete arguments and although ξ is generally a continuous argument, it has to be discretized to be applicable. Hence, a continuous function $h_n(q,\kappa,\xi)$ can neither be derived nor would it be tractable. However, it is apparent that the Eurosystem clusters asset properties, which implies a concentration and negligence of information. This information would be available to elaborate more graduated haircuts as exemplified in gray, which would imply more accurate differentiation. The problems associated with clustering asset properties like liquidity and credit quality is addressed in the following and emphasized in BUITER and SIBERT 2005. The study ascribes low interest spreads on government bonds in the Eurozone prior to the sovereign debt crisis to the application of equal haircuts to government bonds of distinct credit quality.

Figure 4.3: Clustering of asset properties in haircut specification

The figure stylizes the relation between the nominal haircut h_n and asset liquidity κ , asset residual maturity ξ as well as asset credit quality q, as assumed by the Eurosystem in haircut specification. It depicts the broad clustering of asset properties as haircuts are harmonized for e.g. up to seven credit rating notches (S&P scale). Moreover, more graduated haircut specification is indicated in gray.



Liquidity The Eurosystem clusters different asset types into five liquidity categories (LCs). By the end of 2014, LC 1 comprised two asset types, LC 2 four types, LC 3 three types, LC 4 two types and LC 5 only ABSs. Hence, the Eurosystem harmonized haircuts within LCs for up to four asset types, assuming that the assets would be equally liquid. Taking into account that asset types may differ in liquidity, haircuts may have been too large for some assets and too small for others. Generally, the application of harmonized haircuts to assets of different liquidity meets Proposition 4.3, albeit with restrictions.

Residual Maturity Likewise, the Eurosystem consolidates residual maturity into six clusters with a range of up to four years in each cluster (disregarding the cluster ">10"). Figure 4.4 depicts the development of the distribution of assets within clusters from May 2007 to December 2013 measured by the deviation of the median residual maturity from the cluster mean.¹⁵³ The figure indicates that the distribution of assets within clusters was usually right-skewed, i.e. more assets had a residual maturity shorter than the cluster mean. Distribution of assets developed volatile during the crises but returned to be more symmetric again at the end of 2013. The problem arising from the asymmetric distribution of assets originates from the uniform haircut applied within each cluster. If the haircut was specified according to the cluster mean, a rightskewed distribution would imply more assets being deducted by an overly-large haircut. While this would be advantageous from a risk mitigation perspective, it would have adverse effects on collateral availability within the Eurozone. By contrast, the application of a uniform haircut which reflects the cluster mean together with a left-skewed distribution of assets would give rise to insufficient risk control. In general, the application of uniform haircuts to assets of different residual maturity gives rise to haircuts that do not reflect the true collateral risk and thus Proposition 4.5 is only met with restrictions.

Credit Quality Asset properties are clustered to the largest degree for credit quality. The Eurosystem defines three CQSs that comprise credit ratings from "triple A" to "triple B" (cf. Section 3.1.4.3).¹⁵⁴ Figure 4.5 exemplifies the harmonization of credit ratings for the rating scale of S&P. The four credit rating notches from "AAA" to "AA-" are summarized into CQS 1, the three subsequent notches from "A+" to "A-" are collected in CQS 2 and finally the last three notches above non-investment grade, i.e. "BBB+" to "BBB-", are aggregated into CQS 3. For the purpose of haircut application, the Eurosystem aggregates CQS 1 and 2 into one credit quality cluster and uses CQS 3 as the other cluster. Thus, assets of seven and three credit rating notches are subsumed into two credit quality clusters. For instance, an eligible covered bank bond with a fixed-rate coupon and residual maturity of two years that is rated "AAA" and features (according to S&P) a "capacity to meet [the] financial commitment [which] is extremely strong" (cf. S&P 2012, p. 5) receives the same haircut as a covered bank bond with fixed-rate coupon and residual maturity of two years that provide and use the same haircut as a covered bank bond with fixed-rate coupon and residual maturity of the provide and residual maturity of "single A" and is "susceptible to the adverse effects of changes in circumstances and economic conditions" (cf. S&P 2012, p. 5).

 $^{^{153}\,\}mathrm{Cluster}$ ">10" is neglected as an outlier.

¹⁵⁴ The introduction of the ACCs framework and the suspension of the minimum credit rating implied that the Eurosystem implicitly introduced another CQS, i.e. CQS 4.

The figure elaborates on the development of the distribution of eligible marketable assets within the residual maturity clusters from 2007 to 2013. The distribution is important as uniform haircuts are applied within each cluster. It illustrates the deviation of the median residual maturity within each cluster from the cluster mean. It indicates that the distribution within cluster was predominantly right-skewed, i.e. more assets had a residual maturity shorter than the natural cluster mean. Moreover, distributions considerably fluctuated over time.



Source: author's calculation; European Central Bank, Eligible Assets Database

In general, the application of uniform haircuts to assets of different credit quality violates Proposition 4.3, which inter alia calls for higher credit risk to be reflected by larger haircuts. For risk mitigation, a uniform haircut would not be worrisome if it mirrored the risk profile of the lowestrated asset within each cluster. Nevertheless, haircuts would subsequently be too cautious and would restrict collateral availability.

Figure 4.6 reconstructs the distribution of eligible marketable assets across credit rating notches for one day (6 March 2015).¹⁵⁵ Both short- and long-term as well as issue and issuer ratings of eligible marketable assets were collected from Bloomberg and the rules of the Eurosystem for determining the pivotal credit rating were adapted (see Section 3.1.4.3). Accordingly, the first-best rule was applied in case of distinct ratings and issue ratings were preferred over issuer ratings (when both ratings were available).¹⁵⁶ Rating notches are given on the horizontal axis (from low to high) and the vertical axis depicts the frequency of assets for each notch. On 6 March 2015, 76.7% of eligible marketable assets were rated within CQS 1/2 and 15.7% in CQS 3. No issue or issuer rating was available for the residual (7.6%) on the specific day. The figure reveals that the distribution of assets within CQSs considerably varied. Within CQS 3, the frequency of assets increases in credit quality. This pattern is also observable for CQS 1/2, although a large fraction of assets in CQS 1/2 was rated as "A" and only very few assets were rated "AA+" or "AA-". The distribution implies that the application of a uniform haircut to assets in the same CQS leaves a considerable fraction of assets with an overly-small or -large

¹⁵⁵ Data providers like Bloomberg only make available ratings on a daily basis but no historical record is provided. See also Section 3.3.4.

 $^{^{156}\,\}mathrm{Ratings}$ were translated into the rating scale of S&P for the sake of simplicity.

Figure 4.5: Clustering of asset credit quality (exemplified for the rating scale of S&P)

This figure stylizes the clustering of asset credit quality for the rating scale of S&P. It illustrates how the Eurosystem clusters asset credit quality in two steps: first, rating notches are subsumed into CQSs; and second, CQSs are summarized into credit quality clusters.



haircut. If e.g. the uniform haircuts in CQSs 1/2 and 3 were set according to the risk profile of assets rated as "AA" and "BBB", respectively, 33.6% of all assets were deducted by a haircut which would have been too high and 37.2% with an overly-small haircut. The haircut would have hedged collateral default risk correctly for only 21.6% of eligible marketable assets.

Figure 4.6: Distribution of eligible marketable assets within CQSs (on 6 March 2015)

The figure identifies the credit quality of eligible marketable assets in terms of rating notches on 6 March 2015. Accordingly, it reconstructs the distribution of eligible marketable assets across CQSs for one day. Both short- and long-term as well as issue and issuer ratings were collected from Bloomberg and the rules of the Eurosystem for determining the pivotal credit rating were applied. The figure indicates that assets were distributed asymmetrically within CQSs.



4.4 Potential Effects of the Simplified Haircut Specification

Simplified haircut specification gives rise to two potential effects. First, the Eurosystem would increase its risk exposure by lowering collateral criteria if haircuts did not fully address the

additional collateral risk. Second, simplified haircuts can lead to adverse selection of collateral, i.e. counterparties pledging assets of relatively low quality with the Eurosystem and keeping collateral of relatively good quality for alternative utilization. Therefore, the second effect fuels the first one when the Eurosystem relaxes collateral criteria and attracts assets of low quality but insufficiently hedges the additional collateral risk.

Figure 4.7 seizes upon the second potential effect of simplified haircut specification. The figure contrasts the development of the value-weighted average haircut applied to eligible marketable assets and the value-weighted average haircut imposed to collateral pledged with the Eurosystem. It reveals that the average haircut on eligible marketable assets increased from 3% in May 2007 to 6.2% in December 2013. By contrast, the increase of the average haircut on pledged collateral was considerably stronger, from 4.2% in August 2007 to 14.2% in April 2013.¹⁵⁷

Figure 4.7: Average haircut to eligible marketable and pledged assets

The figure shows the development of two haircuts: first, the value-weighted average haircut applied to eligible marketable assets; and second, the value-weighted average haircut imposed to pledged assets. It reveals that average haircuts developed asymmetrically as the average haircut to pledged assets increased more than the average haircut to eligible marketable assets.



Source: author's calculation; European Central Bank, Eligible Assets Database; ECB 2013c.

Although both average haircuts are affected by the Eurosystem's changes to haircuts, their development was very divergent. The possible explanation for this divergent development is twofold. On the one hand, the average haircut on collateral pledged with the Eurosystem also reflects non-marketable assets that have been increasingly pledged over recent years (see Section 5.1). Given that non-marketable assets are subject to relatively large haircuts (see Section 4.2.3), their progressive pledge contributed to the increase in the average haircut. On the other hand, more assets of relatively low quality (and large haircuts) were likely pledged with the Eurosystem, owing to the adverse selection of collateral,¹⁵⁸ which is referred to as Gresham's

 $^{^{157}\,\}mathrm{See}$ ECB 2013c for a detailed description.

¹⁵⁸ This view is supported by CHEUN et al. 2009, FECHT et al. 2015, NYBORG 2015 and ECB 2013c, where the stronger increase of the average haircut on pledged collateral is attributed to "changes in the composition of assets used by counterparties [...]" (ibid., p. 83).

Law of Collateral (GLOC).¹⁵⁹ The following chapter elaborates on the adverse selection of collateral in due consideration of the simplified specification of Eurosystem haircuts.

¹⁵⁹ GLOC is adapted from "Gresham's law", which describes the tendency for bad money to drive out good money. For Gresham's law, see e.g. VON HAYEK 1976 and for GLOC, see EWERHART et al. 2006, CHAILLOUX et al. 2008a,b, EWERHART and TAPKING 2008, SINGH 2013, FECHT et al. 2015 and in particular the following chapter.

5

Collateral Pledged with the Eurosystem¹⁶⁰

This chapter investigates the effects of the development of the Eurosystem collateral framework and risk control. Specifically, the effects on the development of the quality of collateral pledged with the Eurosystem are scrutinized. Information is only published on the quantity rather than quality of pledged collateral. However, the latter is crucial as it is fundamental to the risk mitigation of the Eurosystem. Novel insights are generated based on the estimated credit quality of pledged collateral. As estimates are inter alia derived from the inferred credit quality of eligible marketable assets (Chapter 3), they are restricted to the aggregation of credit quality into CQSs. A model is elaborated that provides further indication concerning the development of collateral quality and facilitates the refinement of collateral quality to credit rating notches, i.e. the distribution of collateral within CQSs. Taking into account certain peculiarities of the Eurosystem collateral framework and risk control, the model shows that the Eurosystem is prone to adverse selection of collateral, which is referred to as Gresham's Law of Collateral (GLOC). According to GLOC, the Eurosystem likely attracts collateral of relatively low quality while good collateral is used in the repo market. Moreover, the model finds that adverse selection of collateral is intensified and beyond the control of the Eurosystem in case of positively distorted credit ratings. The chapter is structured as follows. Section 5.1 refines published information on pledged collateral, estimating how pledged collateral has shifted towards crisis-stricken countries and deteriorated in credit quality in recent years. Section 5.2 elaborates on the institutional background and analyzes collateral criteria and haircuts in the European repo market, the Europystem's main competitor for collateral. Section 5.3 presents the model of adverse selection of collateral. Finally, Section 5.4 revisits the development of collateral quality, taking into account the finding of adverse selection of collateral. It suggests that the risk exposure of the Eurosystem has substantially increased over recent years as collateral of low credit quality for which risk was insufficiently hedged was adversely selected to the Eurosystem.

¹⁶⁰ Section 5.1 bases on EBERL and WEBER 2015 and Sections 5.2 and 5.3 on EBERL and WEBER 2013.

5.1 Development of Collateral Pledged with the Eurosystem

Refinancing credit granted by the Eurosystem has substantially changed in terms of quantity and duration, as depicted in Figure 2.5. Furthermore, Figure 5.1 shows the shift in the geographical distribution of granted refinancing credit for selected countries from May 2007 to December 2014. While the majority of refinancing credit was granted in Germany prior to the financial crisis, it shifted over time towards crisis-stricken countries where banking sectors were hit by both the financial and the sovereign debt crisis. Therefore, refinancing credit considerably changed along three dimensions, i.e. (i) quantity, (ii) duration and (iii) geographical distribution.

Figure 5.1: Eurosystem refinancing loans by country

The figure complements Figure 2.5 and illustrates where Eurosystem refinancing loans were accommodated, indicating a shift in the geographical distribution of refinancing loans from 2007 to 2014. While they were granted predominantly in Germany prior to the financial crisis, the majority were accommodated in Spain, France and Italy thereafter.



Source: author's illustration; National Central Banks.

This change in refinancing credit together with amendments to collateral criteria (see Chapter 3) suggests variation in pledged collateral, although evidence on variation is difficult to ascertain. Scarce information published by the ECB on pledged assets is illustrated in Figure 5.2 (see also ECB 2013c). Like for eligible marketable assets (Figure 3.9), the ECB only published aggregated yearly data prior to 2012 and quarterly data since then. Asset types that were increasingly pledged are uncovered bank bonds, covered bank bonds, ABSs and non-marketable assets. The substantial increase in pledge of non-marketable assets inter alia reflects the introduction of the ACCs framework in December 2011 (see Section 3.2.1). Moreover, the comparison of Figures 3.9 and 5.2 suggests that although the value of eligible government bonds increased, their pledge remained relatively stable. By contrast, ABSs were increasingly pledged although their eligible amount only slightly increased. Likewise, uncovered bank bonds were pledged to a greater extent when financial markets were under distress, which was facilitated by relaxed collateral criteria. Covered bank bonds were increasingly pledged although their eligible amount remained relatively stable.



The figure details the development of pledged collateral by asset type from 2004 to 2014, based on data published by the ECB. It reveals that pledge considerably increased, in line with the expansion of refinancing loans. Increased pledge can be mainly attributed to the intensified use of bank bonds, ABSs and non-marketable assets. Owing to the restriction to yearly data (quarterly data not until 2012), insights from published data are limited.



Source: author's illustration; European Central Bank.

This is all the information that the ECB provides on pledged collateral. However, detailed information would be beneficial owing to the importance of collateral as risk hedge in case of counterparty default. Given the importance of pledged collateral yet the lack of publicly available information, the remainder of this section is devoted to elaborating additional insights into pledged collateral. Specifically, the analysis extends available information illustrated in Figure 5.2 as follows.¹⁶¹ First, it disaggregates information on the yearly development by inferring from the monthly development of refinancing credit. Accordingly, the analysis provides estimates on the monthly development of pledged collateral at the country level by asset type. This facilitates identifying the country-specific effects of amendments to collateral criteria on pledged collateral. Second, the analysis generates novel information on the credit quality of pledged collateral.

The inferred development of collateral pledged with the Eurosystem based on monthly data is depicted in Figure 5.3. It indicates large volatility across months and peaks in collateral pledge during the onset of the financial crisis (October to November 2008) and the implementation of the three-year LTROs ("big bazooka", December 2011 to March 2012). The left panel details the geographical distribution of pledged collateral and the right panel estimates its credit quality. Tautologically, the geographical distribution of pledged collateral resembles the development

¹⁶¹ The analysis is performed as follows. The geographical distribution of refinancing credit is utilized to split aggregate data on pledged collateral across countries, which gives a proxy for collateral pledged at the country level. Insights into the composition of pledged collateral by asset type are gained by inferring from the composition of (i) national collateral pools (Figure 3.13) and (ii) of the Eurosystem collateral pool (Figure 3.10). The unweighted average of both is applied to reflect that collateral pledge at the country level is affected by (i) assets eligible in a country and (ii) assets eligible in other countries owing to cross-country pledge. The credit quality of pledged collateral is likewise estimated by inferring from credit quality in both (i) the national collateral pool (Figure 3.15) and (ii) the Eurosystem collateral pool (Figure 3.14). The analysis has the flaw that it cannot factor in non-marketable assets as no information is available beyond that given in Figure 5.2.

of refinancing credit (Figure 5.1). The left panel suggests that as the majority of refinancing credit was drawn in Germany prior to the crisis, the majority of collateral was also pledged there. Throughout the crisis, refinancing credit shifted towards crisis-stricken countries, as did pledged collateral. The right panel estimates that credit quality of pledged collateral was almost unaffected by the financial crisis but deteriorated throughout the sovereign debt crisis, with a considerable fraction of pledged collateral rated within CQS 3 and later also CQS 4.

Figure 5.3: Inferred pledged collateral by country and by credit quality (estim. monthly data)

The figure depicts the development of the inferred pledge of collateral by country (left panel) and credit quality (right panel) based on monthly data. The panels indicate large volatility of pledge across months, a shift in the geographical distribution and the deterioration of credit quality since the onset of the sovereign debt crisis.



Aggregate information on pledged collateral by asset type is refined to the country level in Figure 5.4. The figure suggests substantial differences across countries with respect to the types of pledged assets. In Germany, pledged collateral is estimated as having declined to almost zero throughout the sovereign debt crisis. Prior to the crisis, central government bonds and bank bonds were predominantly pledged as collateral. By contrast, pledged collateral substantially increased in Spain, particularly throughout 2010 and with the settlement of the two three-year LTROs. About one-third of initially pledged collateral was likely ABSs, although they lost importance over time owing to market developments and tighter collateral criteria. Therefore, government bonds and bank bonds grew in importance. The pledge of assets in France experienced jumps at the beginning of the financial crisis and during the implementation of the three-year LTROs. The majority of pledged collateral was likely government bonds and uncovered bank bonds. Akin, the pledge of collateral in Greece experienced a first jump with the onset of the financial crisis and substantially expanded throughout the sovereign debt crisis. Moreover, the panel reflects the phases during which the Eurosystem revoked the suspension of the minimum credit rating for collateral related to the Greek government. While government bonds accounted for the lion's share of pledged collateral in Greece, uncovered bank bonds grew in importance, especially during the course of 2013. In Ireland, pledged collateral likewise increased during two phases, i.e. throughout the financial crisis and the sovereign debt crisis. It peaked in the end of 2010 and considerably shrunk thereafter. Several types of collateral assets were likely pledged in Ireland, with ABSs playing a major role prior to the tightening of collateral criteria. Collateral pledge surged in Italy throughout 2011 and peaked during the settlement of the two three-year

LTROs. The surge can largely be referred to the pledge of central government and uncovered bank bonds. Finally, the estimated pledge of collateral in Portugal experienced a substantial increase during the sovereign debt crisis. Collateral use surged at the beginning of the crisis and dropped thereafter. Over time, government bonds accounted for the majority of collateral pledged in Portugal, although bank bonds and ABSs gained in importance.

Moreover, Figure 5.5 presents the inferred development of credit quality of pledged collateral at the country level. Figure 5.3 indicated that deterioration in the credit quality of collateral pledged with the Eurosystem intensified throughout the sovereign debt crisis. Figure 5.5 suggests that the deterioration was asymmetric across countries and the largest in countries that increasingly granted refinancing credit since the onset of the sovereign debt crisis. It indicates that credit quality remained high in Germany as well as France but slightly deteriorated in Ireland and Italy throughout the sovereign debt crisis. In Spain, credit quality decreased throughout 2012 as collateral rated within CQS 3 is estimated to have been increasingly pledged. Deterioration was the strongest in Greece and Portugal. It took place in Greece as of mid-2010 when the sovereign debt crisis emerged. Thereafter, collateral almost exclusively rated within CQS 3 is estimated to have been pledged collateral was likely rated within CQS 4, i.e. not complying with the uniform minimum credit rating threshold for marketable assets. In Portugal, the credit quality of collateral pledged started to decline in 2011. Thereafter, about half of pledged collateral is estimated to have been rated within CQS 3.

5.2 The Repo Market as the Eurosystem's Competitor for Collateral

This section sets up the institutional background of the model presented in the next section. Section 2.3 discussed that the Eurosystem provides liquidity predominately as collateralized loans while the private provision of short-term liquidity between banks usually takes the form of repos. Although legal contracts and implementation differ between collateralized loans and repos, they are equivalent from an economic perspective and are considered as close substitutes for borrowers such that the Eurosystem and the repo market are in competition for collateral. The following section addresses general aspects of repos and provides a brief overview of collateral criteria and risk mitigation in the European interbank repo market. Furthermore, similarities and distinctions between private repos and collateralized loans by the Eurosystem are carved out.

5.2.1 General Aspects of Repurchase Agreements

A repo is an agreement between two parties in which one party sells assets to the other and simultaneously agrees to buy back the assets at a later date.¹⁶² The repurchase price usually exceeds the purchase price such that the difference can be regarded as equivalent to the interest paid in collateralized lending.

¹⁶² Therefore, one party is usually called the seller/borrower/collateral provider (COP) and the other is referred to as the buyer/lender/cash provider (CAP).



Figure 5.4: Inferred collateral pledged at the country level by asset type

The figure refines aggregate information on inferred pledged collateral by asset type to the country level, suggesting substantial differences in the types of pledged assets across countries.

Repos hold interest for two reasons. First, they are the main instrument for short-term bank funding (cf. ECB 2012b, 2014c). In December 2014, the total value of contracts outstanding in the European repo market was estimated at EUR 5.5 tn, having slightly declined from EUR 5.8 tn in June 2014 (cf. ICMA 2014). These figures do not include the value of repos transacted with the Eurosystem or central banks outside the Eurosystem. Accordingly, the majority of the repos were performed in the interbank market.¹⁶³ Second, the Eurosystem carries out a fraction of its liquidity-providing operations via repos as one of two possible types of reverse transactions (cf. Section 2.1). Therefore, repo markets occupy first-order relevance for the Eurosystem's refinancing operations.¹⁶⁴

There are different reasons for both the lender and the borrower to enter a repo.¹⁶⁵ In general, the interbank repo market allows liquidity-seeking banks to receive funding from banks with excess liquidity. Borrowers can use idle assets to finance long positions while lenders are able to cover short positions. Interest cost are usually low as repo markets feature high market liquidity. For banks with excess liquidity, repos thus reflect an additional investment option that implies relatively small risk owing to short duration and collateralization ("cash-driven" motive). Moreover, the motive for entering a repo can also be for borrowing certain collateral assets ("security-driven"). Consequently, collateral assets are usually specific in security-driven repos, while cash-driven repos are often conducted as general collateral (GC) repos. In GC repos, lenders accept as collateral a basket/range of assets (i.e. a specific asset type or a specified list of ISIN codes) provided that the assets meet requirements on e.g. credit quality. Security-driven repos are usually referred to as "specials" and no prior restrictions on collateral assets are made.

The following two processes within repo transactions hold interest for the analysis at hand as they are decisive for the specification of collateral criteria and risk control, i.e. (i) the management of collateral and (ii) the clearing of delivery and payment obligations.¹⁶⁶

Collateral Management At the level of collateral management, securities are valued, selected and delivered as collateral from the borrower to the lender. This is undertaken either bilaterally, i.e. by counterparties themselves ("bilateral") or under the involvement of a third party, usually a custodian bank ("tri-party"). Throughout the repo transaction, functions related to collateral management also involve maintaining collateral value and quality. Irrespective of whether collateral management is bilateral or tri-party, collateral risk remains with the lender as the third party does not assume any risk. Repos in which a third party manages collateral are usually cash-driven as the lender and the borrower only negotiate repo conditions but not the delivery of specific collateral assets.

¹⁶³ Cf. BAKK-SIMON et al. 2012. For a detailed investigation of the European interbank repo market, see ECB 2014d as well as MANCINI et al. 2015.

 $^{^{164}}$ See BIS 1999 for an early description of the repo market from the perspective of central banks.

 $^{^{165}\,\}mathrm{For}$ an extensive description of these reasons, see e.g. Choudhry 2006.

¹⁶⁶ Trading and settlement are two further processes in repo transactions. See e.g ECB 2013a for further details.



Figure 5.5: Inferred collateral pledged at the country level by credit quality

The figure generates new information on inferred pledged collateral by investigating the credit quality of pledged collateral at the country level, indicating that it developed asymmetrically



Clearing At the clearing level, delivery and payment obligations are netted among lenders and borrowers.¹⁶⁷ Clearing is performed either bilaterally or multilaterally: in the former, each lender/borrower nets obligations separately with each borrower/lender, while in the latter a CCP interacts between the lender and the borrower and nets obligations. The CCP also becomes the buyer and seller, respectively, of collateral, once the transaction has been agreed between the lender and the borrower. Therefore, the CCP becomes the counterparty to both parties of the repo, i.e. the borrower to the lender and the lender to the borrower, respectively. Repo contracts not cleared by a CCP are less standardized with respect to both collateral requirements and contract terms. However, CCP-cleared repos are usually advantageous to both the lender and the borrower as the CCP is usually a AAA-rated agency and protects both from counterparty default.¹⁶⁸ Furthermore, risk exposure is reduced as the CCP nets transactions between counterparties on a multilateral basis, which implies a smaller net exposure between counterparties relative to non-CCP clearing. The motivation for lenders to enter CCP-cleared repos is usually cash-driven, while it is often security-driven in non-CCP-cleared repos as the smaller degree of standardization allows collateral to be more specific. Table 5.1 details the possible combinations of clearing and collateral management and it is indicated who specifies collateral criteria and imposes the haircut.

Table 5.1: Combinations of clearing and collateral management in private repos

The table detail	s the possible	combinations of	clearing and	collateral mai	nagement. (Combinations
are vital for who	o is exposed to	collateral risk,	hence specify	ing collateral	criteria and	l the haircut.

		Collateral Management		
		BILATERAL	Tri-Party	
Clearing	CCP	collateral criteria and hair- cut imposed by <i>CCP</i>	collateral criteria and hair- cut imposed by <i>CCP</i>	
	Non-CCP	collateral criteria and hair- cut imposed by <i>lender</i>	collateral criteria and hair- cut imposed by <i>lender</i>	

Source: author's compilation.

Risk Mitigation through Collateralization and Haircuts Analogous to the Eurosystem, lenders in the repo market hedge risk in two steps: first, collateral is taken to hedge risk of borrower default (counterparty risk); and second, haircuts are applied to mitigate risk associated with collateral (collateral risk).¹⁶⁹ The eligibility of collateral in the repo market is usually restricted to bonds and other fixed-income instruments and haircuts are applied to the market value of the collateral to mitigate the risk associated with liquidation in case of counterparty default. Table 5.1 reveals that lenders determine eligibility requirements and haircuts applied to col-

¹⁶⁷ Clearing refers to several activities in between trading and settlement, namely trade capture, trade verification, trade matching or affirmation, legal confirmation, reporting to a trade repository, position and payment netting, portfolio compression, novation (a central clearing mechanism) and risk management (cf. ECB 2013a).

¹⁶⁸ CCPs are AAA-rated as exposure is strictly collateralized, backed by reserves and further safeguards like default funds.

¹⁶⁹ GORTON and METRICK 2010, 2012 empirically analyze evolutions in the US repo market throughout the financial crisis. The studies find that haircuts significantly increased from essentially zero in the process of deleveraging, which is referred to as a "run on repo". Further studies on the development of repo markets and haircuts during and after the financial crisis include BRUNNERMEIER 2009 and KRISHNAMURTHY et al. 2014.

lateral in non-CCP-cleared repos irrespective of whether a third party is involved in collateral management. The third party only manages collateral but does not assume any responsibility for it. Thus, risk mitigation is crucial for the lender as he would be left with any loss in case of borrower default. By contrast, in CCP-cleared repos, the CCP buys the collateral and also imposes requirements and haircuts to the collateral. Hence, haircuts are exogenous to lenders in CCP-cleared repos.

5.2.2 Collateral and Haircuts in the European Repo Market

The Eurosystem competes for collateral with private lenders that participate in the European repo market. The total value of repo contracts outstanding in this market amounted to EUR 5.5 th in December 2014, compared to EUR 0.63 th of outstanding refinancing credit from the Eurosystem (cf. ICMA 2014 and Section 2.1).¹⁷⁰ The following collects the scarce available information on the European repo market in terms of collateral and haircuts.

Collateral

The geographical distribution of collateral sold in the European repo market is illustrated in Figure 5.6.¹⁷¹ It indicates the development of the value of outstanding repos and shows that the fraction of assets used for repos in Germany, Spain, France, Greece, Ireland, Italy and Portugal (where the vast majority of collateral that the Europystem receives is pledged) slightly declined over time, i.e. from 56.7% in June 2007 to 48.1% in December 2014. Most of this collateral was sold in Germany, France, Italy and Spain, while only negligible amounts were pledged in Greece, Ireland and Portugal.

In order to gain further insights, the European repo market is segmented along the lines of Table 5.1, i.e. with respect to clearing (CCP-cleared vs. non-CCP-cleared) and collateral management (bilateral vs. tri-party), as additional information is available for CCP-cleared and tri-party repos. The terms of non-CCP-cleared and bilateral repos are usually negotiated directly between counterparties with no information being made available.

According to ibid., 27.4% of repo contracts were cleared with CCP-involvement in 2014.¹⁷² Clearing services in the European repo market were mainly provided by five CCPs by the end of 2014, i.e. CC&G in Italy, BMEClearing in Spain, LCH-Clearnet SA in France, LCH-Clearnet Ltd in the UK, and Eurex Clearing in Germany. Owing to this geographical segmentation and potential differences in risk preference, CCPs deemed eligible different types of collateral. CC&G accepted government bonds from Germany, France and Italy. BMEClearing deemed eligible a variety of securities and government bonds from Austria, Belgium, France, Germany, the Netherlands, Spain, the UK and the US. For LCH-Clearnet SA, government bonds from inter

¹⁷⁰ For the following indicative analysis of the European repo market, the series of the International Capital Market Association (ICMA) repo market survey is consulted. See ICMA 2007, 2008, 2009, 2010, 2011, 2012, 2013, 2014.

¹⁷¹ Analogous to above, the geographical distribution is considered in terms of the country where collateral was issued.

¹⁷² Therefore, the vast majority of repos (72.6% worth EUR 4 tn in 2014, cf. ICMA 2014) was contracted without CCP-involvement. In these repos, contractual details are negotiated bilaterally and no a priori restrictions on collateral are made.



The figure elaborates on the geographical distribution of collateral pledged in the European repo market and shows that the fraction of collateral from Germany, Spain, France, Greece, Ireland, Italy and Portugal (where the Eurosystem receives the lion's share of pledged collateral) slightly declined over time.



Source: author's compilation; ICMA 2007, 2008, 2009, 2010, 2011, 2012, 2013, 2014.

alia Belgium, France, Germany, Italy, the Netherlands, Portugal, Spain, the UK and the US were eligible, as well as government-guaranteed bonds. Likewise, LCH-Clearnet Ltd embraced a variety of government bonds, from e.g. Austria, Belgium, Finland, Germany, Ireland, the Netherlands, Portugal, Slovakia, Slovenia, Spain and the UK. In addition, it deemed eligible corporate bonds that were rated above "single A". Finally, Eurex approved a broad variety of fixed-income securities such as government bonds denominated in euro, Swiss francs as well as other currencies. Moreover, it accepted equities denominated in euro and Swiss francs.

For the tri-party segment of the European repo market, information is available on the type and credit quality of traded collateral. Tri-party repos are contracted directly between the two parties as the involved third party undertakes collateral management but does not assume any risk. INTERNATIONAL CAPITAL MARKET ASSOCIATION 2014 reports that 10.5% of contracts in the European repo market in 2014 were tri-party, having slightly increased over recent years. The development of the type and credit quality of collateral in tri-party repos is depicted in Figure 5.7 from June 2009 to December 2014. The left panel reveals that government bonds, corporate bonds and equities are predominantly sold in tri-party repos. Regional government bonds and equities replaced corporate bonds over time. While these corporate bonds may instead have been pledged with the Eurosystem (cf. Figure 5.3), equities have been deemed ineligible as collateral by the Eurosystem as of May 2005 (cf. Section 3.2.1). The right panel indicates the credit quality of pledged assets (aligned to the Eurosystem's understanding of credit quality) and shows that it remained relatively high, with about 80% to 90% of all assets rated within CQS 1/2. Figure 5.7: Collateral pledged in tri-party repos by asset type and by credit quality

The figure illustrates the development of the composition of collateral pledged in tri-party repos from 2009 to 2014 by asset type (left panel) and credit quality (right panel). The left panel reveals that government bonds, corporate bonds and equities were predominantly pledged in tri-party repos. The right panel indicates that credit quality of pledged assets deteriorated from June 2009 to December 2014.



However, comparing the development of collateral credit quality in tri-party repos and with the Eurosystem (Figure 5.3) reveals differing developments. In June 2009, 92.8% of collateral pledged in tri-party repos was rated within CQS 1/2 while 97.7% of collateral pledged with the Eurosystem is estimated to have been of such high quality. Credit quality deteriorated both in the repo market and with the Eurosystem over time, although the decline was likely stronger for the Eurosystem than in tri-party repos. At the end of 2013, only 78.9% of collateral pledged with the Eurosystem was likely still rated within CQS 1/2 compared to 80.5% in the repo market.¹⁷³

Haircuts

According to Table 5.1, haircuts are imposed by CCPs in CCP-cleared repos but bilaterally negotiated otherwise. In accordance with the European Market Infrastructure Regulation (EMIR), CCPs are required to apply haircuts that are regularly tested (and revised if necessary) and take market conditions into account (EUROPEAN PARLIAMENT 2012). Haircuts have to be determined individually for each collateral asset, considering its relevant properties such as its type and credit quality.¹⁷⁴ Available information on haircuts applied in private repos is scarce, with the little indicative evidence suggesting that haircuts depend on the type of the repo, i.e. whether the repo is GC or special (see above). In certain GC repos (e.g. Eurex Repo Euro GC Pooling, LCH.Clearnet Ltd's RepoClear \in GC), haircuts (and collateral criteria in general) are assumed from the Eurosystem. In less standardized CCP-cleared repos, CCPs impose own haircuts. For instance, BMEClearing states that it applies a "minimum haircut, equal to that of the ECB [...]", which is increased if deemed necessary.¹⁷⁵ Table 5.2 compares average haircuts applied to selected government bonds from January 2011 to December 2013 by two CCPs and

¹⁷⁴Other properties are legal risk, operational risk, duration, historical and hypothetical future price volatility (manifested in liquidity risk), wrong-way risk and exchange-rate risk. See Section 4.2.1 and e.g. ECB 2013a.

¹⁷³ Section 5.3 offers a potential explanation for this indicative evidence of adverse selection of collateral based on the peculiarities of the Eurosystem collateral framework and risk control.

¹⁷⁵ Cf. BMEClearing, "Collateral Valuation," http://www.bmeclearing.es/ing/Collateral/ CollateralProcessing.aspx.

the Eurosystem.¹⁷⁶ The table reveals that the average haircut of the Eurosystem was always lower than that applied by CCPs. Haircut determination by CCPs may also be more differentiated. For instance, CC&G differentiated haircuts according to eleven clusters of residual maturity for Italian government bonds but only for four clusters in case of French or German government bonds. LCH-Clearnet SA used nine clusters of residual maturity for determining haircuts on the variety of government bonds that it accepted.

Table 5.2: Haircuts applied to selected government bonds by CCPs and the Eurosystem

The table compares the average haircut applied to selected government bonds of different residual maturity between January 2011 and December 2013 by two CCPs and the Eurosystem, revealing that average haircuts of the Eurosystem fell short of CCPs' average haircuts for all considered government bonds.

	DURATION	BME^{a}	$CC\&G^a$	Eurosystem ^a
	$\operatorname{Short}^{\mathrm{b}}$	2	3.86	0.5
Germany	MIDDLE ^c	3.56	7.72	2.42
	Long ^d	7.65	20	5.46
FRANCE	Short ^b	2	3.86	0.5
	MIDDLE ^c	3.56	3.86	2.42
	Long ^d	7.65	7.72	5.46
Italy	$\operatorname{Short}^{\mathrm{b}}$	n/a^e	5.95	0.5
	MIDDLE ^c	n/a^e	13.45	2.42
	Long ^d	n/a^{e}	22.51	5.46
	$\operatorname{Short}^{\mathrm{b}}$	2	n/a^e	0.5
Spain	MIDDLE ^c	3.56	n/a^e	2.42
	Long ^d	7.65	n/a^{e}	5.46

^a In percentage points.

^b Defined as a government bond with residual maturity of less than 6 months.

Defined as a government bond with residual maturity of 4 years.

^d Defined as a government bond with residual maturity of 11 years.

Not available as deemed ineligible.

Source: author's compilation; CCPs; European Central Bank.

This indicative evidence suggests that while a broad range of collateral is deemed eligible in both the (different segments of the) repo market and by the Eurosystem, contract conditions such as haircuts may differ. This was discussed for CCP-cleared repos. However, it is intuitive that conditions also differ between non-CCP cleared repos and collateralized loans from the Eurosystem. In these repos, haircuts are subject to negotiation between the lender and the borrower and contracted over the counter in almost three-quarters of repo transactions (ICMA 2014), which makes available data and information very scarce.

Different conditions in private repos and collateralized loans from the Eurosystem are at the heart of the model presented in the following section, investigating the decision of a borrower seeking liquidity either in the repo market or from the Eurosystem. Hence, it applies to all borrowers that have discretion to pledge collateral in the market or with the Eurosystem. Applicability may be limited if a borrower does not have this discretion owing to e.g. counterparty ineligibility (e.g. the borrower is not deemed eligible by the Eurosystem) or collateral ineligibility (e.g. the

 $^{^{176}\,\}mathrm{See}$ also ECB 2014a for a further comparison of haircuts applied by CCPs and the Eurosystem.

collateral is deemed ineligible by the Eurosystem). However, this limitation is de facto loose since the Eurosystem lends to a broad range of counterparties against an ever-broader range of collateral (Chapter 3).

5.3 Gresham's Law of Collateral: A Model of Adverse Selection of Collateral

This section presents a model that elaborates on the behavior of liquidity-seeking banks. Banks can draw this liquidity by borrowing either in the market ("private liquidity") or from the Eurosystem ("public liquidity"). It was discussed above that repos and collateralized loans are equivalent from an economic perspective. Therefore, no distinction is made between the specifics of liquidity provision in the market and by the Eurosystem, which is modeled as repos. The model builds upon the course of action of a repo stylized in Figure 5.8, which depicts the decomposition of a repo into two transactions taking place at distinct points in time, i.e. the opening leg and the closing leg. Within the opening leg, the lender and the borrower agree on the repo as well as its conditions and transfer liquidity in exchange for collateral. Within the closing leg, the borrower repurchases the collateral from the lender and the lender receives back the provided liquidity together with interest implied in a potentially higher repurchase price.



The figure stylizes the course of action for a repo and depicts its decomposition into two transactions taking place at distinct points in time, i.e. the opening leg and the closing leg. The following analysis focuses upon the opening leg.



The model focuses upon the opening leg and the borrower's decision to enter a repo with either another bank ("private lender") or the Eurosystem ("public lender"). Repo terms are shaped by six determinants (see Section 2.3): (i) maturity, (ii) interest rate, (iii) counterparty, (iv) amount, (v) collateral and (vi) haircut. To analyze the decision of the borrower along one dimension, determinants (i) to (v) are assumed to be equal in private and public repos, which makes the haircut the crucial determinant. Specifically, this means that (iii) any bank can borrow (iv) the same amount of liquidity for (i) the same period against (ii) the same interest rate and (v) the same collateral in the market or from the Eurosystem. Therefore, the borrower's decision solely depends on the comparison of the haircut offered in private (in the market) or public (with the Eurosystem) repos.

Lenders impose a haircut to mitigate collateral risk such that the haircut size reflects expectations about the collateral liquidation value. In a certain world, the lender would not bear a loss from borrower default as it would be fully anticipated. The repo would be fully collateralized and the lender would not be worse off from liquidating collateral. However, in reality, borrower default could imply a loss to the lender as expectations about the liquidation value may be incorrect, leading to an under-collateralized repo. Risk mitigation is carried out in the opening leg based on expectations about the collateral liquidation value. The actual outcome of the repo in the closing leg, i.e. whether or not the asset is repurchased, is of second-order as expectations of this outcome are reflected in repo terms realized in the opening leg. Hence, the analysis is narrowed to the opening leg without loss of generality.

The model elaborates on the adverse selection of collateral, which is often referred to as "Gresham's Law of Collateral" (GLOC) in the literature. GLOC is adapted from "Gresham's law", which describes the tendency for bad money to drive out good money. Accordingly, if coins contained metal of different value yet had the same value as legal tender, the coins comprising the cheaper metal ("bad money") would be used for payments while the coins made of the more expensive metal ("good money") would be hoarded (see e.g. MACLEOD 1856 and VON HAYEK 1976). The analogy between Gresham's law for specie and for collateral was established narratively by CHAILLOUX et al. 2008a, b and addressed as the risk of adverse selection of collateral in SINGH 2013. The adverse selection of collateral was first technically analyzed by EWER-HART et al. 2006, who suggest borrowers' preference to pledge illiquid collateral with the central bank owing to different pricing of liquidity risk by the private market and the central bank. EWERHART and TAPKING 2008 provide support that the least liquid and most risky assets are deposited with the central bank, while higher-quality assets are pledged in private (bilateral) repos as this alleviates two-way credit risk. BINDSEIL 2014 seizes on the phenomenon of GLOC and discusses four measures that central banks could in principle consider to counteract the concentration of bad collateral owing to adverse selection. Most recently, FECHT et al. 2015 document adverse selection (referred to as "systemic arbitrage") for German banks' borrowing from the Eurosystem over the 2006-2010 period. In this context, systemic arbitrage is also addressed in NyBORG 2015. The following model differs from previous analyses as it addresses GLOC specifically for the Eurosystem based on the peculiarities of the Eurosystem collateral framework and risk control. Moreover, it emphasizes the role of rating agencies for the degree of adverse selection.

The elaboration of the model is structured as follows. First, the basic model with symmetric information is set up as the benchmark. Subsequently, asymmetric information, a signal of collateral quality and an outside option to borrowers are introduced. The outside option is then further specified and modeled as a stylization of repo condition offered by the Eurosystem in terms of pooling of repo conditions.

5.3.1 Model with Private and Public Liquidity Provision

In the basic model, only the repo market is considered and counterparties can interact in the following simple way: liquidity-seeking banks can use bonds in repos with banks that have excess liquidity. In this environment, four steps of analysis will be taken in which the information structure and the involved players differ. First, borrowers and lenders are symmetrically informed about the quality of bonds that can be used as collateral. Second, the case is considered in

which the borrower has more information than the lender about collateral quality (asymmetric information). Third, the impact of credit ratings as a signal on collateral quality is evaluated. Fourth, an outside option for borrowers is introduced, i.e. the possibility to use collateral outside the repo market, e.g. with the central bank as public lender.

5.3.1.1 Model Setup¹⁷⁷

There are J profit-maximizing borrowers who own one collateral asset each, e.g. a bond. The initial cost of bond creation is assumed to be zero. Bonds differ in the level of quality θ such that bond j is of (unique) quality θ_j , which can be thought of e.g. higher credit quality or liquidity of bond j.¹⁷⁸ Quality is uniformly distributed in the range between the lowest $\underline{\theta}$ and the highest quality level $\overline{\theta}$, i.e. $\theta_j \in [\underline{\theta}, \overline{\theta}] \subset \mathbb{R}$. Different quality levels emerge as risk-to-return ratios of assets underlying the bonds differ such that $0 < \underline{\theta} < \overline{\theta} < \infty$. Selling the bonds gives borrowers liquidity that can be used e.g. to make new loans resulting in a positive return.

Risk-neutral lenders purchase bonds under perfect competition at price $\rho(\theta_j)$. The purchase price is equal to the haircut-adjusted market value of the bond, i.e. $(1 - h_L(\theta_j))\Omega(\theta_j)$, with $0 \le h_L(\theta_j) \le 1$ being the haircut that a lender applies to the market value $\Omega(\theta_j)$ of bond j with quality θ_j .¹⁷⁹ Lenders seek to maximize profit and use an identical production function yielding constant returns to scale with bonds being the only input. For simplicity, each bond generates output equal to θ_j , which can be regarded as return on investment (depending on the quality of the bond), assuming the absence of production costs. The price that lenders take on their output is equal to unity such that their earnings also equate to θ_j .

5.3.1.2 Symmetric Information

If lenders and borrowers both have complete information about the quality of each bond, lenders will offer to pay distinct prices equal to quality $\rho(\theta_j) = \theta_j$. This is what lenders earn from owning the bond and owing to the competitive structure on the demand side, it also gives the equilibrium price. As borrowers miss alternative utilization of the bonds, the set of marketing borrowers is given by $\Theta(\rho) = \{\theta_j : 0 \leq \rho(\theta_j)\}$. As $\rho(\theta_j)$ is equal to θ_j and $\underline{\theta}$ is strictly positive, the set comprises all J borrowers. All bonds are traded and aggregate surplus in the repo market (defined as the sum of rents of borrowers and lenders) is maximized. Figure 5.9 provides a graphical illustration of the equilibrium with symmetric information. Bond quality θ runs on the horizontal axis, with the relevant quality range indicated by $\underline{\theta}$ and $\overline{\theta}$. The 45°-line maps bond quality onto the vertical axis as quality θ_j equals output net of cost and hence earnings from the bond to a lender who is willing to pay a price of the same amount. Therefore, the price is given on the vertical axis. As each bond is traded at a price equal to its quality level, lenders are left with zero rent. Payments to borrowers are reflected by the area underneath the

 $^{^{177}\,\}mathrm{The}$ model setup bases on MAS-COLELL et al. 1995.

¹⁷⁸ Hence, θ represents a broad argument of collateral quality, i.e. it can be interpreted as reflecting collateral success probability q or liquidity κ .

¹⁷⁹ This haircut-adjusted market value will later be referred to as private collateral value, i.e. the value that private lenders assign to collateral.

price curve in the range between $\underline{\theta}$ and $\overline{\theta}$. The area gives aggregate surplus in this market and corresponds to the rent of the supply side, i.e. borrowers.



Figure 5.9: Equilibrium with symmetric information

Technically, rent distribution can be formulated as follows. Lenders are left with zero rent, i.e. $R_L = 0$, while each borrower receives $\rho(\theta_j)$ in exchange for collateral, has zero cost and all bonds are traded such that¹⁸⁰

$$R_B = \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} \tag{5.1}$$

is the rent of borrowers. Aggregate surplus R in the market is equal to the rent of borrowers as R_L is equal to zero, i.e. $R = R_B = \frac{\bar{\theta}^2 - \underline{\theta}^2}{2}$.

5.3.1.3 Asymmetric Information

In the next step, lenders cannot observe the quality of the bond that they purchase (asymmetric information). Rather than distinct prices equal to bond quality, lenders offer the uniform price $\rho(\theta_j) = \rho \forall \theta_j$. As borrowers still lack an alternative to selling bonds on the market, they accept any non-negative price. Accordingly, the set of marketing borrowers is given by $\Theta(\rho) = \{\theta_j : 0 \le \rho\}$. Lenders' demand for bonds ζ is a function of ρ and can be expressed as

$$\zeta(\rho) = \begin{cases} 0 & \text{if } \mu < \rho; \\ [0,\infty) & \text{if } \mu = \rho; \\ \infty & \text{if } \mu > \rho, \end{cases}$$
(5.2)

with μ representing the lenders' uniform belief concerning the average quality of traded bonds. Figure 5.10 illustrates the market with asymmetric information. Lenders anticipate that at any non-negative price, $\Theta(\rho)$ comprises all J borrowers and all bonds in the range from $\underline{\theta}$ to $\overline{\theta}$ are marketed. Lenders' belief concerning the average bond quality in the market, i.e. the expected

 $^{^{180}}$ See the appendix for the derivation of Equation (5.1).

bond quality, is given by $\mu = \mathbb{E}[\theta | \theta \in \Theta(\rho)] = \frac{\bar{\theta} + \theta}{2}$ as θ is uniformly distributed within $[\underline{\theta}, \overline{\theta}]$. Lenders are in perfect competition for bonds and offer a price that simply reprieves them from an expected loss such that $\rho = \frac{\bar{\theta} + \theta}{2}$. While some lenders realize profits and others lose, lenders overall are still left with zero rent, i.e. $R_L = 0$. Therefore, the rent of borrowers is equal to aggregate surplus in the market, amounting to¹⁸¹

$$R_B = R = \frac{\bar{\theta}^2 - \underline{\theta}^2}{2}.$$
(5.3)

Figure 5.10 shows the expenditure of lenders by the rectangle under the horizontal price line (between $\underline{\theta}$ and θ), which is also what borrowers receive as payment for their bonds, i.e. their rent R_B . The earnings of borrowers correspond to the area below the 45°-line (between $\underline{\theta}$ and $\bar{\theta}$). The rents of borrowers and lenders are equal to the benchmark case of symmetric information, as is the aggregate surplus R. However, there are important differences with respect to the distribution of rents within the groups of lenders and borrowers. On the one hand, lenders purchasing (without knowing) bonds of relatively low quality of up to $\frac{\bar{\theta}+\theta}{2}$ take losses, as illustrated by the left shaded triangle. This is because they pay the uniform price $\rho = \frac{\bar{\theta} + \theta}{2}$ but receive bonds of quality $\theta_j < \frac{\bar{\theta} + \theta}{2}$. On the other hand, lenders purchasing higherquality bonds, $\theta_j > \frac{\bar{\theta} + \theta}{2}$, receiving profits that correspond to the right shaded triangle. Only the lender who receives the bond of quality $\theta_j = \frac{\bar{\theta} + \theta}{2}$ for paying $\rho = \frac{\bar{\theta} + \theta}{2}$ makes zero profit, like all lenders did with symmetric information. As the shaded triangles offset in aggregate, R_L remains unaffected. Likewise, borrowers are affected by asymmetric information. Those borrowers selling bonds with $\theta_j < \frac{\theta + \theta}{2}$ receive a higher price than with symmetric information, while borrowers with $\theta_j > \frac{\bar{\theta} + \bar{\theta}}{2}$ are worse off. Accordingly, rent is redistributed from borrowers with high-quality bonds to those with low-quality bonds. Asymmetric information leads to redistribution within the group of lenders and borrowers while aggregate surplus is still maximized.

Figure 5.10: Equilibrium with asymmetric information



¹⁸¹See the appendix for the derivation of Equation (5.3).

5.3.1.4 Signaling under Asymmetric Information

Each lender is now supposed to observe a signal $s(\theta_i)$ on the quality θ of bond j that ranges between $[\theta, \theta]$. Each lender is assumed to believe in this signal as it provides information on bond quality which the lender would otherwise not have. For instance, this signal could be interpreted as a credit rating by a rating agency that is able to assess bond quality.¹⁸² Technically, lenders' belief on bond quality is conditional on the signal and given by $\mathbb{E}[\theta_i|s(\theta_i)] = s(\theta_i)$. Obviously, if the signal reflects true quality, i.e. $s(\theta_j) = \theta_j \forall \theta_j$, asymmetric information would be resolved and the outcome under symmetric information would be mimicked. There are at least three possible explanations for a distorted signal. First, although the rating agency may have the necessary information available to correctly assess bond quality, it delivers a rating that does not fully reflect this information.¹⁸³ Second, borrowers who have to acquire the ratings may selectively disclose ratings, i.e. only disclosing the best rating available. This "ratings shopping" was observed prior to the financial crisis, especially for complex securities.¹⁸⁴ Third, the rating agency may systematically under- or over-value quality when its assessment technology is biased or affected by rating-contingent regulation. With respect to the former, KATZ et al. 2009 point to biased signals as the result of increased competition and the endeavor to increase profits. Accordingly, rating agencies relaxed rating criteria and avoided hiring new staff or investing in costly new databases and rating models prior to the financial crisis. With respect to ratingcontingent regulation, the analysis by OPP et al. 2013 shows that the introduction of such regulation increases the volume of highly-rated securities.¹⁸⁵ This systematic misevaluation of bond quality is addressed in the following.

In this case, the signal reads $s(\theta_j) = \theta_j \pm \epsilon$, with $\epsilon > 0$ being a positive constant reflecting the misevaluation of bond quality.¹⁸⁶ Misevaluation implies that the bond quality perceived by lenders runs parallel to the 45°-line as exemplified in Figure 5.11. The figure depicts the case of a systematic overvaluation of quality, i.e. $s(\theta_j) = \max(\theta_j + \epsilon, \bar{\theta})$.¹⁸⁷ Bonds j of quality $\theta_j \geq \bar{\theta} - \epsilon$ are signaled to be of the highest quality $\bar{\theta}$. Lenders' expectations are $\mathbb{E}[\theta_j|s(\theta_j)] =$ $\max(\theta_j + \epsilon, \bar{\theta})$, which gives the equilibrium price $\rho(\theta_j) = \max(\theta_j + \epsilon, \bar{\theta})$. As borrowers still lack an alternative to selling their bond on the market, they accept any non-negative price and the set of marketing borrowers is again $\Theta(\rho) = \{\theta_j : 0 \leq \rho\}$. All lenders aside from the one buying the bond of quality $\bar{\theta}$ make losses, reflected by the shaded area in Figure 5.11. Specifically,

¹⁸² In reality, credit ratings are not continuous as quality levels are clustered into rating notches. Here, a continuous distribution of θ and $s(\theta)$ is assumed for simplicity. However, the model could likewise be interpreted with discrete quality and signal.

¹⁸³ Rating agencies are private for-profit institutions and bond issuers pay them for ratings. There has been a diversified discussion concerning the conflicted interests of rating agencies in the light of the financial crisis, see e.g. ACKERMANN 2008, ACHARYA and RICHARDSON 2009, BRUNNERMEIER 2009, STOLPER 2009 and WHITE 2010.

¹⁸⁴ On ratings shopping, see Section 4.3.2.1. See also SANGIORGI and SPATT 2015, who find that selective disclosure of ratings lead to inefficient investment decisions. See also GROSSMAN and HART 1980 as well as MILGROM 1981 for the literature on the efficiency of the voluntary disclosure of available information.

¹⁸⁵ This finding is consistent with the view of e.g. CALOMIRIS 2009, who argues that inaccurate ratings are desired by institutional investors as they do not fully internalize the negative effects on ultimate investors.

¹⁸⁶ Misevaluation prevails during the opening leg but is resolved thereafter. Moreover, ϵ is strictly positive but smaller than $\bar{\theta} - \underline{\theta}$.

¹⁸⁷ The maximum function is applied as no bond can be signaled a quality that is higher than $\bar{\theta}$, i.e. the maximum quality in the market. Lenders believe in the multiple signaling of $\bar{\theta}$ without knowing as each lender receives an individual signal and cannot observe the signal provided to other lenders.

each lender incurs a loss to the extent of ϵ up to bonds of quality $\theta_j > \bar{\theta} - \epsilon$ for which losses decrease towards 0 with θ_j increases. Compared to a situation without signal, borrowers are better off as R_B increases by the shaded area. Lenders' rent is given by¹⁸⁸

$$R_L = -(\bar{\theta} - \underline{\theta} - \frac{1}{2}\epsilon)\epsilon < 0, \qquad (5.4)$$

while borrowers receive rent in the amount of 189

$$R_B = \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} - \left(\underbrace{-(\bar{\theta} - \underline{\theta} - \frac{1}{2}\epsilon)\epsilon}_{R_L}\right),\tag{5.5}$$

which is strictly positive for $R_L < 0$ as $\bar{\theta} > \underline{\theta}$. For $\epsilon = 0$, rents are equal to the benchmark as the case of symmetric information would be mimicked. This is intuitive as lenders and borrowers play a zero-sum game, which implies an aggregate surplus to the amount of $R = \frac{\bar{\theta}^2 - \underline{\theta}^2}{2}$, i.e. R remains unchanged. While asymmetric information leads to redistribution within the group of lenders and borrowers, a distorted signal involves a redistribution across groups. With a positively distorted signal, redistribution across groups is from lenders to borrowers. Preliminary findings are summarized in Lemma 5.1.





Lemma 5.1 (Redistribution within and across groups through asymmetric information and distorted signal).

 i) With asymmetric information, rent is redistributed within the groups of lenders and borrowers, namely from borrowers with high-quality bonds to those with low-quality bonds. Likewise, rent is redistributed from lenders purchasing bonds of low quality to those purchasing high-quality bonds.

¹⁸⁸ See the appendix for the derivation of Equation (5.4). R_L is negative as ϵ can at maximum be equal to $\bar{\theta} - \underline{\theta}$. ¹⁸⁹ See the appendix for the derivation of Equation (5.5).

ii) With asymmetric information and a positively distorted signal, rent is redistributed also across groups. Borrowers increase their rent at the expense of lenders.

5.3.1.5 Outside Option under Asymmetric Information

The signal is again neglected to facilitate the stepwise structure of the model. Instead, an outside option to selling bonds on the market is introduced, which is available to borrowers. The outside option could be provided by the central bank, with which borrowers can enter repos to receive liquidity (public repo). This liquidity can be used to make e.g. new loans, resulting in a positive return to borrowers. Despite the beneficial effect of collateralization in terms of mitigating counterparty risk, a private lender purchases a bond to realize a productive effect that arises from e.g. balance sheet effects or the benefit of using the specific bond (see Section 5.2). By contrast, the central bank does not benefit from receiving a specific bond as collateral but is supposed to exclusively mitigate counterparty risk. Hence, private repos are beneficial to both borrowers in terms of additional liquidity and lenders in terms of the productive use of the purchased bond, whereas the benefits of public repos are confined to borrowers.

The outside option gives borrowers a return of $\alpha(\theta_j)$. Like selling the bond on the market, making use of the outside option is costless. Borrowers sell bonds on the market whenever $\alpha(\theta_j) \leq \rho(\theta_j)$ and turn to the outside option otherwise, i.e. selling the bond on the market is preferred over the outside option in case of indifference. Accordingly, the set of marketing borrowers is characterized as $\Theta(\rho) = \{\theta_j : \alpha(\theta_j) \leq \rho(\theta_j)\}$. As long as $\rho \geq \alpha(\theta_j) \forall \theta_j$, all borrowers will dismiss the outside option and the outcome is equal to the one without outside option. However, with $\alpha(\bar{\theta}) > \mathbb{E}[\theta] = \rho$, at least the borrower owing the highest quality bond prefers the outside option over selling it on the market. Lenders would update their belief to $\mu = \mathbb{E}[\theta|\theta_j : \alpha(\theta_j) \leq \rho] < \mathbb{E}[\theta]$ in anticipation that the bond of highest quality will not be traded. Based on the updated belief, lenders offer a lower uniform bond price, which implies further borrowers with bonds of relatively high quality turning to the outside option. The phenomenon of adverse selection with different degrees of market unraveling depending on the outside option is examined in the following.

Outside Option and No Market Unraveling

Figure 5.12 depicts the situation in which the market with asymmetric information is not affected by the outside option as $\alpha(\bar{\theta}) = \mathbb{E}[\theta]$ and $\alpha(\theta) < \mathbb{E}[\theta] \forall \theta_j < \bar{\theta}$. Borrowers sell all bonds on the market at the uniform price $\rho = \mathbb{E}[\theta]$, with rents being equal to those under asymmetric information without outside option.

Outside Option and Partial Market Unraveling

Figure 5.13 illustrates the situation in which adverse selection leads to good-quality bonds being partially driven out of the market. Even if lenders expected all bonds to be sold on the market, some borrowers would turn to the outside option as this would give them more than the expected return in the market. Lenders update their belief and the equilibrium price ρ is characterized



Figure 5.12: Outside option and no market unraveling

by the intersection of the 45°-line and the line indicating expected bond quality. Borrowers for whom $\alpha(\theta_j) \leq \rho$ and hence $\theta_j \leq \theta_{\alpha}$ sell their bonds on the market. θ_{α} is the bond quality for which the owner is simply indifferent between selling the bond on the market and turning to the outside option. Taking into account that $\alpha(\theta) = \theta - \underline{\theta}$ as given in Figure 5.13, lenders realize zero rent while borrowers receive rent that equals aggregate surplus in the market and amounts to¹⁹⁰

$$R_B = R = \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} - \underline{\theta}(\bar{\theta} - \theta_\alpha).$$
(5.6)

While aggregate surplus R was maximized and amounted to $\frac{\bar{\theta}^2 - \underline{\theta}^2}{2}$ in previous steps, it is lower now as $\alpha(\theta_j) < \theta_j \forall \theta_j$, i.e. the outside option yields less than the productive use of bonds by lenders. This is reflected in Equation (5.6) as $\bar{\theta} > \theta_{\alpha}$. Inefficiency owing to the less productive use of bonds with the outside option is given by the hatched area in Figure 5.13. Through the adverse selection, the market size decreases (i.e. it partially unravels) and the outside option does not fully compensate for the unexploited market potential.

Outside Option and Complete Market Unraveling

Complete market unraveling occurs for $\alpha(\underline{\theta}) = \underline{\theta}$ and $\alpha(\theta_j) < \theta_j \forall \theta_j > \underline{\theta}$, which is depicted in Figure 5.14. Progression of the outside option implies that $\rho = \underline{\theta}$ is the only price that would motivate a borrower to sell his bond on the market while being profitable for lenders. The borrower with the bond of quality $\underline{\theta}$ is simply indifferent between selling the bond on the market and turning to the outside option. If this borrower turns to the outside option, the market would completely unravel. In this case, the outcome would be inefficient as aggregate surplus decreases relative to the benchmark case. This inefficiency corresponds to the hatched area in Figure 5.14.

¹⁹⁰ See the appendix for the derivation of Equation (5.6).



Figure 5.13: Outside option and partial market unraveling

The preliminary results of the analysis with respect to information structure, the effect of a signal and consequences of the introduction of an outside option for borrowers are summarized in Proposition 5.1.

Proposition 5.1 (Information structure, signal and outside option).

- i) With symmetric information, all bonds are traded at distinct prices that reflect bond quality and the equilibrium is efficient, i.e. aggregate surplus is maximized. With asymmetric information, the equilibrium remains efficient but all bonds are traded at a uniform price that reflects expected bond quality such that redistribution within the groups of lenders and borrowers takes place.
- ii) An undistorted signal maintains efficiency and prevents redistribution among borrowers and lenders. If the signal was distorted, rents would be redistributed among and across borrowers and lenders as bonds are traded at distinct prices that reflect the distorted signal.
- *iii)* Aggregate surplus decreases when borrowers are offered an outside option that is less productive than the use of bonds by lenders yet sufficiently high to drive better qualities out of the market.

5.3.2 The Eurosystem as Provider of Public Liquidity

The outside option is now modeled in a way that more strongly reflects its interpretation as the possibility to use bonds in public repos, i.e. in repos with the Eurosystem. Therefore, the outside option is modeled according to the Eurosystem's practice of clustering asset properties, which implies the pooling of public repo conditions.



Figure 5.14: Outside option and complete market unraveling

5.3.2.1 Pooling of Public Repo Conditions

Section 4.3.2.4 discussed how the Eurosystem clusters eligible assets in terms of liquidity, duration and credit quality. Furthermore, it was addressed that the application of supplementary haircuts leads to a divergence of nominal and effective haircuts. Section 4.4 introduced shortcomings associated with these two details of the Eurosystem risk control: first, the problem of insufficient hedge of collateral risk was raised; and second, the potential of adverse selection of collateral was addressed. Adverse selection may be the result of pooling public repo conditions, which is understood as the equalization of collateral values across assets of different intrinsic values. The market value of a bond equals the present value of future income flows related to that bond. For a given face value and interest rate, bonds of higher credit quality have a higher market value since their repayment is less uncertain. Consider e.g. two covered bank bonds of different credit quality and hence different market values: one is of high credit quality (AAA) with a market value of 100 and the other is of lower credit quality (A) with a market value of 50. When being pledged as collateral, the AAA-bond should be assigned a higher collateral value than the A-bond, e.g. 90 and 40. The pooling of repo conditions would imply that collateral values of the bonds are aligned, e.g. to 85 and 42.5. Perfectly-pooled repo conditions refer to a situation in which collateral values are uniform across bonds of different credit quality. This pooling of repo conditions is relevant for all eligible assets and particularly for own-use and non-marketable assets that are often subject to theoretical valuation.¹⁹¹ This may reflect the result of the following three steps:

- 1. the application of simplified haircuts that are too large for high-quality collateral and too small for low-quality collateral;
- 2. the application of supplementary haircuts irrespective of credit quality;
- 3. the theoretical valuation of assets.

¹⁹¹ NYBORG 2015 provides a precautionary assessment that about 77% by count or 17% by value of all eligible assets feature theoretical values. Moreover, the author documents that the fraction of theoretically valued assets is larger for lower-quality assets.

Application of Simplified Valuation Haircuts

Chapter 4 analyzed how the Eurosystem applies valuation haircuts to determine collateral values. Furthermore, the caveat was addressed that identical haircuts are applied within the same CQS to assets of different credit quality. In the following, private lenders are assumed to apply haircuts that fully reflect collateral risk. To distinguish the collateral value assigned by the Eurosystem from that assigned by private lenders, the former is labeled Φ_{CB} and the latter Φ_L . The application of simplified haircuts by the Eurosystem has two implications: on the one hand, collateral risk is not fully mitigated; and on the other, a fraction of assets is subsidized relative to the market.

To analyze the pooling of public repo conditions along one dimension of asset properties, bonds are considered as comparable in terms of coupon, duration and liquidity. For instance, Jumbo covered bank bonds with a fixed-rate coupon and residual maturity of two years are considered, which only differ in credit quality. Pooling public repo conditions through clustering asset properties is stylized in Figure 5.15 for fully-informed lenders. The figure illustrates the relation between bond quality θ (horizontal axis) and market value as well as collateral values in private and public repos (vertical axis). The range on the horizontal axis runs from $\hat{\theta}_3 \geq \hat{\theta}$ to $\hat{\theta}_1$ and reflects CQS 3,¹⁹² such that $\hat{\theta}_3$ can be interpreted as the minimum credit rating threshold of the Eurosystem. Fully-informed lenders are willing to pay a price equal to the quality of the bond. Hence, lenders are assumed to apply haircuts h_L that are perfectly differentiated according to bond quality,¹⁹³ such that

$$\Phi_L(\theta_j) = \left(1 - h_L(\theta_j)\right)\Omega(\theta_j) = \rho(\theta_j) = \theta_j.$$
(5.7)

The market value is reflected by $\Omega(\theta)$ with the slope $\partial \Omega(\theta) / \partial \theta = 1$. Likewise, the slope of $\Phi_L(\theta)$ is¹⁹⁴

$$\frac{\partial \Phi_L(\theta)}{\partial \theta} = \frac{\partial \Omega(\theta)}{\partial \theta} - \frac{\partial h_L}{\partial \theta} \Omega(\theta) - \frac{\partial \Omega(\theta)}{\partial \theta} h_L \equiv 1.$$
 (5.8)

Let h_{CB} denote the haircut that the Eurosystem applies to (eligible) bond j of quality θ_j and market value $\Omega(\theta_j)$. The Eurosystem pools repo conditions by applying identical haircuts to bonds of different quality within CQS 1/2 and CQS 3. This implies that $h_{CB}(\theta_j) = h_{CB} \forall \theta_j \in$ $[\hat{\theta}_3, \hat{\theta}_1)$ and the following collateral value assigned by the Eurosystem to bond j:

$$\Phi_{CB}(\theta_j) = (1 - h_{CB}) \,\Omega(\theta_j). \tag{5.9}$$

¹⁹² Note that the interpretation of θ as a broad argument of collateral quality (i.e. reflecting credit quality and liquidity) is narrowed to credit quality for technical convenience, i.e. liquidity is kept constant. CQS 1/2 summarizes eligible assets rated from "triple A" (corresponding to $\bar{\theta}$) to "single A" (corresponding to $\hat{\theta}_1$) while CQS 3 comprises all lower-rated assets down to the rating of "triple B" (corresponding to $\hat{\theta}_3$), cf. Figure 4.5. That is, CQS 1/2 is indexed by 1 and CQS 3 by 3 for the sake of simplicity. Repo conditions are pooled in both CQSs. In Figures 5.15 to 5.17, pooling is exemplified for CQS 3. However, replacing $\hat{\theta}_1$ with $\bar{\theta}$ and $\hat{\theta}_3$ with $\hat{\theta}_1$ would stylize pooling in CQS 1/2.

¹⁹³ Section 5.2.2 provided narrative evidence that lenders in the European repo market cluster collateral properties to a lesser degree than the Eurosystem. Moreover, bilateral repos can be assumed to imply no clustering as repo conditions are negotiated directly between counterparties for each repo.

¹⁹⁴ See Section 5.3.1.2. As $\partial \Omega(\theta) / \partial \theta = 1$, the condition for the identity to hold is $h_L = -(\partial h_L / \partial \theta) \Omega(\theta)$ with $\partial h_L / \partial \theta < 0$.

The slope of $\Phi_{CB}(\theta)$ is given by

$$\frac{\partial \Phi_{CB}(\theta)}{\partial \theta} = \frac{\partial \Omega(\theta)}{\partial \theta} - \frac{\partial h_{CB}}{\partial \theta} \Omega(\theta) - \frac{\partial \Omega(\theta)}{\partial \theta} h_{CB}.$$
(5.10)

Owing to the pooling of public repo conditions, the slope of $\Phi_{CB}(\theta)$ is smaller than that of $\Phi_L(\theta)$, i.e.

$$\frac{\partial \Phi_{CB}(\theta)}{\partial \theta} < 1 = \frac{\partial \Phi_L(\theta)}{\partial \theta}.$$
(5.11)

Therefore, $\Phi_{CB}(\theta)$ is larger than $\Phi_L(\theta)$ for bonds $\theta_j \in [\hat{\theta}_3, \theta_{\tau=0})$ but smaller otherwise. The bond of quality $\theta_{\tau=0}$ marks the threshold bond for which the Eurosystem assigns the same haircut and hence collateral value as private lenders.¹⁹⁵ The Eurosystem offers a subsidy to borrowers whenever repo conditions for a given bond are beneficiary in public repos compared to private repos. The value of this subsidy is given by

$$\Phi_{CB}(\theta_j) - \Phi_L(\theta_j) = \underbrace{\left(h_L(\theta_j) - h_{CB}\right)}_{\tau(\theta_j)} \Omega(\theta_j), \tag{5.12}$$

with $\tau(\theta_j)$ representing the haircut subsidy rate. For all $\theta_j \in [\hat{\theta}_3, \theta_{\tau=0}), \tau(\theta_j) > 0$ which implies that $h_{CB} < h_L(\theta_j)$. Analogously, $\tau(\theta_j) < 0$ for all $\theta_j \in (\theta_{\tau=0}, \hat{\theta}_1)$ and $h_{CB} > h_L(\theta_j)$.

Figure 5.15: Pooling of public repo conditions I: subsidization of collateral



The notion that private lenders are supposed to apply haircuts that reflect true collateral risk has two important implications. The interpretation of the subsidy is twofold: on the one hand, it can be a subsidy in the sense that the Eurosystem provides better conditions than private lenders in the repo market; and on the other, it can be a subsidy in the sense that the Eurosystem applies haircuts that do not reflect true collateral risk. As the market is assumed to impose such

¹⁹⁵ Note that by definition, $\theta_{\tau=0}$ corresponds to θ_{α} as θ_{α} is the bond quality for which the borrower was simply indifferent between the market and the outside option.

haircuts, the first important implication is that both interpretations apply analogously here. Second, adverse selection of collateral leads to increased risk exposure of the Eurosystem.¹⁹⁶ Collateral adversely selected to the Eurosystem increases its risk exposure as risk control is insufficient owing to overly-small haircuts. Lemma 5.2 summarizes the insights on the pooling of public repo conditions by the Eurosystem.

Lemma 5.2 (Subsidization of collateral by the Eurosystem). Owing to the clustering of asset properties in the specification of public repo conditions, the Eurosystem subsidizes a fraction of eligible assets.

- i) A positive subsidy is provided to bonds of relatively low quality within each CQS as smaller haircuts are applied in public compared to private repos.
- *ii)* A negative subsidy is imposed to bonds of relatively high quality within each CQS as larger haircuts are applied in public than in private repos.
- *iii)* For only one bond, no subsidy is provided as the same haircut is applied in a public and a private repo.

Application of Supplementary Haircuts

Section 4.2.3.2 provided an overview of supplementary haircuts (i.e. add-on haircuts and valuation markdowns) that the Eurosystem applies to a broad range of collateral. With the exception of own-use covered bank bonds, supplementary haircuts are not differentiated with respect to credit quality. Hence, the same supplementary haircut is applied to assets within CQSs and even across CQSs. Taking supplementary haircuts into account, Φ_{CB} is expressed as¹⁹⁷

$$\Phi_{CB}(\theta_j) = (1 - h_{CB}^e)\Omega(\theta_j), \qquad (5.13)$$

with

$$h_{CB}^{e} \equiv h_{n} + h_{a} + v - (h_{n} + h_{a}) \cdot v.$$
(5.14)

The last term reflects the interaction term, which biases the effective haircut downwards (see Section 4.3.2.3). This bias is larger for bonds of lower quality, i.e. for which larger haircuts are applied. As the same valuation markdown is applied to theoretically valued assets irrespective of credit quality, assets in CQS 3 are deducted by a relatively smaller effective haircut than those in CQS 1/2. This implies that repo conditions are aligned even across CQSs 1/2 and 3.

Within CQS 3, $h_n(\theta_j) = h_n \forall \theta_j \in [\hat{\theta}_3, \hat{\theta}_1)$ and hence $h_{CB}^e(\theta_j) = h_{CB}^e \forall \theta_j \in [\hat{\theta}_3, \hat{\theta}_1)$. The effective haircut applied to bonds in the same CQS is uniform but larger for any $h_a, v > 0$, i.e. $h_{CB}^e > h_n$. A larger effective haircut lowers the collateral value of a bond with higher market value more than the collateral value of a bond with lower market value. This effect is reflected in Figure 5.16, where Φ_{CB}^e indicates the collateral value that the Eurosystem determines based on

¹⁹⁶ Note that this is not necessarily the case. Risk exposure originates from the relation between the collateral value that the Eurosystem assigns to an asset and its true collateral value. Adverse selection relates to the collateral value assigned by the Eurosystem relative to the one assigned by private lenders. Here, private lenders are supposed to assign the true collateral value to a bond. Hence, adverse selection implies higher risk exposure.

¹⁹⁷ Note that h_{CB}^e is the effective haircut that the central bank applies, h_n is the nominal haircut, h_a the add-on haircut and v the valuation markdown (cf. Section 4.3.2).

the application of the effective haircut (i.e. including supplementary haircuts). As $h_{CB}^e > h_n$, slopes of $\Phi_{CB}(\theta)$ and $\Phi_{CB}^e(\theta)$ differ, i.e.

$$\frac{\partial \Phi_{CB}^{e}(\theta)}{\partial \theta} < \frac{\partial \Phi_{CB}(\theta)}{\partial \theta} < 1 = \frac{\partial \Phi_{L}(\theta)}{\partial \theta} \ \forall \ h_{a}, v > 0.$$
(5.15)

The threshold quality $\theta_{\tau=0}$ is shifted to the left such that more bonds are negatively subsidized and less positively. Hence, the pooling of public repo conditions is fostered for assets to which supplementary haircuts are applied. Accordingly, conditions are not only aligned through uniform valuation haircuts applied to akin assets of different credit quality within the same CQS, but also through the application of supplementary haircuts, which implies a larger uniform effective haircut.

Figure 5.16: Pooling of public repo conditions II: application of supplementary haircuts



Theoretical Valuation of Assets

Section 3.1.4.2 addressed the Eurosystem's principles for collateral valuation, which hold crucial importance for determining collateral values.¹⁹⁸ For non-marketable assets, a theoretical value or simply the outstanding amount is considered. The value of marketable assets is calculated based on the market price. If the last available market price is older than (or has not moved for at least) five days, the Eurosystem assigns a theoretical value. Moreover, theoretical valuation is important for own-use assets, i.e. securities pledged by the issuing counterparty itself as no market price exists. NYBORG 2015 documents that about 77% by count or 17% by value of all eligible assets feature theoretical values. It is addressed in the following that theoretical valuation may foster pooling public repo conditions as it could give rise to valuation errors.

¹⁹⁸ Therefore, valuation of collateral could be interpreted as the de facto seventh property of credit terms (Section 2.3).

In case of valuation errors, the theoretical value $\Psi(\theta_j)$ that the Eurosystem assigns to bond j is given by

$$\Psi(\theta_j) = \left(1 + \varphi(\theta_j)\right) \Omega(\theta_j), \tag{5.16}$$

with $\varphi(\theta_j)$ being the valuation error.¹⁹⁹ The case of valuation errors is stylized in Figure 5.17 under the assumption that all bonds are theoretically valued. All bonds $\theta_j \in [\hat{\theta}_3, \theta_{\tau=0})$ are overvalued and their theoretical value corresponds to $\Psi(\theta_j) = (1 + \varphi(\theta_j))\Omega(\theta_j) > \Omega(\theta_j)$ as $\varphi(\theta_j) > 0$. By contrast, all bonds $\theta_j \in (\theta_{\tau=0}, \hat{\theta}_1)$ are undervalued such that $\Psi(\theta_j) = (1 + \varphi(\theta_j))\Omega(\theta_j) < \Omega(\theta_j)$ with $\varphi(\theta_j) < 0$. Only the valuation of bond $\theta_{\tau=0}$ is unbiased, i.e. $\Psi(\theta_{\tau=0}) = \Omega(\theta_{\tau=0})$ as $\varphi(\theta_{\tau=0}) = 0$. Incorrect valuations may be ascribed to the Eurosystem avoiding disperse theoretical valuation of bonds such that it tends to assign modest theoretical values ("central tendency error", see e.g. GUILFORD 1954 as well as ROSENTHAL and ROSNOW 1991). Moreover, a general misperception of bond quality could prevail.

Figure 5.17: Pooling of public repo conditions III: valuation errors



Valuation errors as described above imply a clockwise rotation of $\Psi(\theta)$ away from a fair theoretical value, which implies a further flattening of $\Phi_{CB}(\theta)$. The larger the misvaluation, the flatter is $\Phi_{CB}(\theta)$. For all bonds $\theta_j \in [\hat{\theta}_3, \theta_{\tau=0})$, the subsidy $\tau(\theta_j)$ increases, while for all other bonds $\theta_j \in (\theta_{\tau=0}, \hat{\theta}_1)$, it becomes more negative. In the most extreme case, this could lead to the slope of $\Phi_{CB}(\theta)$ equal to zero and public repo conditions that are perfectly aligned within CQSs, i.e. $\Phi^e_{CB}(\theta_j) = \Phi^e_{CB}(\theta_{\tau=0}) \forall \theta_j \in [\hat{\theta}_3, \hat{\theta}_1)$ and $\partial \Phi^e_{CB}(\theta)/\partial \theta = 0$.

In conclusion, public repo conditions are pooled within CQSs by clustering asset properties in three steps: (i) the application of simplified haircuts that are too large for high-quality bonds and too small for low-quality bonds; (ii) the application of supplementary haircuts irrespective of credit quality; and (iii) the theoretical valuation of bonds when they lack a market value. However, $\Phi_{CB}(\theta)$ would become perfectly flat $(\partial \Phi_{CB}(\theta)/\partial \theta = 0)$ only for a uniform theoretical valuation of all bonds within the same CQS. Perfectly-pooled repo conditions within CQSs are

¹⁹⁹ $\varphi(\theta_j)$ is characterized as follows: $\partial \varphi(\theta) / \partial \theta < 0$; $\partial^2 \varphi(\theta) / \partial \theta^2 > 0$; $0 < \varphi(0) < 1$; $\varphi(\theta_{\tau=0}) = 0$; $\varphi(\theta_j) > 0 \forall \theta_j \in [0, \theta_{\tau=0})$; $\varphi(\theta_j) < 0 \forall (\theta_{\tau=0}, \infty)$.
assumed as a simplification in the following extension of the basic model to elaborate on the implications of the design of the outside option as offered by the Eurosystem. All implications would qualitatively hold if repo conditions were not perfectly pooled within CQSs, i.e. $0 < \partial \Phi_{CB}(\theta)/\partial \theta < 1 = \partial \Phi_L(\theta)/\partial \theta$.

5.3.2.2 Extended Model With Pooled Public Repo Conditions and Correct Ratings

The basic model with asymmetric information and correct ratings is extended by introducing pooled public repo conditions as derived in the previous section. The stylized case of perfectlypooled public repo conditions within CQSs is modeled hereafter such that

$$\alpha(\theta_j) = \Phi^e_{CB}(\theta_j) = \left(1 - h^e(\theta_j)\right) \Psi(\theta_j)$$

= $\Phi^e_{CB}(\theta_{\alpha_k}) = \left(1 - h^e(\theta_{\alpha_k})\right) \Psi(\theta_{\alpha_k}) = \alpha(\theta_{\alpha_k}) \ \forall \ \theta_j, \theta_{\alpha_k},$ (5.17)

with $k \in \{1,3\}$ indicating the CQS and θ_{α_k} the bond against which public repo conditions are perfectly pooled.²⁰⁰ Bond quality is supposed to be unknown to lenders but signaled correctly by the rating agency such that $s(\theta_j) = \theta_j \forall \theta_j$. The Eurosystem defines CQSs according to credit ratings such that bonds rated between \hat{s}_1 and \bar{s} are arranged in CQS 1/2 and those rated between \hat{s}_3 and \hat{s}_1 are summarized in CQS 3. Moreover, the Eurosystem ties down for each CQS k a rating s_{α_k} , implicitly determining the bond for which the owner is indifferent between entering a private or a public repo. Public repo conditions for bonds rated within the same CQS k are pooled and aligned to the private repo condition of this bond rated s_{α_k} that are given by $\Phi_L(\theta_{\alpha_k}) = \rho(\theta_{\alpha_k}) = s(\theta_{\alpha_k}) = \theta_{\alpha_k}$. Hence, public repo conditions are determined by the Eurosystem such that $\Phi^e_{CB}(\theta_{\alpha_k}) = \alpha(\theta_{\alpha_k}) = \theta_{\alpha_k}$.

In CQS 1/2, $\alpha(\theta_{\alpha_1})$ is further specified as $\alpha(\theta_{\alpha_1}) \equiv \frac{\rho(\bar{s}) + \rho(\hat{s}_1)}{b_1}$ with b_1 implicitly defining the fraction of bonds that is subsidized relative to private repos. b_1 is restricted by $[\underline{b}_1 \equiv \frac{\rho(\bar{s}) + \rho(\hat{s}_1)}{\rho(\bar{s})}, \overline{b}_1 \equiv \frac{\rho(\bar{s}) + \rho(\hat{s}_1)}{\rho(\hat{s}_1)}]$ with \underline{b}_1 being the lower and \overline{b}_1 the upper bound of b_1 . For CQS 3, it holds analogously that $\alpha(\theta_{\alpha_3}) \equiv \frac{\rho(\hat{s}_1) + \rho(\hat{s}_3)}{b_3}$ with $b_3 \in [\underline{b}_3 \equiv \frac{\rho(\hat{s}_1) + \rho(\hat{s}_3)}{\rho(\hat{s}_1)}, \overline{b}_3 \equiv \frac{\rho(\hat{s}_1) + \rho(\hat{s}_3)}{\rho(\hat{s}_3)}]$. Hence, the Eurosystem can implicitly determine the fraction of bonds rated within each CQS k that it subsidizes relative to private repos through the choice of b_k .²⁰¹ The higher b_k , the lower the fraction of bonds in CQS k that are subsidized relative to private repos. As $s(\theta_j) = \theta_j \forall \theta_j$, public repo conditions are given by

$$\alpha(\theta_j) = \begin{cases} 0 & \forall \ \theta_j : s(\theta_j) \in [\underline{s}, \hat{s}_3); \\ \frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3} & \forall \ \theta_j : s(\theta_j) \in [\hat{s}_3, \hat{s}_1); \\ \frac{\bar{\theta} + \hat{\theta}_1}{b_1} & \forall \ \theta_j : s(\theta_j) \in [\hat{s}_1, \bar{s}]. \end{cases}$$
(5.18)

Bonds rated below the minimum rating threshold \hat{s}_3 are ineligible in public repos such that the outside option of borrowers owning these bonds is to leave bonds unexploited and receive nothing. Bonds rated at or above \hat{s}_3 are eligible in public repos. Public repo conditions are

 $[\]overline{}^{200}$ Note again that for the sake of clarity, CQS 1/2 is indexed by 1 (and CQS 3 by 3).

²⁰¹ Note that $b_k > 1$. For $b_k = 2$, the bond rated θ_{α_k} would correspond to the median-rated bond in CQS k such that half of bonds are subsidized and the other half are discriminated relative to private repos.

pooled for bonds of different quality rated within the same CQS k and aligned to the conditions that private lenders would stipulate for the bond of quality θ_{α_k} . Hence, the value of the outside option for bonds rated within CQS 1/2 is $\alpha(\theta_j) = \frac{\bar{\theta} + \hat{\theta}_1}{b_1} \forall \theta_j : s(\theta_j) \in [\hat{s}_1, \bar{s}]$ and $\alpha(\theta_j) = \frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3} \forall \theta_j : s(\theta_j) \in [\hat{s}_3, \hat{s}_1)$ for bonds rated within CQS 3.

The situation is stylized in Figure 5.18 with correct ratings being reflected by the dotted line. Borrowers who own a bond of quality $\theta_j < \hat{\theta}_3$ sell this bond on the market at the price $\rho(\theta_j) = s(\theta_j) = \theta_j$. The same holds for borrowers owning a bond of relatively high-quality in each CQS, i.e. bonds of quality $\theta_j \in [\theta_{\alpha_3}, \hat{\theta}_1) \cap [\theta_{\alpha_1}, \bar{\theta}]$, as the borrowers receive at least as much in the market as in a public repo, i.e. $\rho(\theta_j) \ge \alpha(\theta_j) \forall \theta_j \in [\theta_{\alpha_3}, \hat{\theta}_1) \cap [\theta_{\alpha_1}, \bar{\theta}]$. However, bonds of relatively low quality in each CQS, i.e. bonds of quality $[\hat{\theta}_3, \theta_{\alpha_3}) \cap [\hat{\theta}_1, \theta_{\alpha_1})$ are pledged in public repos with the Eurosystem because the outside option gives the borrowers more liquidity, i.e. $\alpha(\theta_j) > \rho(\theta_j) \forall \theta_j \in [\hat{\theta}_3, \theta_{\alpha_3}) \cap [\hat{\theta}_1, \theta_{\alpha_1})$.





While private lenders are again left with zero rent $(R_L = 0)$, the rent of borrowers corresponds to the aggregate surplus $(R_B = R)$ and is given by²⁰²

$$R_{B} = \frac{\bar{\theta}^{2} - \underline{\theta}^{2}}{2} + \underbrace{\frac{\hat{\theta}_{1}^{2} + \hat{\theta}_{3}^{2}}{2} + \frac{1}{2} \left(\frac{\hat{\theta}_{1} + \hat{\theta}_{3}}{b_{3}}\right)^{2} + \frac{1}{2} \left(\frac{\bar{\theta} + \hat{\theta}_{1}}{b_{1}}\right)^{2} - \frac{\hat{\theta}_{3}(\hat{\theta}_{1} + \hat{\theta}_{3})}{b_{3}} - \frac{\hat{\theta}_{1}(\bar{\theta} + \hat{\theta}_{1})}{b_{1}}}_{additional rent from pooled public repo conditions}.$$
 (5.19)

 R_B is maximized for b_k equal to the lower bound, i.e. $b_1 = \frac{\rho(\bar{s}) + \rho(\hat{s}_1)}{\rho(\bar{s})}$ as well as $b_3 = \frac{\rho(\hat{s}_1) + \rho(\hat{s}_3)}{\rho(\hat{s}_1)}$, and is minimized for b_k equal to the upper bound, i.e. $b_1 = \frac{\rho(\bar{s}) + \rho(\hat{s}_1)}{\rho(\hat{s}_1)}$ and $b_3 = \frac{\rho(\hat{s}_1) + \rho(\hat{s}_3)}{\rho(\hat{s}_3)}$. In the former case, all eligible bonds (except for the bond for which the owner is indifferent) are pledged with the Eurosystem and only ineligible bonds, i.e. $\theta_j \in [\theta, \hat{\theta}_3)$, are sold on the market. In the latter case, no borrower turns to the Eurosystem and all bonds are sold on the market.

 $^{^{202}}$ See the appendix for the derivation of Equation (5.19).

In this case, the benchmark outcome (symmetric information) is mimicked as borrowers receive rent to the amount of $R_B = \frac{\bar{\theta}^2 - \bar{\theta}^2}{2}$.

For any b_k smaller than the upper bound, the Eurosystem provides additional rent to borrowers, as reflected in Equation (5.19) and indicated by the shaded areas in Figure 5.18. This provision is at the risk of the Eurosystem as it turns into a loss of the Eurosystem if the borrower fails to repurchase the bond in the closing leg. Hence, the Eurosystem bears risk that is equal to the additional rent provided to borrowers and corresponds to

$$R_{CB} = -\left(\frac{\hat{\theta}_1^2 + \hat{\theta}_3^2}{2} + \frac{1}{2}\left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3}\right)^2 + \frac{1}{2}\left(\frac{\bar{\theta} + \hat{\theta}_1}{b_1}\right)^2 - \frac{\hat{\theta}_3(\hat{\theta}_1 + \hat{\theta}_3)}{b_3} - \frac{\hat{\theta}_1(\bar{\theta} + \hat{\theta}_1)}{b_1}\right).$$
 (5.20)

The Eurosystem pays prices $\alpha(\theta_{\alpha_k})$ and receives bonds that have to be liquidated in case of borrower default. Liquidation would give the Eurosystem θ_j for each bond j such that it would be left with a loss to the amount of $\alpha(\theta_{\alpha_k}) - \theta_j$ for each bond j. R_{CB} positively depends on b_k since an increase in b_k implies fewer bonds being subsidized by the Eurosystem and more borrowers turning to the market.

This represents a rationale for GLOC according to which there is adverse selection of collateral. The Eurosystem receives relatively low-quality bonds within each CQS, while bonds of relatively high quality are sold on the market. The reason for this finding is the subsidization of relatively low-quality bonds in each CQS through the pooling of repo conditions. Upon first glance, aggregate surplus in the market is increased beyond the benchmark case with symmetric information. Some borrowers enjoy such good conditions in public repos that they can realize a higher return than they would when selling the bonds on the market. The rent of lenders remains zero. Upon further consideration, additional rent for borrowers is at the cost of the Eurosystem, which bears risk in terms of a potential loss (illustrated by the shaded areas in Figure 5.18). In case of borrower default, bonds would have to be liquidated rather than being repurchased in the closing leg. The Eurosystem could then realize a liquidation value smaller than the liquidity provided in the opening leg. In case of this adverse outcome, the additional rent of borrowers in the opening leg is matched by a loss of the Eurosystem thereafter. This finding is summarized in the following proposition.

Proposition 5.2 (Pledge of low-quality bonds in public repos). By offering a uniform outside option to bonds within each CQS, the Eurosystem attracts low-quality bonds within each CQS.

- i) While bonds of relatively high quality within each CQS are sold on the market,
- *ii)* bonds of relatively low quality within each CQS are used in public repos with the Eurosystem.
- *iii)* Rent of borrowers owning bonds of relatively low quality within each CQS is increased at the risk of the Eurosystem.

The finding shows that borrowers turn to the central bank despite information on bond quality being de facto symmetric. Essentially, a borrower is assumed to seek the most profitable yet feasible utilization for his bond. Whenever the conditions offered by the central bank are better than in the repo market and the bond is eligible for a public repo, the borrower will turn to the central bank. Therefore, pooling of public repo conditions is one explanation for the observation that collateral was increasingly pledged with the Eurosystem rather than being sold on the market. Another possible explanation is that it is not feasible for borrowers to sell their bond on the market. STIGLITZ and WEISS 1981 describe such a constraint as a credit rationing of the supply side. Contrasting the view that the underprovision of credit represents a temporary disequilibrium through an exogenous shock, a loan market may in equilibrium be characterized by credit rationing. Similarly, in the present model, the possibility to sell bonds on the market could be limited and borrowers would have to turn to the outside option. However, with pooling of public repo conditions, the Eurosystem would still underbid the market and provide better conditions than private lenders for at least a fraction of bonds.

5.3.2.3 Extended Model with Pooled Public Repo Conditions and Distorted Ratings

In the last step of the analysis, ratings of bond quality are no longer assumed to be correct but rather positively proportionally distorted in the opening leg.²⁰³ This corresponds to the case in which the quality assessment technology of the rating agency is systematically biased owing e.g. to overly optimistic market sentiment.

With positive proportional distortion, bonds are rated $s(\theta_j) = \max[(\theta_j + \epsilon), \bar{\theta}]$ with $\epsilon > 0$. Distorted ratings result in a bias of $\Omega(\theta_j)$ and of $\Phi_L(\theta_j)$ such that $\Phi_L(\theta_j) = \rho(\theta_j) = s(\theta_j) = \max[(\theta_j + \epsilon), \bar{\theta}]$. Distorted ratings are depicted by the dotted line in Figure 5.19. As the Eurosystem defines rating thresholds for eligibility and between CQSs in terms of minimum ratings, i.e. \hat{s}_k for $k \in \{1, 3\}$, bond quality levels that are incorporated into CQSs de facto range from $\hat{\theta}_1 - \epsilon$ to $\bar{\theta}$ in CQS 1/2 and from $\hat{\theta}_3 - \epsilon$ to $\hat{\theta}_1 - \epsilon$ in CQS 3 as $s(\hat{\theta}_1 - \epsilon) = \hat{s}_1$ and $s(\hat{\theta}_3 - \epsilon) = \hat{s}_3$.

The borrower who is simply indifferent between entering a private and a public repo is no longer the owner of the bond of quality θ_{α_k} because bond quality implying indifference decreases due to the distortion to $\theta_{\alpha_k} - \epsilon$. This implies that the value of the outside option in CQS k is endogenous to the Eurosystem and decreases by ϵ . While the number of subsidized bonds is unaffected (as $\theta_{\alpha_k} - \epsilon - \hat{\theta}_k + \epsilon = \theta_{\alpha_k} - \hat{\theta}_k$), subsidized bonds are of lower quality. Bonds of quality higher than $\hat{\theta}_3 - \epsilon$ are signaled to be of quality $\theta_j \geq \hat{s}_3$ and are eligible in a public repo at pooled conditions $\alpha(\theta_{\alpha_1}) = \alpha(\theta_j) \forall \theta_j \in [\hat{\theta}_1 - \epsilon, \bar{\theta}]$, i.e. within CQS 1/2, and $\alpha(\theta_{\alpha_3}) = \alpha(\theta_j) \forall \theta_j \in [\hat{\theta}_3 - \epsilon, \hat{\theta}_1 - \epsilon)$ within CQS 3.

Figure 5.19 illustrates that bonds of quality $\theta_j \in [\hat{\theta}_3 - \epsilon, \theta_{\alpha_3} - \epsilon) \cup [\hat{\theta}_1 - \epsilon, \theta_{\alpha_1} - \epsilon)$ are pledged in public repos with the Eurosystem rather than being sold on the market. CQS 3 is effectively shifted to bonds of lower quality, while CQS 1/2 is extended to the extent of the distortion. As shown in Section 5.3.1.4, the positive distortion of ratings implies higher rents for borrowers at the expense of lenders who believe in the signal and are willing to pay prices that are too high

²⁰³ See EBERL and WEBER 2013 for the case of negatively proportionally distorted ratings. In this case, the subsidized bonds are of higher quality. The Eurosystem bears less risk as it receives bonds of higher quality. If the Eurosystem believes in market prices that are too low relative to true bond quality owing to overly prudent ratings, it can reduce the price distortion and offer a price that better reflects the true quality. However, this implies a loss in aggregate surplus as borrowers turn to the Eurosystem, which brings them more than in the market (owing to the negatively distorted signal) but on aggregate gives less than the productive use of bonds by private lenders.

for the given bond quality. For lenders receiving bonds of quality $\theta_j > \bar{\theta} - \epsilon$, the adverse effect of distorted ratings decreases towards zero with increasing bond quality.





Lenders realize rent to the amount of^{204}

$$R_L = -\left(\bar{\theta} - \underline{\theta} - \frac{1}{2}\epsilon + \hat{\theta}_1 + \hat{\theta}_3 - \frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3} - \frac{\bar{\theta} + \hat{\theta}_1}{b_1}\right)\epsilon,\tag{5.21}$$

which negatively depends on b_k and vanishes for $\epsilon = 0$. Moreover, Equation (5.21) resembles Equation (5.4) for b_k set at the upper bound, while R_L is smaller for any smaller b_k as more collateral is pledged in public repos. By contrast, borrowers receive rent that is given by²⁰⁵

$$R_{B} = \frac{\bar{\theta}^{2} - \underline{\theta}^{2}}{2} + \underbrace{\frac{\hat{\theta}_{1}^{2} + \hat{\theta}_{3}^{2}}{2} + \frac{1}{2} \left(\frac{\hat{\theta}_{1} + \hat{\theta}_{3}}{b_{3}}\right)^{2} + \frac{1}{2} \left(\frac{\bar{\theta} + \hat{\theta}_{1}}{b_{1}}\right)^{2} - \frac{\hat{\theta}_{3}(\hat{\theta}_{1} + \hat{\theta}_{3})}{b_{3}} - \frac{\hat{\theta}_{1}(\bar{\theta} + \hat{\theta}_{1})}{b_{1}}}_{\text{add. rent from pooled public repo conditions}}$$

$$+ \underbrace{\left(\bar{\theta} - \underline{\theta} - \frac{1}{2}\epsilon\right)\epsilon}_{\text{add. rent from rating distortion}}$$
(5.22)

add. rent from rating distortion

which is convexly decreasing in b_k . For $\epsilon = 0$, Equation (5.22) resembles Equation (5.19) as the case of correct ratings would be mimicked. Plugging in the upper bound of b_k , i.e. $\bar{b}_1 = \frac{\bar{\theta} + \hat{\theta}_1}{\hat{\theta}_1}$ and $\bar{b}_3 = \frac{\hat{\theta}_1 + \hat{\theta}_3}{\hat{\theta}_3}$, gives $R_B = \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} + (\bar{\theta} + \underline{\theta} - \frac{1}{2}\epsilon)\epsilon$, which mimics R_B under asymmetric information and a positively distorted signal but without an outside option as all bonds are sold on the market. Likewise, $R_L = -(\bar{\theta} - \underline{\theta} - \frac{1}{2}\epsilon)\epsilon$, i.e. R_L under asymmetric information and positively distorted signal but without outside option. An increase in the value of the outside option from lowering b_k down to $\underline{b}_1 = \frac{\bar{\theta} + \hat{\theta}_1}{\bar{\theta}}$ and $\underline{b}_3 = \frac{\hat{\theta}_1 + \hat{\theta}_3}{\hat{\theta}_1}$ makes R_L less negative toward

 $^{^{204}}$ See the appendix for the derivation of Equation (5.21).

 $^{^{205}}$ See the appendix for the derivation of Equation (5.22).

 $R_L = -(\hat{\theta}_3 - \underline{\theta} - \frac{1}{2}\epsilon)\epsilon$, while R_B increases convexly. Aggregate surplus in the market corresponds to

$$R = \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} + \frac{\hat{\theta}_1^2 + \hat{\theta}_3^2}{2} + \frac{1}{2} \left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3} \right)^2 + \frac{1}{2} \left(\frac{\bar{\theta} + \hat{\theta}_1}{b_1} \right)^2 - \frac{\hat{\theta}_3(\hat{\theta}_1 + \hat{\theta}_3)}{b_3} - \frac{\hat{\theta}_1(\bar{\theta} + \hat{\theta}_1)}{b_1} + \left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3} + \frac{\bar{\theta} + \hat{\theta}_1}{b_1} - \hat{\theta}_3 - \hat{\theta}_1 \right) \epsilon,$$
(5.23)

which differs from aggregate surplus with correct ratings by the last term, i.e. for $\epsilon > 0$.

Risk to the Eurosystem is illustrated by the dark-shaded areas in Figure 5.19 and given by 206

$$R_{CB} = -\left(\frac{\hat{\theta}_{1}^{2} + \hat{\theta}_{3}^{2}}{2} + \frac{1}{2}\left(\frac{\hat{\theta}_{1} + \hat{\theta}_{3}}{b_{3}}\right)^{2} + \frac{1}{2}\left(\frac{\bar{\theta} + \hat{\theta}_{1}}{b_{1}}\right)^{2} - \frac{\hat{\theta}_{3}(\hat{\theta}_{1} + \hat{\theta}_{3})}{b_{3}} - \frac{\hat{\theta}_{1}(\bar{\theta} + \hat{\theta}_{1})}{b_{1}}\right) - \epsilon\left(\frac{\hat{\theta}_{1} + \hat{\theta}_{3}}{b_{3}} + \frac{\bar{\theta} + \hat{\theta}_{1}}{b_{1}} - \hat{\theta}_{3} - \hat{\theta}_{1}\right),$$
(5.24)

which again positively depends on b_k and mimics risk to the Eurosystem in case of correct ratings, i.e. Equation (5.20), for $\epsilon = 0$. As the value of the outside option for eligible bonds is implicitly determined by private repo conditions, it is upward-distorted, just like prices on the market. In addition, the Eurosystem deems eligible bonds of quality lower than its initial quality threshold $\hat{\theta}_3$ to the extent of $\hat{\theta}_3 - \epsilon$. Therefore, the quality of bonds pledged with the Eurosystem is lower as CQS 3 is effectively shifted to bonds of lower quality and CQS 1/2 is extended to the extent of the rating distortion. The coincidence of pooling of public repo conditions and upward distortion of credit ratings amplifies risk to the Eurosystem. For each bond pledged with the Eurosystem, the risk corresponds to the distance between the price the Eurosystem pays $\alpha(\theta_j)$ and true quality θ_j as in case of counterparty default the Eurosystem has to liquidate the bond. The Eurosystem generates additional rent for borrowers exceeding the additional rent that private lenders already provide owing to distorted ratings (light-shaded area). The findings of this last step of analysis are summarized in Proposition 5.3.

Proposition 5.3 (Intensified pledge of low-quality bonds in public repos owing to positively distorted ratings). With the systematic overvaluation of bond quality owing to a positively distorted signal, the Eurosystem's attraction of low-quality bonds is intensified. Both CQSs are expanded beyond the control of the Eurosystem to the extent of the rating distortion.

- 1. While bonds of relatively high quality in each CQS are sold on the market, eligible bonds of relatively low quality in each CQS are used in public repos with the Eurosystem. The quality of bonds attracted by the Eurosystem is lower than in the case without rating distortion.
- 2. Additional rent for borrowers comes at the risk of private lenders owing to the distorted signal. For borrowers owning bonds of relatively low quality within each CQS, the rent is even further increased at the risk of the Eurosystem.

 $^{^{206}\,\}mathrm{See}$ the appendix for the derivation of Equation (5.24).

5.4 Credit Quality of Collateral Pledged with the Eurosystem Revisited

The model elaborated on the adverse selection of collateral, finding that the central bank would attract assets of relatively low quality if it pooled repo conditions for collateral of different quality. This attraction of low-quality assets is intensified for positively distorted ratings as collateral eligibility is extended and beyond the control of the Eurosystem. Thus, pooling of public repo conditions is a potential means of the central bank to expand liquidity provision to mitigate crisis. However, crisis mitigation comes along with cost of adverse qualitative effects that imply increase in collateral risk. Collateral risk was already addressed in Section 5.1 when the credit quality of pledged assets was investigated (right panel of Figure 5.3). However, this investigation was superficial as credit quality was considered in terms of CQSs, which comprise up to seven credit rating notches. No insight was provided into the distribution of collateral quality within CQSs, i.e. across credit rating notches. The following revisits the distribution of collateral and refines the right panel of Figure 5.3 to the distribution of collateral across credit rating notches. Four stylized cases are considered: (i) uniform distribution across rating notches, (ii) distribution reflecting the one of eligible marketable assets on 6 March 2015, (iii) distribution taking into account linear adverse selection of collateral and (iv) distribution allowing for progressive adverse selection.²⁰⁷

Initially, it is supposed that the distribution of collateral is uniform across rating notches. The development of collateral credit quality for this scenario is depicted in the left panel of Figure 5.20. However, Figure 4.6 exemplified the distribution of eligible marketable assets for one specific day, which was not uniform across rating notches. The right panel of Figure 5.20 projects this anecdotal evidence of the distribution of eligible marketable assets on pledged collateral. Compared to the previous scenario, the fraction of "AAA" and "AA"-rated collateral increases. Moreover, it is supposed that more assets with relatively low rating such as "A" and "BBB+" were pledged.

However, these scenarios neglect the implications of GLOC derived within the model of the previous section. According to GLOC, collateral of relatively low quality in each CQS is pledged with the central bank. This implies that in CQS 1/2, assets rated "A-" are more likely pledged than assets rated "A", which in turn are more likely pledged than "A+"-rated assets et cetera.²⁰⁸ However, this "pecking order" may not be fully observed in reality for the following reasons. First, the model made borrowers' decision where to draw liquidity solely conditional upon haircuts, which were considered the exclusive determinant of repo conditions. Nonetheless, in reality, the six factors examined in Section 2.3 jointly determine repo conditions. Second, it was assumed that a continuum of borrowers was equipped with one bond each. In reality, borrowers usually own a portfolio of collateral at disposal and the decision concerning which of the assets to pledge may deviate from the pecking order derived above. Third, the repo market considerably contracted (GORTON and METRICK 2012, KRISHNAMURTHY et al. 2014) and tightened collateral criteria (CHEUN et al. 2009, ECB 2013a, MANCINI et al. 2015) during times of financial

²⁰⁷ Implications of adverse selection are derived for the credit quality of collateral in the following. Likewise, adverse selection is at work with respect to other asset properties such as liquidity.

²⁰⁸ Likewise, in CQS 3, "BBB-"-rated assets are more likely pledged than "BBB"-rated assets, which in turn are more likely used than assets rated "BBB+".

Figure 5.20: Refined credit quality of inferred pledged collateral I

The figure refines the right panel in Figure 5.3 by elaborating on the distribution of collateral within CQSs, i.e. across credit rating notches. The left panel supposes that this distribution is uniform across rating notches and the right panel mirrors anecdotal evidence on the distribution of eligible marketable assets across rating notches (Figure 4.6). By comparison, the fraction of both high- ("AAA" and "AA") and lower-quality collateral ("A" and "BBB+") is larger in the right than the left panel.



Source: author's calculation; National Central Banks; European Central Bank, Eligible Assets Database.

distress, with the consequence that borrowers could have been constrained to pledge assets with the Eurosystem that they would have pledged in the market in normal times. Fourth, if there was no demand for specific assets in special repos, borrowers would have to pledge assets with the Eurosystem. Finally, the stigma of borrowing from the central bank may discourage owners of low-quality collateral to pledge with the central bank.²⁰⁹ Hence, it is important to understand GLOC as a basic principle that gives rise to some degree of adverse selection of collateral rather than being fully observable in reality.

Therefore, Figure 5.21 presents different degrees of adverse selection of collateral. The left panel assumes adverse selection that is linear across credit rating notches, i.e. the fraction of pledged collateral decreases linearly with increasing credit quality within CQSs. The right panel illustrates the scenario in which adverse selection is progressive, i.e. the fraction of pledged collateral decreases degressively with increasing credit quality within CQSs. Average credit quality is lower in the latter scenario as the majority of pledged collateral is considered relatively low quality within each CQSs (i.e. "A-" and "BBB-"). Compared to the previous scenarios (Figure 5.20), the average credit quality of pledged collateral is estimated to have deteriorated owing to the adverse selection of collateral.

Sections 4.4 and 5.2.2 presented indicative evidence on adverse selection of collateral for the Eurosystem.²¹⁰ Figure 5.22 reaffirms the adverse selection between 2007 and 2013.²¹¹ The

²⁰⁹ On the stigma of central bank borrowing, see e.g. Allen and MOESSNER 2013, ARMANTIER et al. 2013, DEUTSCHE BUNDESBANK 2014 as well as DOMANSKI et al. 2014.

²¹⁰ The former revealed that the average haircut (as a proxy of asset quality) on pledged collateral was considerably stronger than for eligible marketable assets, while the latter indicated that the decline of credit quality of pledged collateral was stronger for the European repo market.

²¹¹ See also BINDSEIL and PAPADIA 2006, CHAILLOUX et al. 2008b and NYBORG 2015 for anecdotal evidence over different periods.

Figure 5.21: Refined credit quality of inferred pledged collateral II

The figure elaborates on the distribution of collateral across rating notches in due consideration of different degrees of adverse selection of collateral. The left panel assumes adverse selection that is linear across credit rating notches, i.e. the fraction of pledged collateral decreases linearly with increasing credit quality within CQSs. The right panel illustrates the scenario in which adverse selection is progressive, i.e. the fraction of pledged collateral decreases degressively with increasing credit quality within CQSs.



Source: author's calculation; National Central Banks; European Central Bank, Eligible Assets Database.

blue line details the average probability of default of eligible marketable assets and the red line illustrates the average probability of default of pledged collateral. The figure reveals that while the average default probability of eligible marketable increased subtly yet steadily from 0.1% in May 2007 to 0.14% in December 2013, the increase in the default probability of pledged collateral was stronger from 0.04% in August 2007 to 0.15% in April 2013. The average default probability of pledged collateral has outvalued that of eligible marketable assets as of October 2011. The potential narrative of this development is twofold: on the one hand, adverse selection of collateral gives an intuitive explanation for the asymmetric development of average default probabilities; and on the other, non-marketable assets have been increasingly pledged (Section 5.1). As these assets are not reflected by the blue line, the asymmetric development of default probabilities may to some extent be ascribed to the increased pledge of non-marketable assets. Therefore, the figure reveals that the default probability of pledged collateral substantially increased from 2007 to 2013. This increase can inter alia be attributed to relaxations of Eurosystem collateral criteria, the effects of which were amplified by the adverse selection of collateral.

The central bank attraction of specific tranches of eligible assets constitutes an effective market intervention with two broad effects (CHAILLOUX et al. 2008b, NYBORG 2015). First, it affects asset markets as it influences commercial banks' asset-liability management decisions and tends to change relative prices for collateral assets. While the former could provide incentives for a larger level of liquidity leverage (CHAILLOUX et al. 2008b), the latter gives rise to the overproduction of low-quality assets (NYBORG 2015).²¹² Second, it makes central bank finances more prone to adverse shocks. The attraction of low-quality collateral would increase the Eurosys-

²¹² NYBORG 2015 argues that the overproduction of illiquid assets is inefficient and may amplify business cycles. The author concludes that a central bank promoting this overproduction in a depressed economy impedes its recovery.

Figure 5.22: Avg. default probability (1-year horizon) of eligible marketable and pledged assets

The figure depicts the development of the average probability of default (over a one-year horizon) of eligible marketable assets (blue line) and pledged assets (red line). It provides indicative evidence concerning the adverse selection of collateral as the increase in the average default probability was larger for pledged collateral (from 0.04% in August 2007 to 0.15% in April 2013) than for eligible marketable assets (from 0.1% in May 2007 to 0.14% in December 2013).



tem's risk exposure if additional collateral risk was not sufficiently hedged. Moreover, it affects the overall liquidity of the central bank balance sheet, which becomes less flexible. As central bank and government finances are naturally intertwined, the accumulation of low-quality assets in the central bank balance sheet has the potential to adversely affect government finances and hence it bears fiscal implications. Accordingly, fiscal implications of the Eurosystem collateral framework are addressed in depth in the following.

6

Framework for Analyzing the Fiscal Implications of Central Bank Collateral Criteria

This chapter introduces the framework for analyzing the fiscal implications of the Eurosystem collateral framework. These fiscal implications arise as amendments to collateral criteria induce dynamics in the government liability matrix, i.e. they cause the occurrence and transition of government liabilities. This is important in institutional settings with one government and one central bank. Importance is fostered in a monetary union where collateral criteria can have cross-country fiscal implications. Fiscal implications are analyzed within the framework of fiscal sustainability. The common framework of fiscal sustainability analysis under certainty is extended to uncertainty to factor in fiscal risk. Moreover, an intuitive indicator for assessing fiscal implications is derived. This indicator facilitates identifying sustainable and unsustainable fiscal policy stances and reflects necessary adjustments to fiscal policy induced by amendments to collateral criteria. The chapter is structured as follows. Section 6.1 introduces the concept of fiscal sustainability and argues that this concept is suitable for investigating the fiscal implications of collateral criteria. However, fiscal sustainability under certainty, as addressed in Section 6.2, fails to incorporate fiscal risk, which is vital to the analysis at hand and thus discussed in Section 6.3. Fiscal risk is introduced into the extended framework of fiscal sustainability analysis under uncertainty in Section 6.4. Moreover, this section derives an intuitive indicator for assessing fiscal implications of the collateral framework. Section 6.5 provides an overview of the fiscal implications of the Eurosystem collateral framework that are examined in depth in the following chapters.

The rationale for investigating the fiscal implications of central bank collateral criteria is the close (financial) connection between the central bank and the government. This connection originates from the institutional arrangement in which the government assigns banknote issuance to an operationally independent and specialized agency, i.e. the central bank. The central bank is linked to the government as the latter is usually the only or main shareholder of the former.²¹³ Accordingly, the central bank receives the right to issue the legal tender, while in return the government is entitled to receive (part of) the central bank's profit.²¹⁴ At the same time, the financial link also proceeds from the government to the central bank as the government implicitly backs the central bank through its ability to tax, which gives it the power to always recapitalize the central bank.²¹⁵ Based on this close connection, part of the literature calls for a consolidation of government and central bank budgets (BUITER 2004, 2007, BUITER and RAHBARI 2012a).

The fiscal implications of collateral criteria may vary according to the time horizon. Consider e.g. potential effects of counterparty default on refinancing credit that is not sufficiently backed by collateral: the immediate effect on the government budget would be short-term as write-offs of central bank interest income implies a lower transfer to the government when write-offs are realized. However, the effect on the government budget would also be long term as counterparty default lastingly affects central bank interest income. The central bank cannot compensate for written-off refinancing credit without inducing inflation risk. Therefore, the short-term loss in interest income implies a long-term loss as interest income is permanently decreased for a given inflation rate.

Fiscal sustainability analysis provides a convenient technical framework to elaborate on the fiscal implications of collateral criteria over both the short- and long-term horizon. It examines whether the current fiscal policy stance can be maintained over time or whether adjustment is necessary to avoid debt levels that the government cannot bear. Therefore, the focus is usually not on default itself but rather on the policy changes necessary to avoid default. Consideration of the transfer from the central bank to the government facilitates the analysis of the fiscal implications of collateral criteria. Moreover, fiscal sustainability provides a suitable partial equilibrium framework to respect both the short- and long-term nature of the implications. The advantage of the partial equilibrium approach is that it highlights potential reliefs or constraints that amendments to collateral criteria may impose on fiscal policy. However, the application of this

²¹³ For instance, the shareholders of NCBs in the Eurosystem are the sovereigns, except for the Banque Nationale de Belgique, the Bank of Greece and the Banca D'Italia, which are listed on the stock market. See BUITER and SIBERT 2005, BUITER 2008a, COUR-THIMANN 2013 and SINN 2015a. Potentially adverse effects of the close connection are investigated in e.g. IZE 2006 as well as IZE and OULIDI 2009. The separation must not be compromised in conducting monetary and fiscal policy. Central banks may be legally and financially linked to the government (and vice versa) but monetary policy is considered optimal when it is independent from fiscal policy, see ROGOFF 1985, PERSSON and TABELLINI 1993, ALESINA and GATTI 1995, FISCHER 1995, WALSH 1995 and BERGER et al. 2001.

²¹⁴ See HAWKINS 2003 as well as SINCLAIR and MILTON 2011. As a discretionary distribution of central bank revenue to the government could be regarded as state financing through the central bank, rules are usually formally laid down for the transfer. See e.g. HAWKINS 2003 for an overview of how this transfer is determined in several countries.

²¹⁵ The link is implicit as while the government is entitled to receive parts of central bank's profits, the study by LÖNNBERG and STELLA 2008 finds that governments are usually not explicitly responsible for central bank obligations. This asymmetric treatment of profits and losses is further addressed in Chapter 7.

framework is connected with the common drawbacks of partial equilibrium, i.e. general equilibrium effects are neglected (BALASSONE and FRANCO 2000, GIAMMARIOLI et al. 2007, ECB 2011c). Specifically, the analysis leaves aside interactions between fiscal policy and economic variables.²¹⁶

The concept of fiscal sustainability evolved from the extensive discussion of the effects of government debt on the economy, dating back to David Hume, Adam Smith and David Ricardo.²¹⁷ Concerns about the potential adverse effects of government debt were also raised by DOMAR 1944, who articulated that "[...] continuous government borrowing results in an ever-rising public debt, the servicing of which will require higher and higher taxes; and that the latter will eventually destroy our economy or result in outright repudiation of the debt" (cf. ibid., p. 799). Furthermore, ibid. was the first to formalize the idea of fiscal sustainability within a partial equilibrium framework in which the growth rate of the economy and the interest rate are exogenous.²¹⁸ ibid. investigates how the tax rate is affected by a perpetual public deficit that leads to a permanently increasing and unbounded stock of public debt. Taxes are exclusively levied for interest payments on issued bonds. The tax rate is applied to taxable income, defined as the sum GDP, and interest receipts on debt as the latter are also subject to taxation. The core result is that the debt-to-Gross Domestic Product (GDP) ratio converges to some constant value, which is determined by the relationship between the deficit (as a given fraction of GDP) and the growth rate of the economy. Moreover, even a perpetually growing level of debt does not lead to an ever-increasing tax rate. However, the crucial assumption is that the deficit is a constant fraction of GDP. As the growth rate of debt is finitely limited by the growth rate of GDP, the relation of debt to GDP is also finite. This relation assures the boundedness of the tax rate. The core result of ibid. can be regarded as the first formulation of a condition for fiscal sustainability based on which further approaches were subsequently formulated.

Table 6.1 provides an overview of the most common definitions of fiscal sustainability, which revolve around the idea that government solvency has to be maintained in the long run. While there is general agreement that solvency refers to the government's ability to service its debt obligations without explicitly defaulting, the government is considered to achieve long-term solvency in two ways: on the one hand, solvency is assured if fiscal policy gives rise to a bounded debt ratio in the long run, i.e. the debt ratio must not diverge. On the other hand, the government is considered solvent if it does not run a Ponzi scheme, i.e. a scheme in which it rolls over its entire debt in every period by issuing new debt. Two types of fiscal sustainability conditions can be identified in the literature based on the different interpretations of government solvency. According to BLANCHARD et al. 1990, the two types are summarized as follows:

- *i*) "[...] a sustainable fiscal policy [is] [...] a policy such that the ratio of debt to Gross National Product (GNP) eventually converges back to its initial level" (p. 11);
- *ii)* "[...] the present discounted value of the ratio of primary deficits to GNP [...] is equal to the negative of the current level of debt to GNP [...]" (p. 12).

²¹⁶ There is no general agreement on the theory governing this interaction. See e.g. BALASSONE and FRANCO 2000 as well as CHALK and HEMMING 2000 for surveys of the literature.

 $^{^{217}\,\}mathrm{See}$ e.g. Shaviro 1997 as well as Balassone and Franco 2000 for tracings of the discussion.

²¹⁸ This assumption neglects the notion that both factors depend on fiscal policy itself through e.g. crowding-out of private investment. Moreover, monetary considerations are left out as a constant price level is assumed.

These two conditions are treated as equivalent in BLANCHARD et al. 1990, although the second condition requires the present value of the debt ratio to converge to zero, which is consistent with the first condition as well as with any other finite debt ratio. Therefore, while the first condition implies the second, the second is necessary yet not sufficient for the first. It is apparent that the first condition is closely related to the core result in DOMAR 1944, i.e. the convergence of the debt ratio to some finite level. However, it is stricter as it requires the boundedness of the debt ratio to the initial level.

Table 6.1: Definitions of fiscal sustainability

The table surveys the most common definitions of fiscal sustainability, which revolve around the idea that government solvency has to be maintained in the long run.

AUTHOR(S)	DEFINITION	
Buiter 1985	Sustainable policy defined as one maintaining "a constant $[]$ ratio of public sector net worth to $[]$ output $[]$." (p. 37)	
Blanchard et al. 1990	"Sustainability is basically about good housekeeping. It is essentially about whether, based on the policy currently on the books, a government is headed towards excessive debt accumulation." (p. 8)	
EC 2012	"Ability of a government to assume the financial burden of its debt in the future. Fiscal policy is not sustainable if it implies an excessive accumulation of government debt over time and ever increasing debt service." (p. 1)	
ECB 2011c	"[] [A] government's capacity to service its debt obligations in the long term." (p. 62)	
IMF 2002	"An entity's liability position is sustainable if it satisfies the present value budget constraint without a major correction in the balance of income and expenditure given the costs of financing $[]$." (p. 5)	
OECD 2014	"Ability of a government to maintain public finances at a credible and serviceable position over the long term." (p. 50)	

Source: author's compilation.

The (broad) second condition for fiscal sustainability as formulated in BLANCHARD et al. 1990 is applied in the following to elaborate on the fiscal implication of collateral criteria. This fiscal implication is investigated in terms of necessary changes to fiscal policy to sustain government solvency. The condition evolves from simple accounting identities of the government budget and is technically derived in the following.^{219,220}

²¹⁹ The budget identity of the government has been a familiar component of dynamic macroeconomic models at least since the late-1960s; see e.g. CHRIST 1968 as well as BLINDER and SOLOW 1973. It was earlier recognized within a broader context in e.g. PATINKIN 1956, HANSEN 1958 and MUSGRAVE 1959. Later, BARRO 1979 formulated the version of the budget identity that is still used in current fiscal sustainability analyses.

²²⁰ It is necessary to define variables to subsume empirical counterparts when elaborating on accounting identities. For instance, the scope of the government sector as well as the adequate debt measure need to be defined. See e.g. BALASSONE and FRANCO 2000 for a discussion and DIPPELSMAN et al. 2012 for an overview of different scopes of the government sector as well as debt measures.

6.2 Fiscal Sustainability under Certainty

Four steps are taken in the following to elaborate an intuitive measure according to which qualitative statements are made concerning the fiscal implications of amendments to collateral criteria. First, the budget identity of the government, i.e. the identity of revenue and expenditure in a given period, is specified. Second, the period budget identity is solved recursively to derive the relation between past, present and future budgetary items. The identity is formulated in present-value terms to make expressions comparable over different periods. Third, imposing a transversality condition (TVC) (or no-Ponzi game condition (NPG), see e.g. BLANCHARD and FISCHER 1989, CHALK and HEMMING 2000 and GIAMMARIOLI et al. 2007) to the government turns the present-value budget identity into a *constraint* to the government. The intuitive measure for assessing the fiscal implications of collateral criteria is derived from the present-value budget constraint (PVBC) in the final step.

Period Budget Identity of the Government

The analysis is performed within the budgetary framework of a government in a closed economy.²²¹ The budget identity of the form still used in fiscal sustainability analysis was developed by BARRO 1979. It is extended to incorporate the transfer from the central bank to the government.²²² The government has to generate revenue to finance its expenditure G, e.g. in providing services that the market fails to efficiently provide.²²³ The government generates revenue from three sources. Foremost, revenue is raised via taxation, which gives tax revenue to the amount of T. Second, the government can borrow by issuing one-period bonds to the amount of D against paying the interest rate i > 0. Third, the government receives transfer from the central bank, denoted by Γ . The flow budget identity of the government in period t + 1 is given by

$$G_{t+1} + i_{t+1}D_t = T_{t+1} + (D_{t+1} - D_t) + \Gamma_{t+1}.$$
(6.1)

Rearranging Equation (6.1) gives the law of motion of the government debt-to-GDP ratio.²²⁴ With $Y_{t+1} = (1 + g_{t+1})Y_t$ and $P_{t+1} = (1 + \pi_{t+1})P_t$ where Y is GDP, P the price level, g the growth rate of GDP and π the inflation rate, this law of motion reads as

$$\frac{D_{t+1}}{P_{t+1}Y_{t+1}} = \frac{(1+i_{t+1})}{(1+\pi_{t+1})(1+g_{t+1})} \frac{D_t}{P_t Y_t} - \frac{(T_{t+1}-G_{t+1})}{P_{t+1}Y_{t+1}} - \frac{\Gamma_{t+1}}{P_{t+1}Y_{t+1}}.$$
(6.2)

²²¹ The assumption of a closed economy rules out concerns about complications created by external debt, i.e. debt owed to non-residents, as well as debt denominated in foreign currency.

²²² See among others UCTUM and WICKENS 2000, BURNSIDE 2005a, b and LEY 2010. This transfer mainly originates from interest income from money creation, which has been an important source of government revenue, see e.g. DORNBUSCH 1988, GIAVAZZI and GIOVANNINI 1989, CUKIERMAN et al. 1992, GROS 1993, SINN and FEIST 1997, 2000, SIMS 2004 and VERGOTE et al. 2010. The importance of central bank interest income in the context of the European sovereign debt crisis has been emphasized by BUITER and RAHBARI 2012a, SINN 2012, 2014b as well as PÂRIS and WYPLOSZ 2014.

²²³ According to the "Selection Principle" (SINN 1997), governments should provide those services that are unsuitable for private markets such as public infrastructure goods and social insurance

²²⁴ The importance of analyzing relative measures of government debt was first recognized in HANSEN 1941 and DOMAR 1944.

As the real interest rate is given by $1 + r \equiv \frac{1+i}{1+\pi}$, Equation (6.2) can be simplified to²²⁵

$$d_{t+1} = \frac{1 + r_{t+1}}{1 + g_{t+1}} d_t - pb_{t+1} - \gamma_{t+1}, \tag{6.3}$$

with pb_{t+1} being the primary balance, i.e. tax revenue minus government expenditure relative to nominal GDP, and d as well as γ the corresponding upper-case variables relative to nominal GDP. Accordingly, γ_{t+1} gives the transfer from the central bank to the government in period t + 1. Equation (6.3) always holds as it is derived from accounting identities. Per se, there is no reason for r and g to be exogenous, but since the framework is partial equilibrium, effects of fiscal policy on r and g are neglected.²²⁶

Present-Value Budget Identity of the Government

While the flow budget identity is useful for assessing the stance of fiscal policy in a given period, it fails to highlight the dynamic nature of government budgeting (see e.g. POTERBA 1997). Therefore, Equation (6.3) is solved recursively to derive the deterministic present-value budget identity. Hence, the present value of the debt ratio in period t + j is given by the following standard solution of the first-order difference equation:

$$\prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}}\right)^{-1} d_{t+j} = d_t - \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}}\right)^{-1} p b_{t+j} - \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}}\right)^{-1} \gamma_{t+j}.$$
 (6.4)

Equation (6.4) shows that d_{t+j} is determined by the following three factors: (i) initial debt (i.e. the debt ratio in period t), (ii) the present value of primary balances and (iii) the present value of transfers from the central bank.

Present-Value Budget Constraint (PVBC) of the Government

The present-value budget *constraint* of the government is subsequently derived in two steps. First, Equation (6.4) is rearranged under the assumption that time goes to infinity, which gives

$$\lim_{j \to \infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} d_{t+j} = d_t - \lim_{j \to \infty} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} p b_{t+j} - \lim_{j \to \infty} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} \gamma_{t+j}.$$
(6.5)

A TVC is imposed to the left-hand side of Equation (6.5), i.e. the present value of the debt ratio when times goes to infinity, to receive a constraint for the government budget. In principle, this

²²⁵ The real interest rate can be expressed as $r = (i - \pi)/(1 + \pi)$. It is known from the Fisher equation that this is accurately approximated by $r \approx i - \pi$ when both *i* and π are sufficiently small. To circumvent a tedious discussion about dynamic (in)efficiency, it is assumed that r > g in the remainder, i.e. the case of dynamic efficiency is considered. A dynamically inefficient economy would never have a sustainability problem as the growth rate of the economy would exceed the interest rate on government debt. See e.g. PHELPS 1961, ALLAIS 1962 and VON WEIZSAECKER 1962 on dynamic (in)efficiency. See e.g. DIAMOND 1965, SPAVENTA 1987, BLANCHARD 1990, BLANCHARD et al. 1990, BARTOLINI and COTTARELLI 1994, BOHN 1995 and BLANCHARD and WEIL 2001 for the role of dynamic (in)efficiency in the context of public debt and fiscal sustainability.

²²⁶ The limitations are put forward by e.g. MUSGRAVE and MUSGRAVE 1984, MASSON 1985 and WEIL 1987.

present value could be (i) smaller than, (ii) greater than or (iii) equal to zero. The first case would imply that when time goes to infinity, the government holds an asset as private agents are indebted with the government. This can be ruled out for a rational government. Instead, (ii) would imply that the government runs a Ponzi scheme in which it rolls over its debt in every period by issuing new debt.²²⁷ O'CONNELL and ZELDES 1988 show that this is not feasible with a finite number of agents as it would imply that some individual holds government debt at some infinite point in the future, whereas this individual would be better off in at least one period by not holding such debt. Therefore, holding debt that is continuously rolled over is strictly dominated by holding no debt at all, which prohibits the government from running a Ponzi scheme. The imposition of the TVC requires the left-hand side of Equation (6.5) to be equal to zero, i.e.

$$\lim_{j \to \infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} d_{t+j} = 0.$$
(6.6)

Requiring that the TVC holds is generally regarded as the conventional condition for the government remaining solvent in infinite time. Imposing the TVC transforms the present-value budget *identity* to the deterministic present-value budget *constraint*, given by

$$d_t = \lim_{j \to \infty} \sum_{j=0}^{\infty} \prod_{k=1}^j \left(\frac{1+r_{t+k}}{1+g_{t+k}}\right)^{-1} p b_{t+j} + \lim_{j \to \infty} \sum_{j=0}^{\infty} \prod_{k=1}^j \left(\frac{1+r_{t+k}}{1+g_{t+k}}\right)^{-1} \gamma_{t+j}.$$
 (6.7)

Equation (6.7) stipulates the initial debt ratio to be offset by the present value of all future revenue of the government.²²⁸ Accordingly, the government has to finance its initial debt from (i) primary balance and (ii) transfers from the central bank. Running temporary primary deficits would be consistent with Equation (6.7) as long as they are compensated for by sufficiently high surpluses in other periods.²²⁹

The fiscal theory of the price level (FTPL) interprets Equation (6.7) from a different perspective:²³⁰ if initial debt and the present value of future surpluses do not match in real terms for a given price level, then the price level adjusts to restore equilibrium as government spending induces a change in the price level through its effect on aggregate demand. Hence, the FTPL turns a constraint on the government budget into a macroeconomic equilibrium condition. However, this theory is subject to persistent critique (e.g. BUITER 2002, 2005, and BOHN 2008). It is claimed that the PVBC should always be satisfied (and not only in equilibrium) and that the FTPL implicitly allows the government to pursue time-inconsistent policy.

²²⁷ Ponzi scheme refers to a scheme in which individuals pay out funds to some parties by borrowing these funds from others. It dates back to Charles Ponzi, who ran such a scheme in the early 20th century (e.g. RUSSEL 1973).

²²⁸ A weaker condition of fiscal sustainability demands stationarity of the debt ratio, cf. e.g. BARTOLINI and COTTARELLI 1994 as well as HABER and NECK 2008.

²²⁹ Abstracting from central bank transfers, MCCALLUM 1984 points out that permanent primary deficits would be inconsistent with the PVBC if initial debt is positive, as this would imply a Ponzi scheme. Likewise, the government could not run a small primary deficit followed by a primary balance equal to zero.

²³⁰ See e.g. LEEPER 1991, SIMS 1994, WOODFORD 1995, CANZONERI et al. 2001 and COCHRANE 2001 for the FTPL.

Reinterpretation of Equation (6.7) within the FTPL is closely connected to the general literature on the relationship between fiscal and monetary policy.²³¹ The literature recognizes that both the fiscal and the monetary authority can contribute to satisfying the PVBC. The fiscal authority can adjust pb and the monetary authority can alter γ . A situation in which fiscal policy is expected to adjust its policy stance to ensure that the PVBC is balanced while the monetary authority sets its policy freely (i.e. γ in this context) is referred to as monetary dominance or a *Ricardian* regime. By contrast, a non-Ricardian regime with fiscal dominance assumes the monetary authority to ensure that the PVBC is met by adapting the transfer to the government. Developments in the Eurozone since the onset of the sovereign debt crisis suggest that such a clear-cut distinction between the two regimes may not be observable in reality.²³² Developments indicate a (at least) partial shift from monetary dominance to a (partial non-Ricardian) regime of quasi-fiscal dominance as the Eurosystem became a major player in managing the debt crisis, adapting its monetary policy to assume quasi-fiscal activities and support government budgets under distress.²³³

Evaluation of Fiscal Sustainability

The PVBC given by Equation (6.7) serves as the benchmark against which fiscal sustainability is qualitatively evaluated in the following.²³⁴ Fiscal sustainability is evaluated for an economy in a steady state, i.e. (i) the real interest rate and the growth rate of the economy are constant $(r_{t+k} = r \text{ and } g_{t+k} = g)$, (ii) the primary balance is time-invariant $(pb_{t+j} = pb)$ and (iii) the transfer from the central bank to the government is constant, i.e. $\gamma_{t+j} = \gamma$. Hence, Equation (6.7) can be simplified to

$$d \le \sum_{j=1}^{\infty} \left(\frac{1+r}{1+g}\right)^{-j} \left(pb+\gamma\right). \tag{6.8}$$

Fiscal policy is considered sustainable if the sum of present values of primary balance and transfer from the central bank are equal to or greater than the debt ratio. Under the assumption of dynamic inefficiency (r > g), steady-state dynamics of the debt ratio are implied by the sustainability condition²³⁵

$$d \le \frac{1+g}{r-g} \left(pb + \gamma \right). \tag{6.9}$$

²³¹ See among others AIYAGARI and GERTLER 1985, LEEPER 1991, SARGENT and WALLACE 1981, KING and PLOSSER 1985, COCHRANE 2001, AFONSO 2008, IZE and OULIDI 2009 as well as WALSH 2010.

 $^{^{232}}$ See e.g. Afonso 2008 for an empirical investigation and a comprehensive overview of previous studies.

²³³ Therefore, weak fiscal dominance is considered in Chapter 7 when the preferences of the government over the level of collateral criteria are examined. If the monetary authority takes into account the government's preferences, the government still seeks to adjust primary balance to meet the PVBC but is supported by the monetary authority.

²³⁴ The PVBC is also central to the quantitative assessment of fiscal sustainability within two extensive strands of literature. On the one hand, it has served as the basis of various indicators of fiscal sustainability. See among many others BLANCHARD 1990, BLANCHARD et al. 1990, LANGENUS 2006, GIAMMARIOLI et al. 2007 and BALASSONE et al. 2009. On the other hand, empirical tests have been elaborated based on the PVBC, see e.g. HAMILTON and FLAVIN 1986, TREHAN and WALSH 1991, WILCOX 1989, HAKKIO and RUSH 1991, BOHN 1998, PAPADOPOULOS and SIDIROPOULOS 1999, AFONSO 2005, BURNSIDE 2005a, BOHN 2008, MENDOZA and OSTRY 2008 as well as POTRAFKE and REISCHMANN 2015. For discussions of both strands, see e.g. BALASSONE and FRANCO 2000, CHALK and HEMMING 2000 as well as NECK and STURM 2008. However, no quantitative assessment of fiscal sustainability will be made here.

²³⁵ Outside steady state, Equation (6.9) could still be equivalent to Equation (6.7) if steady-state values are interpreted as averages (BURNSIDE 2005a). Moreover, $\sum_{j=1}^{\infty} ((1+r)/(1+g))^{-j}$ only converges for r > g but diverges otherwise.

The interpretation of $pb + \gamma$ as the government's *capacity to pay* facilitates the analysis of debt dynamics inherent in Condition (6.9) (cf. BARTOLINI and DIXIT 1991, MILLER and ZHANG 2000) that are stylized in Figure 6.1. The sustainability frontier determined by Condition (6.9) and reflected by the solid line is decisive for evaluating fiscal sustainability. Combinations of $pb + \gamma$ and d to the right of the solid line are considered sustainable as the debt ratio converges (white area) while all combinations to the left are deemed unsustainable as the debt ratio diverges (gray area). The gray area is inter alia determined by the growth rate-interest rate differential r - g. The smaller r - g, the steeper the sustainability frontier and hence the smaller the gray area, i.e. less combinations of $pb + \gamma$ and d are deemed unsustainable. By contrast, the sustainability frontier is flat for large r - g, such that more combinations of $pb + \gamma$ and d are considered unsustainable.

Figure 6.1: Fiscal sustainability and debt dynamics under certainty

The figure reflects Condition (6.9) and illustrates inherent debt dynamics. The sustainability frontier given by the solid line is decisive for evaluating fiscal sustainability. Combinations of $pb + \gamma$ and d to the right are considered sustainable as the debt ratio converges (white area), while all combinations to the left are deemed unsustainable as the debt ratio diverges (gray area).



Moreover, Condition (6.9) can be used to evaluate fiscal sustainability in two ways (BURNSIDE 2005a). First, the steady-state level of d can be determined based on assumptions for r, g, pb and γ . Government finances are deemed unsustainable if the actual d exceeds this steady-state level. Second, under the assumption that a rationale government chooses the lowest primary balance deemed sustainable, Condition (6.9) holds with equality and can be solved for the primary balance such that

$$pb^* = \frac{r-g}{1+g}d - \gamma.$$
 (6.10)

Equation (6.10) serves as a straightforward indicator of fiscal sustainability.²³⁶ It determines the primary balance that is at least necessary to ensure fiscal sustainability in steady state, i.e.

²³⁶ Fiscal sustainability indicators of the EC are akin to pb^* . The indicators measure the gap between the debtstabilizing and the actual primary balance for different time horizons. See BALASSONE et al. 2009 and EC 2012. See BARTOLINI and COTTARELLI 1994, PAPADOPOULOS and SIDIROPOULOS 1999, BARNHILL and KOPITS 2005, BURNSIDE 2005a as well as BALASSONE et al. 2009 for further applications.

the primary balance necessary to stabilize the debt ratio. It depends on the interest rate-growth rate differential, the debt ratio and the transfer from the central bank. The larger the interest rate-growth rate differential and the initial debt ratio or the lower the transfer from the central bank, the larger the required debt-stabilizing primary balance.

The indicator provided by Equation (6.10) is used in the following to analyze the fiscal implications of central bank collateral criteria in terms of the *qualitative* development of pb^* . No statement on the *quantitative* development of pb^* is made. This restriction is due to acknowledging the simple form of the indicator and the assumptions necessary to derive it. Furthermore, insights are subject to the flaws of any fiscal sustainability indicator (see GIAMMARIOLI et al. 2007), e.g. the necessity to quantify factors over an infinite time horizon, usually based on exogenous assumptions on macroeconomic variables.

Another shortcoming of sustainability indicators developed in a deterministic environment is that they fail to recognize potential adverse effects to fiscal sustainability under uncertainty. Policies that appear sustainable in a certain world may no longer be so under uncertainty (BOHN 1991, 1995). However, the effects of uncertainty are particularly important for fiscal sustainability owing to the infinite time horizon inherent to the analysis.

6.3 The Concept of Fiscal Risk

Thus far, the government has been assumed to have deterministic income and expenditure flows such that the fiscal outcome is fully predictable and expectations on this outcome are always met. This may no longer hold when uncertainty is introduced, giving rise to fiscal risk. CEBOTARI et al. 2009 define fiscal risk as the possibility of fiscal outcomes to deviate from what was expected. Deviations can be the result of variations in both income and expenditure flows. For the purpose of the analysis, the focus is placed upon variations in expenditure flows.²³⁷ This section addresses the concept of fiscal risk, which evolves around the *government liability matrix* dating back to BRIXI 1998.²³⁸ The concept is presented and discussed in general and with respect to the qualitative analysis of the fiscal implications of the collateral framework.²³⁹

Government Liability Matrix

The government liability matrix is depicted in Table 6.2 and recognizes the uncertainty of government liabilities with respect to amount and maturity. Uncertainty with respect to amount is implied by the differentiation between "explicit" and "implicit" liabilities, while uncertainty with respect to maturity is considered by distinguishing between "non-contingent" (or direct)

²³⁷ See BRIXI and MODY 2002 for a comprehensive analysis based on an extended asset and liability management framework. For the extended government balance sheet upon which this framework bases, see e.g. CASSARD and FOLKERTS-LANDAU 1997, CURRIE and VELANDIA 2002 as well as WORLD BANK and IMF 2014.

²³⁸ The matrix is a common approach to structure liabilities of the government, see e.g. BRIXI and MODY 2002, BRIXI 2005, GIAMMARIOLI et al. 2007, DAS et al. 2012 as well as IMF 2013. See IMF 2005 and DIPPELSMAN et al. 2012 for another systematization with respect to the degree of certainty of government liabilities.

²³⁹ The discussion is limited to the extent that it does not present a framework for the disclosure as well as the statistical and accounting treatment of uncertain government income and expenditure flows. See IMF 2014b for the statistical treatment and IFAC 2014 for accounting. Furthermore, the question of fiscal risk management is only addressed conceptually, see BRIXI and MODY 2002, CEBOTARI et al. 2009 as well as IMF 2013 for in-depth investigations.

and "contingent" liabilities. Differentiation between explicit and implicit liabilities bases on the origin of liabilities. Explicit liabilities are contractually agreed such that the amount can be determined with certainty, corresponding to the face value stipulated by contract. By contrast, implicit liabilities are not contractually agreed but originate from expectations by e.g. the public or interest groups. Hence, while the government is not legally obliged to act upon such liabilities, it is expected to do so. As expectations are usually based on past practices, implicit liabilities are particularly important in the long run and must not be neglected in fiscal sustainability analysis. Differentiation between non-contingent and contingent liabilities bases on maturity. Non-contingent liabilities are liabilities that mature in any case. However, liabilities can also mature depending on the occurrence of a specific event. These liabilities are called contingent liabilities and they are usually neglected in conventional government budget analysis and accounting.

The combination of the two dimensions of uncertainty gives the four types of liabilities stated in Table 6.2, i.e. (i) non-contingent explicit liabilities, (ii) non-contingent implicit liabilities, (iii) contingent explicit liabilities and (iv) contingent implicit liabilities. Non-contingent explicit liabilities are liabilities that are legally enforceable by the obligee as they are stipulated by contract or law and that mature in any case, i.e. maturity does not depend on particular events. Government debt, i.e. debt instruments issued by the government, is the conventional noncontingent explicit liability.²⁴⁰ In addition, this type of government liability comprises legally enforceable government expenditure commitments established e.g. in the annual government budget plan. In many countries, such commitments also include unfunded and clearly defined pension payments. By contrast, non-contingent implicit liabilities mature in any case but are not legally enforceable by the obligee as they are not stipulated by law or contract. Despite not being legally binding, such liabilities are expected to be settled in certain cases, such as future social welfare payments. Social welfare payments stemming from a pay-as-you-go pension scheme are the prime example of a non-contingent implicit liability. Therefore, they are often considered in determining the "implicit debt" of the government (see SINN 2000 as well as MOOG and RAFFELHÜSCHEN 2012; cf. AUERBACH et al. 1999 as well as KOTLIKOFF and RAFFELHÜSCHEN 1999 for estimates of implicit debt). Another type is payments related to recurrent operations such as the refurbishment of the economy's capital stock.

By contrast, contingent liabilities only mature in case a specific event occurs. Therefore, they are often considered as fiscal risk, treated as off-balance-sheet items and undiscounted in conventional budgeting.²⁴¹ In case of maturity, contingent liabilities may pose an adverse shock to the government budget. The clear-cut example of *contingent explicit liabilities* is government individual guarantees on debt issued by public and private entities. Moreover, explicit contingent liabilities can result from government umbrella guarantees and insurance schemes on e.g.

²⁴⁰ A different view is presented in GROSSMAN and VAN HUYCK 1988, where government debt is considered uncertain as the government may have to reduce or suspend debt service in bad states of the world, i.e. if certain contingencies arise. The government can "excusably" default, i.e. default that is justifiable and accepted by creditors owing to the contingency, without damaging capital market access. By contrast, it would be punished if default was unjustifiable and inexcusable.

²⁴¹ Even if they are taken into account in budgeting, contingent liabilities are complex arrangements that are difficult to quantify. Therefore, no single measurement approach can fit all situations and comprehensive approaches are still evolving, see IMF 2013.

Table 6.2: Government liability matrix

The table implies the four types of government liabilities that originate from the combination of the two dimensions of uncertainty, i.e. (i) non-contingent explicit liabilities, (ii) non-contingent implicit liabilities, (iii) contingent explicit liabilities and (iv) contingent implicit liabilities.

Type of Liability	Non-Contingent not depending on realization of particular event	Contingent depending on realization of particular event
EXPLICIT legally binding	 non-contingent but legally binding liabilities like debt instruments issued^a expenditure commitments established in the annual budget plan 	 both contingent and legally binding liabilities from government individual guarantees (on debt issued by public or private entities)^b government umbrella guarantees (e.g. on household mort-gages) government insurance schemes (e.g. on bank bonds, bank deposits, returns from private pension funds)^b
IMPLICIT not legally binding but originating from public and/or interest group expectations and pressure	neither contingent nor legally binding liabilities like • future social welfare payments • future expenditures related to recurrent opera- tions	 contingent but not legally binding liabilities from bailout of defaulting public or private sector entities^b central bank bailout^c disaster relief environmental damage military financing

^a addressed in Chapters 7 and 8.
^b addressed in Chapter 8.
^c addressed in Chapter 7.
Source: author's compilation; adapted from BRIXI 1998 and DAS et al. 2012. See IMF 2013 for a comprehensive version of the matrix.

bank deposits and returns of private pension funds. *Contingent implicit liabilities* are usually triggered by the default of a public or private entity that the government has to bail out. Furthermore, the government may have to assume financial responsibility for the central bank when the central bank balance sheet is under distress. It is also expected to bear the cost of disaster relief and environmental damage, as well as financially backing the military. Contingent explicit and implicit liabilities as fiscal risk are further examined in the following given their importance to the analysis.

Contingent Explicit Liabilities Contingent explicit liabilities are defined as legal or contractual obligations that give rise to payments if one or more stipulated conditions arise (UN 2009). Owing to the legal or contractual a priori agreement, only the occurrence of the contingent explicit liability is uncertain, while the value at risk is known to the government. Contingent explicit liabilities have repeatedly proven a severe threat to the government budget in the past. Contingent explicit liabilities often take the form of guarantees to public and private sector entities. Different types of government guarantees exist.²⁴² While there are government individual guarantees granted to debt instruments of beneficiaries on an individual basis, the government also gives umbrella guarantees to specific types of debt instruments and/or specific issuer groups such as household mortgages and student loans. Moreover, a very broad form of guarantee is issued by the government as insurance schemes. Such schemes can take the form of insurance of bank deposits, insurance against losses from e.g. crop failures and natural disasters, insurance of a minimum return from private pension funds or war-risk insurance. Typically, government insurance schemes cover risk deemed uninsurable by the private insurance industry, specifically low-probability high-impact events (see KUNREUTHER 1997, CUTLER and ZECKHAUSER 1999 as well as EBERL and JUS 2012). In particular, government guarantees and insurance issued to the financial sector has proved a substantial source of fiscal risk in terms of contingent explicit liabilities, as the recent financial crisis revealed.²⁴³ For the recent financial crisis (2007 to 2009), LAEVEN and VALENCIA 2014 estimate the median direct fiscal costs, i.e. fiscal outlays committed to the financial sector that capture direct fiscal implications of public intervention, in various advanced and emerging economies of 4.9% of GDP and the median increase in the debt ratio at 23.9%. Therefore, direct fiscal costs were lower (compared to 10% of GDP for crises between 1970 and 2006), although the increase in the debt ratio was larger (relative to 16.3%). However, the fiscal costs of financial crises were not exclusively due to explicit government action but also the result of government contingent implicit liabilities.

Contingent Implicit Liabilities Contingent implicit liabilities are obligations that do not arise from a legal or contractual source but are recognized after a condition or event is realized. These liabilities often exceed explicit ones if due as the government and the beneficiary did not agree on a specific value at risk. The guarantee has no face value and its value depends on

²⁴² See IMF 2005 for an extensive study of government guarantees as fiscal risk. See IMF 2013 for a condensed overview of the different types of government guarantees.

²⁴³ See e.g. LAEVEN and VALENCIA 2008, 2012, 2014, CLAESSENS et al. 2014 as well as REINHART and ROGOFF 2014 for analyses of the impact of the recent yet also historic systemic banking crises inter alia on the government budget.

past practice and expectations. Therefore, both occurrence and value at risk are uncertain.²⁴⁴ Most importantly, contingent implicit liabilities emerge from the bailout of defaulting public and private sector entities. This is especially the case for the financial sector, for which bailing out illiquid/insolvent banks contributed to the fiscal costs of crisis and the rise in debt ratios (see above). Bailout beyond contractual obligations is extensively discussed with respect to financial institutions that are considered too big to fail (TBTF).²⁴⁵ Moreover, the government does not have the legal obligation to financially support its central bank in times of financial distress. However, if the central bank might struggle to fulfill its monetary policy mandate, the government may provide financial support.²⁴⁶ Akin guarantees that may result in a bailout of companies are implicitly given to certain industries. The clear-cut example is the nuclear industry, for which governments usually assume financial responsibility beyond contractual or legal obligations in case of disaster.²⁴⁷ Accordingly, governments are expected to assume financial responsibility for disaster relief and environment damage.

Liabilities are not static within Table 6.2 as movements within the matrix are possible. For instance, the government could reduce the tax contribution to the social security system, which ceteris paribus would transform explicit into non-contingent implicit liabilities. Likewise, the government often has to issue new debt to finance the social security system, which would turn non-contingent implicit liabilities explicit. A systemic banking crisis can result in implicit guarantees to the financial sector being made explicit as well as financial institutions being bailed out by the government. If the government bore the cost by issuing new debt, the former contingent explicit and implicit liabilities would lead to non-contingent explicit liabilities. Dynamics within the matrix triggered by government intervention to bail out the central bank and mitigate implicit guarantees to the financial sector by issuing new debt are addressed in Chapters 7 and 8, respectively.

6.4 Fiscal Sustainability under Uncertainty

The basic framework of fiscal sustainability analysis is extended to incorporate fiscal risk in terms of contingent explicit and implicit liabilities. The framework is modified in two ways when uncertainty is taken into account.²⁴⁸ First, future values of variables may vary across different states of the world such that values are expressed in terms of expected values. Second, contingent explicit and implicit liabilities are factored in.

The literature proposes different approaches to investigate fiscal sustainability under uncertainty, differing with respect to the task assigned to fiscal sustainability analysis.²⁴⁹ MENDOZA and

²⁴⁴ Behavioral implications of implicit guarantees such as moral hazard from de facto limited liability may increase the value at risk, as addressed in e.g. SINN 1980, 1983 and SHAVELL 1986.

²⁴⁵ For discussions and estimates for different countries and time periods, see BAKER and MCARTHUR 2009, KELLY et al. 2012, UEDA and WEDER DI MAURO 2012, TSESMELIDAKIS and MERTON 2012, ARATEN and TURNER 2013, ACHARYA et al. 2014a as well as IMF 2014.

²⁴⁶ The contingent implicit liability from central bank bailout is addressed in depth in Chapter 7.

 ²⁴⁷ See EBERL and JUS 2012 for the implicit guarantee to the nuclear industry and a proposal for optimal regulation.
 ²⁴⁸ BOHN 1995 proposes a third way, claiming that the marginal rate of substitution between different points in time has to be used for discounting under uncertainty, see also CHALK and HEMMING 2000. However, this issue is predominantly relevant for empirical investigations and is neglected here for the sake of technical convenience.

 $^{^{249}\,\}mathrm{See}\,$ Burnside 2004 for a discussion and assessment of various approaches.

OVIEDO 2006 consider uncertainty in revenue flows of governments in emerging market economies and investigate how this affects the government's ability to borrow. XU and GHEZZI 2003 model flows in the government budget identity as stochastic processes to estimate default probabilities. Approaches by the IMF such as IMF 2003 and 2013 allow for uncertainty to generally enhance the assessment of prior fiscal policy records and discussion of future policy choices. Finally, BARNHILL and KOPITS 2005 assess fiscal risk from contingent liabilities by measuring government net worth from a stochastic process. Based on the valuation of contingent liabilities and their effect on the government budget, the study evaluates the probability of the net worth becoming negative.

A more general and stylized approach to fiscal sustainability analysis under uncertainty is developed in the following. The deterministic PVBC given by Equation (6.7) is formulated in expected value terms and contingent liabilities are taken into account to derive the stochastic PVBC. Based on this stochastic formulation, the simple indicator for assessing fiscal sustainability is extended to reflect uncertainty.

Present-Value Budget Constraint Revisited

Based on Equation (6.3), which gives the government budget identity under certainty, the budget identity under uncertainty is given by taking expectations as from time t + 1 such that

$$\mathbb{E}_t[d_{t+1}] = \frac{1 + \mathbb{E}_t[r_{t+1}]}{1 + \mathbb{E}_t[g_{t+1}]} d_t - \mathbb{E}_t[pb_{t+1}] - \mathbb{E}_t[\gamma_{t+1}],$$
(6.11)

with $\mathbb{E}_t[\cdot]$ denoting the expectations operator conditional on information available at time t. Solving Equation (6.11) recursively and assuming infinite time gives

$$\lim_{j \to \infty} \mathbb{E}_{t} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} d_{t+j} = d_{t} - \lim_{j \to \infty} \mathbb{E}_{t} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} p b_{t+j} - \lim_{j \to \infty} \mathbb{E}_{t} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} \gamma_{t+j},$$
(6.12)

which is the analogue of Equation (6.5) under uncertainty, i.e. the present-value budget identity with uncertainty about future realizations of debt, primary balances and transfers from the central bank.

Imposing the TVC, i.e. assuming that the left-hand side of Equation (6.12) is equal to zero,²⁵⁰ gives the stochastic PVBC as

$$d_{t} = \lim_{j \to \infty} \mathbb{E}_{t} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} p b_{t+j} + \lim_{j \to \infty} \mathbb{E}_{t} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} \gamma_{t+j}.$$
 (6.13)

²⁵⁰ See BOHN 1991, BARTOLINI and COTTARELLI 1994, BOHN 1995 as well as BLANCHARD and WEIL 2001 for the TVC/NPG under uncertainty.

Beyond the formulation of future values in expected value terms, uncertainty implies the existence of contingent liabilities, which are introduced into the technical framework in the following.²⁵¹

Contingent Liabilities in the Present-Value Budget Constraint

The government budget may be subject to various contingent explicit and implicit liabilities that were addressed in the previous section. To keep the technical analysis tractable, a "representative" contingent explicit and implicit liability is modeled, which are assumed to enter the analysis in terms of their expected value.²⁵² In case of budgeting contingent liabilities according to the expected value, two properties of the contingent liability are relevant, namely (i) its value and (ii) its probability of occurrence.

The value of the liability is denoted by c_{t+j}^e in case of the contingent explicit liability and c_{t+j}^i for the contingent implicit liability. The explicit liability is contractually agreed and hence its value is assumed to be known. For the implicit liability, no face value is available such that the value is itself subject to uncertainty.²⁵³

The probability of occurrence is characterized for both the explicit (e) and implicit (i) contingent liability by p^m with $m \in \{e, i\}$. It can generally be distributed according to any discrete probability distribution.²⁵⁴ The literature often assumes that the probability of occurrence of contingent liabilities is distributed according to a Poisson process of parameter λ (see HAIGHT 1967, PANJER and WILLMOT 1992, COHEN 2002, KLUGMAN et al. 2012). For instance, the probability of occurrence of e.g. a government guarantee or a low-probability high-impact event is assumed to follow a Poisson process in insurance economics (see PANJER and WILLMOT 1992, COHEN 2002, NGUYEN 2007, CUMMINS and MAHUL 2009). The Poisson distribution is a discrete distribution with the probability of occurrence being given by the density function $p_{\lambda}^{m}(n) = \frac{\lambda^{n}}{n!} \cdot e^{-\lambda}$ for $n \in \{1, 2, ...\}$ where λ is the expected frequency of the event triggering the contingent liability and n the frequency of triggering events within a given time horizon. Figure 6.2 gives the probability distribution according to the Poisson process for different values of λ . The frequency n is given on the horizontal axis and the corresponding probability of occurrence on the vertical axis. The right panel indicates that for sufficiently large values of λ , the Poisson distribution is approximated by a normal distribution. By contrast, the other panels suggest that small values of λ generate probability distributions that are suitable for describing the probability of occurrence of less frequent contingent explicit and implicit liabilities.

²⁵¹ Moreover, uncertainty would give rise to implicit debt, which is uncertain in terms of the value of the liability. Nonetheless, implicit debt is neglected in the following for convenience.

²⁵² See COHEN 2002 for an analysis arguing that expected costs are not the right measure for budgeting contingent liabilities if the time horizon over which the liability is expected to mature is very short or discounting is very low.

²⁵³ The analysis could be enriched by taking into account this additional layer of uncertainty. The focus is placed upon uncertainty with respect to maturity in the following for the sake of clarity.

²⁵⁴ See e.g. Ross 2010 and KLUGMAN et al. 2012 for overviews of discrete probability distributions.

Figure 6.2: Probability distributions according to Poisson process

The figure depicts the probability distribution according to the Poisson process for different values of λ . Although the Poisson distribution is approximated by a normal distribution for sufficiently large values of λ (right panel), small values of λ generate distributions that are suitable for describing the probability of occurrence of less frequent contingent government liabilities.



Hence, the probability of occurrence of both types of contingent liabilities is assumed to be described by a Poisson process in the following, i.e. $p_{\lambda}^m \sim Po(\lambda)$. Therefore, the expected value of the explicit and implicit liability, respectively, in period t + j is given by

$$\mathbb{E}[c_{t+j}^m] = \begin{cases} c_{t+j}^m & \text{with } p_{t+j,\lambda}^m; \\ 0 & \text{with } 1 - p_{t+j,\lambda}^m, \end{cases}$$
(6.14)

such that the expected value of the liability is either c_{t+j}^e and c_{t+j}^i , respectively, with probability $p_{t+j,\lambda}^e$ and $p_{t+j,\lambda}^i$, respectively, or zero otherwise.

When taking both representative contingent liabilities into account, the government budget identity in period t + 1 reads as

$$\mathbb{E}_{t}[d_{t+1}] = \frac{1 + \mathbb{E}_{t}[r_{t+1}]}{1 + \mathbb{E}_{t}[g_{t+1}]} d_{t} + \underbrace{p_{t+1,\lambda}^{e} c_{t+1}^{e}}_{\mathbb{E}_{t}[c_{t+1}^{e}]} + \underbrace{p_{t+1,\lambda}^{i} c_{t+1}^{i}}_{\mathbb{E}_{t}[c_{t+1}^{e}]} - \mathbb{E}_{t}[pb_{t+1}] - \mathbb{E}_{t}[\gamma_{t+1}].$$
(6.15)

Formulating the present-value budget identity analogous to above and imposing the TVC gives the *extended* stochastic PVBC, which represents the comprehensive constraint on the government budget as it incorporates (both explicit and implicit) contingent liabilities together with non-contingent explicit liabilities. It is given by

$$d_{t} = \lim_{j \to \infty} \mathbb{E}_{t} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} pb_{t+j} + \lim_{j \to \infty} \mathbb{E}_{t} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} \gamma_{t+j} - \lim_{j \to \infty} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} p_{t+j,\lambda}^{e} c_{t+j}^{e} - \lim_{j \to \infty} \sum_{j=0}^{\infty} \prod_{k=1}^{j} \left(\frac{1+r_{t+k}}{1+g_{t+k}} \right)^{-1} p_{t+j,\lambda}^{i} c_{t+j}^{i}.$$

$$(6.16)$$

Evaluation of Fiscal Sustainability under Uncertainty

Equation (6.16) as the extended stochastic PVBC serves as the benchmark for evaluating fiscal sustainability under uncertainty. Analogous to the approach under certainty, fiscal sustainability is evaluated for an economy in a (stochastic) steady state. This simplifies Equation (6.16) and fiscal policy is considered sustainable if $d \leq \frac{1+g}{r-g} (\mathbb{E}[pb] + \mathbb{E}[\gamma] - p_{\lambda}^e c^e - p_{\lambda}^i c^i)$. It is assumed in the

Figure 6.3: Fiscal sustainability and debt dynamics under uncertainty

The figure is the analogue to Figure 6.1 and shows that the fiscal sustainability frontier is shifted to the right under uncertainty, meaning that uncertainty restricts fiscal policy. Fiscal policy stances that are deemed unsustainable under uncertainty yet sustainable under certainty are indicated by the dark-gray area. Stances deemed unsustainable under both certainty and uncertainty are reflected in light-gray.



remainder of this chapter that the primary balance and the transfer from the central bank are at the government's discretion such that there is no uncertainty related to these two budgetary items.²⁵⁵ Technically, this implies that $\mathbb{E}[pb] = pb$ and $\mathbb{E}[\gamma] = \gamma$ and fiscal policy is deemed sustainable if

$$d \le \frac{1+g}{r-g} \left(pb + \gamma - p_{\lambda}^{e} c^{e} - p_{\lambda}^{i} c^{i} \right).$$
(6.17)

Interpreting again the term in brackets in Condition (6.17) as the government's capacity to pay (see above) shows that this capacity is constrained under uncertainty compared to the situation under certainty owing to the potential occurrence of contingent liabilities. This effect on the capacity to pay and debt dynamics is depicted in Figure 6.3, which is the analogue to Figure 6.1 under uncertainty. The recognition of contingent liabilities shifts the fiscal sustainability frontier determined by Condition (6.17) to the right. The fiscal sustainability frontier under certainty is represented by the dashed line and that under uncertainty by the solid line. Owing to this shift, more fiscal policy stances are deemed unsustainable. Fiscal policy stances that are deemed unsustainable under uncertainty yet sustainable under certainty are indicated by the dark-gray area. Stances deemed unsustainable under both certainty and uncertainty are reflected in lightgray. Therefore, the consideration of uncertainty in fiscal sustainability analysis restricts fiscal policy if sustainable fiscal policy is required.

 $^{^{255}\,\}mathrm{This}$ assumption is relaxed in Chapter 7.

Under the assumption of a rational government, i.e. choosing the smallest sustainable primary balance such that Condition (6.17) holds with equality, the sustainable primary balance pb^{**} under uncertainty is given by

$$pb^{**} = \frac{r-g}{1+g}d - \gamma + p_{\lambda}^{e}c^{e} + p_{\lambda}^{i}c^{i} = pb^{*} + p_{\lambda}^{e}c^{e} + p_{\lambda}^{i}c^{i}.$$
(6.18)

It is straightforward that $pb^{**} > pb^*$ for any positive probability that contingent liabilities mature such that the necessary primary balance ensuring sustainability is strictly larger under uncertainty than under certainty.

6.5 Overview of the Fiscal Implications of Eurosystem Collateral Criteria

The remainder of this chapter provides an overview of fiscal implications of central bank collateral criteria. These implications are important as amendments to collateral criteria give rise to new government liabilities and induce dynamics in the government liability matrix (Table 6.2), i.e. movements and transitions of liabilities. Three fiscal implications of collateral criteria are briefly reviewed hereafter and thoroughly investigated in Chapters 7 and 8: first, the relationship between collateral criteria and government debt as a non-contingent explicit liability is considered; second, the importance of collateral criteria for potential central bank recapitalization (contingent implicit liability) is examined; and third, it is summarized how collateral criteria shape government incentives to give explicit guarantees (contingent explicit liability).

6.5.1 Collateral Criteria and Government Debt

Collateral criteria bear fiscal implications as they affect the issuance of government bonds, i.e. the assumption of non-contingent explicit liabilities. Usually, government bonds have been considered risk-free and hence as a prototype of collateral as default and liquidity risk was deemed absent. Credit ratings of several governments depreciated during the sovereign debt crisis, even below the minimum credit rating of the Eurosystem in some cases. Consequently, the Eurosystem suspended the minimum credit rating for several countries (see Section 3.2.2) to deem eligible bonds issued by governments of these countries. Relaxed collateral criteria facilitate government indebtedness by lowering the cost of bond issuance in three ways.²⁵⁶

First, extended collateral eligibility gives (government) bond holders additional potential utilization (despite holding the bond to maturity) such that eligible assets are usually traded at an eligibility premium (BINDSEIL and PAPADIA 2006, RULE 2012, BINDSEIL 2014 and BIS 2015). Therefore, eligibility implies higher bond prices and lower interest rates, which facilitates government bonds issuance.

Second, when government bonds are pledged as collateral with the Eurosystem, the issuing government de facto pays a fraction of interest to itself in case of counterparty default owing to

²⁵⁶ Moreover, this gives rise to the crowding-out of private investment as banks would funnel larger amounts of liquidity towards the government rather than profitable investments outside the government sector.

the distribution of interest income in the Eurosystem. This lowers the expected cost of bond issuance and facilitates government indebtedness.

Third, the relaxation of collateral criteria enhances the liquidity of counterparties that can be used to acquire additional government bonds, which can subsequently again be pledged as collateral.²⁵⁷ This process is self-sustaining and fosters government indebtedness as long as government bonds are pledgeable with the central bank. In this case, the government would benefit from an alleviating effect on the interest rate that it would have to pay on its debt as demand for government bonds would increase. Banks tend to disproportionately expose themselves to government bonds irrespective of central bank collateral criteria. This exposure largely comprises government bonds of the banks' home country, a phenomenon, referred to as home bias in government debt.²⁵⁸ This home bias is reported to have intensified in the Eurozone since the financial crisis. ARSLANALP and TSUDA 2014 document that the fraction of the stock of Eurozone government bonds that banks held from their own government increased on average from 57% in 2007 to 69% in 2011. In Greece, Italy and Spain, it was even close to 100% (see also COEURDACIER and REY 2013).²⁵⁹ Anecdotal evidence for the period of the three-year LTROs launched by the Eurosystem reveals that Spanish banks used additional liquidity to acquire additional government bonds to the amount of EUR 85 bn during this period (SINN 2015a). However, this behavior appears to contradict standard finance theory, which would suggest that banks holding a large stock of domestic government bonds strive to invest in assets with uncorrelated risk to diversify their portfolio. Observable intensification in exposure to domestic government debt can be explained by the following reasons.

BATTISTINI et al. 2013, 2014, ACHARYA and STEFFEN 2015 as well as DRECHSLER et al. 2015 address risk shifting to explain the increase in government bond holdings. Risk shifting, i.e. the externalization of risk, in the banking sector can be referred to three reasons. First, limited liability may induce undercapitalized banks to invest in risky assets. If liability is limited and the downside risk of failing to redeem liabilities is low, banks are incentivized to take excessive risk as they win if the gamble succeeds while their creditors lose otherwise. SINN 1980, 1983 refers to this phenomenon as the BLOOS rule, alluding to the term "you can't get blood out of a stone" (SINN 1983, p. 163).²⁶⁰ Second, the study by DIAMOND and RAJAN 2011 advances the hypothesis that undercapitalized banks allocate investment to illiquid assets such as risky government bonds, given that the state of the world in which the bonds default coincides with the state in which banks themselves default. Hence, portfolio concentration would be rationale as the entire risk is with a state of the world that would be catastrophic regardless. Finally, banks may engage in risk shifting by increasing their stock of risky (domestic) government debt with the endeavor to remain or become systemically important. The government's inability to credibly commit

²⁵⁷ On the increased importance of government debt as collateral for refinancing (ASONUMA et al. 2015) in the market, see BOLTON and JEANNE 2011 and from the central bank, see Chapters 3 and 5.

²⁵⁸ The tendency of investors to predominantly invest in their home country was first documented by FRENCH and POTERBA 1991 as well as TESAR and WERNER 1995 for equities. See e.g. ACHARYA et al. 2012, 2014b for the home bias in government debt.

²⁵⁹ Repatriation of government bond holdings is confirmed by MERLER and PISANI-FERRY 2012, ANDRITZKY 2012, LEVY and LEVY 2014, VAN RIET 2014 as well as ACHARYA and STEFFEN 2015. According to ANDRITZKY 2012, HAAS and VAN HOREN 2012 as well as GIANNETTI and LAEVEN 2012, repatriation in investment is typical pattern in the aftermath of crisis.

²⁶⁰ See STIGLITZ and WEISS 1981 for the phenomenon in a credit market equilibrium context. Others termed it a "gamble for resurrection", see DEWATRIPONT and TIROLE 1993, 1994 as well as SINN 2008.

not to bail out systemically important banks incentivizes banks to excessively and collectively invest in assets that decline during crisis, such as government debt (ACHARYA and YORULMAZER 2007, FARHI and TIROLE 2012, 2014 as well as GENNAIOLI et al. 2014). Owing to excessive and collective investment, banks maximize the value of the implicit government guarantee. Hence, risk shifting can explain that (undercapitalized) banks scale up holdings of domestic government debt.

Moreover, the hypothesis of moral suasion by the government is put forward to explain portfolio concentration to government debt (e.g. ACHARYA and STEFFEN 2015). Accordingly, governments of crisis-stricken countries appeal to domestic banks to increase the absorption of their debt. MARCO and MACCHIAVELLI 2014 provide a systematic analysis of moral suasion by governments in recent years, documenting the extent of moral suasion and showing that during the European sovereign debt crisis and upon receiving government support banks significantly increased exposure to domestic government debt only if they had strong political affiliations.²⁶¹ This effect appears to be twice as large for banks located in the European periphery. According to BUITER and RAHBARI 2012b, many banks were subject to moral suasion by governments to draw credit from the Europystem and park the funds in sovereign debt. BATTISTINI et al. 2013, 2014 claim that several national supervisors temporarily promoted a home bias in bank exposure.

Finally, PORTES et al. 2001 argue that it is principally advantageous to hold government bonds as they respond less to information frictions than corporate bonds or equity. ARSLANALP and TSUDA 2014 quote general factors such as the global recession and new financial regulation in explaining the intensification of domestic government bond holdings after the financial crisis. Weak economic growth in the aftermath of the financial crisis reduced demand for bank loans, prompting banks to increase government bond holdings. Moreover, banks likely intensify government bond holdings to meet new regulation such as Basel III, which demands stricter capital and liquidity standards (ibid.). In addition, financial regulation in place assigning a zero-risk weight to government bonds induces undercapitalized banks to invest in additional government bonds to make use of regulatory arbitrage.

6.5.2 Collateral Criteria and Central Bank Finances

Collateral criteria bear fiscal implications as they are crucial for central bank finances, which are intertwined with the government budget (see Chapter 7). Relaxed collateral criteria may be detrimental to central bank finances owing to three effects. While the first effect is general, the others are specifically related to the Eurosystem. First, central banks predominantly realize income from accumulating interest-bearing assets, which they receive in exchange for self-created money. When collateral criteria are strict, the flow of interest income is steady and safe. Section 7.3.1 shows that when collateral criteria are relaxed, central bank interest income becomes vulnerable. Pledged assets that the central bank would have to liquidate in case of

²⁶¹ The authors measure political affiliation as the total percentage of shares held by the government or by political foundations prior to the crisis. Moral suasion of the banking sector by governments is not restricted to the recent past but originates from the close relation between governments and banks. See BUCK 2013 for the historical perspective.

counterparty default bear a higher risk of default and are less liquid when collateral criteria are lax. Consequently, the flow of interest income may become less reliable, which can put central bank finances at risk. Second, Section 7.3.2 argues that central bank income in the Eurozone is particularly put at risk by relaxed collateral criteria and the provision of ELA. ELA is provided by NCBs on their own account to solvent banks that lack collateral complying with the (relaxed) uniform Eurosystem collateral criteria. Hence, any relaxation of uniform collateral criteria implicitly lowers collateral criteria applied in ELA. This puts finances of the responsible NCB at risk. Likewise, ELA puts at risk finances of all other NCBs as the responsible NCB is de jure liable for all losses from ELA while liability is de facto limited. Therefore, ELA losses would be shared among NCBs if they exceeded the financial means available to the responsible NCB. Finally, finances of NCBs in the Eurosystem may come under distress owing to intra-Eurosystem imbalances (Section 7.3.3). The relaxation of collateral criteria has resulted in the accumulation of huge claims and liabilities of NCBs towards the ECB. If an NCB fails to settle its liability with the ECB, all NCBs share the write-off losses of the ECB.

Distressed central bank finances may raise the issue of central bank recapitalization ("bailout"), which can take three forms (Section 7.4): first, the central bank can retain part of its profit to bolster financial strength ("internal recapitalization"); second, the central bank can call the government for capital injection ("external recapitalization"); and finally, a third (but governmentrelated) party may divert distress from the central bank balance sheet ("circumvented recapitalization"). Irrespective of the peculiarities of recapitalization, the government bears the contingent implicit liability of central bank bailout, which would adversely affect its fiscal position if it matured as the government would have to divert existing financial means or issue new debt. In the latter case, the contingent implicit liability would transform into a non-contingent explicit liability. Section 7.5 elaborates on the fiscally desired level of collateral criteria, which is the outcome from trading off the cost and benefits from amending collateral criteria from a fiscal perspective. It shows that while the fiscally desired level of collateral criteria is unique in a one-country one-central bank setting, divergent optimal levels can emerge in a multi-country one-central bank setting. It is shown that divergent fiscally desired levels of collateral criteria give rise to cross-country fiscal implications, depending on the uniform level of collateral criteria that the central bank implements. Therefore, it is argued that the relaxation of collateral criteria by the Eurosystem may have resulted in a transfer of fiscal sustainability across countries as collateral criteria are too strict for some countries yet too lax for others.

6.5.3 Collateral Criteria and Government Guarantees

Finally, Chapter 8 shows that collateral criteria shape government incentives to assume contingent explicit liabilities in terms of explicit individual guarantees and insurance schemes to the banking sector (see Table 6.2). Section 3.2.2 analyzed that the Eurosystem relaxed collateral criteria for government-guaranteed assets after the onset of the financial crisis. This endowed governments the discretion to free up collateral as explicit government guarantees may have three effects on assets: first, the guarantee can be basal for eligibility, e.g. in case of own-use uncovered bank bonds; second, the guarantee would augment the collateral value of an eligible asset if it improved the credit assessment deemed decisive for haircut determination; and third, the guarantee augments the collateral value by increasing the market value of the asset. If governments make use of this discretion, collateral criteria bear fiscal implications as they imply dynamics in the government liability matrix as governments assume additional contingent explicit liabilities or make existing implicit guarantees explicit. Chapter 8 indicates that governments of several crisis-stricken countries have made use of their discretion in recent years. They have initiated new and extended existing explicit guarantee schemes as well as granting individual explicit guarantees to facilitate the refinancing of local banks with the Eurosystem. This finding motivates a novel rationale for governments to give guarantees to the banking sector beyond the common rationale, i.e. preventing panic-driven runs on solvent yet illiquid financial institutions (DIAMOND and DYBVIG 1983). The novel rationale explains government incentive to give guarantees based on their discretion to provide collateral to banks, which is inherent in the central bank collateral framework. Moreover, the novel rationale can contribute to explain the development of explicit government guarantees to eligible marketable assets observable in the Eurozone.

7

Fiscal Implication from Collateral Criteria and Central Bank Finances

This chapter elaborates on the fiscal implication of collateral criteria that evolves from the close relation between collateral criteria, central bank finances and the government budget. The relaxation of collateral criteria can give rise to central bank financial straits, prompting the government to provide financial support to the central bank ("central bank bailout"). The chapter argues that collateral criteria have a bearing on central bank finances in three ways, the first of which is general, while the second and third are specific to the Eurosystem. First, amendments to collateral criteria alter the expected value of central bank income from money creation. Consequently, the expected income from money creation to the Eurosystem has deteriorated over recent years owing to relaxed collateral criteria. This effect has been reinforced by the adverse selection of collateral. Second, the provision of Emergency Liquidity Assistance (ELA) puts central bank income in the Eurosystem further at risk as it is granted against collateral of lower quality than stipulated by the collateral framework. Third, the accumulation of intra-Eurosystem balances was facilitated by relaxed collateral criteria, thus jeopardizing central bank finances in the Eurosystem. Amendments to collateral criteria hence affect central bank finances and the likelihood of central bank bailout. The fiscally desired level of collateral criteria balances the costs and benefits to the Treasury from amending collateral criteria. It is shown that uniform collateral criteria bear fiscal implications in a stylized form of the Eurozone when Treasuries prefer distinct levels of collateral criteria. Depending on their level, uniform collateral criteria give rise to risk-sharing and the transfer of fiscal sustainability across countries. The chapter is structured as follows. Section 7.1 emphasizes the relevance of central bank finances and Section 7.2 presents stylized facts on central bank finances in the Eurosystem. Section 7.3 elaborates on the importance of collateral criteria for central bank finances in due consideration of developments in the Eurosystem. Section 7.4 argues that financially-stricken central banks may call the government for recapitalization. Finally, Section 7.5 presents a model to characterize the fiscally desired level of collateral criteria and elaborates on the fiscal implications of collateral criteria via their effect on central bank finances.

7.1 Relevance of Central Bank Finances

While the issue of central bank finances has long been confined to developing countries, it has been placed on the agenda in advanced economies in the recent past as central banks took more active policies, thus rendering their finances more vulnerable.²⁶² For instance, the ECB claimed additional capital from the NCBs to the amount of EUR 5 bn in December 2010, accounting inter alia for growing credit risk.²⁶³ Hence, capital of the ECB surged from EUR 5.76 bn to EUR 10.76 bn. This anecdotal evidence suggests that central bank finances are a topical issue, which is emphasized by LÖNNBERG and STELLA 2008, SCHOBERT 2008, as well as LEHMBECKER 2009.²⁶⁴ LÖNNBERG and STELLA 2008 present evidence on fifteen Central and South American central banks that experienced losses for five or more years between 1987 and 2005. SCHOBERT 2008 documents that 43 out of 108 central banks bore losses from 1984 to 2005. LEHMBECKER 2009 reports that half of a sample of 62 central banks recorded losses at least in one year between 1990 and 2003.

The major reason why central bank finances experienced a surge in attention is the non-standard measures by which central banks all over the world intervened as lender of last resort (LOLR) to unprecedented extent. LOLR-lending was identified as a major cause of central bank financial straits (HAWKINS 2003 and CUKIERMAN 2011) as it is associated with greater risk to central banks' balance sheets.²⁶⁵ This concern was particularly raised with respect to NCBs of the Eurosystem, which were increasingly exposed to risk from purchases of and lending against lower quality assets (e.g. SIBERT 2009, BUITER and RAHBARI 2012b as well as SINN 2012, 2014b).

Why do central banks hold capital? The primary reason for any bank is to absorb potential losses in adverse states of the world, which would otherwise imply negative capital and balance-sheet insolvency.^{266,267} A commercial bank deemed balance-sheet insolvent is forced to quit operating and is liquidated. By contrast, a balance-sheet insolvent central bank cannot be forced to quit operating as it is typically excluded from company law and, where applicable, it is explicitly protected from bankruptcy or related proceedings (ARCHER and MOSER-BOEHM 2013). Hence, a central bank can never be considered balance-sheet insolvent from a legal perspective. In addition, there is also a repeatedly adduced economic reason (e.g. DE GRAUWE and JI 2013).²⁶⁸ The central bank usually possesses the monopoly right to create the economy's legal tender,

²⁶² Accordingly, literature on central bank finances has been limited yet growing in recent years. See ADLER et al. 2012, ARCHER and MOSER-BOEHM 2013, as well as SCHWARZ et al. 2014 for recent contributions.

²⁶³ Furthermore, the recapitalization aimed at enabling the ECB to augment its risk buffer as the provision and reserves are statutorily limited to the level of paid-up capital.

²⁶⁴ For earlier evidence, see e.g. LEONE 1994 for a study providing data on central bank losses for a group of fourteen Latin American countries between 1987 and 1992. The author observes that some countries report fiscal deficits smaller than central bank losses.

²⁶⁵ CUKIERMAN 2011 e.g. argues that monetary policy actions taken during the crisis increased the risk profile of assets held by the central banks, whereby such policy action might create large capital losses, potentially leading to low or negative levels of capital, including in the longer run.

 $^{^{266}}$ Moreover, capital may be intended to cover start-up costs (STELLA 1997).

²⁶⁷ "Balance-sheet insolvency" refers to the situation in which a bank holds liabilities in excess of assets. The other concept of bank insolvency is "cash-flow insolvency" (illiquidity), in which the bank is unable to pay obligations as they fall due (BUITER 2008a and LASTRA 2011).

²⁶⁸ It refers to situations in which liabilities are predominately nominally denominated (i.e. not index-linked) or denominated in domestic currency. If denomination is nominal or in foreign currency, a central bank would be increasingly exposed to the threat of cash-flow insolvency.

i.e. money. Money is created at virtually no cost while the central bank receives interest in exchange for it. BUITER 2008a argues that the central bank is not restricted in terms of money creation, thus implying what ibid. terms the "short-term deep pockets" of the central bank, which complement the "long-term deep pockets" of the government (see also CHAILLOUX et al. 2008b). The short-term deep pockets give rise to the central bank franchise value²⁶⁹, which was claimed to be included as an intangible asset in a comprehensive central bank balance sheet (BUITER 2008a, BUITER and RAHBARI 2012a,b).²⁷⁰ The residual in this comprehensive balance sheet would be central bank net worth, which could substantially differ from central bank capital (residual of the conventional central bank balance sheet). When central bank capital is negative while net worth remains positive, the central bank would be considered balance-sheet insolvent based on its conventional balance sheet but solvent with respect to its comprehensive balance sheet.

While this reasoning sounds appealing as it allows for the power of central banks to generate income by money creation, it is misleading from an accounting perspective.²⁷¹ Money creation is neutral from an accounting perspective and no net worth can be created out of the blue (CARUANA 2012) as it gives rise to a corresponding accumulation of interest-bearing assets. The stream of interest income that the central bank receives is exactly equal in present-value terms to the value of assets in the balance sheet. Hence, the central bank franchise value is fully matched by the current (reflected in the conventional and comprehensive balance sheet) and future value (reflected only in the comprehensive balance sheet) of base money. Hence, the net worth of the central bank cannot be augmented by accounting wrinkles as future base money also has to be accounted for in the comprehensive balance sheet in terms of an intangible liability.

Beyond accounting subtleties, anecdotal evidence and empirical analyses suggest that central banks care about their financial position and perform monetary policy under due consideration of their finances.²⁷² For instance, FRIEDMAN and SCHWARTZ 1963 claim that balance sheet concerns were a factor crucial in preventing the Federal Reserve from an aggressive expansionary response to the Great Depression in the 1920s. More recently, CARGILL 2005, BENECKÁ et al. 2012 and NYBORG 2015 requote policy-makers of the Bank of Japan, the Swiss National Bank and the Dutch Central Bank, expressing concern about the potential deterioration of finances induced by non-standard monetary policy measures. The ECB itself states that "an NCB should always be sufficiently capitalized. In particular, any situation should be avoided whereby for a prolonged period of time an NCB's net equity is below the level of its statutory capital or

²⁶⁹ See Fry 1993, Stella 1997, Hawkins 2003, Pringle 2003a,b on the franchise value of central banks.

²⁷⁰ In addition to the franchise value, measured as the present value of income from money creation, intangible liabilities such as the present value of the cost running the central bank and the present value of payments to the government may be considered in the comprehensive balance sheet.

²⁷¹ See also LEHMBECKER 2009 for a critical appraisal, where it is argued that reliance on the franchise value of the central bank could be misleading as it is impossible to assess.

²⁷² BINDSEIL et al. 2004 review the literature, finding that sufficient capital is regarded as a crucial prerequisite for a central bank's ability to achieve monetary stability. More recent surveys are provided by ADLER et al. 2012 as well as ARCHER and MOSER-BOEHM 2013. SIMS 2004 provides a theoretical approach showing that a central bank concerned about its independence could refrain from implementing expansive monetary policy owing to risk to its balance sheet. Likewise, JEANNE and SVENSSON 2007 argue that central banks care about their financial position and pursue suboptimal policies to minimize the risk of losing independence, which could be impaired by financial weakness. SINCLAIR and MILTON 2011 claim that central bank capital is relevant as potential fiscal back-up is both costly and uncertain.
is even negative. [...] Therefore, the event of an NCB's net equity becoming less than its statutory capital or even negative would require that the respective Member State provides the NCB with an appropriate amount of capital [...] within a reasonable period of time" (ECB 2012a, pp. 25-6). Moreover, empirical analyses explore the effects of central bank finances on monetary policy outcomes such as inflation. IZE 2006 divides a sample of 87 central banks into subsamples with either positive or negative structural profits and finds average inflation in the former subsample to be 3.5% but considerably higher in the latter (9.5%). STELLA 2003 applies an akin approach and produces similar results for a different sample and different years. STELLA 2011 investigates an artificial indicator for central bank financial strength for a wider sample of central banks in a different set of years, confirming that central banks with weak finances tend to have higher inflation outcomes. However, these studies fail to control for the presence of other factors that may affect policy outcomes. Taking an econometric approach to control for such factors, KLÜH and STELLA 2008 find a negative relationship between central bank financial strength and inflation outcomes, which is robust to the choice of alternative country samples, control variables, estimation strategies and concepts of central bank financial strength. ADLER et al. 2012 have recently investigated the effects of central bank finances on deviations of the interest rate from its "optimal" level (in the spirit of Taylor), finding statistically significant effects.

Various causes for deterioration in central bank finances have been revealed in the past,²⁷³ such as (*i*) periods of credibility build-up (HAWKINS 2003, CUKIERMAN 2011), (*ii*) interventions to stabilize the exchange rate (DALTON and DZIOBEK 2005, CUKIERMAN 2011), (*iii*) quasi-fiscal operations (STELLA 1997, HAWKINS 2003, MARTÍNEZ-RESANO 2004, GROS 2004, STELLA 2008, BUITER and RAHBARI 2012a) and (*iv*) deterioration in profitability (DALTON and DZIOBEK 2005, PAPI 2011).²⁷⁴ It is argued in the following that recent non-standard monetary policy measures of the Eurosystem put at risk central bank finances owing to a combination of (*iii*) and (*iv*). Quasi-fiscal activities often take the form of central bank support to the financial sector in terms of subsidized loans and the purchase of overvalued or non-performing assets.²⁷⁵ This support often proves to have detrimental effects on central bank profitability (HAWKINS 2003, STELLA 1997, CUKIERMAN 2011), particularly when performed at low interest rates and against potentially inadequate collateral (ERNHAGEN et al. 2002).

7.2 Stylized Facts on Central Bank Finances in the Eurosystem

This section elaborates three stylized facts on central bank finances in the Eurosystem that were affected by non-standard monetary policy measures performed by the Eurosystem since the financial crisis.²⁷⁶

²⁷³ See LEONE 1994, ERNHAGEN et al. 2002, DALTON and DZIOBEK 2005 as well as PAPI 2011 for overviews.

²⁷⁴ Furthermore, DARBYSHIRE 2011 suggests that the implementation of more rigorous central bank accounting standards has amplified central bank financial straits.

²⁷⁵ GROS 2004 investigates the effect of central bank quasi-fiscal activities on its inflation performance and finds a statistically significant correlation between the leanness of the central bank balance sheet and average inflation for a sample of European central banks between 1967 to 1990; namely, the less quasi-fiscal activities that a central bank had to perform, the better its inflation performance.

²⁷⁶ See e.g. COUR-THIMANN and WINKLER 2013, CLAEYS 2014, DOMANSKI et al. 2014 as well as Chapter 2 for the non-standard measures and ARCHER and MOSER-BOEHM 2013 for effects on central bank financial strength.

Stylized Fact 7.1: Sustained profitability and transfers. The left panel of Figure 7.1 indicates that selected NCBs have remained profitable over recent years, with the exception of the Bank of Greece, which showed losses in 2012 and 2014. Prior to the financial crisis, the ECB showed losses in 2003 and 2004, as well as zero-profits from 2005 to 2007, although profits increased during the financial crisis in 2008 and 2009. By contrast, profits of the Deutsche Bundesbank considerably declined between 2008 and 2012, while the Banca d'Italia increased and sustained high profits as of 2008. Accordingly, transfers to governments (right panel) remained high and even increased for most NCBs, while the Deutsche Bundesbank had to considerably reduce its transfers after the financial crisis between 2008 and 2012 owing to low profits.

Figure 7.1: Annual profits of selected NCBs and the ECB as well as transfers to governments

The figure shows the development of central bank annual profits and transfers to governments. The left panel indicates that NCBs remained profitable over recent years, with the exception of the Bank of Greece, which showed losses in 2012 and 2014. Accordingly, transfers to governments (right panel) remained high or even increased for most NCB, while the Deutsche Bundesbank had to reduce its transfers after the financial crisis.



Stylized Fact 7.2: Profitability was sustained owing to riskier monetary policy. Sustained profitability can be mainly ascribed to the Eurosystem having engaged as LOLR since the onset of the financial crisis by substantially increasing liquidity provision (Chapter 2). IZE and OULIDI 2009 find that more active central banks (or central banks operating in more turbulent macroeconomic environments) become more exposed to extraordinary profits. This suggests balance sheet volatility as a good predictor of central bank profitability. In normal times, central banks strive for a lean balance sheet, characterized by interest-free banknotes representing the majority of liabilities and hence of the balance sheet total. In times of crisis, central banks tend to become more active and perform non-standard measures, which implies the accumulation of liabilities other than banknotes. The balance sheet leanness indicator (BSLI) provides information on balance sheet volatility and puts the balance sheet total in relation to banknotes in circulation.²⁷⁷ It is given by

$$BSLI = \frac{balance sheet total}{banknotes in circulation}.$$
 (7.1)

²⁷⁷ See e.g. BINDSEIL 2014 for the BSLI. A similar indicator is applied in GROS 2004, reflecting the ratio of total liabilities over the monetary base.

A BSLI equal to unity indicates a perfectly lean balance sheet, i.e. interest-free banknotes are the sole liability of the central bank. BINDSEIL 2014 argues that such a balance sheet suggests that a central bank is focused on its core mandate of price stability and that financial markets function well because the central bank is neither used as intermediary by the banking system nor is it engaged in special measures, e.g. sterilizing asset purchases. A balance sheet that exceeds unity largely indicates that a central bank is not entirely focused on price stability but engages in non-standard measures. Figure 7.2 illustrates the development of the BSLI for selected NCBs from 2002 to 2014. It reveals a convergence of the BSLI except for the Bank for Ireland prior to the financial crisis towards a BSLI value of 2.4. The BSLI diverged as of 2008 and resumed to converge as of 2013. Crisis-related increases were most prominent for the ECB with a jump in 2008, and for the Bank of Greece as well as the Banco de Portugal, for which the BSLI substantially increased from 2007 to 2012. The BSLI of the Banco de España remained low after the outset of the financial crisis but considerably increased during the sovereign debt crisis as of 2010. The Central Bank of Ireland has the most volatile BSLI: already running at a high level prior to the financial crisis, it considerably increased during the financial crisis and peaked at 16.6 in 2010, before continuously decreasing thereafter. This development indicates that the sustained profitability of NCBs was associated with the extension of the monetary base by extended liquidity provision. While this contributed to preserving central bank sustainability, it makes NCBs and the ECB prone to extraordinary losses. This is emphasized by IZE and OULIDI 2009, who find that more active central banks also become more exposed to extraordinary losses.

Figure 7.2: Balance sheet leanness indicator for selected NCBs

The figure illustrates the development of the BSLI for selected NCBs from 2002 to 2014. It reveals convergence of the BSLI prior to the financial crisis, aside from the Bank for Ireland, which has the most volatile BSLI. The BSLI diverged as of 2008 and resumed to converge as of 2012.



Source: author's calculation; National Central Banks; European Central Bank.

Stylized Fact 7.3: NCBs moderately adapted risk buffer. NCBs responded to the increase in risk exposure by adapting buffers against balance sheet shocks. A risk buffer comprises (*i*) capital and reserves, (*ii*) revaluation accounts and (*iii*) provisions and its development is

revealed in Figure 7.3 for selected NCBs. The development of capital and reserves relative to the balance sheet total is depicted in the upper-left panel, showing that the ratio of capital and reserves remained low or shrunk for most NCBs owing to an expansion in balance sheets. The overall average ratio steadily declined from 4.1% in 2002 to 2.6% in the pre-crisis year of 2007 to 2.1% in 2014. The ratio of capital and reserves of the ECB experienced a drop in 2008 but recovered thereafter. Another component of the risk buffer is revaluation accounts, which are held as protection against changes in asset valuation. Financial means retained for potential revaluation relative to the balance sheet total are displayed in the upper-right panel of Figure 7.3. Large volatility is due to changes in both absolute revaluation accounts and balance sheet sizes. The average ratio increased from 5.9% in 2002 to 6.4% in 2007 to 9.5% in 2014. Remarkably, the Central Bank of Ireland has substantially increased its revaluation accounts since 2012 and the Banca d'Italia accumulated revaluation accounts at a peak of 21.1% of the balance sheet total in 2010, although it sharply decreased thereafter. The bottom-left panel gives the ratio of provisions to the balance sheet total. The average provisions ratio decreased from 4.3% in 2002 to 2.2% in 2008, with the Banca d'Italia and the Banco de Portugal holding above-average relative provisions. The average ratio advanced to 3.3% in 2014 despite the additional risk that central banks have subsequently assumed with the non-standard measures. While relative risk provisions dropped for the Banca d'Italia and the Banco de España, provisions of the Bank of Greece experienced a marked growth.²⁷⁸ Mandatory "special provisions for monetary policy operations" are included in provisions, given in the bottom-right panel. These provisions were accumulated by NCBs according to capital keys with the ECB to counteract increased counterparty risk in collateralized lending, which culminated in the default of five counterparties in the fall of 2008 (Section 7.3.1.3). This cushion was accumulated in 2008 and steadily decreased thereafter until it was dissolved in 2013.

This section has revealed that while NCBs has remained profitable over recent years, support to the financial system by expanded liquidity provision through non-standard policy measures has been associated with higher risk to their balance sheets. NCBs only moderately adapted risk buffers, which left them prone to the potential effects of adverse shocks to the balance sheet. The next section elaborates on three potential sources of shocks to central bank balance sheets and particularly concerning the importance of collateral criteria in this context.

7.3 Collateral Criteria and Central Bank Finances

This section elaborates on the importance of collateral criteria for central bank finances. It argues that collateral criteria are crucial to central bank income from money creation as collateral serves as a risk hedge in case of counterparty default (Section 7.3.1). The relaxation of collateral criteria together with insufficient risk control subsequently puts at risk central bank income from money creation, whereby this effect is amplified in case of the adverse selection of collateral. Moreover, it is addressed that collateral criteria in the Eurosystem and particularly their relaxation were decisive for two developments in the Eurozone, i.e. the provision of ELA (Section 7.3.2) and the

 $^{^{278}}$ This growth was due to increases in the absolute amount of provisions, which changed from EUR 1.1 bn in 2008 to EUR 6.8 bn in 2014.



The figure details the development of a risk buffer in the Eurosystem from 2002 to 2014, comprising (i) capital and reserves, (ii) revaluation accounts and (iii) provisions. Moreover, the bottom-right panel illustrates "special provisions for monetary policy operations", which were temporarily accumulated in 2008. The overall impression is that the Eurosystem only moderately adapted a risk buffer despite higher risk from non-standard monetary policy measures.



accumulation of intra-Eurosystem imbalances (Section 7.3.3). While the relevance of collateral criteria for income from money creation is general to central banks, the relevance for ELA and intra-Eurosystem imbalances is specific to the Eurosystem.

7.3.1 Collateral Criteria and the Income from Money Creation

Income that the central bank realizes by using self-created money (base money) for lending or purchases is referred to as "seignorage".²⁷⁹ While base money is accounted as a liability in the central bank balance sheet, it is not a true liability in the economic sense for two reasons (cf. LANGE 1995, SINN and FEIST 1997 as well as DE GRAUWE and JI 2013). First, demand for base money increases in a growing economy, such that redemption claims against the central bank

²⁷⁹ In its historical sense, "seignorage" refers to the income of the "seigneur" who issued money against its face value. For early analyses of seignorage and sources of central bank income, see e.g. THORNTON 1802, BRESCIANI-TURRONI 1937 and CAGAN 1956.

are unlikely.²⁸⁰ Second, base money does not give interest to its holder such that it is interestfree to the central bank. Hence, the central bank realizes interest income on the assets that it receives in exchange for base money. Interest income represents a substantial part of central bank income, as Figure 7.4 reveals for selected NCBs. The left panel details interest income from both lending and purchases relative to total net income. It reveals that the importance of interest income differs across NCBs, whereby it grew from 2005 to 2014 for the NCBs of Spain, France, Greece and Portugal yet diminished in Germany and Ireland. The right panel shows absolute interest income from lending and purchases for the NCBs of Germany, Spain, Greece, Ireland and Portugal.²⁸¹ It points to the growing importance of interest income from asset purchases and reveals that interest income became volatile as of 2008 despite growth for all NCBs prior to the financial crisis. Since then, three phases can be identified. First, all NCBs aside from the Bank of Greece experienced a contraction in interest income after the onset of the financial crisis (2008-2009). Second, this contraction was compensated during the sovereign debt crisis (2010-2012), predominantly owing to larger interest income from asset purchases. Compensation substantially varied across NCBs, with the NCBs of Spain, France, Greece and Portugal being the main profiteers. Third, interest income deteriorated again as of 2012.

Figure 7.4: Importance of interest income from central bank loans and asset purchases²⁸²

The figure reveals that interest income represents a substantial part of central bank income for selected NCBs. The left panel details the development of interest income relative to total net income from 2005 to 2014, suggesting that interest income differs in its importance to NCBs. The right panel differentiates between interest income from loans and purchases, suggesting that interest income became more dependent on asset purchases.



The following aims to explain the developments observable in Figure 7.4. It argues that interest income is associated with a larger degree of uncertainty when collateral criteria are relaxed.

²⁸⁰ A lack of redemption alone does not suffice to disqualify base money from being a true liability because, like a perpetuity, potential interest yields would imply a negative (positive) present value for the debtor (creditor), LANGE 1995.

²⁸¹ The Banca d'Italia does not publish detailed information on income from asset purchases for monetary policy operations.

7.3.1.1 Measuring Income from Money Creation

No commonly-accepted measure of seignorage exists as the income from money creation. Three measures that have been most prominently proposed are briefly reviewed in the following,²⁸³ i.e. (i) opportunity cost seignorage, (ii) monetary seignorage and (iii) fiscal seignorage.²⁸⁴ The analysis makes the following simplifications without loss of generality. First, the central bank only receives interest income from loans and purchased assets, which abstracts from e.g. income from other central bank assets such as gold or foreign exchange reserves. Second, costs of money creation like material, production and maintenance expenses are neglected as they are negligible in today's fiat money system. Hence, the concepts presented in the following are defined as gross concepts.²⁸⁵ Third, minimum reserves that central banks usually require commercial banks to hold are interest-free such that the economic nature of minimum reserves is similar to that of currency and the entire monetary base is not remunerated. Fourth, the central bank distributes its entire earnings to the government and does not retain any interest income.²⁸⁶

Opportunity Cost Seignorage The concept of opportunity cost seignorage centers upon the feature of base money being interest-free.²⁸⁷ The private sector holds base money because it either provides utility or is forced to do so by the central bank in terms of reserve requirements. Irrespective of the motivation, base money is an interest-free asset to the private sector such that seignorage is measured in terms of foregone interest income, i.e. the opportunity cost of holding money, of the private sector.²⁸⁸ Opportunity cost seignorage δ_{opp} is formalized as

$$\delta_{opp} = \frac{iM_t}{P},\tag{7.2}$$

with *i* being the market interest rate, M_t the monetary base in period *t* and *P* the time-invariant price level. However, the application of the concept of opportunity cost seignorage causes two problems. On the one hand, opportunity cost seignorage may not equal the actual income from base money creation whenever the structure of the asset portfolio of the central bank differs from that preferred by private investors (NEUMANN 1992).²⁸⁹ On the other hand, the time path of the true nominal interest rate *i* has to be chosen (SINN and FEIST 2000). A reasonable yet errorprone approach would be to determine the weighted average of observable asset returns that meaningfully approximates the true opportunity cost of money holders (KLEIN and NEUMANN 1990). In order to avoid these problems, empirical studies have focused on the following concept of seignorage (FISCHER 1982 and MANKIW 1987).

²⁸³ See DRAZEN 1985, KING and PLOSSER 1985, KLEIN and NEUMANN 1990, NEUMANN 1992, GROS 1993, JORDAN 1994, LANGE 1995, NEUMANN 1996, HONOHAN 1996 and BUITER 2007 for further reviews.

²⁸⁴ A fourth—albeit less frequently applied—approach is to multiply base money by the inflation rate, see e.g. BAILEY 1956, FRIEDMAN 1956 as well as POTERBA and ROTEMBERG 1990.

 $^{^{285}\,\}mathrm{See}$ e.g. Lange 1995 for net concepts.

 $^{^{286}}$ This simplification is relaxed in Section 7.4.

²⁸⁷ See BAILEY 1956, PHELPS 1973, AUERNHEIMER 1974 and MARTY 1978 for standard references. For applications of the concept, see e.g. among others BARRO 1982 for the US and LANGE 1995 for Germany.

²⁸⁸ An alternative interpretation is to consider the monetary base as interest-free credit from the private to the public sector. In this case, opportunity cost seignorage gives the real value of per-period saved interest payments by the government (GROS 1989). Both interpretations are only equivalent if opportunity cost of the private and the public sector are equal.

²⁸⁹ This argument may not hold if opportunity cost seignorage is interpreted as the interest savings for the government from issuing base money, see ibid.

Monetary seignorage The concept of monetary seignorage refers to seignorage as the net change in base money outstanding.²⁹⁰ Therefore, it measures the actual wealth transfer that the private sector has to make to the public sector to receive base money. Monetary seignorage δ_{mon} is formalized as

$$\delta_{mon} = \frac{M_t - M_{t-1}}{P} = \frac{\Delta M}{P},\tag{7.3}$$

where ΔM represents the change in base money between periods t-1 and t. Monetary seignorage is a flow concept associated with certain shortcomings. For instance, it can only arise if the monetary base is increased, which conflicts with any positive interest income that may accrue even with a constant monetary base from prior base money creation. Moreover, as monetary seignorage considers the creation of base money, it may deviate from interest income that does not accrue before the assets backing the monetary base pay interest.²⁹¹

The flow concept of monetary seignorage and the stock concept of opportunity cost seignorage are interrelated as follows. From an economic perspective, there is little difference between focusing on interest wealth (stock) or interest income (flow) because the accumulated stock of interest-bearing assets equals the present value of interest income. This present value equals today's market value of the assets that generate interest. Therefore, stock and flow approaches are equivalent under the prerequisite that the interest-bearing assets of the central bank are evaluated at their market value.²⁹² Abstracting from inflation, a one-shot increase in base money today (t = 0) is exactly equal to monetary seignorage and induces an infinite interest flow on assets accumulated owing to the additional base money, such that

$$\delta_{mon} = \frac{\Delta M}{P} = \frac{1}{P} \sum_{t=1}^{\infty} \frac{i\Delta M}{(1+\beta)^t} = V(\delta_{opp}), \tag{7.4}$$

where β is the discount rate and $V(\cdot)$ denotes the present value. This infinite interest flow corresponds to the present value of seignorage measured by the opportunity cost concept. The stock concept (opportunity cost concept) of seignorage hence corresponds to the flow concept (monetary seignorage) when the former is measured in present value terms.

Both concepts only implicitly consider the actual source of income from money creation, i.e. the assets backing the monetary base. Hence, both concepts abstract from the actual process of base money creation with the result that they may fail to measure the actual amount of seignorage.²⁹³ KLEIN and NEUMANN 1990 coin the term "fiscal seignorage" as an alternative approach to measure seignorage at its actual source.

²⁹⁰ For monetary seignorage, see CAGAN 1956, MARTY 1967, DORNBUSCH 1988, GRILLI 1988 and KLEIN and NEUMANN 1990. FRIEDMAN 1971, CALVO 1978 as well as FISCHER et al. 2002 use the concept to determine the income-maximizing inflation tax, where the inflation tax is the reduction in the real value of the monetary base caused by inflation. See also EASTERLY et al. 1995.

²⁹¹ This shortcoming does not apply to opportunity cost seignorage, which is a stock concept and measures seignorage at the time when interest income accrues to the central bank.

²⁹² Central banks are usually required to comply with international financial reporting standards that account assets according to their market value (mark-to-market accounting), see HAWKINS 2003 and PRINGLE 2003a. The equivalence of stock and flow approaches fails to hold (irrespective of accounting) if some assets of the central bank do not generate income (see SINN and FEIST 1997).

²⁹³ NEUMANN 1992 assumes this point of criticism and presents an approach that takes into account the structure of assets backing the monetary base. This approach extends monetary seignorage by including the interest income on the assets backing the monetary base, i.e. $\delta_{mon,ext} = \Delta M/P + i(M_t/P)$.

Fiscal seignorage Fiscal seignorage measures the income that the central bank receives from interest-bearing assets backing base money.²⁹⁴ While KLEIN and NEUMANN 1990 consider fiscal seignorage in a sources-and-uses format such that it is determined residually in either monetary or opportunity cost seignorage, NEUMANN 1992 as well as LANGE 1995 take into account the actual source of seignorage, i.e. the assets that the central bank receives in exchange for base money. Under the assumption that base money is solely created by lending to commercial banks, fiscal seignorage can be formalized as

$$\delta_{fisc} = \frac{i_L L_t}{P},\tag{7.5}$$

where L_t are loans granted to the private sector in period t and i_L is the (uniform) interest rate on these loans.

The concepts of seignorage presented thus far explain shortfalls in seignorage over time only by a decreasing interest rate or monetary base. However, shortfalls may occur for a non-decreasing interest rate or monetary base as the assets backing the monetary base may default. The assets backing the monetary base have been implicitly assumed to be perfectly safe thus far. The negligence of default risks leaves aside valuable insights on sustaining seignorage as an important source of central bank income. An alternative concept of seignorage is developed in the following, aiming to correctly grasp both the actual process of base money creation and the default risk of assets backing the monetary base. This alternative concept is closely related to fiscal seignorage and it explicitly considers the asset side of the central bank balance sheet.

7.3.1.2 Asset-Side Approach to Income from Money Creation

This approach focuses on the asset side of the central bank balance sheet as the assets backing the monetary base are the source of income from money creation. By lending or selling base money to the private sector, the central bank accumulates assets in its balance sheet, as stylized in Figure 7.5.²⁹⁵ Accordingly, the process of base money creation is accompanied by the accumulation of interest-bearing assets on the left-hand side of the balance sheet and a non-interest-bearing liability on the right-hand side. Assets that the central bank accumulates are loans to the private sector (L) as well as public and private securities (A) purchased in the market. The total value of assets is equal to the stock of base money M (comprising banknotes and reserves of commercial banks held with the central bank).

Despite serving as an ideal starting point for elaborating the alternative concept of seignorage, accounting principles and the stylized balance sheet are not well suited to describing the phenomenon of creating wealth in terms of seignorage through the issuance of base money. The reason is that base money is listed on the liability side and the assets received in exchange

²⁹⁴ See Klein and Neumann 1990, Neumann 1992, Lange 1995 and Neumann 1996 for further analyses.

²⁹⁵ Figure 7.5 depicts a stylized central bank balance sheet based on several simplifications. For instance, no gold or foreign exchange reserves are taken into account and non-monetary liabilities as well as central bank equity are neglected. Central bank balance sheets have always been surrounded by some mystique, mainly due to the accounting of the monetary base as a liability despite it de facto being no true economic liability. See STELLA 1997, HAWKINS 2003, MARTÍNEZ-RESANO 2004, BUITER 2007, 2008a, CARUANA 2012, REIS 2013 as well as MILES and SCHANZ 2014 for analyses and discussions of central bank balance sheets. See VERGOTE et al. 2010, BUITER and RAHBARI 2012a, b and REIS 2013 for discussions in the context of the Eurosystem.

А	SSETS	LIABILITIES
	loans (<i>L</i>) purchased securities (<i>A</i>)	monetary base (<i>M</i>) banknotes reserves of commcercial banks

Figure 7.5: Stylized central bank balance sheet

Source: author's illustration.

for base money are listed on the asset side. Therefore, the issuance of base money is neutral from an accounting perspective: as both the asset and the liability side grow simultaneously, no wealth is generated. However, while the assets are remunerated, base money is no true liability (see above). Therefore, the central bank receives interest payments on the assets without any repayment obligation for the liabilities. The asset-side approach to measure the income from money creation is developed as follows. First, the approach is presented under certainty, before uncertainty is introduced as the assets backing the monetary base are subject to default risk. It is shown that seignorage may be volatile and lower than under certainty as potential write-off losses would ceteris paribus imply a permanent deterioration in interest income.

Income from Lending under Certainty

When the central bank grants loans with a nominal value L and duration T, the relation between the present value of all income flows from the loans (V_L) and its nominal value is described by²⁹⁶

$$V_L = i_L L\left(\frac{1 - (1 + \beta)^{-T}}{\beta}\right) + L(1 + \beta)^{-T},$$
(7.6)

with i_L being the interest rate that the central bank charges on the loans.²⁹⁷ With $i_L = \beta$ (i.e. the central bank charges an interest rate that is equal to the discount rate), the present value of interest payments from the loan corresponds to its nominal value, i.e. $V_L = L$. However, if the central bank charges an interest rate lower than the discount rate, i.e. $i_L < \beta$, the present value of interest payments is smaller than the nominal value, i.e. $V_L < L$. The case $V_L < L$ holds particular interest as the difference between V_L and L can be interpreted as an implicit interest subsidy that the central bank provides to its counterparties by charging an overly-low interest rate.²⁹⁸

The central bank grants a chain of loans in which replacement loans are given upon the maturity of prior loans. Consider the simple chain of one-period loans (i.e. T = 1) as depicted in Figure 7.6.

 $^{^{296}}$ See the appendix for the derivation of Equation (7.6).

²⁹⁷ For the Eurosystem, i_L is given by the interest rate applied in open market operations, i.e. the main refinancing rate (MRR), see VERGOTE et al. 2010. The MRR is uniform to all counterparties such that it does not reflect counterparty risk (see Section 4.2).

²⁹⁸ See e.g. KLEIN and NEUMANN 1990, VAN EWIJK 1992 and BUITER 2007. If the central bank lends at short term and earns interest free of default risk, β can be proxied by the risk-free rate and the interest subsidy is described by $(\beta - i_L) \left[(1 - (1 + \beta)^{-T}) / \beta \right]$. Note that this may only constitute part of the subsidy, whereas the total subsidy is measured by the difference between the interest rate that counterparties would have to pay for a comparable loan in the interbank market and the central bank interest rate. BUITER and RAHBARI 2012b estimate the implicit subsidy from the Eurosystem longer-term refinancing operations as EUR 30 bn per year.

Figure 7.6: Infinite chain of central bank loans under certainty



Let L_0 denote the nominal value of initial loans granted at time t = 0 and L_j with $j = \{1, ..., \infty\}$ the constant nominal values of all replacement loans. The present value of payments generated by the chain of loans, i.e. V_L , is determined by using T = 1 and solving Equation (7.6) recursively such that²⁹⁹

$$V_L = i_L L_0 \left(\frac{1 - (1 + \beta)^{-1}}{\beta}\right) \sum_{j=0}^n (1 + \beta)^{-j} + L_j (1 + \beta)^{-(j+1)}.$$
(7.7)

When the chain of loans is considered to be infinite, i.e. $j \to \infty$, V_L simplifies to

$$V_L = \frac{i_L}{\beta} L_0. \tag{7.8}$$

Therefore, the present value of granting loans to the private sector over an infinite horizon pays the central bank an interest rate stream in terms of a perpetuity. With $i_L = \beta$, this perpetuity is exactly equal to the nominal value of the loan.

Income from Asset Purchases under Certainty

Likewise, the central bank can create base money by purchasing both public and private securities to the amount of A_0 , i.e. government bonds and private securities such as covered bank bonds or asset-backed securities. As the central bank usually acquires assets traded in the secondary market, a market value exists for the assets. This market value can be expressed in terms of the present value of income flows on the securities V_A analogous to above, i.e.

$$V_A = i_A A_0 \left(\frac{1 - (1 + \beta)^{-1}}{\beta}\right) \sum_{j=0}^n (1 + \beta)^{-j} + A_j (1 + \beta)^{-(j+1)}.$$
(7.9)

If securities are infinitely replaced by newly purchased securities (i.e. $T \to \infty$) for the purpose of steady base money creation, the present value simplifies to

$$V_A = \frac{i_A}{\beta} A_0, \tag{7.10}$$

with i_A being the interest rate on purchased assets.

Income from Money Creation under Certainty

It was argued above that accounting principles cannot contribute to revealing the income from money creation. However, they help to explore the relation between the conventional liabilityside approaches presented above and the asset-side approach. Seignorage measured by the

 $^{^{299}}$ See the appendix for the derivation of Equation (7.7).

asset-side approach (δ_{as}) comprises the sum of interest flows from both loans and purchased assets in present value terms, i.e.

$$\delta_{as} = V_L + V_A. \tag{7.11}$$

In a static calculation, seignorage corresponds to the present value of interest flows from loans granted and assets purchased to back the current monetary base, as well as all replacement loans and assets. Therefore, seignorage in a static consideration amounts to

$$\delta_{stat}^{as} = V_L + V_A = \frac{i_L}{\beta} L_0 + \frac{i_A}{\beta} A_0 = (i_L L_0 + i_A A_0) \frac{1}{\beta}.$$
(7.12)

In all future periods, the central bank can expand the monetary base to account for e.g. economic growth and receives additional interest-bearing assets. The present value of additional income that these assets generate is taken into account in a dynamic calculation. Under the assumption that the economy grows at the constant rate g, which is smaller than the discount rate, i.e. $\beta > g$, dynamic seignorage δ_{as}^{dyn} comprises static seignorage (i.e. seignorage accruing from today's stock of interest-bearing assets) and the present value of interest payments from loans issued and assets purchased to back future base money issuance, i.e.³⁰⁰

$$\delta_{as}^{dyn} = \delta_{as}^{stat} + \sum_{t=1}^{\infty} \frac{g\left(\frac{i_L}{\beta}L_{t-1} + \frac{i_A}{\beta}A_{t-1}\right)}{(1+\beta)^t} = (i_L L_0 + i_A A_0) \frac{1}{\beta - g}.$$
 (7.13)

Therefore, seignorage in a dynamic calculation accounts for growth in the monetary base, which leads to additional income equal to the present value of interest on loans granted and securities bought until today and in the future. Proposition 7.1 summarizes the results for seignorage under certainty.

Proposition 7.1 (Seignorage under certainty). In a certain world in which the assets backing the monetary base are not subject to default risk, static seignorage measured by the asset-side approach is given by Equation (7.12), and dynamic seignorage, i.e. allowing for future growth of the monetary base, is determined by Equation (7.13). Hence, seignorage under certainty is crucially determined by (i) the value of loans granted and securities purchased, as well as (ii) the interest rate on these loans and securities.

Thus far, it has been assumed that the central bank acts in a certain world, i.e. loans that it grants and assets that it purchases are free of default risk. Accordingly, it was implicitly assumed that the success probability of granted loans and purchased assets was equal to unity. However, in reality, loans and assets are subject to default risk, which has to be borne by the central bank. In this case, $0 \le p_L < 1$ and $0 \le p_A < 1$, with p_L being the success probability of a loan and p_A the success probability of a purchased asset. If future cash flows are uncertain, literature proposes two methods to adjust the present value of interest flows received by the risk-neutral central bank (ROBICHECK and MYERS 1966, RUBINSTEIN 1973, FAMA 1977):

- i) certainty equivalent method,
- *ii)* risk-adjusted discount rates method.

 $^{^{300}}$ See the appendix for the derivation of Equation (7.13).

While the former method replaces expected cash flows by certainty equivalent cash flows in the present value formulation, the latter calls for adjustment of the discount factor. The former method is applied in the following as it follows directly from microeconomic utility theory and matches the purpose of the analysis. Hence, the certainty equivalent CE of the cash flow in each period $t \ CF_t$ is determined, i.e. the payment that makes the central bank indifferent between risk-free and risky cash flows. As the central bank is considered risk-neutral, $CE(CF_t)$ corresponds to the expected value of the cash flow, $\mathbb{E}(CF_t)$, which implies the following expression for the present value of uncertain cash flows:

$$V = \sum_{t=1}^{T} \frac{CE(CF_t)}{(1+\beta)^t} = \sum_{t=1}^{T} \frac{\mathbb{E}(CF_t)}{(1+\beta)^t}.$$
(7.14)

Income from Lending under Uncertainty

Initially, the situation is considered in which the central bank grants uncollateralized loans such that it does not receive anything in case of default. The present value of a T-period uncollateralized loan of value L is expressed in terms of its uncertain payment flows such that the analogous to Equation (7.6) under uncertainty is

$$V_{L} = \sum_{t=1}^{T} \frac{\mathbb{E}(i_{L}L)}{(1+\beta)^{t}} + \frac{\mathbb{E}(L)}{(1+\beta)^{T}} = \sum_{t=1}^{T} \frac{p_{L}^{t}i_{L}L}{(1+\beta)^{t}} + \frac{p_{L}^{T}L}{(1+\beta)^{T}}$$
$$= i_{L}L \left(\frac{p_{L} - (1+\beta)^{-T}p_{L}^{1+T}}{1+\beta - p_{L}}\right) + p_{L}^{T}L(1+\beta)^{-T},$$
(7.15)

with p_L being the constant success probability.³⁰¹

Consider a chain of one-period loans, i.e. T = 1, in which the initial loan of value L_0 is repeatedly replaced. The initial loan L_0 is granted to the counterparty with the probability of success p_L . Whenever the loan does not default and is fully repaid, the central bank can inflation-neutrally grant a new loan in the same nominal value, abstracting from economic growth for the moment. However, when the loan defaults, the central bank is left with nothing and no replacement loan is granted. The chain of loans is depicted in Figure 7.7.

Figure 7.7: Infinite chain of uncollateralized central bank loans under uncertainty



The present value of this chain of loans V_L is determined analogously to Equation (7.7), i.e.³⁰²

$$V_L = i_L L_0 \left(\frac{p_L - (1+\beta)^{-1} p_L^2}{1+\beta - p_L} \right) \sum_{j=0}^n p_L^j (1+\beta)^{-j} + p_L^{j+1} L_j (1+\beta)^{-(j+1)}.$$
(7.16)

³⁰¹ The case of certainty is mimicked for $p_L = 1$, i.e. Equations (7.15) and (7.6) coincide.

 $^{^{302}\,\}mathrm{See}$ the appendix for the derivation of Equation (7.16).

With an infinite chain of loans $(j \to \infty)$, V_L simplifies to

$$V_L = i_L L_0 \frac{p_L}{1 + \beta - p_L}.$$
(7.17)

Partial derivation of V_L with respect to p_L gives

$$\frac{\partial V_L}{\partial p_L} = i_L L_0 \frac{(1+\beta)}{(1+\beta-p_L)^2} > 0, \tag{7.18}$$

which represents the marginal effect of a variation in the success probability on the present value of the loan. The relation is strictly positive, which reveals that the present value increases in the success probability of the loan.

Thus far, uncollateralized lending has been examined, although previous analysis however showed that NCBs grant refinancing credit against collateral. In case of loan default, the central bank is left with collateral, which can be liquidated for compensation. Moreover, collateral assets are subject to default risk. Let q indicate the credit risk of collateral, i.e. q is the success probability of collateral. Let $\mathbb{E}(\Phi)$ be the expected collateral value, i.e. the value that the central bank expects to receive from liquidating the collateral. For now, this value is assumed to be L if the collateral survives but zero otherwise, i.e.

$$\Phi = \begin{cases} L & \text{with } q; \\ 0 & \text{with } 1 - q. \end{cases}$$
(7.19)

The infinite chain of collateralized loans and its relation to the path of the monetary base is depicted in Figure 7.8. The path of the monetary base is fully determined by the initial loan L_0 as (if loans were the only means of base money creation) the monetary base is equal to the nominal value of the loan. Therefore, the monetary base soars with the granting of L_0 and remains constant in this static consideration, i.e. the initial loan is replaced by loans of equal volume. In the static case and irrespective of loan default, the monetary base is equal to the nominal value of the initial loan. Whether the present value of interest payments is also equal to the nominal value (and hence the monetary base) depends on loan default: if loans are repaid in every period, the present value of interest payments is equal to the monetary base. The same is true for loan default if the nominal value can be recovered from collateral liquidation. However, if both the loan and the collateral default, the central bank would lose all current and future interest payments.





The present value of interest payments on the initial loan is equal to

$$V_L = p_L \left(\frac{i_L L_0}{1+\beta}\right) + (1-p_L)q \left(\frac{i_L L_0}{1+\beta}\right).$$
(7.20)

It has been assumed thus far that if collateral survived, the expected liquidation value would be exactly equal to the interest payment. However, in reality, the liquidation value may fall short of the nominal value of the loan, e.g. by the fraction $1 - \kappa$. Therefore, the liquidation value of collateral changes to

$$\Phi = \begin{cases} \kappa L \text{ with } q; \\ 0 \text{ with } 1 - q, \end{cases}$$
(7.21)

with $0 \le \kappa \le 1$ reflecting liquidity of collateral (see above) or the recovery rate, respectively. Hence, the present value of interest payments on the initial one-period loan L_0 becomes

$$V_L = p_L \left(\frac{i_L L_0}{1+\beta}\right) + (1-p_L)q \left(\frac{\kappa i_L L_0}{1+\beta}\right).$$
(7.22)

Figure 7.9 illustrates the infinite chain of collateral central bank loans, the present value of interest payments from it and the path of the monetary base. Without default, the present value of interest flows coincides with the path of the monetary base. The dashed curve represents all potential liquidation values κL_1 in case of loan default. A situation with $\kappa L_1 < L_1$ is depicted in dark blue, confirming that a one-time write-off to the amount of $(1 - \kappa)L_0$ on the loan leads to a permanent loss in interest income. While the path of the monetary base is unaffected (being equal to the present value of interest income that the central bank would have received in the absence of default), the present value of interest income is permanently reduced since replacement loans to the amount of $\kappa L_1 < L_0$ can be granted after default without causing inflation risk. In case of default in future periods, the loss in seignorage would be aggravated.

The present value of interest payments of the n-period chain of one-period collateralized loans under uncertainty reads as

$$V_{L} = \underbrace{i_{L}L_{0} \frac{p_{L} + (1 - p_{L})q\kappa}{(1 + \beta)}}_{\text{initial loan}} + \underbrace{i_{L}L_{0} \frac{p_{L}^{2} + p(1 - p_{L})q\kappa + p_{L}(1 - p_{L})q\kappa + (1 - p_{L})^{2}q^{2}\kappa^{2}}{(1 + \beta)^{2}}}_{\text{first replacement loan}} + \dots$$

$$= i_{L}L_{0}\sum_{j=1}^{n} \frac{(p_{L} + (1 - p_{L})q\kappa)^{j}}{(1 + \beta)^{j}}.$$
(7.23)

The consideration of an infinite chain of refinancing loans (i.e. $j \to \infty$) simplifies V_L to³⁰³

$$V_L = i_L L_0 \frac{p_L + (1 - p_L)q\kappa}{1 + \beta - p_L - (1 - p_L)q\kappa}.$$
(7.24)

³⁰³ Plugging in $p_L = 1$ replicates the case of certainty, i.e. Equation (7.8).



Figure 7.9: Infinite static chain of collat. central bank loans under uncertainty; $\kappa \in [0, 1]$

Income from Asset Purchases under Uncertainty

The central bank can also buy public and private assets in the market to create base money. Equation (7.10) describes the present value of purchased securities, respectively, under uncertainty as a perpetuity of interest payments, i.e.

$$V_A = i_A A_0 \frac{p_A}{1 + \beta - p_A},\tag{7.25}$$

with p_A being the success probability of a purchased security. A comparison of Equations (7.25) and (7.17) indicates that the present value of purchased assets is equivalent in its structure to that of an uncollateralized loan.

Income from Money Creation under Uncertainty

Seignorage under uncertainty is determined analogously to the case under certainty described above. Static seignorage from granting collateralized loans and purchasing public and private securities that are subject to default risk reads as

$$\delta_{as}^{stat} = V_L + V_A = i_L L_0 \frac{p_L + (1 - p_L)q\kappa}{1 + \beta - p_L - (1 - p_L)q\kappa} + i_A A_0 \frac{p_A}{1 + \beta - p_A}.$$
(7.26)

In a dynamic consideration, the central bank realizes interest income to the amount of

$$\delta_{as}^{dyn} = \delta_{as}^{stat} + \sum_{t=1}^{\infty} \frac{g \left(i_L L_{t-1} \frac{p_L + (1-p_L)q\kappa}{1+\beta - p_L - (1-p_L)q\kappa} + i_A A_{t-1} \frac{p_A}{1+\beta - p_A} \right)}{(1+\beta)^t}$$

$$= \left(i_L L_0 \frac{p_L + (1-p_L)q\kappa}{1+\beta - p_L - (1-p_L)q\kappa} + i_A A_0 \frac{p_A}{1+\beta - p_A} \right) \frac{\beta}{\beta - g}.$$
(7.27)

The amplified importance of counterparty and collateral quality for a growing economy is revealed by Figure 7.10, which is the counterpart to Figure 7.9 for the dynamic case in which the monetary base (which is assumed to be generated by collateralized lending only, for clarity) steadily grows in lockstep with the economy. While write-off to the initial or any replacement loan implied a permanent yet constant loss in interest income in the static case, the loss is no longer constant in the dynamic case. As the present value of interest income without default would grow in line with the path of the monetary base, any write-off implies a loss that aggravates in every period.

Figure 7.10: Infinite dynamic chain of collat. central bank loans under uncertainty; $\kappa \in [0, 1]$



Proposition 7.2 outlines the results of the analysis of the income from money creation that can be claimed by the central bank under uncertainty.

Proposition 7.2 (Seignorage under uncertainty). In an uncertain world in which the assets backing the monetary base are subject to default risk, static seignorage measured by the asset-side approach is given by Equation (7.26) and dynamic seignorage by Equation (7.27). Under uncertainty, seignorage is additionally determined by (i) counterparty default risk, (ii) collateral default risk and (iii) collateral liquidity risk.

7.3.1.3 Evaluation of Income from Money Creation for the Eurosystem

The previous section identified five determinants of interest income from money creation, i.e. (i) the volume of collateralized loans and securities granted/purchased by the central bank; (ii) the interest rate on these loans and securities; (iii) the default risk of loans and purchased securities; (iv) the default risk of collateral; and (v) the liquidity risk of collateral. This section further elaborates on the determinants in two ways: first, general effects of variations in the determinants on the present value of interest income are derived; and second, the development of the determinants is evaluated for the Eurosystem.^{304,305}

Volume and Interest Rate of Collateralized Loans and Purchased Securities

Comparative statistics with respect to variations in volume and interest rate give

$$\frac{\partial \delta_{as}^{stat}}{\partial L_0} = i_L \frac{p_L + (1-p_L)q\kappa}{1+\beta - p_L - (1-p_L)q\kappa} > 0; \quad \frac{\partial \delta_{as}^{stat}}{\partial i_L} = L_0 \frac{p_L + (1-p_L)q\kappa}{1+\beta - p_L - (1-p_L)q\kappa} > 0;$$

³⁰⁴ General effects are derived for static seignorage under uncertainty, as given by Equation (7.26). Qualitative results hold analogously for dynamic seignorage. The growth rate of the economy is another determinant of seignorage in a dynamic consideration, although this is not at discretion of the central bank.

³⁰⁵ The evaluation of determinants is largely restricted to collateralized loans as necessary information on purchased securities is not available.

$$\frac{\partial \delta_{as}^{stat}}{\partial A_0} = i_A \frac{p_A}{1+\beta-p_A} > 0; \qquad \qquad \frac{\partial \delta_{as}^{stat}}{\partial i_A} = A_0 \frac{p_A}{1+\beta-p_A} > 0.$$

Changes in both the volume and the interest rate positively affect seignorage.³⁰⁶ Moreover, changes in volume or interest rate have ceteris paribus larger marginal effects on seignorage for collateralized loans than for purchased securities.³⁰⁷ This is particularly important as the central bank has discretion to determine the interest rate in collateralized loans. VERGOTE et al. 2010 emphasize that the volume of loans and the interest rate have been crucial for the development of central bank income in the Eurosystem in recent years. Figure 2.2 revealed that the Eurosystem considerably increased liquidity provision in terms of collateralized loans and security purchases, which should have boosted income. However, extended liquidity provision by collateralized lending was accompanied by a historically low interest rate (cf. Figure 2.6).

Counterparty Risk

Variation in counterparty risk involved in loans and purchased securities bears the following marginal effects on seignorage:

$$\frac{\partial \delta_{as}^{stat}}{\partial p_L} = i_L L_0 \frac{(1+\beta)(1-q\kappa)}{(1+\beta-p_L-(1-p_L)q\kappa)^2} > 0;$$
(7.28)

$$\frac{\partial \delta_{as}^{stat}}{\partial p_A} = i_A A_0 \frac{(1+\beta)}{(1+\beta-p_A)^2} > 0.$$

$$(7.29)$$

Equations (7.28) and (7.29) reflect the importance of the risk-hedging effect of collateral. The acquisition of securities de facto inherits the same risk implications as *uncollateralized* loans. The central bank only has to write off *collateralized* loans in case of double default, i.e. the default of both counterparty and collateral, although it already has to write off purchased securities in case of (single) counterparty default.

This general assertion is only valid for the implicit assumption that the success probabilities of the loan and collateral are independent. However, in reality, the success probabilities may be correlated owing to links between the counterparty and the collateral issuer ("wrong-way risk", see Section 4.2). Figure 7.11 shows the course of action in case of a collateralized loan (see also above). The loan is repaid with probability p_L and collateral is irrelevant in such a case. With loan default ("LD"), the central bank has to liquidate collateral in case of collateral survival ("CS") and it has to fully write off the loan otherwise. Figure 7.11 highlights the vital course of events for the conditionality of success probabilities. Three different cases that differ in the extent of wrong-way risk are considered in the following, the first of which recapitulates independence, the second "close links" and the third "own-use".

³⁰⁶ While the numerator is strictly positive, the condition for a positive denominator in case of collateralized loans is $1+\beta > p_L + (1-p_L)q\kappa$, which holds for all $p_L, q, \kappa \in [0, 1]$ and $\beta > 0$. The equivalent condition for purchased securities is $1+\beta > p_A$, which holds for $p_A \in [0, 1]$ and $\beta > 0$.

³⁰⁷ This holds given equal counterparty risk as the numerator is larger and the denominator smaller for collateralized loans than for purchased securities with $q, \kappa \in [0, 1]$ and $\beta > 0$.

Figure 7.11: Course of events for collateralized loan under uncertainty



Case 1 ("independence"): In this case, success probabilities are independent such that the probability of both events LD and CS taking place is given by

$$Prob(CS \cap LD) = Prob(LD) \cdot Prob(CS) = (1 - p_L) \cdot q, \tag{7.30}$$

where q is again the success probability of collateral such that the present value is equal to that derived in Equation (7.24), i.e.

$$V_{1,L} = i_L L_0 \frac{p_L + (1 - p_L)q\kappa}{1 + \beta - p_L - (1 - p_L)q\kappa}.$$
(7.31)

Case 2 ("close links"): This case assumes close links between the counterparty and the collateral issuer. Close links would make collateral default more likely if the counterparty defaulted such that the success probability of collateral is conditional on the counterparty success probability. The conditional probability that collateral survives after loan default is

$$Prob(CS|LD) = \frac{Prob(CS \cap LD)}{Prob(LD)} = \frac{Prob(CS \cap LD)}{1 - p_L} \equiv \chi$$
(7.32)

such that the probability of CS taking place after LD is

$$Prob(CS \cap LD) = (1 - p_L)\chi,\tag{7.33}$$

where χ is the probability of collateral survival after loan default with close links between the counterparty and the collateral issuer. In this case, the present value reads as

$$V_{2,L} = i_L L_0 \frac{p_L + (1 - p_L)\chi\kappa}{1 + \beta - p_L - (1 - p_L)\chi\kappa},$$
(7.34)

which is smaller than the present value in case of independence, i.e. $V_{2,L} < V_{1,L}$, for $\chi < q$. This holds as the success probability of collateral after counterparty default is lower for close links than in case of independence.

Case 3 ("own-use"): Own-use of collateral describes the situation in which the counterparty pledges own-use uncovered bank bonds as collateral. Therefore, collateral defaults accordingly in case of counterparty default. The probability that collateral survives after counterparty default is zero in case of own-use, which corresponds to

$$\chi = 0. \tag{7.35}$$

The present value is subsequently given by

$$V_{3,L} = i_L L_0 \frac{p_L}{1 + \beta - p_L},\tag{7.36}$$

which resembles the present value of an uncollateralized loan, cf. Equation (7.17). This reveals that the risk profile of loans collateralized by own-use uncovered bank bonds is similar to that of uncollateralized loans and asset purchases, respectively, as the role of collateral as risk hedge de facto vanishes.³⁰⁸ Despite collateralization, the central bank already has to write off these loans in case of single default. This indicates how the Eurosystem gradually approved larger exposure to counterparty risk. The analysis shows that counterparty risk is the smallest for collateralized loans in which counterparty and collateral issuer are independent, although it surges and approaches counterparty risk in asset purchases for loans collateralized by own-use uncovered bank bonds. The Eurosystem carried out this course as it increased counterparty risk exposure in two steps: first, it relaxed collateral criteria down to the eligibility of own-use uncovered bank bonds; and second, it initiated asset purchases. The duration of liquidity provision usually distinguishes counterparty risk exposure in collateralized loans from that in asset purchases as collateralized loans are usually short-term while the residual maturity of purchased assets is medium or long-term. However, Chapter 2 revealed that the Eurosystem increased the duration of collateralized loans from initially up to three months to three years and counterparties appreciated this possibility of long-term funding (Figure 2.5). Since this extension, the duration of collateralized loans is only slightly below the average residual maturity of e.g. government bonds purchased under the SMP (Table 2.4). Therefore, the Eurosystem gradually increased its counterparty risk exposure by watering down the risk hedge in collateralized lending and extending the duration of collateralized loans close to that of purchased assets.

The notion that counterparty risk is relevant to the Eurosystem was revealed in the fall of 2008 when five counterparties defaulted on their loans to the total value of EUR 10.3 bn.³⁰⁹ Complementing this anecdotal evidence, Figure 7.12 provides an indication concerning the development of counterparty risk from 2008 to 2014 based on the EU banks credit default swap (CDS) index (5Y).³¹⁰ Despite being a measure for the entire EU banking sector, the index serves as an appropriate proxy to counterparty risk as the Eurosystem gives loans to a very broad range of counterparties.³¹¹ The figure reveals that counterparty risk has been volatile over recent years, increasing during the fall of 2008 and throughout the sovereign debt crisis.

³⁰⁸ See also COUR-THIMANN and WINKLER 2013, CLAEYS et al. 2014 as well as ECB 2015c on comparisons of the (risk) implications of collateralized loans and asset purchases. The financial risk of central bank asset purchases is unraveled in STONE et al. 2011.

³⁰⁹ See ECB 2009, "Eurosystem Monetary Policy Operations in 2008," Press Release, 5 March 2009, SIBERT 2009 as well as BELKE and POLLEIT 2010. According to the ECB, loans under default were mainly collateralized by highly complex ABSs of limited liquidity. The Deutsche Bundesbank was the largest creditor with a total value of EUR 8.5 bn. Liquidation of collateral took more than four years and brought EUR 7.4 bn, leaving the Deutsche Bundesbank as a creditor in the Lehman Brothers bankruptcy proceedings with a residual claim of EUR 1.9 bn (including interest), see Deutsche Bundesbank, "Conclusion of Resolution of Lehman Collateral," Press Release, 20 February 2013, as well as BINDSEIL and JABLECKI 2013. See also Deutsche Bundesbank, "Bundesbank hat letzte Zahlung aus dem deutschen Lehman-Insolvenzverfahren erhalten," Press Release, 10 February 2015.

³¹⁰ The index refers to counterparty risk as it measures the premium that EU banks had to pay to obtain a CDS as protection against default: accordingly, the higher the index, the larger the market anticipation of default.

³¹¹ By the end of 2014, the ECB listed 8,296 monetary financial institutions in the EU, 5,555 of which were deemed eligible counterparties (ECB 2011d). See also Section 2.3.



The figure depicts the development of the EU banks CDS index (5Y) from 2008 to 2014. The index provides an indication concerning the development of counterparty risk to the Eurosystem, suggesting that it increased over time and peaked throughout the financial crisis and the sovereign debt crisis.



Collateral Quality

Counterparty default would only pose a risk to interest income from lending if collateral quality was insufficient. When collateral defaults or is illiquid, the central bank cannot recover its financial position in case of counterparty default, i.e. the position that it would have had if the loan had been repaid. Instead, the central bank has to write off (at least a fraction) of the loan. Hence, collateral quality that manifests in default risk and liquidity risk of collateral is basal. The following analysis considers collateral default risk in terms of the (unconditional) success probability q and collateral liquidity risk in terms of liquidity κ .

Collateral Default Risk The marginal effect of a change in default risk on seignorage is characterized by

$$\frac{\partial \delta_{as}^{stat}}{\partial q} = i_L L_0 \frac{(1+\beta)(1-p_L)\kappa}{(1+\beta-p_L-(1-p_L)q\kappa)^2} > 0.$$
(7.37)

Hence, a rise in default risk (i.e. a decrease in q) implies lower seignorage.³¹² The development of default risk in the Eurosystem was already extensively addressed in Sections 3.3.4 (eligible marketable assets), 5.1 (pledged collateral) and 5.4 (pledged collateral revisited). It was revealed that the default risk of eligible marketable assets increased as of October 2008 when the minimum credit rating threshold was relaxed and that the probability of default of eligible marketable and pledged collateral has substantially increased since then (Figure 5.22). Moreover, it was

³¹² In case of close links, q has to be replaced by χ in Equation (7.37). As $\chi < q$ (see above), the marginal effect is still positive but larger with close links than with independence.

estimated that the credit quality of pledged collateral overproportionally deteriorated owing to the adverse selection of collateral and the increased pledge of non-marketable assets.

Collateral Liquidity Risk The adverse selection of collateral is also relevant for changes in collateral liquidity risk, the marginal effect of which is determined by

$$\frac{\partial \delta_{stat}^{as}}{\partial \kappa} = i_L L_0 \frac{(1+\beta)(1-p_L)q}{(1+\beta-p_L-(1-p_L)q\kappa)^2} > 0.$$
(7.38)

Analogous to default risk, the marginal effect is positive such that a boost of liquidity risk (i.e. a decrease in κ) implies a decrease in seignorage. The development of liquidity risk of eligible marketable assets is illustrated in Figure 7.13. The figure depicts the evolution of the median bid-ask spread of eligible marketable assets from May 2007 to December 2013 as a measure of (il)liquidity.³¹³ The bid-ask spread measures the difference of the ask price over the bid price relative to the ask price in percent. The larger the bid-ask spread, the less liquid the asset. The figure suggests that the liquidity of eligible marketable assets was very volatile with collateral being particularly illiquid throughout 2010 and 2013, as well as at the end of 2011. Hence, liquidity only slightly deteriorated after the outbreak of the financial crisis but considerably throughout the sovereign debt crisis, which can be attributed to the large fraction of government bonds in the eligible collateral pool. Hence, the development resembles that for collateral default risk.

Figure 7.13: Liquidity risk of eligible marketable assets

The figure depicts the development of the median bid-ask spread of eligible marketable assets from 2007 to 2013 as a measure of (il)liquidity. It indicates that the liquidity of eligible marketable assets was very volatile, with eligible assets being particularly illiquid throughout 2010 and 2013 as well as at the end of 2011.



³¹³ On the bid-ask spread as the standard measure of (il)liquidity, see e.g. EDWARDS et al. 2007 and GOLDSTEIN et al. 2007. The author collected bid prices and ask prices from Bloomberg for eligible marketable assets, where available.

This section has revealed that uncertainty adversely affects seignorage and hence central bank finances. Default on refinancing loans and purchased assets imply a permanent deterioration of central bank income. Evaluating the determinants of seignorage for the Eurosystem indicates that it was at risk owing to a combination of (i) a low interest rate, (ii) increased counterparty risk and (iii) deteriorating collateral quality in terms of higher default and liquidity risk. While write-off losses have been limited to date, the deterioration of collateral quality (inter alia owing to adverse selection of collateral) is particularly alarming. While potential write-off losses owing to low-quality collateral would be relatively small in a low interest rate environment, they would not be restricted to the short term but would rather induce a permanent decrease in seignorage. This puts at risk interest income from money creation as a major source of central bank income.

7.3.2 Collateral Criteria and Emergency Liquidity Assistance

This section argues that Emergency Liquidity Assistance (ELA) provided in the Eurozone is closely linked to the Eurosystem collateral framework and intensifies risk to central bank income. ELA is the facility through which NCBs can fund local counterparties that have lost market access and are unable to participate in standard refinancing operations under specific conditions and against collateral.³¹⁴ It is grounded in Article 14.4 of the Statue of the ESCB and the ECB. The article decrees that ELA is "performed on the responsibility and liability of NCBs" (ECB 2012e, p. 237) as it is not considered part of the single monetary policy of the Eurosystem. Therefore, potential losses from ELA are de jure not shared among NCBs (like for ordinary refinancing operations) but fully borne by the NCB that provides it. Moreover, the article determines that the ECB Governing Council can veto the provision of ELA by a majority of two-thirds. Further specifications for ELA procedures were published by the ECB in October 2013, although they aimed at ensuring adequate information flow within the Eurosystem, i.e. from NCBs to the ECB, rather than improving public disclosure of ELA (ECB 2013b).

ELA is presumably provided against different conditions than conventional refinancing loans as the responsible NCB has discretion over the maturity, interest rate, collateral criteria and the haircut (albeit which the ECB can veto). While the interest rate is supposed to be higher than in conventional refinancing operations as a penalty mark-up is charged,³¹⁵ collateral criteria are supposed to be laxer; otherwise, ELA would be unnecessary if the counterparty possessed sufficient collateral complying with the uniform collateral criteria under the full allotment of refinancing credit. This makes the collateral framework and ELA inherently linked. On the one hand, the collateral framework steers collateral availability, whereby low collateral availability owing to a strict framework makes it more likely that counterparties strive for ELA. On the other hand, the responsible NCB has discretion to determine collateral criteria for ELA such that a relaxed uniform collateral framework would imply even lower collateral criteria for ELA.

³¹⁴ See BUITER et al. 2011, LATTUGA and VALLI 2012, SINN 2012, 2014b, ECB 2013b, ILLING and KÖNIG 2014 and WHELAN 2014b for further information and analyses of ELA. See FUEST and SINN 2015 for a discussion of the risk implications.

³¹⁵ This mark-up (difference between the ELA interest rate and the MRR) is shared among NCBs, see ECB 2011a and SINN 2014b.

The NCBs of Belgium, Germany, Greece, Ireland and Cyprus have most likely granted ELA in recent years. Accordingly, they have supported local banks that lacked collateral complying with the Eurosystem collateral framework despite its extensive relaxation.³¹⁶ With NCBs of Greece, Ireland and Cyprus, these were NCBs of crisis-stricken countries for which the previous analysis suggested temporary collateral scarcity (see Chapter 1). Tracking the development of the ELA volume is difficult as its quantity is not specifically disclosed, whereby only a proxy is available from monthly financial statements of NCBs.³¹⁷ The approximated development is illustrated in Figure 7.14. January 2008 is used as the reference month to capture the effects of the crises. The Deutsche Bundesbank and the Banque Nationale de Belgique provided ELA in specific situations for limited periods: the Deutsche Bundesbank provided EUR 35 bn to the Hypo Real Estate and the Banque Nationale de Belgique to Fortis to the peak amount of EUR 51.3 bn. ELA given by the Bank of Greece surged to a level slightly above EUR 50 bn during the second half of 2011 and jumped to EUR 109.4 bn in February 2012 when the Eurosystem temporarily revoked the suspension of the minimum credit rating for collateral linked to the Greek government (Section 3.2.2). ELA in Greece developed in a volatile manner thereafter but remained on a high level throughout 2012 and declined until the end of 2014. The Central Bank of Ireland provided ELA almost exclusively to the Irish bad bank, i.e. the National Asset Management Agency (NAMA), such that it was given earlier and on a smaller scale (peak amount of EUR 70.3 bn in February 2012). Finally, the Central Bank of Cyprus provided ELA as of April 2012 culminating to EUR 11.4 bn in March 2013. After the Eurosystem waived the minimum credit rating requirement for Cypriot government-linked collateral in May 2013, provision of ELA steadily contracted in Cyprus.

ELA bears risk to central bank finances as it is granted to financially-stricken banks against collateral of quality lower than stipulated by the Eurosystem collateral framework. Therefore, collateral risk (and likewise counterparty risk) is deemed to be higher for ELA than in standard refinancing operations. The tolerance of higher collateral risk is justified by the de jure provision that NCBs grant ELA on their own responsibility, i.e. they have to bear potential losses. This has two implications: on the one hand, it poses substantial risk to the financial strength of the respective NCB, which makes recapitalization more likely; and on the other hand, it constitutes a de jure provision that may not be sustainable in reality. The liability of NCBs is de facto limited to their capital in addition to their entitlement to the present value of interest income of the Eurosystem (neglecting potential minimum reserves, see SINN 2014b). Accordingly, write-off losses on ELA in terms of permanent foregone interest income (see above) would be shared among NCBs if they exceeded the liability amount of the NCB that has granted ELA.

Hence, ELA potentially affects not only the financial strength of the NCB that provided it and de jure bears all losses from the provision, but also the other NCBs owing to the de facto limited liability of the responsible NCB. The general reasoning of the previous section that income from money creation is put at risk by relaxations of collateral criteria is fostered for ELA. As ELA

³¹⁶ Data for the entire Eurosystem suggests that the ECB tolerated ELA to the peak total amount of EUR 250.6 bn (June 2012). At the end of 2014, ELA amounted to 58.7 bn.

³¹⁷ ELA was accounted as the balance sheet item "other claims on Euro Area credit institutions denominated in euro" after April 2012 when accounting standards were harmonized. Prior to this harmonization, ELA was classified as "other claims on Euro Area credit institutions denominated in euro" and "other assets."

Figure 7.14: Approximated provision of Emergency Liquidity Assistance by NCBs

The figure approximates the development of ELA provided by NCBs from 2008 to 2014. January 2008 is the reference month to reflect the effects of the crises on the provision of ELA. The figure shows that the Deutsche Bundesbank and the Banque Nationale de Belgique provided ELA for limited periods, while in particular the Bank of Greece and the Central Bank of Ireland gave ELA over longer periods and on a larger scale.



is provided against collateral of even lower quality, any relaxation of uniform collateral criteria implicitly lowers the collateral criteria imposed in ELA and fosters risk to the interest income from money creation.³¹⁸

7.3.3 Collateral Criteria and Intra-Eurosystem Claims/Liabilities

Intra-Eurosystem balances are a further issue for which collateral criteria are key and which imply risk to central bank finances in the Eurosystem.³¹⁹ Claims and liabilities of NCBs towards the ECB were relatively balanced prior to the financial crisis but subsequently turned into huge intra-Eurosystem imbalances. The imbalances mainly stem from Target claims and liabilities that measure the relocation of refinancing credit across Eurozone countries, reflected in the balance sheets of NCBs.³²⁰ The relaxation of collateral criteria was key to the accumulation of Target balances, enabling NCBs to liberally and fully allot refinancing loans to liquidity-seeking

³¹⁸ However, relaxations of uniform collateral criteria similarly reduce the likelihood that ELA has to be provided. Relaxations sustain de jure risk-sharing from ordinary refinancing operations rather than transforming it into de facto risk-sharing under ELA (owing to limited liability of NCBs).

³¹⁹ Intra-Eurosystem balances are closely related to the previous issue of ELA as the provision of the latter gives rise to the accumulation of the latter, see FUEST and SINN 2015.

³²⁰ Target is the acronym for Trans-European Automated Real-Time Gross Settlement Express Transfer System (Target), i.e. the internal payment settlement system of the Eurosystem. Its growing importance during the crisis was unveiled by SINN and WOLLMERSHÄUSER 2012 and is comprehensively discussed in SINN 2012, 2014b. Target balances are not addressed to the extent warranted here as focus on interest here is placed upon the fiscal implications of intra-Eurosystem balances, of which Target balances are one component. Therefore, important issues like causes of Target imbalances and their interpretation are neglected. For further discussions, see SCHLESINGER 2012, NEUMANN 2012, BERNHOLZ 2012, BINDSEIL et al. 2012, BINDSEIL and KÖNIG 2012, BINDSEIL and WINKLER 2013, PISANI-FERRY 2013, AUER 2014, POTRAFKE and REISCHMANN 2014 as well as WHELAN 2014a. A comprehensive survey of the literature is provided in SINN 2012, 2014b and COUR-THIMANN 2013.

banks. Hence, the presence of Target balances is closely related to the exceptional measures taken by the Eurosystem, i.e. the full allotment of very long-term refinancing credit at a fixed rate against low-quality collateral (see e.g. SINN 2012, 2014b, SINN and WOLLMERSHÄUSER 2012, NEUMANN 2012 and COUR-THIMANN 2013).

However, Target balances are just one component of intra-Eurosystem balances recorded in the balance sheet of NCBs. Intra-Eurosystem balances comprise several claims/liabilities, most importantly net claims/liabilities related to the allocation of euro banknotes within the Eurosystem besides net claims/liabilities arising from Target accounts.³²¹

Intra-Eurosystem claims and liabilities resulting from underproportional or overproportional banknote issuance are similar to Target claims and liabilities in many ways, despite not being included in official national balance-of-payments statistics (WHITTAKER 2011, SINN 2012, 2014b, SINN and WOLLMERSHÄUSER 2012). The over-/underproportional issuance of banknotes measures the relation between banknotes entered into circulation by an NCB and the amount of banknotes that the NCB is conceded to issue according to its share in ECB capital. In case of overproportional issuance, an outflow of physical cash to other countries is alleged, which is comparable to the outflow of book money measured by a Target claim. Hence, the overproportional issuance of banknotes incurs an intra-Eurosystem liability while an underproportional issuance of banknotes incurs an intra-Eurosystem claim.

Figure 7.15 illustrates the development of intra-Eurosystem claims and liabilities for selected countries. Positive values indicate a claim while negative values correspond to a liability. Prior to the financial crisis, Target balances were close to zero such that intra-Eurosystem balances developed stably on a small scale. The German liability originated from overproportional banknote issuance and the French claim from underproportional one.³²² Since the outset of the financial crisis, the development of intra-Eurosystem claims and liabilities has become more disperse, with Germany turning its liability into a huge claim. While liabilities were concentrated to Greece and Ireland until mid-2011, Spain and Italy converted their initial claim into a liability, which steadily grew until mid-2012. Subsequently, intra-Eurosystem claims and liabilities contracted again, although the figure reveals enormous imbalances that still prevailed at the end of 2014.

The imbalances are noteworthy as they involve liabilities of NCBs against the ECB, which are subject to potential write off. The ECB may have to write off its claim against an NCB if the latter cannot settle its liability, e.g. because the country is insolvent. This is independent from whether the insolvent country would remain in or exit the Eurozone as intra-Eurosystem claims and liabilities remain in place (SINN 2015b).³²³ However, government default would adversely

³²¹ Other claims are participating interest in the ECB, claims equivalent to the transfer of foreign reserves to the ECB and other net claims with the Eurosystem. Other liabilities are liabilities related to promissory notes backing the issuance of ECB debt certificates as well as other net liabilities within the Eurosystem (ECB 2012c).

³²² According to SINN and FEIST 1997 as well as SINN 2014b, Germany's overproportional banknote issuance resulted from the enormous stock of deutschmark banknotes that circulated outside of Germany prior to the introduction of the euro. Seignorage wealth from this stock was shared in the Eurosystem and counted as a liability of the Deutsche Bundesbank. Therefore, overproportional banknote issuance developed apace with economic growth. According to COUR-THIMANN 2013, Germany has overproportional issuance because extra Eurozone demand for banknotes is often addressed to German banks, as well as because German tourists withdraw cash at home and spend it abroad.

 $^{^{323}}$ Nevertheless, potential losses to NCBs may differ in both cases, see SINN 2015b.



The figure details the development of intra-Eurosystem balances for selected countries from 2002 to 2014. Positive values indicate a claim while negative values correspond to a liability. Target balances were close to zero prior to the financial crisis such that intra-Eurosystem balances developed stably on a small scale. Huge liabilities were accumulated thereafter by crisis-stricken countries.



affect NCB finances as government bonds, which are largely pledged as collateral (see Figure 5.2), would lose in value. This would increase the likelihood that the borrowing NCB fails to meet its intra-Eurosystem liability such that the ECB would have to write off the liability.³²⁴ As NCBs would have to share the ECB loss, NCB default on its intra-Eurosystem liability would adversely affect the other NCBs in terms of an adverse shock to their balance sheet.

Consider the stylized case of two countries: the NCB of country 1 has an intra-Eurosystem claim, while the NCB of country 2 bears a corresponding intra-Eurosystem liability.³²⁵ The stylized balance sheet of country 1's NCB is depicted in the left panel of Figure 7.16, while the right panel details the stylized balance sheet of country 2's NCB.³²⁶ If the NCB could not redeem its liability towards the ECB, the NCB of country 1 would participate in this loss and it would have to write off a fraction of its claim, corresponding to the gray area. This write-off loss would wipe out capital of country 1's NCB and it would be left with liabilities exceeding assets, i.e. negative capital.

This example reveals how partial write-offs of intra-Eurosystem claims may already weaken central bank finances, which is relevant owing to the magnitude of intra-Eurosystem imbalances relative to central bank capital and risk cushion (see Section 7.2). In the following, it is discussed

³²⁴ See e.g. SINN 2014a,b, WHELAN 2014a and SINN 2015b for discussions and calculations of different scenarios of government default and exits of the Eurozone.

³²⁵ Accordingly, the implicit assumption is made that the other NCBs have a zero intra-Eurosystem claim and liability, respectively.

³²⁶ The balance sheets in Figure 7.16 are extended in comparison to that given in Figure 7.5 by "other assets" and "other liabilities". Other central bank assets are e.g. foreign exchange reserves as well as gold while other liabilities are e.g. non-monetary liabilities such as central bank securities.

ASSETS

SSETS	NCB of c	country 1 LIABILITII	ES	ASSETS	NCB of a	country 2	LIABILITIES
loans purchased secur	rities	monetary base banknotes reserves of commcercial banks	3	loans purchased securitie	es	monetary base banknotes reserves of com	mcercial banks
other assets intra-Eurosystem net claim		other lightlities				other liabilities	
		positive capital	-	other assets		intra-Eurosystem net liability	
and the second sec			-				

Figure 7.16: Stylized central bank balance sheet with intra-Eurosystem claim/liability

how the central bank could strive for recapitalization in case of weakened finances, thus adversely affecting the government budget.

7.4 Central Bank Finances and Central Bank Recapitalization

The relaxation of collateral criteria implied risk to central bank interest income from money creation (Section 7.3.1), which is reinforced by the adverse selection of collateral. Moreover, the relaxation of collateral criteria facilitated the accumulation of intra-Eurosystem imbalances, which augment the risk of potential write-offs (Section 7.3.3). Despite the relaxation of collateral criteria, a lack of eligible collateral can lead to NCBs providing ELA to distressed banks against the collateral of even lower quality, which intensifies risk to interest income from money creation (Section 7.3.2). These three risks to the central bank balance sheet could—individually or jointly—trigger a shock to central bank finances. In the first-best scenario, the shock does not materialize. If the shock materialized and the balance sheet was impaired, central bank capital would vanish. In this case, the central bank could either operate with zero or negative capital and withstand potential adverse consequences or strive for rebuilding capital.³²⁷ The central bank can boost its finances in two ways:³²⁸ on the one hand, it can retain income to bolster its capital, referred to as *internal* recapitalization; and on the other, it can ask the government for support in terms of *external* recapitalization.³²⁹ Moreover, the shock could be diverted from the central bank balance sheet and assumed by a third party ("circumvented recapitalization").

7.4.1 Internal Recapitalization

Thus far, the analysis has implicitly assumed that the central bank transfers the entirety of its income from money creation to the government. However, the transfer may fall short of total income as central banks usually retain some fraction of income, e.g. for the accumulation of capital and a risk cushion (see Section 7.1). Retention may be particularly high when central bank finances are distressed and the central bank aims to bolster capital via internal recapital-

³²⁷ MARTÍNEZ-RESANO 2004 cherishes that adequate central bank capital is paramount for effective central bank independence and ultimately monetary stability. See also Section 7.1 for the importance of central bank capital.

³²⁸ See VAEZ-ZADEH 1991 and LEHMBECKER 2009 for overviews of remedies for central bank losses. The central bank could also lower its operating cost which are however bounded to zero.

³²⁹ Of course, external and internal recapitalization may take place jointly. Qualitative effects would be equivalent, i.e. a negative shock to the government budget, such that separate examination simplifies the analysis without loss of generality.

ization. The retention of central bank income when abstracting from sources other than money creation can be expressed as

$$\delta_{as} = \Gamma + \nu, \tag{7.39}$$

with Γ being the transfer from the central bank to the government (see Chapter 6) and ν the part of central bank income that is retained. Thus far, it has been implicitly assumed that $\nu = 0$ such that $\delta_{as} = \Gamma$, i.e. total seignorage was transferred to the government. In case of internal recapitalization, the central bank transfers the amount Γ_{int} that is smaller than the initial transfer Γ (i.e. without retention) and retains income to the amount of

$$\nu = \delta_{as} - \Gamma_{int} \ge 0. \tag{7.40}$$

How the transfer from the central bank to the government (and hence income retention) is determined varies across countries as distribution rules for central bank income differ. HAWKINS 2003 summarizes that distribution can be laid down in *(i)* law/statute, it can be at *(ii)* the discretion of the central bank or *(iii)* the government or subject to *(iv)* negotiation between the central bank and the government. Most central banks transfer more than half of their profits to the government (ibid., KURTZIG and MANDER 2003, STELLA 2005).³³⁰ Vice versa, central banks retain about half of their profits to bolster financial strength.

However, internal recapitalization is associated with three flaws. First, it may be beyond power of the central bank when profit retention is prohibited by law. Second, the bolstering of central bank capital via profit retention can be too slow as it takes place gradually. According to e.g. MARSHALL 2003, the restoration of a central bank balance sheet could last several decades. While this may be adequate in some cases, it may be insufficient if central bank financial weakness is pressing. Finally, financial means that are available for internal recapitalization may be restricted either explicitly by law or implicitly by the central bank's inflation target (STELLA 1997, SIMS 2005, CUKIERMAN 2011, DEL NEGRO and SIMS 2014).³³¹ According to LÖNNBERG and STELLA 2008, inflationary pressure from extending the monetary base to generate additional income could render the central bank "policy insolvent", i.e. it can only assure long-run profitability by increasing the monetary base at a rate inconsistent with the inflation target. Hence, the central bank may find itself in the paradoxical situation in which it tries to reestablish credibility by strengthening its financial position while undermining credibility owing to induced inflationary pressure (GOODFRIEND 2007). STELLA 1997 argues that the central bank could offset the inflationary pressure by offering assets from its own portfolio bearing a market return in exchange for base money. However, such sterilization would balance additional income and could be restricted by central bank's supply of liquid assets. Moreover, the central bank could

³³⁰ In a survey of 44 central banks' accounting practices, KURTZIG and MANDER 2003 find that central banks transferred from 0% to more than 60% of profits to the government. Moreover, the authors claim that an increasing number of central banks predetermine declarable profits to support government revenue.

³³¹ DEL NEGRO and SIMS 2014 develop a model in which central bank's inflation objectives and the behavior of interest income under high inflation are crucial in determining whether the central bank can finance itself via higher interest income or if it is in need of external recapitalization. Hence, there are clear limits to the central bank's ability to credibly commit to an inflation target in the absence of a fiscal anchor for the central bank, as shown in SIMS 2005.

issue own debt instruments, although this would put it at risk of accumulating unsustainable debt.

7.4.2 External Recapitalization

Let financial resources available for internal recapitalization be (explicitly or implicitly) restricted to $\bar{\nu}$ while resources necessary to restore central bank finances amount to ν_{reg} . The central bank has to ask for government support if $\bar{\nu} < \nu_{req}$, i.e. if the income that the central bank could retain would be insufficient. In this case, the government is expected (but is usually not legally obliged) to provide support, which gives rise to a contingent implicit liability.³³² If the contingent implicit liability is called, the government can meet it by diverting existing funds or issuing additional government debt. In both cases, external recapitalization would result in a transfer from the government to the central bank such that central bank holdings of government debt increase (IZE 2005). However, external recapitalization is associated with shortcomings (see STELLA 2005). It is related to loss of central bank independence, i.e. the central bank becomes reliant on the government.³³³ CUKIERMAN 2011 argues that maintaining a sufficiently high level of capital is essentially a (partial) insurance against states of nature in which the central bank's ability to resist the pressure of political authorities is weakened. The author asserts that the relation between central bank independence and the level of central bank capital is likely to be discontinuous in the sense that below a certain threshold of negative capital, the central bank will be seriously limited by the government. Furthermore, there is usually no legal obligation for the government to cover losses of the central bank, despite legal provisions for how central bank income is allocated between the central bank and the government.³³⁴ Irrespective of legal obligations for the government to cover losses, the government's ability to do so, especially on a timely basis, is called into question by e.g. STELLA 1997, 2005, LÖNNBERG and STELLA 2008. The government's ability could also be limited by complex budgetary procedures that restrict discretion to provide means to the central bank and the potential coincidence of states of the world in which central bank finances are in distress with those of fiscal distress.

7.4.3 Circumvented Recapitalization

A third, less obvious form of recapitalization is one in which potential losses do not affect the central bank balance sheet as they are assumed by a third party, e.g. a supranational institution. Such circumvented recapitalization took place in the Eurozone, whereby financial means of fiscal rescue packages in e.g. Greece and Ireland were used to redeem refinancing loans, which reduced the exposure of the Eurosystem (SINN 2015a). Furthermore, an intergovernmental

 ³³² For instance, Article 33.2 of the Statute does not consider external recapitalization of the ECB (ECB 2012e).
 ³³³ See among others STELLA 1997, BUITER 2008b, LÖNNBERG and STELLA 2008, CUKIERMAN 2011, PAPI 2011 as well as HALL and REIS 2015. See BERGER et al. 2001 for a survey.

³³⁴ See HAWKINS 2003 as well as LÖNNBERG and STELLA 2008 for overviews of legal treatments of central bank profits and losses. ibid. examines data on 135 central bank laws with respect to central bank recapitalization provisions. The authors show that even in cases in which the government is legally obliged to maintain central bank finances, it may do so in a purely cosmetic fashion. VAEZ-ZADEH 1991 criticizes the lack of arrangements with respect to central bank losses, which are thus often ignored until the sheer size of the problem renders ignoring it impossible.

rescue package worth EUR 10 bn was jointly provided by the European Stability Mechanism (ESM) and the IMF to Cyprus in April 2013. The aim of this rescue package was to protect the Eurosystem from write-off losses on ELA granted by the Central Bank of Cyprus to the already-insolvent Laiki Bank (see SINN 2014a,b). While this may quantitatively alter the sharing of losses by governments, qualitative effects remain unchanged if a supranational institution backed by one or a multitude of governments assumes risks and diverts the potential shock from the central bank balance sheet, i.e. the government has to ultimately bear the cost.

7.5 Optimal Level of Collateral Criteria from a Fiscal Perspective

Collateral criteria bear a fiscal implication via their effect on central bank finances. Strict criteria imply low risk and stable income from money creation. For the Eurosystem, strict collateral criteria also hamper the accumulation of large Target balances and implicitly specify relatively strict collateral criteria for ELA. By contrast, loose criteria give rise to higher risk and unstable income with adverse effects on central bank finances. Potential adverse effects are fostered for the Eurosystem as loose collateral criteria facilitate the accumulation of large Target balances and imply loose collateral criteria for ELA. Therefore, the relaxation of collateral criteria amplifies the likelihood that the central bank calls for financial support, which would impair the government budget, irrespective of whether central bank recapitalization is internal, external or circumvented. The following elaborates on the fiscal implication of collateral criteria from a fiscal perspective, i.e. the fiscally desired level of collateral criteria.^{335,336}

The fiscally desired level of collateral criteria is characterized as the solution to the trade-off between two opposing effects that amendments to collateral criteria bear on the government budget. On the one hand, a relaxation of collateral criteria affects the interest rate on government debt (Section 6.5.1), while on the other hand it impinges on central bank bailout (Sections 6.5.2, 7.3, 7.4). Consider a relaxation of collateral criteria in terms of a lowering of the minimum credit rating threshold \hat{q} .³³⁷ First, this would lower the interest rate on government debt. Relaxed collateral criteria free up collateral that banks can use to draw additional refinancing loans from the central bank (as observed for the Eurosystem e.g. in early 2012, see Sections 2.3 and 6.5.1). The more collateral that is freed up, the more additional refinancing loans can be drawn. Additional liquidity is (at least partially) invested in government bonds, whereby this increased

³³⁵ Of course, the fiscal perspective on central bank collateral criteria is only one of many as collateral criteria can also be considered in terms of monetary policy transmission and financial stability (BINDSEIL 2014). However, the fiscal perspective must not be left out of consideration as central bank collateral criteria are first and foremost a means of risk mitigation to the central bank, the finances of which are closely intertwined with the government budget.

³³⁶ Consideration of this fiscal implication of collateral criteria acknowledges the partial regime shift from monetary dominance to a (non-Ricardian) regime of quasi-fiscal dominance observable in the Eurozone, whereby the Eurosystem became a major crisis manager (Chapter 2 and Section 6.2). Monetary dominance would imply that the central bank ignores the fiscal implication of collateral criteria on the government budget.

³³⁷ Accordingly, collateral criteria are approached in terms of credit quality, i.e. *q*. Likewise, they could be interpreted in terms of e.g. liquidity.

demand drives up the bond price and lowers the interest rate and government's borrowing cost ("borrowing cost effect"), i.e.³³⁸

$$i \equiv i(\hat{q}),\tag{7.41}$$

with $\partial i/\partial \hat{q} > 0$, $\partial^2 i/\partial \hat{q}^2 > 0$, $\partial^3 i/\partial \hat{q}^3 = 0$. The interest rate *i* positively depends on \hat{q} and this effect is assumed to be convex. Accordingly, the initial lowering of \hat{q} would reduce the interest rate to a larger extent than any further lowering as it would free up more eligible assets.³³⁹ Second, the previous sections addressed in detail the notion that a relaxation of collateral criteria increases the likelihood of central bank bailout ("bailout cost effect"). Central bank bailout is considered in terms of a one-time contingent implicit liability to the government, i.e. the Treasury has to divert resources to the amount of c^i to the central bank with probability

$$p^i_{\lambda} \equiv p^i_{\lambda}(\hat{q}). \tag{7.42}$$

The probability p_{λ}^{i} negatively depends on \hat{q} , i.e. a lower \hat{q} increases the likelihood that the government has to bail out the central bank such that $\partial p_{\lambda}^{i}/\partial \hat{q} < 0$ with $\partial^{2} p_{\lambda}^{i}/\partial \hat{q}^{2} > 0$, $\partial^{3} p_{\lambda}^{i}/\partial \hat{q}^{3} = 0$. Hence, the likelihood is assumed to grow overproportionally with a lower \hat{q} , i.e. the initial lowering of \hat{q} only slightly increases p_{λ}^{i} but any further lowering leads to an overproportional increase as central bank finances are further impaired owing to imperfect risk control and the adverse selection of collateral. Moreover, the better the risk control, the lower runs p_{λ}^{i} (and $\partial p_{\lambda}^{i}/\partial \hat{q}$) as additional collateral risk from lowering collateral criteria is hedged to a larger extent, which makes central bank bailout less likely. According to Section 6.4, the expected value of the contingent implicit liability of central bank bailout reads as

$$\mathbb{E}[c^i] = p^i_\lambda(\hat{q})c^i. \tag{7.43}$$

Taking into account the two opposing effects of an amendment to collateral criteria, the primary balance that ensures fiscal sustainability hitherto given by Equation (6.18) reads as^{340}

$$pb = i(\hat{q})d + p_{\lambda}^{i}(\hat{q})c^{i}.$$

$$(7.44)$$

The remainder of this section characterizes the fiscally desired level of collateral criteria based on Equation (7.44) in two different environments, i.e. (i) one with one country and one central bank and (ii) one with two countries and one central bank (as a stylized form of monetary union).

³³⁸ See ACHARYA and STEFFEN 2015 and ASONUMA et al. 2015 on the negative relationship between domestic government bond holdings and government borrowing cost. See e.g. CRUCES and TREBESCH 2013 as well as CHAMON et al. 2014 on determinants of government borrowing cost.

³³⁹ Figure 4.6 provides anecdotal evidence on the distribution of marketable assets according to credit quality. Compare the hypothetical scenario in which the Eurosystem would lower the minimum credit rating threshold first from "A-" to "BBB+" and subsequently from "BBB+" to "BBB". Given that the first lowering would free up more eligible assets than the second lowering, the alleviating effect of the lowering on the interest rate would be stronger for the initial amendment.

³⁴⁰ It is abstracted from any positive transfer from the central bank to the government, i.e. $\gamma = 0$. Moreover, economic growth and inflation are neglected, which simplifies the analysis along two dimensions: first, the real interest rate is approximately equal to the nominal interest rate; and second, potential crowding-out of private investment, which would affect economic growth, can be ignored.

7.5.1 Fiscally Desired Level of Collateral Criteria with One Country and One Central Bank

First, the fiscally desired level of collateral criteria is examined in the institutional setting with one country and one central bank, i.e. there is one country, whose Treasury backs the central bank. This setting resembles the situation that prevailed e.g. in Eurozone countries prior to the introduction of the euro. Moreover, it would describe the situation of the Eurozone as a fiscal union in which the common Treasury backs the common central bank. The fiscally desired level of collateral criteria evolves from balancing the two opposing effects of an amendment to \hat{q} . It is given by

$$\hat{q}^* = \underset{\hat{q}}{\operatorname{arg\,min}} pb(\hat{q}). \tag{7.45}$$

 \hat{q}^* is the credit rating threshold that minimizes the primary balance that has to be generated to keep fiscal policy sustainable. Hence, the central bank is assumed to support the government in fulfilling its present-value budget constraint (PVBC), which implies a deviation from the Ricardian regime of monetary dominance towards a (non-Ricardian) regime of (quasi-) fiscal dominance (Section 6.2). The optimal solution \hat{q}^* to the trade-off is characterized by

exp. marginal bailout cost

$$\frac{\partial i}{\partial \hat{q}^*} d = - \frac{\partial p_{\lambda}^i}{\partial \hat{q}^*} c^i$$
(7.46)
exp. marginal borrowing cost

and is visualized in Figure 7.17. In the optimum, the level of collateral criteria balances the expected marginal borrowing cost of amending \hat{q} to the expected marginal bailout cost of doing so and the minimum primary balance that ensures the sustainability of fiscal policy is implemented.³⁴¹ Therefore, the central bank can amend collateral criteria to affect the fiscal position.





 $^{341}\,\mathrm{For}$ the given functional forms, \hat{q}^* is unique.

7.5.2 Fiscally Desired Levels of Collateral Criteria with *Two* Countries and One Central Bank

Second, the institutional setting is analyzed in which two countries delegate monetary policy to a common central bank, i.e. the central bank implements a level of collateral criteria in terms of a minimum credit rating \hat{q} that is uniform across both countries. Divergent preferences over the level of collateral criteria from a fiscal perspective emerge in such an environment for two reasons. On the one hand, countries are considered heterogeneous with respect to their debt level, which implies different interest rates that they have to pay on their debt. On the other hand, the countries contribute asymmetrically to potential central bank bailout. This situation resembles a stylized form of the current Eurozone in which countries are heterogeneous with respect to their fiscal positions and would share central bank bailout costs unequally. The fiscal implication of collateral criteria differs for the two countries depending on the uniform level of \hat{q} implemented by the common central bank. For the Treasury of country $l \in \{1, 2\}$, the individually optimal level of collateral criteria is determined by $\hat{q}_l^* = \arg \min_{\hat{q}} p b_l(\hat{q})$, which gives

$$\frac{\partial i_l}{\partial \hat{q}_l^*} d_l = -\frac{\partial p_\lambda^i}{\partial \hat{q}_l^*} k_l c^i, \tag{7.47}$$

with k_l being the country-specific share according to which the Treasury of country l would have to contribute to the bailout of the common central bank.³⁴² The fiscally desired level of collateral criteria can differ across countries owing to differences in

- *i)* the expected marginal borrowing cost: the expected marginal borrowing cost given on the left-hand side of Equation (7.47) differs across countries owing to differences in the interest rate (i_l) and debt ratio (d_l) ;
- *ii)* the expected marginal bailout cost: the expected marginal bailout cost reflected on the right-hand side of Equation (7.47) differs across countries owing to different contributions to central bank bailout (k_l) .

Asymmetric Marginal Borrowing Cost The two countries are asymmetrically affected by amendments to collateral criteria when they feature distinct fiscal situations (captured by the debt ratio and the interest rate). This is exemplified for the case in which country 2 bears a higher debt ratio and a higher interest rate than country 1 such that $\frac{\partial i_1}{\partial \hat{q}}d_1 < \frac{\partial i_2}{\partial \hat{q}}d_2$ for any given \hat{q} . For $k_1 = k_2$, the left-hand side of Equation (7.47) differs for the two countries, which implies $\hat{q}_1^* \neq \hat{q}_2^*$. The left panel of Figure 7.18 illustrates that in this case the Treasury of country 2 prefers a lower level of collateral criteria than the Treasury of country 1. Consider the case in which the fiscal situation in both countries is initially equal and corresponds to that of country 1 in the left panel of Figure 7.18. Subsequently, country 2 experiences a shock to its fiscal position, which shifts expected marginal borrowing cost upwards for any level of \hat{q} . In this case, the initial level of collateral criteria given by \hat{q}_1^* is no longer optimal for country 2 as the expected marginal borrowing costs exceed the expected marginal bailout costs. When \hat{q} is relaxed, the expected

³⁴² For the Eurosystem, k_l can be interpreted as the capital key of an NCB, i.e. the fraction according to which it contributed to ECB capital, with $\sum_l k_l = 1$. In the two-country setting, $k_1 + k_2 = 1$.

marginal borrowing costs decrease while the expected marginal bailout costs increase until \hat{q}_2^* is characterized by $\frac{\partial i_2}{\partial \hat{q}_2^*} d_2 = -\frac{\partial p_\lambda^i}{\partial \hat{q}_2^*} k_{1=2} c^i$ in the optimum. This level of collateral criteria now desired by the Treasury of country 2 is lower than the initial level, which is still desired by the Treasury of country 1, i.e. $\hat{q}_2^* < \hat{q}_1^*$.

Asymmetric Marginal Bailout Cost Divergent preferences over \hat{q} may also originate from differences in the expected marginal bailout cost of amending collateral criteria. A bailout of the common central bank constitutes a joint contingent implicit liability to the Treasuries, i.e. the Treasuries share the risk of central bank bailout. The expected marginal bailout costs from amending \hat{q} are equal to both Treasuries when they are liable for the bailout in equal shares. In this case, both Treasuries would ceteris paribus prefer the same level of \hat{q} . However, if the Treasuries did not share bailout costs equally, the Treasury bearing the smaller cost would ceteris paribus prefer a lower level of \hat{q} . While this Treasury fully enjoys the benefits of lower collateral criteria, it underproportionally contributes to a potential bailout.³⁴³ This situation is illustrated in the right panel of Figure 7.18, with the Treasury of country 2 contributing less to potential bailout than the Treasury of country 1, i.e. $k_1 > k_2$. Both Treasuries prefer a lower level of collateral criteria compared to the situation in which they would bear the entire cost (characterized by the intersection of $\frac{\partial i_{1=2}}{\partial \hat{q}} d_{1=2}$ and the dotted line). For this level of \hat{q} but with sharing of bailout costs, the expected marginal borrowing costs would exceed the expected marginal bailout costs for both Treasuries, i.e. $\frac{\partial i_{1=2}}{\partial \hat{q}} d_{1=2} > \frac{\partial p_{\lambda}^{i}}{\partial \hat{q}} k_{l} c^{i}$. Optimality is achieved by trading off the expected marginal borrowing costs and expected marginal bailout costs, i.e. by lowering \hat{q} down to $\frac{\partial i_{1=2}}{\partial \hat{q}_{i}^{*}} d_{1=2} = \frac{\partial p_{\lambda}^{i}}{\partial \hat{q}_{i}^{*}} k_{l} c^{i}$ with $\hat{q}_{1}^{*} > \hat{q}_{2}^{*}$. The intuition is straightforward. Although central bank bailout is a joint contingent implicit liability, Treasuries are asymmetrically liable. The beneficial effect of a lower level of collateral criteria in terms of lower interest payments on government debt is fully enjoyed by each Treasury at asymmetric costs. It is optimal for the Treasury of country 2 to prefer a lower level of collateral criteria while the Treasury of country 1 prefers a higher level, implying a lower probability of central bank bailout.



Figure 7.18: Fiscally desired levels of collateral criteria: two countries and one central bank

The extent of risk-sharing among countries depends on \hat{q} , i.e. the uniform level of collateral criteria implemented by the central bank. The stricter \hat{q} , the smaller the extent of risk-sharing. Figure 7.19 illustrates the extent of risk-sharing among two countries with distinct fiscal positions

³⁴³ See SINN 2014b for another description of this externality.

and asymmetric sharing of central bank bailout cost for three levels of collateral criteria. In the first case, the central bank chooses \hat{q}_1^* as the uniform level such that the extent of risk-sharing is rather small. In the second case, the central bank implements \hat{q}_2^* and risk-sharing is extended. The third case represents a situation in which the central bank implements the level of collateral criteria that a social planner with utilitarian preferences would choose from a fiscal perspective, i.e. the social planner solves $\hat{q}_{sp}^* = \arg \min_{\hat{q}} \sum_l p b_l(\hat{q})$.³⁴⁴ This optimal level of collateral criteria \hat{q}_{sp}^* is characterized by $\sum_l \frac{\partial i_l}{\partial \hat{q}_{sp}^*} d_l = -\frac{\partial p_{\lambda}^i}{\partial \hat{q}_{sp}^*} c^i$, i.e. total expected marginal borrowing costs are balanced to total expected marginal bailout costs in the optimum. Figure 7.19 shows that \hat{q}_{sp}^* is in between \hat{q}_1^* and \hat{q}_2^* and the extent of risk-sharing is mediocre as it is larger than for \hat{q}_1^* but smaller than for \hat{q}_2^* .

Figure 7.19: Uniform level of collateral criteria and extent of risk-sharing



In this context, the fiscal implication of collateral criteria originates from the effect of the uniform level \hat{q} on the sustainability-ensuring primary balance pb_l for $l = \{1, 2\}$. This primary balance is given for both Treasuries and the three addressed levels of \hat{q} in Table 7.1: (i) the level preferred by the Treasury of country 1, i.e. \hat{q}_1^* ; (ii) the level preferred by the Treasury of country 2, i.e. \hat{q}_2^* ; and *(iii)* the level that the social planner would prefer, i.e. \hat{q}_{sp}^* with $\hat{q}_2^* < \hat{q}_{sp}^* < \hat{q}_1^*$. The last column seizes upon the fiscal implication of collateral criteria by comparing the sustainabilityensuring primary balance pb_l for the two countries and the three possible levels of collateral criteria. The table indicates that each Treasury would have to generate the lowest primary balance if the central bank implemented the optimal level of collateral criteria for the respective Treasury. By contrast, if the central bank implemented the level preferred by the respective other Treasury, both Treasuries would have to generate the highest primary balance. If the central bank chose the social-planner level, collateral criteria would be too lax for the Treasury of country 1 and too strict for that of country 2. Accordingly, the central bank can enforce risk-sharing and transfer fiscal sustainability across countries depending on the initial level of \hat{q} . The relaxation of collateral criteria from \hat{q}_1^* to \hat{q}_{sp}^* would increase risk-sharing and transfer fiscal sustainability from country 1 to country 2 since the sustainability-ensuring primary balance decreases in country 2 but increases in 1. Analogously, an increase in collateral criteria from

³⁴⁴ Note that the case with a social planner differs from that with a fiscal union between the countries, whereby a single fiscal authority would back the central bank in the latter case. Accordingly, the case of fiscal union would resemble the situation with one country and one central bank described in Section 7.5.1. See BORDO et al. 2011, FUEST and PEICHL 2012, DOLLS et al. 2014 for further information on fiscal integration and fiscal union in the context of the Eurozone.
\hat{q}_2^* to \hat{q}_{sp}^* would reduce risk-sharing and transfer fiscal sustainability from country 2 to country 1. This illustrates how collateral criteria can be used as a means to affect fiscal positions and transfer fiscal sustainability across countries. In a monetary union of heterogeneous countries, the central bank can improve the fiscal situation of one country at the expense of another via the choice of the level of collateral criteria.

l	\hat{q}_1^*	\hat{q}_2^*	\hat{q}_{sp}^{*}	FISCAL IMPLICATION
1	$pb_1(\hat{q}_1^*)$	$pb_1(\hat{q}_2^*)$	$pb_1(\hat{q}_{sp}^*)$	$pb_1(\hat{q}_1^*) < pb_1(\hat{q}_{sp}^*) < pb_1(\hat{q}_2^*)$
2	$pb_2(\hat{q}_1^*)$	$pb_2(\hat{q}_2^*)$	$pb_2(\hat{q}_{sp}^*)$	$pb_2(\hat{q}_2^*) < pb_2(\hat{q}_{sp}^*) < pb_2(\hat{q}_1^*)$

Table 7.1: Fiscal implications of different levels of \hat{q}

Source: author's compilation.

Interpretation for the Eurozone

The Eurosystem defined uniform collateral criteria with the introduction of the Single List in January 2007 (Section 3.1). Section 3.2 revealed that uniform collateral criteria were relaxed and that the Eurosystem deviated from uniformity after the onset of the financial crisis. The financial crisis also triggered substantial increases in the debt ratios of Eurozone countries, culimating in the sovereign debt crisis. Interest rates that Eurozone governments had to pay on their debt converged prior to but spread during the sovereign debt crisis (e.g. SINN 2012, 2014b). Within the simple technical framework, the sovereign debt crisis induced a shock to borrowing costs, which differed across countries and induced asymmetric changes to levels of collateral criteria preferred from a fiscal perspective. Different preferences were amplified by disparate contribution rates to central bank bailout (cf. right panel of Figure 7.18) in the Eurosystem. Some countries suffering from the largest surge in their debt ratio and interest rate were among those that would have only had to contribute little to a central bank bailout.

Fiscally desired levels of collateral criteria for selected Eurozone countries are stylized in Equation (7.48). The fraction on the right-hand side of Equation (7.48) implicitly characterizes desired levels based on Equation (7.47) in terms of k_l and d_l . c^i is considered equal to unity for simplicitly but without loss of generality:

$$-\frac{\partial i_l/\partial \hat{q}_l^*}{\partial p_{\lambda}^i/\partial \hat{q}_l^*} = \frac{k_l}{d_l}.$$
(7.48)

As, ceteris paribus, a large k_l and/or a small d_l implies a preference for strict collateral criteria, countries with large k_l/d_l prefer strict collateral criteria while countries with low k_l/d_l desire loose collateral criteria. The development of the implicitly characterized fiscally preferred level of collateral criteria is stylized in Figure 7.20 for selected Eurozone countries from 2007 to 2014. It suggests substantial differences in preferences and that preferences varied slightly from 2007 to 2014. As k_l remained relatively stable,³⁴⁵ variations in preferences were driven by

³⁴⁵ Paid-up shares changed five times but only slightly in January 2008 (Cyprus and Malta joined EMU), January 2009 (Slovakia joined EMU), January 2011 (Estonia joined EMU), July 2013 (Croatia joined EU) and January 2014 (Latvia joined EMU).

asymmetric developments in public indebtedness. The figure indicates that Germany preferred the strictest collateral criteria, while Greece, Ireland and Portugal had a preference for loose collateral criteria.

Figure 7.20: Indication on the fiscally preferred level of collateral criteria in selected countries

The figure seizes upon the development of fiscally preferred levels of collateral criteria by detailing the evolution of preferences implicitly characterized by Equation (7.48). Countries with large values tend to prefer relatively strict collateral criteria, while those with low values desire relatively loose collateral criteria. It suggests differences in preferences across countries, which varied slightly from 2007 to 2014.



Source: author's calculation; European Central Bank; Eurostat.

The intuition can be exemplified by a comparison of the indicators for Germany and Portugal. The indicator was 0.46 for Germany and 0.036 for Portugal in 2007. As both countries had comparable debt ratios at that time (63.7% in Germany and 68.4% in Portugal), differences in the contribution to a potential central bank bailout accounted for the substantial difference in the implicitly characterized preference over the level of collateral criteria: while Germany would have had to pay 29.52% of such a bailout, Portugal would have only had to contribute 2.47%. By the end of 2014, the implicitly characterized preference decreased to 0.34 in Germany and 0.019 in Portugal. Although the contribution rate for Germany dropped to 25.72% in 2014 while it remained relatively stable in Portugal (2.49%), the development is attributed to asymmetric evolutions of fiscal positions, given that the debt ratio increased by 11 percentage points (74.7% in 2014) in Germany and surged by 61.8 percentage points (130.2% in 2014) in Portugal.

The effects of asymmetric developments in fiscal situations in Germany and Portugal on \hat{q}_l^* are stylized in Figure 7.21. Initially, expected marginal borrowing costs were almost identical for country 1 (Germany) and country 2 (Portugal) as interest rates were approximately equal $(i_1 \approx i_2 \text{ owing to interest-rate convergence prior to the crisis, cf. SINN 2012, 2014b})$, while the debt ratios only slightly differed $(d_1 < d_2)$. However, the expected marginal bailout costs differed as both countries would have had to contribute to a central bank bailout to differing extents $(k_1 > k_2)$. This implies that Portugal already preferred laxer collateral criteria than Germany prior to the crisis i.e. $\hat{q}_2^* < \hat{q}_1^*$. The crisis-induced shock to public finances $(d'_2 \gg d_2, i'_2 \gg i_2)$ is stylized to have hit Portugal such that the expected marginal borrowing costs were shifted upward.³⁴⁶ This induced a decrease in the level of collateral criteria preferred by Portugal to $\hat{q}_2^{**} < \hat{q}_2^* < \hat{q}_1^*$. The range of fiscally desired levels of collateral criteria within which the common central bank specifies uniform criteria widens owing to the crisis-induced drop in the level preferred by Portugal. Depending on the uniform level of collateral criteria, Germany could be worse off owing to the deterioration of Portugal's fiscal position. If the central bank acted e.g. as an utilitarian social planner, a shock to the fiscal situation in country 2 would result in a lower level of uniform collateral criteria, thus implying a larger extent of risk-sharing (see above). This lower level of collateral criteria would improve the fiscal situation in country 2 at the expense of the fiscal situation in country 1. As argued above, fiscal sustainability would be transferred across countries.

Figure 7.21: Sovereign debt crisis and fiscally desired levels of collateral criteria



This qualitative discussion of the (cross-country) fiscal implication of amendments to collateral criteria has attempted to reason the observable relaxations of the Eurosystem collateral framework throughout recent years. Moreover, the investigation of divergent preferences over collateral criteria across countries exemplified for Germany and Portugal sheds light on the endeavor of several NCBs to relax collateral criteria below the uniform level. This endeavor predominantly manifested in the ACCs framework, in which the Banco de Portugal was one of seven NCBs that deemed eligible assets of quality lower than stipulated by the Eurosystem collateral framework (Section 3.2.1).

The analysis suggests that collateral criteria can serve as a tool to accommodate fiscal shocks rather than mere risk protection.³⁴⁷ The immanent difference between conventional monetary policy tools such as the interest rate and collateral criteria bases on their distinct purpose. For the Eurosystem, the interest rate is a policy instrument, i.e. it does not address counterparty risk but is the same for all counterparties (Section 4.2). Counterparty risk is hedged by imposing restrictions on counterparties (see above) and by stipulating collateral. Relaxing collateral crite-

³⁴⁶ The situation is assumed to have remained unchanged in Germany, where the higher debt ratio was compensated by decreases in the interest rate.

³⁴⁷ See BINDSEIL 2013 for an assessment concerning how central banks' collateral framework can be interpreted as a policy instrument to sustain financial stability and implement monetary policy rather than as risk protection. See also CASSOLA and KOULISCHER 2014, KOULISCHER and STRUYVEN 2014, KOULISCHER 2015 as well as WEBER 2015 for elaborated approaches. Moreover, BRUNNERMEIER 2012 proffers to use collateral policy (in terms of haircuts) to improve monetary policy in a suboptimal monetary union.

ria to accommodate fiscal shocks would alienate collateral from its primary purpose of hedging risk (on condition of ensured counterparty access) as the Eurosystem's risk exposure would increase if additional collateral risk was insufficiently addressed by haircuts. In this case, collateral criteria could be amended to accommodate fiscal shocks at the cost of lower risk protection.

Fiscal Implication from Collateral Criteria and Government Guarantees

The previous chapter addressed the cross-country fiscal implication of collateral criteria resulting from divergent preferences over the uniform level of collateral criteria. By contrast, this chapter elaborates on the fiscal implication of collateral criteria evolving from the close relation between collateral criteria and government guarantees to the financial sector. It shows that the Eurosystem collateral framework involves government discretion to free up new collateral by giving explicit guarantees. This shapes the government's incentive to grant explicit guarantees. It is shown that governments contributed to the provision of additional collateral, which was deemed necessary during the financial and sovereign debt crisis to accommodate collateral scarcity. Hence, the collateral framework of the Eurosystem was key to the accumulation of government guarantees, i.e. contingent explicit government liabilities, and intensified the nexus between governments and the financial sector, which is often considered vicious (FARHI and TIROLE 2014, SINN 2014b, ALLEN et al. 2015, NYBORG 2015). The chapter is structured as follows. Section 8.1 elaborates how the Eurosystem collateral framework involves government discretion to create collateral. Stylized facts on government guarantees to eligible marketable assets in the Eurozone over recent years are derived in Section 8.2. Section 8.3 reviews the common rationale for government guarantees, before Section 8.4 finally presents the novel rationale that characterizes government incentives to give explicit guarantees from a fiscal perspective in due consideration of collateral criteria. This novel rationale can contribute to explain the development observable in the Eurozone of explicit government guarantees to eligible marketable assets.

8.1 Government Discretion to Create Collateral

The Eurosystem repeatedly relaxed collateral criteria related to government-guaranteed marketable assets (Section 3.2 and in particular Figure 3.3). Most importantly, it introduced the eligibility of government-guaranteed own-use uncovered bank bonds in February 2009. This amendment is crucial as it made the eligibility of such bonds endogenous to the government, i.e. it gave the government discretion over eligibility and freeing up collateral.³⁴⁸ This is particularly important under full allotment as collateral availability constitutes the de facto limit to borrowing from the Eurosystem. Furthermore, the minimum rating threshold was repeatedly suspended for assets guaranteed by governments of crisis-stricken countries.

Governments have discretion to create collateral as their guarantees can affect collateral in three ways:³⁴⁹ first, the guarantee can be basal for eligibility, e.g. in case of own-use uncovered bank bonds; second, the guarantee augments the collateral value of an eligible asset when it improves the credit assessment deemed decisive for haircut determination; and third, the guarantee augments the collateral value of the asset.

Therefore, the collateral framework involves discretion for governments to free up new collateral, i.e. to create eligible assets that financial institutions can pledge with the Eurosystem. Figure 8.1 indicates whether selected governments have made use of this discretion. The bars indicate the nominal value of newly guaranteed eligible assets (given on the left axis), differentiated by type of guaranteed asset. The change in refinancing credit (relative to the previous month) granted by the respective NCB is also depicted on the left axis. Moreover, the figure measures the average residual maturity of newly guaranteed eligible assets on the right axis.

The figure indicates that governments have made use of their discretion to create collateral via government guarantees to differing extents.³⁵⁰ In Germany, government guarantees were predominantly given to uncovered bank bonds during the financial crisis and later to other marketable assets. The residual maturity of newly guaranteed assets fluctuated around the average of 4.08 years. The Spanish government predominantly guaranteed uncovered bank bonds with relatively low residual maturity (average residual maturity of guaranteed assets between February 2009 and December 2013: 6.65 years) as of February 2009. Moreover, Spanish government guarantees were given to ABSs prior to the financial crisis, to uncovered bank bonds during the financial crisis as well as during the period of extended liquidity provision by the Eurosystem ("big bazooka" period), and to corporate bonds in the beginning of 2013. The panel also reveals that Spanish banks extensively drew credit from the Eurosystem during the "big bazooka" period, a fraction of which was likely collateralized by guaranteed assets that feature an average residual maturity of 4.24 years (January 2012 to March 2012), i.e. slightly above

³⁴⁸ The fact that a fraction of pledged uncovered bank bonds was own-use and hence only eligible in case of government guarantee can be inferred from the Eurosystem's repeated efforts to restrict their pledge (Figure 3.3).

³⁴⁹ See Section 3.1.4.3 and BRENDEL et al. 2015 for how the Eurosystem takes government guarantees into account for the credit assessment of collateral. Moreover, LEVY and SCHICH 2010, LEVY and ZAGHINI 2010 as well as DAVIES and NG 2011 show that the costs of issuing a guaranteed debt instrument are mainly determined by the creditworthiness of its guarantor.

³⁵⁰ The notion that some governments made use of their discretion to provide collateral via guarantees is also documented in SINN 2014b and NYBORG 2015.

Figure 8.1: New government guarantees to eligible marketable assets in selected countries

The figure indicates whether selected governments made use of the discretion to create collateral. The bars reflect the nominal value of newly guaranteed eligible assets on the left axis, differentiated by asset type. The flow of refinancing credit (relative to the previous month) granted by the respective NCB is depicted on the left axis. Moreover, the figure measures the (unweighted) average residual maturity of newly guaranteed eligible assets on the right axis. It indicates that governments made use of their discretion to create collateral via government guarantees to differing extents.



Source: author's calculation; European Central Bank, Eligible Assets Database.

the duration of the three-year LTROs and lower than the overall average (excluding ABSs) of 6.65 years.³⁵¹ In Italy, government guarantees were almost exclusively given to uncovered bank bonds during the "big bazooka" period. As in Spain, Italian banks extensively borrowed from the Eurosystem during this period. The panel suggests that a large portion of new refinancing credit was collateralized by uncovered bank bonds, which were newly eligible with a government guarantee.³⁵² With 3.18 years on average, assets guaranteed between December 2011 and July 2012 had a residual maturity almost matching the duration of the three-year LTROs. Likewise, the Portuguese government extended government guarantees to newly eligible marketable assets

³⁵¹ For this purpose, the Spanish government reintroduced the "Spanish Guarantee Scheme" in January 2012, which was initiated in October 2008, see EC 2012, "State Aid SA. 34224 (2012/N)–Spain. Reintroduction of the Spanish Guarantee Scheme," 9 February 2012.

³⁵² The government of Italy implemented its first guarantee scheme to the financial sector in December 2011, see Clifford Chance, "Italian Government Guarantees. Easing the Pressure on Italian Banks," *Client Briefing*, 23 December 2011.

(primarily uncovered bank bonds) during the "big bazooka" period (esp. in March 2012).³⁵³ Between December 2011 and March 2012, the Portuguese government guaranteed newly eligible assets with average residual maturity of 4.23 years.

Further indication on whether governments made use of the discretion to create eligible collateral is corroborated by Table 8.1, which compares monthly averages of new government guarantees over the pre-"big bazooka" period (June 2006 to November 2011), the "big bazooka" period (December 2011 to March 2012) and the post-"big bazooka" period (April 2012 to December 2013) for selected countries. It shows that all governments aside from the German substantially extended guarantees during the "big bazooka" period.

Table 8.1: Monthly average value of new government guarantees in selected countries

The table compares the monthly average value of new government guarantees to eligible marketable assets for selected countries prior to, during and after the "big bazooka" of the Eurosystem. It illustrates that new government guarantees to marketable assets considerably increased in all countries except for Germany during the "big bazooka" period.

	Monthly Average (Total) ^a				
	Рке-"Від Ваzоока" 06/2007 - 11/2011	"Від Ваzоока" 12/2011 - 03/2012	Post-"Big Bazooka" 04/2012 - 12/2013		
GERMANY	21.1 (1137)	15.1 (60.4)	11.1 (233.7)		
SPAIN	2.7(144.9)	10.7 (42.7)	4.3(89.6)		
FRANCE	2.8(150)	16.9(67.5)	8.5 (178.1)		
GREECE	1.3(72)	4.5 (18.1)	2(41.8)		
IRELAND	5(271.5)	10.5(42)	2.6(53.6)		
ITALY	0.01 (0.8)	31.6(126.5)	0.1 (1.9)		
Portugal	0.6 (30.1)	3.4(13.6)	0.4 (8.5)		

^a in EUR bn; deviations possible owing to rounding.

Source: author's calculation; European Central Bank, Eligible Assets Database.

This development of new government guarantees to eligible marketable assets suggests a close tie between central bank collateral criteria government incentive to grant explicit guarantees. This connection is further elaborated in the following.

8.2 Stylized Facts on Government Guarantees in the Eurozone³⁵⁴

Government support to the financial sector can take the form of deposit insurance, explicit and implicit guarantees as well as general guarantee schemes.³⁵⁵ The present analysis focuses on explicit government guarantees to eligible marketable assets and derives two stylized facts on their development in the Eurozone from 2007 to 2013.

 $^{^{353}}$ The Portuguese government did not set up a new government guarantee scheme as the one implemented in October 2008 remained in place, see MULLER et al. 2012.

 $^{^{354}}$ This section bases on EBERL and WEBER 2015.

³⁵⁵ On the importance of government support to the financial sector, especially in times of financial distress, see DIAMOND and DYBVIG 1983, GORTON 1988, JACKLIN and BHATTACHARYA 1988, ALLEN and GALE 1998 as well as ALLEN et al. 2015. See SINN 2010a, ALLEN et al. 2011, MULLER et al. 2012 as well as ALLEN et al. 2015 for government interventions in the Eurozone during the recent financial crisis. Despite its calming effect, government intervention potentially features high costs as well as moral hazard effects, see Section 6.3 and below.

Figure 8.2 seizes upon the finding of Section 3.3, namely that government guarantees to eligible marketable assets grew in importance between May 2007 and December 2013. The left panel depicts the development of the total nominal value of government-guaranteed eligible marketable assets (line) together with the nominal value of new government guarantees in each month (bars). It reveals that the value of new government guarantees substantially increased throughout 2009 and at the turn of 2011/2012.³⁵⁶ This development explains the swift rise in the stock of government guarantees from EUR 22 bn in May 2007 to EUR 1293 bn in December 2013. The right panel stylizes the development of the total nominal value of government-guaranteed assets relative to GDP for selected countries, showing that the value of government guarantees relative to GDP substantially increased in all countries. For all countries except for Italy and Greece, the increase can be attributed to the onset of the financial crisis in the fall of 2008. This particularly holds for Ireland, where government guarantees to eligible assets reached 60%of GDP.³⁵⁷ Government guarantees increased to about 20% of GDP in Germany as well as Spain and peaked at some 40% in Greece. In Italy, government guarantees did not play a role throughout the financial crisis and the early phase of the sovereign debt crisis. The Italian government started to guarantee eligible marketable assets as of the turn of 2011/2012 (see Section 8.1) to the amount of up to 8% of GDP.

Stylized Fact 8.1: Government guarantees grew in importance. Government guarantees to eligible marketable assets grew in importance in particular during two phases, i.e. during (i) the financial crisis (11/2008 to 12/2009) and during (ii) the phase of increased liquidity provision by the Eurosystem ("big bazooka" period, 12/2011 to 03/2012).

Figure 8.2: Government guarantees to eligible marketable assets

The left panel depicts the development of the stock of government guarantees to eligible marketable assets (line) from 2007 to 2013, together with the flow of government guarantees in each month (bars). The right panel stylizes the development of the stock of government guarantees relative to GDP for selected countries. The figure confirms the finding of Section 3.3.7 that importance of government guarantees for eligible marketable assets has increased over recent years.



Considering the increased importance of government guarantees during the two periods, the remainder of this section investigates the structure of newly guaranteed eligible marketable

³⁵⁶ Additionally, there was a sharp increase in guarantees in January 2008 as the German government guaranteed issuances of government-linked financial institutions.

³⁵⁷ This figure refers to government guarantees to marketable assets eligible with the Eurosystem. Overall, Irish government guarantees to the financial sector rose to about 200% of GDP (cf. ALLEN et al. 2015).

assets in terms of: (i) the eligibility of guaranteed assets, i.e. already vs. newly eligible assets; (ii) the type of guaranteed assets; and (iii) governments that guaranteed assets.

According to the first dimension, government guarantees can be given to (i) assets that were already eligible or (ii) assets that became newly eligible (respective or irrespective of the guarantee). This differentiation is illustrated in Figure 8.3. The figure reveals that at the outset of the financial crisis, a considerable fraction of new government guarantees was directed to already-eligible marketable assets. However, new government guarantees were predominantly granted to newly eligible assets over time and throughout the sovereign debt crisis.

Figure 8.3: Structure of new gov. guarantees: guarantees to already- vs. newly eligible assets

The figure elaborates on the blue bars in left panel of Figure 8.2 and differentiates between government guarantees to (i) assets that were already eligible and (ii) assets that became newly eligible. While a considerable fraction of new government guarantees was directed to already-eligible marketable assets at the outset of the financial crisis, new government guarantees were predominantly granted to newly eligible assets thereafter.



Source: author's calculation; European Central Bank, Eligible Assets Database.

The left panel of Figure 8.4 provides indications concerning the second dimension, i.e. the type of assets that received a new government guarantee. The panel indicates that the vast majority of guarantees were given to uncovered bank bonds. Moreover, corporate bonds and other marketable assets were substantially guaranteed, while covered bank bonds only occupied importance in January 2008 and July 2009. ABSs played a minor role for government guarantees. The right panel of Figure 8.4 provides evidence on the geographical distribution of government guarantees, revealing that the German government gave the lion's share of new guarantees prior to and throughout the financial crisis. New guarantees in considerable value were also granted by the governments of France, Ireland, Spain, Denmark and the UK (the latter two are comprised in "other") throughout 2009. During the European sovereign debt crisis, Greece and Italy also became active in granting guarantees to eligible marketable assets.

Stylized Fact 8.2: The structure of newly guaranteed assets changed. During the financial crisis, government guarantees were given to both already and newly eligible assets

Figure 8.4: Structure of new gov. guarantees: guarantees by asset type and by country

This figure further investigates the blue bars in the left panel of Figure 8.2 in terms of types of guaranteed assets (left panel) and the geographical distribution (right panel). During the beginning of the financial crisis, Germany gave new guarantees predominantly to eligible uncovered bank bonds. Throughout the European sovereign debt crisis, mainly uncovered bank bonds as well as corporate bonds and other marketable assets were guaranteed by governments of crisis-stricken countries.



(39% and 61% of total nominal value, respectively), which were largely uncovered bank bonds (79%), and guarantees were spread across various countries (only 14% of guarantees were granted in countries later stricken by the sovereign debt crisis). During the "big bazooka" period, new government guarantees were still predominantly granted to uncovered bank bonds (77%), although these were newly eligible (85%) and issued in crisis-stricken countries (61%) to a large extent.

8.3 Common Rationale for Government Guarantees

The common rationale for government guarantees dates back to the seminal contribution by DIAMOND and DYBVIG 1983.³⁵⁸ Government guarantees are rationalized as support to illiquid financial institutions. These institutions face a potential run on their deposits owing to panic among depositors, based on the self-fulfilling belief of deposit withdrawal. The government can intervene and provide liquidity insurance to rule out panic-driven bank runs. Accordingly, government guarantees would be a costless and fully effective means to prevent bank runs. However, this result crucially relies on three assumptions: (*i*) bank runs are solely caused by irrational panic, i.e. banks may be illiquid but solvent; (*ii*) guarantee schemes are fully credible as the government possesses sufficient financial means; and (*iii*) guarantees are costless to the government as any positive probability of bank run is ruled out by the announcement of and the commitment to the guarantee.³⁵⁹

However, the recent financial crisis suggests that these assumptions may be misleading. Bank runs may not be driven by irrational panic rather caused by deterioration of fundamentals such

 $^{^{358}\,\}mathrm{See}$ e. g. Allen et al. 2015, 2011 for discussions.

³⁵⁹ Moreover, DIAMOND and DYBVIG 1983 abstract from potential moral hazard effects of guarantees (ALLEN et al. 2015, 2011).

as economic conditions and the value of bank assets, which gives rise to insolvencies of banks.³⁶⁰ Moreover, governments experienced hard times in providing sufficient resources for necessary guarantees.³⁶¹ In this case, government guarantees cannot fully foreclose the possibility of bank runs. Under these circumstances, government guarantees (*i*) may not be fully effective in preventing the occurrence of bank runs and (*ii*) can entail substantial costs, thus rendering them (*iii*) a potential threat to the government budget.³⁶²

Recent literature has investigated the close link between the government budget and financial stability of the financial sector via government guarantees, see COOPER and NIKOLOV 2013 as well as ACHARYA et al. 2014b. The government is considered to have limited resources such that extended government support to the financial sector tightens the government budget. This gives rise to a feedback loop between the government and the financial sector in which any deterioration of the position of the financial sector spills over to the government and vice versa.³⁶³ As government guarantees place pressure on the government budget, beneficiaries of the guarantee start to question the government's ability to honor it. This reduces the credibility of the guarantee and likewise its effectiveness in preventing bank runs. Consequently, instability in the financial sector increases, thus calling for further public intervention.

8.4 Novel Rationale: Importance of Collateral Criteria for Government Guarantees

When the announcement of government guarantees does not preclude the possibility of bank runs, the expected costs and benefits of a guarantee have to be taken into account. The next section presents a novel yet simple rationale for the government decision to guarantee (eligible) assets based on the trade-off between the costs and benefits from a fiscal perspective. Specifically, the approach recognizes the government as an agent with limited resources and considers the government's PVBC as the benchmark (see Section 6.4). Government guarantees are considered a powerful means to mitigate banking crisis as they can calm panic-fueled depositors. Collateral eligibility amplifies this power of government guarantees. Depending on the strictness of collateral criteria, government guarantees can have beneficial effects beyond calming depositors as they augment liquidity in the financial sector when they free up collateral. While the common rationale does not explain why the structure of guaranteed assets has changed over time, the novel rationale can expound that explicit guarantees to eligible marketable assets have gained in importance over time, as well as explaining how the structure of newly guaranteed assets developed. Moreover, while the common rationale can only rationalize government guarantees to illiquid but solvent banks, the novel rationale can also back up government guarantees to liquid banks that seek to restructure borrowing by pledging uncovered bank bonds with the

³⁶⁰ See CHARI and JAGANNATHAN 1988, JACKLIN and BHATTACHARYA 1988 as well as ALLEN and GALE 1998 for this fundamentals-based approach to bank runs. For a comparison of panics vs. fundamentals as the reason for bank runs, see GOLDSTEIN 2012.

³⁶¹ In general terms, Eurozone governments lack the possibility of monetizing guarantees. However, monetizing would be associated with the cost of inflation.

³⁶² Accounting for the possibility that government guarantees can be costly to the government also calls for an examination of the funding structure of the guarantee scheme. As the government cannot generate resources without distortions, the cost may offset any benefit.

 $^{^{363}}$ This feedback loop has intensified in recent years owing to a rise in home bias in government bond holdings (Section 6.5.1).

Eurosystem. Accordingly, this can elucidate the observable growing importance of government guarantees over recent years.

Three types of bonds that the government guarantees can be differentiated: first, bonds that are eligible irrespective of the government guarantee; second, bonds that are only eligible with the government guarantee; and third, bonds that are ineligible irrespective of the government guarantee. The government would grant guarantees to bonds of the second type if it sought to increase liquidity in the financial sector by freeing up collateral.

The approach is elaborated in three steps. First, the expected costs that accrue to the government from the guarantee are examined. It is argued that they differ between guarantees to bonds that are (i) ineligible and (ii) eligible as collateral with the central bank. Second, the benefits of the guarantee to the government are characterized, whereby it is discussed that the benefits likewise differ between guarantees given to eligible and ineligible bonds. Third, the optimal level of government guarantees is characterized by solving the trade-off between the costs and benefits of guarantees. It is shown that governments have strong incentives to guarantee eligible bonds owing to the lower expected costs and larger benefits.

8.4.1 Expected Costs of a Government Guarantee

Government guarantees on pledged assets would only be called in case of "double default", i.e. if both the counterparty and the collateral default. This is illustrated in Table 8.2, in which the independent probabilities of counterparty and collateral success are again denoted by p_L and q, respectively. When only one of the counterparty or the collateral defaults (or neither), the government does not appear on the scene and the payout is zero. The government is called upon in case of double default, which happens with probability $(1 - p_L) \cdot (1 - q)$. With respect to the framework of fiscal sustainability analysis, this probability is subsumed as $p_{\lambda}^e \equiv (1-p_L) \cdot (1-q)$, i.e. p_{λ}^{e} gives the probability that the explicit government guarantee is called and is distributed according to a Poisson process (see Section 6.4). In case of close links between the counterparty and the collateral issuer (see Section 7.3.1), the probability is given by $p_{\lambda}^e \equiv (1-p_L) \cdot (1-\chi)$ as the success probability of collateral χ is contingent on success of the counterparty. For own-use uncovered bank bonds, $\chi = 0$ such that $p_{\lambda}^e \equiv (1 - p_L)$. When the guarantee is called, the government has to pay c^e if the collateral is eligible (and pledged) and \overline{c}^e if the bond is ineligible with the Eurosystem but used as collateral in a private repo. The next two paragraphs argue that expected costs differ, i.e. $\mathbb{E}[\overline{c}^e] \neq \mathbb{E}[c^e]$, depending on whether the guaranteed bond is pledged as collateral in a private or public repo.

Guaranteed Bond Used as Collateral in Private Repo

Figure 8.5 illustrates the course of action for a government-guaranteed bond used as collateral in a private repo. All solid lines represent actions that take place irrespective of counterparty/collateral default. The borrower issues a bond that is guaranteed by the government and pledged in a repo with a private lender for liquidity.³⁶⁴ By contrast, the dashed line depicts action by

³⁶⁴ As in Chapter 5, only the opening leg of the repo is considered for simplicity.

Event	Probability	Payout
counterparty and collateral default	$(1-p_L)\cdot(1-q)\equiv p_{\lambda}^e$	c^e or \overline{c}^e
counterparty defaults but collateral survives	$(1-p_L)\cdot q$	0
counterparty survives but collateral defaults	$p_L \cdot (1-q)$	0
counterparty and collateral survive	$p_L \cdot q$	0

Table 8.2: Probability of government guarantee to be called (case of independence)

Source: author's compilation.

the government that would only be required if the guarantee is called, i.e. it takes place with probability p_{λ}^{e} . In this case, the government pays any outstanding principal and interest to the private lender.





Consider the payment of government l to the private lender to be given by $\overline{c}^e = (1 + i_L)\Phi$, with i_L being the repo rate and Φ principal.³⁶⁵ Hence, the expected cost for government l from guaranteeing this bond are

$$\mathbb{E}[\overline{c}^e] = p_{\lambda}^e \underbrace{(1+i_L)\Phi}_{\overline{c}^e}.$$
(8.1)

Guaranteed Bond Used as Collateral with the Eurosystem

The course of action in case of a government guarantee to a bond that is eligible and pledged as collateral in a public repo with the Eurosystem is depicted in Figure 8.6. Actions are similar except for the potential payment from the Eurosystem to the government. Owing to the sharing of profits and losses in the Eurosystem, the government de facto becomes a beneficiary of the payment that it has to make when the guarantee is called. In this case, it pays principal and interest to the Eurosystem and receives back a fraction of this payment via its NCB ("partial repayment of guarantee").³⁶⁶

Let k_l be the fraction according to which country l participates in profits of the Eurosystem, with $\sum_l k_l = 1$. Hence, the government would have to pay principal and interest, i.e. $(1 + i_L)\Phi$,

 $^{^{365}\,\}Phi$ resembles the collateral value used in previous chapters as haircuts are neglected here.

 $^{^{366}}$ The NCB of country *l* receives a fraction of the Eurosystem's profit according to paid-up capital, see ECB 2012e and Chapter 7. For simplicity, profit retention by the ECB and NCBs is neglected and all payments are assumed to take place in the same period.



Figure 8.6: Course of action for bond pledged in *public* repo

if the guarantee was called but it would de facto receive back a fraction of its payment, i.e. $k_l i_L \Phi$.³⁶⁷ The expected costs of the guarantee are given by

$$\mathbb{E}[c^e] = p_{\lambda}^e \underbrace{(1-k_l)(1+i_L)\Phi}_{c^e}.$$
(8.2)

Comparison of Expected Cost

The comparison of Equations (8.1) and (8.2) reveals that ceteris paribus, $\mathbb{E}[c^e] < \mathbb{E}[\overline{c}^e]$ as $0 < k_l < 1$. This shows that the expected cost from guaranteeing a bond pledged as collateral with the Eurosystem is lower than guaranteeing a bond used as collateral in a private repo. This strengthens the incentive to guarantee assets that comply with the Eurosystem collateral framework.³⁶⁸ Based on data on eligible collateral in the Eurozone, the maximum hypothetical total savings in expected costs from granting new guarantees to assets that can be pledged as collateral with the Eurosystem rather than ineligible assets are estimated for the period from May 2007 to December 2013. Using information on the nominal value of new government guarantees to eligible marketable assets for Φ , the year-average refinancing rate for i_L and the shares of paid-up capital for k_l , maximum total hypothetical savings are estimated at EUR 430.1 bn for Germany, EUR 85.5 bn for France, EUR 33.6 bn for Spain, EUR 23.4 bn for Italy, EUR 6.5 bn for Greece, EUR 6.1 bn for Ireland and EUR 1.8 bn for Portugal.³⁶⁹

8.4.2 Benefits of a Government Guarantee

The government may benefit from granting explicit guarantees to eligible bonds via two channels, i.e. (i) an *implicit guarantee channel* and (ii) a *borrowing cost channel*. The two channels are briefly explained in the following and qualitatively introduced into the technical approach. Hence, qualitative assumptions are made concerning the effects of explicit guarantees without making assertions on quantitative effects.

³⁶⁷ Note that the financial means that the government pays to itself is not restricted to the fraction of interest payments. Over the infinite horizon, the government receives back the fraction of its *entire* payment, i.e. $k_l(1 + i_L)\Phi$, since the payment sustains the monetary base to be backed by interest-bearing assets. The present value of interest payments on these assets over the infinite horizon is equal to principal.

 $^{^{368}}$ Note that an akin narrative holds for government bonds used as collateral with the Eurosystem (see Section 6.5.1).

³⁶⁹ Of course, the actual savings are substantially lower as eligible guaranteed assets have to be pledged and the probability of double default as well as profit retention by the ECB and NCBs have to be taken into account.

Implicit Guarantee Channel

Explicit government guarantee schemes are usually restricted to banks fulfilling certain institution criteria such as systemic importance or a minimum market share and they require minimum bond issue volumes.³⁷⁰ Therefore, institution criteria as well as the minimum bond value imply the prerequisite that banks intending to benefit from an explicit guarantee should be systemically important. The implicit guarantee channel refers to the benefit that the government derives from the alleviating effect that an additional *explicit* guarantee has on the *implicit* guarantee, which systemically important banks already enjoy owing to their TBTF status (Section 6.3). This triggers dynamics within the government liability matrix (Table 6.2) as a contingent implicit liability is transformed to a contingent explicit liability. This effect is considered in the technical analysis by considering the cost of the implicit guarantee c^i as being negatively dependent on Φ , i.e. the larger the explicit guarantee, the lower the implicit guarantee. The expected cost of the implicit guarantee to the banking sector is subsequently given by

$$\mathbb{E}[c^i] = \sigma p^i_{\lambda} c^i(\Phi), \tag{8.3}$$

with $\partial c^i/\partial \Phi < 0$, $\partial^2 c^i/\partial \Phi^2 > 0$ and $\partial^3 c^i/\partial \Phi^3 = 0$ such that c^i decreases with explicit guarantees Φ and this marginal effect decreases the larger that Φ is. Furthermore, $\mathbb{E}[c^i]$ depends on $\sigma \in (0,1)$, which reflects the impatience/panic of depositors. The higher σ , the more impatient depositors are and the lower their belief in the implicit guarantee. Therefore, σ refers to a panic component in the spirit of DIAMOND and DYBVIG 1983. According to ibid., any sufficient explicit guarantee would imply $\sigma = 0$ such that the government would be able to perfectly mitigate the expected cost of the implicit guarantee. However, owing to the drawbacks of government guarantees (e.g. solvency concerns and moral hazard, see Section 8.3), the government is deemed unable to do so and σ is strictly positive.

Borrowing Cost Channel

This channel was already backed up in Section 6.5.1 and elaborated in Section 7.5 in a slightly different context. It originates from the close tie between collateral criteria and government guarantees and the observable development that banks increased their holdings of (domestic) government debt after drawing liquidity from the Eurosystem. This behavior can be explained by several reasons, most importantly risk shifting and moral suasion (Section 6.5.1). Banks in particular engaged in this behavior during the "big bazooka" period in which extended liquidity provision by the Eurosystem was made possible by governments providing collateral to banks in their territory via explicit government guarantees. The negative relationship between domestic government bond holdings and government borrowing cost (see ACHARYA and STEFFEN 2015 and ASONUMA et al. 2015) suggests that governments were able to reduce borrowing costs

³⁷⁰ See MULLER et al. 2012 for a comprehensive survey of government guarantee schemes initiated in Europe in response to the financial crisis. For instance, institution criteria comprise the requirement for the systemic importance of supported credit institutions (see e.g. the Irish government guarantee scheme). For the prerequisite of a minimum market share, see e.g. the Spanish government guarantee scheme, and the requirement of a minimum value, see e.g. the joint guarantee scheme of Belgium, France and Luxembourg.

by granting explicit guarantees.³⁷¹ The beneficial effect of explicit guarantees on government borrowing cost is incorporated into the analysis by allowing for a negative relationship between Φ and the interest rate on government bonds *i* such that^{372,373}

$$i \equiv i(\Phi), \tag{8.4}$$

with $\partial i/\partial \Phi < 0$, $\partial^2 i/\partial \Phi^2 > 0$ and $\partial^3 i/\partial \Phi^3 < 0$. Hence, more explicit guarantees implicitly induce (at least some) additional demand for government bonds, which gives rise to the negative relationship between Φ on i. The marginal effect on i decreases the larger that Φ is.

8.4.3 The Optimal Level of Government Guarantees

The government's decision to grant explicit guarantees evolves around the trade-off of the costs and benefits of the guarantee, which differ between different types of assets, i.e. newly eligible, already-eligible and ineligible assets. For any type of assets, the government would enjoy lower expected costs of the implicit guarantee from the explicit guarantee. Moreover, the government would benefit from a lower interest rate on its debt if the explicit guarantee was given to newly eligible assets. However, the explicit guarantee constitutes an explicit fiscal risk, which would adversely affect its budget if the risk materialized. This potentially adverse effect is larger for ineligible rather than eligible assets, as the expected cost differ.

The application of the framework for analyzing fiscal sustainability (Section 6.4) facilitates characterizing the government's optimal choice regarding the level of explicit guarantees within the given environment. Abstracting for simplicity from economic growth and inflation,³⁷⁴ the stylized sustainability indicator given by Equation (6.18) can be exemplified for *newly eligible* assets as³⁷⁵

$$pb = i(\Phi)d - \gamma + p_{\lambda}^{e}(1 - k_{l})(1 + i_{L})\Phi + \sigma p_{\lambda}^{i}c^{i}(\Phi).$$

$$(8.5)$$

The government's optimal choice with respect to the level of explicit guarantees to newly eligible assets is determined by

$$\Phi^* = \underset{\Phi \in [0,\overline{\Phi}]}{\operatorname{arg\,min}} pb(\Phi). \tag{8.6}$$

³⁷¹ This implies that the assessment of governments' financial standing is unaffected by explicit guarantees, e.g. because the explicit guarantees replaced already-existing implicit guarantees.

³⁷² In the previous chapter, *i* was negatively dependent on the level of collateral criteria. In this chapter, the level of collateral criteria is taken as given but the government can decide on the amount of explicit guarantees Φ . ³⁷³ Only a qualitative but no quantitative statement is made on the relation between Φ and *i*. The negative relationship holds for any (small) additional demand for government bonds from extra liquidity that was freed up by explicit guarantees. Granting the explicit guarantee has per se no effect on the interest rate as government's financial standing is unaffected, e.g. because an implicit guarantee is made explicit.

³⁷⁴ This simplifies the analysis as the real interest rate is approximated by the nominal interest rate and potential crowding-out of private investment is neglected, see Section 7.5.

³⁷⁵ The analysis can be enriched by considering γ to be dependent on Φ . If explicit government guarantees freed up collateral and increased borrowing from the Eurosystem, central bank income would increase, thus resulting in a higher γ . The qualitative results would remain unchanged.

 Φ is assumed to be restricted for newly eligible assets to the interval $[0, \overline{\Phi}]$ with $\overline{\Phi}$ representing the upper limit on the amount of pledgeable government-guaranteed bonds, i.e. $\overline{\Phi}$ is specified by the collateral framework.³⁷⁶ The optimal level of government guarantees is characterized by

$$-\underbrace{\left[\frac{\partial i}{\partial \Phi^*}d + \sigma \frac{\partial c^i}{\partial \Phi^*}p^i_{\lambda}\right]}_{\text{exp. marginal benefit}} = \underbrace{p^e_{\lambda}(1-k_l)(1+i_L)}_{\text{exp. marginal cost}}$$
(8.7)

The government chooses Φ^* by balancing the expected marginal benefit in terms of lower interest payments on its debt (first term on the left-hand side) and a lower implicit fiscal risk (second term on the left-hand side) against the expected marginal cost in terms of an explicit fiscal risk (right-hand side).

Table 8.3 provides an overview of the expected marginal costs and benefits of explicit guarantees to the three assets types. As the expected marginal costs and benefits differ between asset types, optimal levels of explicit guarantees differ. The government has the strongest incentive to guarantee newly eligible assets as it only enjoys the borrowing cost effect by guaranteeing such assets. It has the second strongest incentive to guarantee assets that are eligible irrespective of the guarantee as although the borrowing cost effect vanishes, the expected costs of the guarantee are still lower than for assets that are ineligible. Accordingly, the government has the weakest incentive to guarantee ineligible assets.

Table 8.3: Marginal costs and benefits of explicit government guarantees by asset type

Asset Type	EXP. MARGINAL BENEFITS	EXP. MARGINAL COSTS
NEWLY/RENEWED ELIGIBLE ASSET e.g. own-use uncovered bank bond	$-[\tfrac{\partial i}{\partial \Phi}d+\sigma \tfrac{\partial c^{i}}{\partial \Phi}p_{\lambda}^{i}]$	$p_{\lambda}^{e}(1+i_{L})(1-k_{l})$
ALREADY-ELIGIBLE ASSET e.g. covered bank bond	$-\sigma rac{\partial c^i}{\partial \Phi} p^i_\lambda$	$p_{\lambda}^{e}(1+i_{L})(1-k_{l})$
INELIGIBLE ASSET e.g. subordinated bond	$-\sigma rac{\partial c^i}{\partial \Phi} p^i_\lambda$	$p^e_\lambda(1+i_L)$

Source: author's compilation.

Optimal levels of explicit guarantees to the different asset types are illustrated in Figure 8.7, which reflects the expected marginal costs and benefits as summarized in Table 8.3. The expected marginal benefit of guaranteeing newly eligible assets is given by the downward-sloped curve comprising both the borrowing cost effect and the implicit guarantee effect. The downward-sloped line represents the expected marginal benefit of guarantee effect. The downward-sloped assets and reflects only the implicit guarantee effect. The upper horizontal line gives the (higher) expected marginal cost of guaranteeing ineligible assets while the lower horizontal line depicts the (lower) cost of giving an explicit guarantee to eligible assets. The figure confirms the finding that the optimal level of explicit guarantees is the largest for newly eligible assets, followed by already-eligible assets and finally ineligible assets.

³⁷⁶ For instance, the Eurosystem limited the amount of pledgeable government-guaranteed bank bonds with close links in July 2012, see Section 3.2.2.



Figure 8.7: Optimal levels of explicit government guarantees by asset type

The findings contribute to explain the observable development of explicit government guarantees in the Eurozone, as summarized by the Stylized Facts 8.1 and 8.2. The financial and the sovereign debt crisis had distinct effects on government incentives to grant explicit guarantees. While turmoil in financial markets predominantly affected the impatience/panic of depositors, the surge in public indebtedness increased debt ratios and interest rates on government debt. Variations in three determinants of government incentives to grant explicit guarantees are elaborated in the following, i.e. (i) increased depositor impatience/panic, (ii) deterioration in the fiscal position and (iii) the relaxation of collateral criteria. It is shown that variation in these determinants can facilitate explaining the observable development of government guarantees in the Eurozone.

Increase in Depositor Impatience/Panic Several Eurozone countries' banking sectors faced a surge in depositor impatience during the financial crisis, which culminated in panic in several countries (see e.g. SINN 2010a). As a result, governments deemed it necessary to grant explicit guarantees in large amounts to already and newly eligible marketable assets of struggling financial institutions (Figure 8.3 and MULLER et al. 2012). The surge in depositor impatience corresponds to an increase of σ , which in turn makes existing implicit guarantees ceteris paribus more expensive as depositors contest the credibility of the implicit guarantees. The three panels of Figure 8.8 reveal how a shock to σ increases government incentives to grant explicit guarantees to all three types of assets. This contributes to explain the observation that explicit government guarantees grew in importance throughout the financial crisis (Stylized Fact 8.1) and were likewise given to newly as well as already-eligible marketable assets (Stylized Fact 8.2).



Figure 8.8: Increase in depositor impatience and optimal levels of expl. gov. guarantees

Deterioration in the Fiscal Position By contrast, explicit government guarantees were almost exclusively given to newly eligible marketable assets during the "big bazooka" period with extended liquidity provision by the Eurosystem (Stylized Fact 8.2). This period was preceded by the deterioration of fiscal positions in several Eurozone countries in terms of a surge in debt ratios and interest rates. The rise in the debt ratio and the interest rate augments the beneficial effect of explicit guarantees to newly eligible assets (through the borrowing cost channel). The three panels of Figure 8.9 depict the impact of the exogenous rise of d and i on the optimal level of explicit guarantees to the three types of assets. The figure reveals that the shock only affects the incentive to explicitly guarantee newly eligible assets. This could explain the stylized fact that the structure of new explicit government guarantees to eligible marketable assets changed, i.e. guarantees were predominantly given in crisis-stricken countries to newly eligible marketable assets during the "big bazooka" period (Stylized Fact 8.2).





Relaxation of Collateral Criteria Over recent years, the Eurosystem has facilitated the growing importance of government guarantees by relaxing collateral criteria inter alia with respect to government-guaranteed assets, i.e. it increased the limit on the amount of pledgeable government-guaranteed bonds $\overline{\Phi}$. Whether such an increase in $\overline{\Phi}$ affects the optimal level of government guarantees depends on whether the collateral constraint implicitly imposed by the central bank is binding. It is non-binding for $\Phi^* \leq \overline{\Phi}$ but binding otherwise. The case of a non-binding collateral constraint is illustrated in the left panel of Figure 8.10. To the left of the red line, explicit guarantees are given to newly eligible bonds such that the expected marginal benefits are decreasing and convex and expected marginal costs are low. The collateral constraint induces a discontinuity in both the expected marginal benefits and costs. To the right of the red line, the expected marginal benefits are lower and linearly falling (as it solely reflects the implicit guarantee effect) and expected marginal costs are higher as guaranteed assets are ineligible. It is obvious that a relaxation of collateral criteria, which would shift $\overline{\Phi}$ to the right, would not affect Φ^* , which is already optimal given the initial $\overline{\Phi}$. By contrast, the right panel of Figure 8.10 depicts the case of a binding collateral constraint. The initial collateral constraint is depicted by the dashed line and the government chooses $\Phi = \overline{\Phi}_1 < \Phi^*$ prior to any relaxation of collateral criteria. The relaxation of collateral criteria increases $\overline{\Phi}_1$ to $\overline{\Phi}_2$ and the government responds optimally by extending guarantees to Φ^* . This contributes to explain the observation that government guarantees grew in importance during the financial crisis when the Eurosystem relaxed collateral criteria related to government-guaranteed assets (Stylized Fact 8.1). Moreover, it supports understanding why the majority of explicit government guarantees

were given to newly eligible assets and particularly uncovered bank bonds that were deemed eligible if own-use provided that they were government-guaranteed (Stylized Fact 8.2).



Figure 8.10: Relaxation of collateral criteria and optimal level of expl. gov. guarantees

This chapter has provided a novel rationale for government guarantees, extending the common rationale that justifies government guarantees based on their capability to avoid bank runs (DIAMOND and DYBVIG 1983). The novel rationale was derived from a fiscal perspective in due consideration of the close tie between collateral criteria and government guarantees. Government incentives to provide explicit guarantees were characterized based on the costs and benefits of guarantees, which differ between (i) newly eligible, (ii) already-eligible and (iii) ineligible assets. Moreover, the importance of variations in (i) depositor impatience, (ii) the fiscal position and (*iii*) collateral criteria by explicit government guarantees and their structure were emphasized. Collateral criteria thus bear a fiscal implication as they affect government incentives to grant explicit guarantees, i.e. to assume contingent explicit liabilities.³⁷⁷ Despite the beneficial effect of government guarantees from a fiscal perspective, governments' discretion to free up collateral should be limited for the following reasons. First, the anticipation of explicit guarantees can give rise to moral hazard and risk shifting such that financial institutions are incentivized to take excessive risk. Second, while explicit government guarantees to illiquid but solvent banks can be beneficial, governments may fail to identify those banks and provide guarantees to illiquid and insolvent banks. This would be unfavorable from a fiscal perspective as it would increase future costs to the government from bank bailouts. Third, governments' discretion to free up collateral results in governments striving for the provision of explicit guarantees to assets that would not have been eligible otherwise. An ever-looser collateral framework thus incentivizes governments to guarantee assets of ever-decreasing quality. Finally, the amplified incentive for governments to give explicit guarantees fosters the nexus between governments and the financial sector, particularly in times when the nexus unfolds its unfavorable effects (see Section 8.3).

³⁷⁷ VAN BEKKUM et al. 2015 address another fiscal implication of collateral criteria in terms of government guarantees. The authors examine the effects of Eurosystem collateral policy in the Netherlands regarding the quality of assets backing RMBS. They find that deterioration of the quality of underlying assets is only present for loans with government guarantees and refer this finding to undesired risk shifting from banks to the government.

Conclusion

9.1 Summary: the Eurosystem Collateral Framework and its Fiscal Implications

The dissertation has elaborated on the Eurosystem collateral framework and its fiscal implications.³⁷⁸ A narrative database of amendments to the collateral framework was compiled, revealing that the Eurosystem predominantly relaxed collateral criteria, particularly in response to the financial and the sovereign debt crisis. It was elaborated that the Eurosystem thereby broadened the pool of eligible marketable assets ("collateral pool") both horizontally and vertically at the aggregated and national level. It was shown that amendments to collateral criteria affected the Eurosystem collateral pool with respect to geographical composition, composition by asset type, the credit quality of eligible marketable assets, the denomination of eligible marketable assets, markets in which eligible assets were traded, the fraction of government-guaranteed marketable assets and the residual maturity of eligible marketable assets.

The Eurosystem adjusted its risk control in response to the development of the collateral pool. However, valuation haircuts as the major risk mitigation tool were applied in a simplified way and only adjusted infrequently. The application of simplified haircuts aligns collateral values that are assigned to assets with different properties. Such an alignment gives rise to imperfect risk control and the adverse selection of collateral, i.e. the Eurosystem likely attracts collateral of relatively low quality while high-quality collateral is pledged in the market. The attraction of low-quality collateral would be intensified and beyond the control of the central bank if credit ratings were positively distorted. The adverse selection of collateral facilitates the understanding of empirical evidence that default probabilities of eligible and pledged collateral have developed asymmetrically in the Eurozone over recent years. Hence, the Eurosystem's risk exposure has substantially increased owing to relaxations of collateral criteria but imperfect risk control. Relaxations implied a deterioration in collateral quality, which was reinforced by the adverse

 $^{^{378}\,\}mathrm{See}$ Section 1.2 for a detailed overview of the main findings.

selection of collateral, while the associated increase in collateral risk was insufficiently hedged by simplified haircuts.

Two fiscal implications of amendments to collateral criteria were addressed. First, the implementation of uniform collateral criteria in a monetary union of heterogeneous countries gives rise to risk-sharing among countries owing to the mutualization of potential costs of a central bank bailout. If the common central bank relaxed uniform collateral criteria, risk-sharing would be extended and fiscal positions would be affected from a transfer of fiscal sustainability across countries. Second, amendments to collateral criteria directed at the eligibility of governmentguaranteed assets incentivize governments to initiate new or extend existing explicit guarantees, i.e. to assume contingent liabilities that are not contractually agreed.

The following final section presents a stylized proposal for an improved collateral framework from a fiscal perspective that specifically addresses the previously-identified flaws of Eurosystem collateral criteria.

9.2 An Improved Collateral Framework from a Fiscal Perspective

This section proposes an improved framework from a fiscal perspective that takes up one of the subtleties of the current framework and bases on past practice. Moreover, nine principles are carved out for the proposed framework that are formulated as to circumvent shortcomings identified for the current framework. Taking into account the proposed principles would improve risk control and curtail the adverse selection of collateral such that additional collateral risk from relaxed collateral criteria would be better hedged.³⁷⁹

9.2.1 Structure of the Proposed Collateral Framework

The proposed framework features a two-tier structure that resembles the design of the initial collateral framework. Its major advantage is that it would sidestep excessive risk-sharing among countries, which would give rise to the transfer of fiscal sustainability. The first tier comprises collateral criteria that are uniform across the Eurozone, while the second tier leaves room for country-specific collateral provisions.³⁸⁰ The general structure of the proposed two-tier collateral framework is illustrated in Figure 9.1 and further explained in the following.

The uniform Eurozone-wide collateral criteria of the first tier are laid down by the ECB. Improved risk control owing to the principles specified in Section 9.2.2 would imply Treasuries tolerating a lower level of collateral criteria as the additional collateral risk would be better

³⁷⁹ In the light of the model of Section 7.5, it would imply that the marginal probability of central bank bailout $\partial p_{\lambda}^{i}/\partial \hat{q}$ would progress flatter (owing to the restraint of the adverse selection) and on a lower level (owing to improved risk control).

³⁸⁰ Note that the proposed framework predominately addresses risk-sharing in the monetary income of the Eurosystem. While Chapter 7 elaborated that collateral criteria are also vital for the accumulation of intra-Eurosystem balances and the provision of ELA, these two issues should be separately addressed. While the accumulation of intra-Eurosystem balances should be tackled by implementing an appropriate settlement mechanism (see e.g. SINN 2012, 2014b), the proposed framework would render ELA redundant as the provision of refinancing loans based on country-specific collateral criteria on the condition that governments are solvent would resemble ELA while being more transparent.



Figure 9.1: Structure of the proposed two-tier collateral framework

hedged. The proposed framework still involves risk-sharing among countries, whereby the degree depends on the level of uniform collateral criteria. First, a cautious ECB would implement a level of uniform collateral criteria that corresponds to the lowest common denominator of the countries, i.e. the strictest level preferred by one of the countries. This level would be the most consistent with the primary purpose of collateral as a risk hedge. Moreover, it would imply a small extent of risk-sharing and a small transfer of fiscal sustainability. A less cautious ECB acting as an utilitarian social planner with the aim of implementing the optimal level of uniform collateral criteria for the entire Eurozone from the fiscal perspective would specify an intermediate level (Section 7.5). This level would be stricter than the lowest preferred level but still looser than the strictest preferred level. It would give rise to a larger extent of risk-sharing relative to the previous level and a larger transfer of fiscal sustainability. Finally, the level of collateral criteria could be subject to democratic decision-making in the ECB Council, which would likewise give rise to an intermediate level of collateral criteria. Specifically, agreement on the level of collateral criteria in the Council should be based on voting power reflecting contribution to potential losses and central bank bailout, respectively, rather than on a one-country one-vote basis.

Heterogeneous countries fiscally desire distinct levels of collateral criteria. Distinct preferences can be amplified in case of asymmetric shocks to e.g. fiscal positions. If country-specific preferences were answered by uniform collateral criteria in the first tier, risk-sharing among countries would be intensified. Therefore, the second tier of the proposed collateral framework enables NCBs of solvent countries to formulate country-specific collateral criteria.³⁸¹ However, it exempts assets pledged under these criteria from risk-sharing in the Eurosystem. Accordingly, risk-sharing is restricted to assets pledged under the uniform tier-one collateral criteria but is suspended for assets pledged according to country-specific tier-two criteria. From a fiscal perspective, the country-specific level of collateral criteria would optimally solve the trade-off between the expected marginal costs and benefits of amending collateral criteria for each country.

The proposed two-tier framework mimics the design of the initial Eurosystem collateral framework (Section 3.1.1). Moreover, it takes up a subtlety of the current framework by resembling

³⁸¹ The establishment of country-specific collateral criteria is restricted to NCBs of solvent countries; otherwise, country-specific collateral criteria would not be justifiable from a fiscal perspective as an insolvent government could not back its NCB. However, government backup is vital as the finances of the affected NCB would be weakened in case of counterparty and collateral default and without loss-sharing, given that it would have to compensate the other NCBs for the permanent loss in interest income. See the discussion on income from money creation in Section 7.3.1 and the discussion on ELA in Section 7.3.2.

the risk-sharing features of ACCs (Section 3.2.1). The initial framework featured two tiers of collateral criteria, i.e. a tier comprising uniform collateral criteria and a tier including country-specific provisions. However, potential losses were shared irrespective of the collateral criteria according to which assets were pledged.³⁸² By contrast, potential write-off losses on assets pledged under the ACCs framework were not mutualized but would have to be borne by the affected NCB.

The proposed framework would offer the advantage of circumventing excessive risk-sharing and the transfer of fiscal sustainability. Moreover, it would be based on the fundamental economic principle of liability (provided that governments are and remain solvent). However, it would oppose the Eurosystem's endeavor to facilitate the integration of financial markets via uniform collateral criteria. Instead, it would give rise to a collateral framework with criteria differing across Eurozone countries. The proposed framework would reduce the complexity of the current framework, which is characterized by a myriad of temporary and country-specific provisions, resulting in excessive risk-sharing and transfer of fiscal sustainability. The proposed framework would lay down clear-cut ex ante rules with respect to uniform and country-specific collateral criteria. Moreover, the restriction to risk-sharing would likely render country-specific collateral redundant. The proposed framework can be regarded as a transient framework until a European political and fiscal union is established. A European fiscal authority backing the common central bank would justify a uniform level of collateral criteria involving a larger degree of risk-sharing.

9.2.2 Principles of the Proposed Collateral Framework

Finally, the following formulates nine principles evolving from the previous analysis of the Eurosystem collateral framework that collateral criteria specified for both tiers of the proposed framework should respect to improve risk control.³⁸³

1. Restriction of Collateral with Close Links Collateral criteria should restrict the eligibility of assets with close links (or even own-use) as the risk-hedging effect is diminished when these assets are pledged as collateral (Section 7.3.1).

2. Restriction of the Role of Government Guarantees It was shown in Chapter 8 that amendments to the collateral criteria of government-guaranteed assets affect government incentives to give explicit guarantees. Although Section 8.4 revealed that government guarantees can be beneficial to the government from a fiscal perspective, the section also outlined reasons why government discretion to free up eligible assets via explicit guarantees should be restricted.

3. Restriction of Theoretical Valuation of Assets Eligible assets are valued theoretically when they do not feature a market value (e.g. in case of non-marketable or own-use assets) or when

³⁸² Another difference between the initial two-tier framework and the proposed one is that in the initial framework assets included in the country-specific tier were not deemed eligible for purchases.

 $^{^{383}\}operatorname{See}$ also NyBorg 2015 for shortcomings of the current Eurosystem collateral framework.

their market price has not moved for at least five days (Section 3.1.4.2).³⁸⁴ However, theoretical values give rise to potential valuation errors, which would increase the risk exposure of the Eurosystem (Section 5.3.2.1). Therefore, the leeway of theoretical valuation should be restricted.

4. Restriction of Non-Marketable Assets Theoretical valuation owing to a lacking market value is particularly important for non-marketable assets that also feature low liquidity. Therefore, the eligibility of non-marketable assets should be restricted in due consideration of their increased pledge with the Eurosystem over recent years (Section 5.1).

5. Restriction of Assets Traded on Non-Regulated Markets Assets traded on non-regulated markets often lack an up-to-date market value and have to be theoretically valued (NYBORG 2015). Section 3.2.3 raised concerns about the assessment of non-regulated markets with respect to the stipulated principles of safety, transparency and accessibility. Specifically, the application of the principle of transparency was exemplarily questioned for the STEP market (Section 3.2.3). Owing to the lack of up-to-date market values and their opacity, which contradicts the principle of transparency, the eligibility of assets traded on non-regulated markets should be restricted.

6. More Graduated Haircuts Haircuts are the most important risk control measure to the Eurosystem and they should be as graduated as possible.³⁸⁵ The guiding principle of haircut specification should be that they reflect all available information and are based on market input (ibid.). Therefore, haircuts should mirror asset properties such as coupon, liquidity, credit quality and residual maturity as accurately as possible (Section 4.3.2). More graduated haircuts (as indicated in Figure 4.3) would give rise to differentiated rather than pooled refinancing conditions based on asset properties. This would minimize the peril of adverse selection of collateral as the subsidization of low-quality collateral would be circumvented.

7. More Frequent Revision of Haircuts Section 4.2.2 indicated that haircut revision was infrequent and often delayed or unrelated to events exogenous and endogenous to the Eurosystem. However, given that haircuts are the most important risk control measure of the Eurosystem, they should be revised frequently and in instantaneous response to relevant events. Furthermore, more frequent, timely and accurate revision of haircuts is indispensable to minimize the peril of adverse selection of collateral.

8. Simultaneous Application of Risk Control Measures Furthermore, the bias in haircuts from the sequential application of risk control measures (Section 4.3.2.3) should be resolved. Risk control via valuation haircuts and supplementary haircuts in terms of valuation markdowns should thus occur simultaneously.

 $^{^{384}}$ NyBorg 2015 reports that about 77% by count or 17% by value of eligible assets feature theoretical values.

³⁸⁵ If the Eurosystem applied oversimplified haircuts to improve the liquidity of certain assets or facilitate liquidity provision, it would become redundant within the proposed framework as NCBs could lay down country-specific collateral criteria to mimic the envisaged effect and bear the emerging risk.

9. Rejection of First-Best Rule The Eurosystem applies the first-best rule to eligible assets other than ABSs for determining of the pivotal credit rating (Section 3.2.1). Section 4.3.2.1 reasoned that the application of the first-best rule is associated with three flaws, i.e. (i) information available on collateral credit quality is potentially neglected; (ii) rating agencies are incentivized to provide the pivotal credit rating, which could give rose to a "race to the top" of ratings irrespective of actual credit quality; and (iii) "ratings shopping" of counterparties is facilitated. A rejection of the first-best rule and stipulation and consultation of more than one credit rating would avoid these flaws.

Appendix to Chapter 5

Derivation of Equation (5.1)

Let the binary variable $I(\theta_j)$ be either 1 if bond j is traded on the market or 0 otherwise. Rent of borrowers R_B is given by the area underneath the price curve in Figure 5.9. As all bonds are traded at the distinct price $\rho(\theta_j) = \theta_j$, R_B reads as

$$R_B = \int_{\underline{\theta}}^{\overline{\theta}} I(\theta)\rho(\theta) \ d\theta = \int_{\underline{\theta}}^{\overline{\theta}} I(\theta)\theta \ d\theta = \int_{\underline{\theta}}^{\overline{\theta}} \theta \ d\theta = \left[\frac{1}{2}\theta^2\right]_{\underline{\theta}}^{\overline{\theta}} = \frac{\overline{\theta}^2 - \underline{\theta}^2}{2}$$

Derivation of Equation (5.3)

Analogous to the derivation of Equation (5.1) and with the uniform price $\rho = \frac{\bar{\theta} + \underline{\theta}}{2}$, R_B and R, respectively, are given by

$$R_B = R = \int_{\underline{\theta}}^{\overline{\theta}} I(\theta) \rho \ d\theta = \int_{\underline{\theta}}^{\overline{\theta}} \frac{\overline{\theta} + \underline{\theta}}{2} \ d\theta = \left[\frac{(\overline{\theta} + \underline{\theta})\theta}{2} \right]_{\underline{\theta}}^{\overline{\theta}} = \frac{\overline{\theta}^2 - \underline{\theta}^2}{2}.$$

Derivation of Equation (5.4)

With a positively distorted signal and all bonds being traded with $\rho(\theta_j) = \max\left(\theta_j + \epsilon, \bar{\theta}\right)$, lenders suffer a loss in the amount of

$$R_{L} = \int_{\underline{\theta}}^{\overline{\theta}} I(\theta) \left(\theta - \rho(\theta)\right) d\theta = \int_{\underline{\theta}}^{\overline{\theta} - \epsilon} I(\theta) \left(\theta - (\theta + \epsilon)\right) d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta}} I(\theta) (\theta - \overline{\theta}) d\theta$$
$$= \left[-\epsilon\theta\right]_{\underline{\theta}}^{\overline{\theta} - \epsilon} + \left[\frac{1}{2}\theta^{2} - \overline{\theta}\theta\right]_{\overline{\theta} - \epsilon}^{\overline{\theta}} = -(\overline{\theta} - \underline{\theta} - \frac{1}{2}\epsilon)\epsilon < 0.$$

Derivation of Equation (5.5)

Analogous to Equation (5.4), borrowers receive rent in case of a positively distorted signal given by

$$R_B = \int_{\underline{\theta}}^{\overline{\theta}} I(\theta)\rho(\theta) \ d\theta = \int_{\underline{\theta}}^{\overline{\theta}-\epsilon} I(\theta)(\theta+\epsilon)d\theta + \int_{\overline{\theta}-\epsilon}^{\overline{\theta}} I(\theta)\overline{\theta}d\theta$$
$$= \left[\frac{1}{2}\theta^2 + \theta\epsilon\right]_{\underline{\theta}}^{\overline{\theta}-\epsilon} + \left[\overline{\theta}\theta\right]_{\overline{\theta}-\epsilon}^{\overline{\theta}} = \frac{\overline{\theta}^2 - \underline{\theta}^2}{2} - \left(\underbrace{-(\overline{\theta}-\underline{\theta}-\frac{1}{2}\epsilon)\epsilon}_{R_L}\right)$$

Derivation of Equation (5.6)

Considering $\alpha(\theta) = \theta - \underline{\theta}$ and that a fraction of bonds is traded in the market while the remainder is used with the outside option, rent of borrowers corresponding to aggregate surplus in the market is given by

$$R_B = R = \int_{\underline{\theta}}^{\overline{\theta}} I(\theta) \rho \ d\theta + \int_{\underline{\theta}}^{\overline{\theta}} \left(1 - I(\theta) \right) \alpha(\theta) \ d\theta = \int_{\underline{\theta}}^{\theta_{\alpha}} \frac{\underline{\theta} + \theta_{\alpha}}{2} \ d\theta + \int_{\theta_{\alpha}}^{\overline{\theta}} \left(\theta - \underline{\theta} \right) \ d\theta$$
$$= \left[\frac{(\underline{\theta} + \theta_{\alpha})\theta}{2} \right]_{\underline{\theta}}^{\theta_{\alpha}} + \left[\frac{1}{2} \theta^2 - \theta \underline{\theta} \right]_{\theta_{\alpha}}^{\overline{\theta}} = \frac{\theta^2 - \underline{\theta}^2}{2} - \underline{\theta}(\overline{\theta} - \theta_{\alpha}).$$

Derivation of Equation (5.19)

Rent of borrowers with pooled public repo conditions and correct ratings is given by

$$\begin{split} R_{B} &= \int_{\underline{\theta}}^{\overline{\theta}} I(\theta)\rho(\theta) \ d\theta + \int_{\underline{\theta}}^{\overline{\theta}} \left(1 - I(\theta)\right)\alpha(\theta) \ d\theta \\ &= \int_{\underline{\theta}}^{\hat{\theta}_{3}} s(\theta) \ d\theta + \int_{\hat{\theta}_{3}}^{\theta_{\alpha_{3}}} \alpha(\theta) \ d\theta + \int_{\theta_{\alpha_{3}}}^{\hat{\theta}_{1}} s(\theta) \ d\theta + \int_{\hat{\theta}_{1}}^{\theta_{\alpha_{1}}} \alpha(\theta) \ d\theta + \int_{\theta_{\alpha_{1}}}^{\overline{\theta}} s(\theta) \ d\theta \\ &= \int_{\underline{\theta}}^{\hat{\theta}_{3}} \theta \ d\theta + \int_{\hat{\theta}_{3}}^{\theta_{\alpha_{3}}} \frac{\hat{\theta}_{1} + \hat{\theta}_{3}}{b_{3}} \ d\theta + \int_{\theta_{\alpha_{3}}}^{\hat{\theta}_{1}} \theta \ d\theta + \int_{\hat{\theta}_{1}}^{\theta_{\alpha_{1}}} \frac{\overline{\theta} + \hat{\theta}_{1}}{b_{1}} \ d\theta + \int_{\theta_{\alpha_{1}}}^{\overline{\theta}} \theta \ d\theta \\ &= \left[\frac{1}{2}\theta^{2}\right]_{\underline{\theta}}^{\hat{\theta}_{3}} + \left[\frac{(\hat{\theta}_{1} + \hat{\theta}_{3})\theta}{b_{3}}\right]_{\hat{\theta}_{3}}^{\theta_{\alpha_{3}}} + \left[\frac{1}{2}\theta^{2}\right]_{\theta_{\alpha_{3}}}^{\hat{\theta}_{1}} + \left[\frac{(\bar{\theta} + \hat{\theta}_{1})\theta}{b_{1}}\right]_{\hat{\theta}_{1}}^{\theta_{\alpha_{1}}} + \left[\frac{1}{2}\theta^{2}\right]_{\theta_{\alpha_{1}}}^{\overline{\theta}} \\ &= \frac{\overline{\theta}^{2} - \underline{\theta}^{2}}{2} + \underbrace{\frac{\hat{\theta}_{1}^{2} + \hat{\theta}_{3}^{2}}{2} + \frac{1}{2}\left(\frac{\hat{\theta}_{1} + \hat{\theta}_{3}}{b_{3}}\right)^{2} + \frac{1}{2}\left(\frac{\overline{\theta} + \hat{\theta}_{1}}{b_{1}}\right)^{2} - \frac{\hat{\theta}_{3}(\hat{\theta}_{1} + \hat{\theta}_{3})}{b_{3}} - \frac{\hat{\theta}_{1}(\bar{\theta} + \hat{\theta}_{1})}{b_{1}} \right]_{\theta_{1}}^{\theta_{1}} + \underbrace{\left[\frac{1}{2}\theta^{2}\right]_{\theta_{1}}^{\theta_{1}}}{b_{1}} + \underbrace{\left[\frac{1}{2}\theta^{2}\right]_{\theta_{1}}^{\theta_{1}}}{b_{1}} + \underbrace{\left[\frac{1}{2}\theta^{2}\right]_{\theta_{1}}^{\theta_{1}}} + \underbrace{\left[\frac{1}{2}\theta^{2}\right]_{\theta_{1}}^{\theta_{1}} + \underbrace{\left[\frac{1}{2}\theta^{2}\right]_{\theta_{1}}^{\theta$$

additional rent from pooled public repo conditions

Derivation of Equation (5.21)

Lenders sustain a loss with perfectly-pooled repo conditions and positively distorted ratings in the amount of

$$\begin{split} R_L &= \int_{\underline{\theta}}^{\overline{\theta}} I(\theta) \Big(\theta - \rho(\theta) \Big) \ d\theta = \int_{\underline{\theta}}^{\hat{\theta}_3 - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\theta_{\alpha_3} - \epsilon}^{\hat{\theta}_1 - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\theta_{\alpha_1} - \epsilon}^{\overline{\theta}} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\theta_{\alpha_3} - \epsilon}^{\overline{\theta}} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\theta_{\alpha_3} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\theta_{\alpha_4} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta}} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta}} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big(\theta - s(\theta) \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon}^{\overline{\theta} - \epsilon} \Big) \ d\theta + \int_{\overline{\theta} - \epsilon$$

Derivation of Equation (5.22)

By contrast, borrowers benefit from positively distorted ratings and realize rent given by

$$\begin{split} R_B &= \int_{\underline{\theta}}^{\theta} I(\theta)\rho(\theta) \ d\theta + \int_{\underline{\theta}}^{\theta} \left(1 - I(\theta)\right)\alpha(\theta) \ d\theta \\ &= \int_{\underline{\theta}}^{\hat{\theta}_3 - \epsilon} s(\theta) \ d\theta + \int_{\hat{\theta}_3 - \epsilon}^{\hat{\theta}_{\alpha_3} - \epsilon} \alpha(\theta) \ d\theta + \int_{\hat{\theta}_{\alpha_3} - \epsilon}^{\hat{\theta}_1 - \epsilon} s(\theta) \ d\theta + \int_{\hat{\theta}_1 - \epsilon}^{\theta} \alpha(\theta) \ d\theta + \int_{\hat{\theta}_{\alpha_1} - \epsilon}^{\theta} s(\theta) \ d\theta \\ &= \int_{\underline{\theta}}^{\hat{\theta}_3 - \epsilon} (\theta + \epsilon) \ d\theta + \int_{\hat{\theta}_3 - \epsilon}^{\theta} \frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3} \ d\theta + \int_{\theta_{\alpha_3} - \epsilon}^{\hat{\theta}_1 - \epsilon} (\theta + \epsilon) \ d\theta + \int_{\hat{\theta}_1 - \epsilon}^{\theta} \frac{\hat{\theta}_1 + \hat{\theta}_1}{b_1} \ d\theta \\ &+ \int_{\theta_{\alpha_1} - \epsilon}^{\bar{\theta} - \epsilon} (\theta + \epsilon) \ d\theta + \int_{\bar{\theta} - \epsilon}^{\bar{\theta}} \bar{\theta} \ d\theta \\ &= \left[\frac{1}{2}\theta^2 + \theta\epsilon\right]_{\underline{\theta}}^{\hat{\theta}_3 - \epsilon} + \left[\frac{(\hat{\theta}_1 + \hat{\theta}_3)\theta}{b_3}\right]_{\hat{\theta}_3 - \epsilon}^{\theta - \epsilon} + \left[\frac{1}{2}\theta^2 + \theta\epsilon\right]_{\theta_{\alpha_3} - \epsilon}^{\hat{\theta}_1 - \epsilon} + \left[\frac{(\hat{\theta} + \hat{\theta}_1)\theta}{b_1}\right]_{\hat{\theta}_1 - \epsilon}^{\theta - \epsilon} \\ &+ \left[\frac{1}{2}\theta^2 + \theta\epsilon\right]_{\theta_{\alpha_1} - \epsilon}^{\bar{\theta} - \epsilon} + \left[\bar{\theta}\theta\right]_{\bar{\theta} - \epsilon}^{\bar{\theta}} \\ &= \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} + \underbrace{\frac{\hat{\theta}_1^2 + \hat{\theta}_3^2}{2} + \frac{1}{2}\left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3}\right)^2 + \frac{1}{2}\left(\frac{\bar{\theta} + \hat{\theta}_1}{b_1}\right)^2 - \frac{\hat{\theta}_3(\hat{\theta}_1 + \hat{\theta}_3)}{b_3} - \frac{\hat{\theta}_1(\bar{\theta} + \hat{\theta}_1)}{b_1}} \right]_{\theta_1}^{\theta_1(\bar{\theta} + \hat{\theta}_1)} \\ &= \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} + \underbrace{\frac{\hat{\theta}_1^2 + \hat{\theta}_3^2}{2} + \frac{1}{2}\left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3}\right)^2 + \frac{1}{2}\left(\frac{\bar{\theta} + \hat{\theta}_1}{b_1}\right)^2 - \frac{\hat{\theta}_3(\hat{\theta}_1 + \hat{\theta}_3)}{b_3} - \frac{\hat{\theta}_1(\bar{\theta} + \hat{\theta}_1)}{b_1}} \right]_{\theta_1}^{\theta_2(\bar{\theta} + \hat{\theta}_1)} \\ &= \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} + \underbrace{\frac{\hat{\theta}_1^2 + \hat{\theta}_3^2}{2} + \frac{1}{2}\left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3}\right)^2 + \frac{1}{2}\left(\frac{\bar{\theta} + \hat{\theta}_1}{b_1}\right)^2 - \frac{\hat{\theta}_3(\hat{\theta}_1 + \hat{\theta}_3)}{b_3} - \frac{\hat{\theta}_1(\bar{\theta} + \hat{\theta}_1)}{b_1}} \right]_{\theta_1}^{\theta_2(\bar{\theta} + \hat{\theta}_1)} \\ &= \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} + \underbrace{\frac{\hat{\theta}_1^2 + \hat{\theta}_3^2}{2} + \frac{1}{2}\left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3}\right)^2 + \frac{1}{2}\left(\frac{\bar{\theta} + \hat{\theta}_1}{b_1}\right)^2 - \frac{\hat{\theta}_3(\hat{\theta}_1 + \hat{\theta}_3)}{b_3} - \frac{\hat{\theta}_1(\bar{\theta} + \hat{\theta}_1)}{b_1}} \right]_{\theta_1}^{\theta_1(\bar{\theta} + \hat{\theta}_1)} \\ &= \frac{\bar{\theta}^2 - \underline{\theta}^2}{2} + \underbrace{\frac{\hat{\theta}_1^2 + \hat{\theta}_3^2}{2} + \frac{1}{2}\left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3}\right)^2 + \frac{1}{2}\left(\frac{\bar{\theta}_1 + \hat{\theta}_1}{b_1}\right)^2 - \frac{\hat{\theta}_3(\hat{\theta}_1 + \hat{\theta}_3)}{b_3} - \frac{\hat{\theta}_1(\bar{\theta} + \hat{\theta}_1)}{b_1}} \right]_{\theta_1}^{\theta_1(\bar{\theta} + \hat{\theta}_1)}$$

Derivation of Equation (5.24)

Risk to the central bank from offering perfectly-pooled public repo conditions when ratings are positively distorted corresponds to

$$\begin{aligned} R_{CB} &= \int_{\underline{\theta}}^{\overline{\theta}} \left(1 - I(\theta) \left(\theta - \alpha(\theta) \right) = \int_{\hat{\theta}_3}^{\theta_{\alpha_3}} \left(\theta - \alpha(\theta) \right) \, d\theta + \int_{\hat{\theta}_1}^{\theta_{\alpha_1}} \left(\theta - \alpha(\theta) \right) \, d\theta \\ &= \left[\frac{1}{2} \theta^2 - \frac{(\hat{\theta}_1 + \hat{\theta}_3) \theta}{b_3} \right]_{\hat{\theta}_3 - \epsilon}^{\theta_{\alpha_3} - \epsilon} + \left[\frac{1}{2} \theta^2 - \frac{(\overline{\theta} + \hat{\theta}_1) \theta}{b_1} \right]_{\theta_{\alpha_1} - \epsilon}^{\hat{\theta}_1 - \epsilon} \\ &= - \left(\frac{\hat{\theta}_1^2 + \hat{\theta}_3^2}{2} + \frac{1}{2} \left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3} \right)^2 + \frac{1}{2} \left(\frac{\overline{\theta} + \hat{\theta}_1}{b_1} \right)^2 - \frac{\hat{\theta}_3(\hat{\theta}_1 + \hat{\theta}_3)}{b_3} - \frac{\hat{\theta}_1(\overline{\theta} + \hat{\theta}_1)}{b_1} \right) \\ &- \epsilon \left(\frac{\hat{\theta}_1 + \hat{\theta}_3}{b_3} + \frac{\overline{\theta} + \hat{\theta}_1}{b_1} - \hat{\theta}_3 - \hat{\theta}_1 \right). \end{aligned}$$

Appendix to Chapter 7

Derivation of Equation (7.6)

The present value of income flows from a certain central bank loan with nominal value L and duration T is given by

$$V_L = \frac{i_L L}{(1+\beta)} + \frac{i_L L}{(1+\beta)^2} + \dots + \frac{i_L L}{(1+\beta)^T} + \frac{L}{(1+\beta)^T}$$
$$= \sum_{t=1}^T \frac{i_L L}{(1+\beta)^t} + \frac{L}{(1+\beta)^T}$$
$$= i_L L \left(\frac{1-(1+\beta)^{-T}}{\beta}\right) + L(1+\beta)^{-T}.$$

Derivation of Equation (7.7)

The present value of income flows from a chain of one-period central bank loans under certainty is obtained by solving Equation (7.6) recursively such that

$$\begin{aligned} V_L &= i_L L_0 \left(\frac{1 - (1 + \beta)^{-1}}{\beta} \right) + L_0 (1 + \beta)^{-1} \\ &= i_L L_0 \left(\frac{1 - (1 + \beta)^{-1}}{\beta} \right) + \left(i_L L_0 \left(\frac{1 - (1 + \beta)^{-1}}{\beta} \right) + L_1 (1 + \beta)^{-1} \right) (1 + \beta)^{-1} \\ &= i_L L_0 \left(\frac{1 - (1 + \beta)^{-1}}{\beta} \right) \left(1 + (1 + \beta)^{-1} \right) + L_1 (1 + \beta)^{-2} \\ &= \dots \\ &= i_L L_0 \left(\frac{1 - (1 + \beta)^{-1}}{\beta} \right) \sum_{j=0}^n (1 + \beta)^{-j} + L_j (1 + \beta)^{-(j+1)}, \end{aligned}$$

where L_0 is the nominal value of the initial loan and L_j the constant nominal value of all replacement loans.

Derivation of Equation (7.13)

Dynamic seignorage under certainty allows for growth in the monetary base. Assuming that the monetary base grows in line with the economy at the rate g, dynamic seignorage comprises static

seignorage and the present value of interest payments from loans issued and assets purchased to back future base money issuance, i.e.

$$\begin{split} \delta_{as}^{dyn} &= \delta_{as}^{stat} + \sum_{t=1}^{\infty} \frac{g\left(\frac{i_L}{\beta}L_{t-1} + \frac{i_A}{\beta}A_{t-1}\right)}{(1+\beta)^t} \\ &= \frac{i_L}{\beta}L_0 + \frac{i_A}{\beta}A_0 + \sum_{t=1}^{\infty} \frac{g(1+g)^{t-1}\left(\frac{i_L}{\beta}L_0 + \frac{i_A}{\beta}A_0\right)}{(1+\beta)^t} \\ &= \frac{i_L}{\beta}L_0 + \frac{i_A}{\beta}A_0 + \left(\frac{i_L}{\beta}L_0 + \frac{i_A}{\beta}A_0\right)\sum_{t=1}^{\infty} \frac{g(1+g)^{t-1}}{(1+\beta)^t} \\ &= \frac{i_L}{\beta}L_0 + \frac{i_A}{\beta}A_0 + \left(\frac{i_L}{\beta}L_0 + \frac{i_A}{\beta}A_0\right)\frac{g}{\beta-g} \\ &= (i_L L_0 + i_A A_0)\frac{1}{\beta-g}. \end{split}$$

Derivation of Equation (7.16)

The present value of income flows from a chain of one-period central bank loans under uncertainty is derived analogously to Equation (7.7) and given by

$$\begin{aligned} V_L &= i_L L_0 \left(\frac{p_L - (1+\beta)^{-1} p_L^2}{1+\beta - p_L} \right) + p_L L_0 (1+\beta)^{-1} \\ &= i_L L_0 \left(\frac{p_L - (1+\beta)^{-1} p_L^2}{1+\beta - p_L} \right) \left(1 + p_L (1+\beta)^{-1} \right) + p_L L_1 (1+\beta)^{-2} \\ &= \dots \\ &= i_L L_0 \left(\frac{p_L - (1+\beta)^{-1} p_L^2}{1+\beta - p_L} \right) \sum_{j=0}^n p_L^j (1+\beta)^{-j} + p_L^{j+1} L_j (1+\beta)^{-(j+1)}. \end{aligned}$$

References

- ACHARYA, V. V., D. ANGINER, and A. J. WARBURTON (2014a), "The End of Market Discipline? Investor Expectations of Implicit Government Guarantees," *mimeo*.
- ACHARYA, V. V., I. DRECHSLER, and P. SCHNABL (2012), "A Tale of Two Overhangs. The Nexus of Financial Sector and Sovereign Credit Risks," *Financial Stability Review* 16, 51–6.
- (2014b), "A Pyrrhic Victory? Bank Bailouts and Sovereign Credit Risk," *Journal of Finance* 69, 2689–739.
- ACHARYA, V. V. and M. RICHARDSON (2009), Restoring Financial Stability. How to Repair a Failed System, John Wiley & Sons: Hoboken, NJ.
- ACHARYA, V. V. and S. STEFFEN (2015), "The 'Greatest' Carry Trade Ever? Understanding Eurozone Bank Risks," *Journal of Financial Economics* 115, 215–36.
- ACHARYA, V. V. and T. YORULMAZER (2007), "Too Many to Fail. An Analysis of Time-Inconsistency in Bank Closure Policies," *Journal of Financial Intermediation* 16, 1–31.
- ACKERMANN, J. (2008), "The Subprime Crisis and its Consequences," Journal of Financial Stability 4, 329–37.
- ADLER, G., P. CASTRO, and C. E. TOVAR (2012), "Does Central Bank Capital Matter for Monetary Policy?" *IMF Working Paper* 12/60, February 2012.
- ADRIAN, T., B. BEGALLE, A. COPELAND, and A. MARTIN (2013), "Repo and Securities Lending," Federal Reserve Bank of New York Staff Report 529, February 2013.
- AFONSO, A. (2005), "Fiscal Sustainability. The Unpleasant European Case," *FinanzArchiv/Public Finance Analysis* 61, 19–44.
 - (2008), "Ricardian Fiscal Regimes in the European Union," *Empirica* 35, 313–34.
- AIYAGARI, S. R. and M. GERTLER (1985), "The Backing of Government Bonds and Monetarism," *Journal of Monetary Economics* 16, 19–44.
- ALESINA, A. and R. GATTI (1995), "Independent Central Banks. Low Inflation at No Cost?" American Economic Review 85, 196–200.
- ALLAIS, M. (1962), "The Influence of the Capital-Output Ratio on Real National Income," Econometrica 30, 700–28.
- ALLEN, F., E. CARLETTI, I. GOLDSTEIN, and A. LEONELLO (2015), "Moral Hazard and Government Guarantees in the Banking Industry," *Journal of Financial Regulation* 1, 30–50.
- ALLEN, F., E. CARLETTI, and A. LEONELLO (2011), "Deposit Insurance and Risk Taking," Oxford Review of Economic Policy 27, 464–78.
- ALLEN, F. and D. GALE (1998), "Optimal Financial Crises," Journal of Finance 53, 1245-84.
- ALLEN, W. A. and R. MOESSNER (2013), "The Liquidity Consequences of the Euro Area Sovereign Debt Crisis," BIS Working Papers 390, March 2013.
- ANDERSON, J. and J. STALLINGS (2013), "Greece. Déjà Vu All Over Again?" *IIF Research Note*, 27 November 2013.

- ANDRITZKY, J. R. (2012), "Government Bonds and Their Investors. What Are the Facts and Do They Matter?" *IMF Working Paper* 12/158, June 2012.
- ARATEN, M. and C. TURNER (2013), "Understanding the Funding Cost Differences Between Global Systemically Important Banks (GSIBs) and Non-GSIBs in the USA," *Journal of Risk Management in Financial Institutions* 6, 387–410.
- ARCHER, D. and P. MOSER-BOEHM (2013), "Central Bank Finances," *BIS Papers* 71, April 2013.
- ARMANTIER, O., E. GHYSELS, A. SARKAR, and J. SHRADER (2013), "Discount Window Stigma During the 2007-2008 Financial Crisis," *Federal Reserve Bank of New York Staff Report* 483, September 2013.
- ARSLANALP, S. and T. TSUDA (2014), "Tracking Global Demand for Advanced Economy Sovereign Debt," *IMF Economic Review* 62, 430–64.
- ASHCRAFT, A., N. GÂRLEANU, and L. H. PEDERSEN (2011), "Two Monetary Tools. Interest Rates and Haircuts," in: ACEMOGLU, D. and M. WOODFORD (eds.), NBER Macroeconomics Annual 2010, Volume 25, University of Chicago Press: Chicago, IL, 143–80.
- ASONUMA, T., S. BAKHACHE, and H. HESSE (2015), "Is Banks' Home Bias Good or Bad for Public Debt Sustainability?" *IMF Working Paper* 15/44, February 2015.
- AUER, R. A. (2014), "What Drives Target2 Balances? Evidence from a Panel Analysis," Economic Policy 29, 139–97.
- AUERBACH, A. J., L. J. KOTLIKOFF, and W. LEIBFRITZ (1999), *Generational Accounting around* the World, University of Chicago Press: Chicago, IL.
- AUERNHEIMER, L. (1974), "The Honest Government's Guide to the Revenue from the Creation of Money," *Journal of Political Economy* 82, 598–606.
- BAGEHOT, W. (1873), Lombard Street. A Description of the Money Market, Henry S. King and Co.: London.
- BAILEY, M. J. (1956), "The Welfare Cost of Inflationary Finance," *Journal of Political Economy* 64, 93–110.
- BAKER, D. and T. MCARTHUR (2009), "The Value of the 'Too Big To Fail' Big Bank Subsidy," *CEPR Issue Brief*, September 2009.
- BAKK-SIMON, K., S. BORGIOLI, C. GIRON, H. HEMPELL, A. MADDALONI, F. RECINE, and S. ROSATI (2012), "Shadow Banking in the Euro Area. An Overview," *ECB Occasional Paper* 133, April 2012.
- BALASSONE, F., J. CUNHA, G. LANGENUS, B. MANZKE, J. PAVOT, D. PRAMMER, and P. TOM-MASINO (2009), "Fiscal Sustainability and Policy Implications for the Euro Area," Bundesbank Discussion Paper 04/2009, January 2009.
- BALASSONE, F. and D. FRANCO (2000), "Assessing Fiscal Sustainability. A Review of Methods with a View to EMU," in: BANK OF ITALY (ed.), *Fiscal Sustainability. Essays presented at the Bank of Italy Workshop.* Bank of Italy: Rome, 21–60.
- BANK FOR INTERNATIONAL SETTLEMENTS (1999), Implications of Repo Markets for Central Banks, BIS: Basle.
- (2010), "The Role of Margin Requirements and Haircuts in Procyclicality," *BIS CGFS Papers* 36, March 2010.
- (2013a), "Asset Encumbrance, Financial Reform and the Demand for Collateral Assets," BIS CGFS Papers 49, May 2013.

- BANK FOR INTERNATIONAL SETTLEMENTS (2013b), "Central Bank Collateral Frameworks and Practices," *BIS Markets Committee*, March 2013.
- (2015), "Central Bank Operating Frameworks and Collateral Markets," *BIS CGFS Papers* 53, March 2015.
- BARCLAYS CAPITAL (2012), "ECB SMP. Marking to Market," Interest Rates Research, 6 January 2012.
- BARNHILL, T. M. and G. KOPITS (2005), "Assessing Fiscal Sustainability Under Uncertainty," International Journal of Public Budget 55, 85–124.
- BARRO, R. J. (1979), "On the Determination of the Public Debt," *Journal of Political Economy* 87, 940–71.
- (1982), "Measuring the Fed's Revenue from Money Creation," *Economics Letters* 10, 327–32.
- BARTOLINI, L. and C. COTTARELLI (1994), "Government Ponzi Games and the Sustainability of Public Deficits Under Uncertainty," *Ricerche Economiche* 48, 1–22.
- BARTOLINI, L. and A. DIXIT (1991), "Market Valuation of Illiquid Debt and Implications for Conflicts among Creditors," *IMF Staff Papers* 38, 828–49.
- BATTISTINI, N., M. PAGANO, and S. SIMONELLI (2013), "Systemic Risk and Home Bias in the Euro Area," *European Economy Economic Papers* 494, April 2013.
 - (2014), "Systemic Risk, Sovereign Yields and Bank Exposures in the Euro Crisis," *Economic Policy* 29, 203–51.
- BECKER, B. and T. MILBOURN (2011), "How Did Increased Competition Affect Credit Ratings?" Journal of Financial Economics 101, 493–514.
- BEIRNE, J., L. DALITZ, J. EJSING, M. GROTHE, S. MANGANELLI, F. MONAR, B. SAHEL, M. SUSEC, J. TAPKING, and T. VONG (2011), "The Impact of the Eurosystem's Covered Bond Purchase Programme on the Primary and Secondary Markets," *ECB Occasional Paper* 122, January 2011.
- BELKE, A. (2015), "Eurosystem Collateral Policy and Framework. Post-Lehman Time as a New Collateral Space," *Intereconomics* 2, 82–90.
- BELKE, A. and T. POLLEIT (2010), "How much Fiscal Backing Must the ECB Have? The Euro Area is not (yet) the Philippines," *International Economics* 124, 5–30.
- BENECKÁ, S., T. HOLUB, N. L. KADLCÁKOVÁ, and I. KUBICOVÁ (2012), "Does Central Bank Financial Strength Matter for Inflation? An Empirical Analysis," CNB Working Paper 3/2012, May 2012.
- BERGER, H., J. DE HAAN, and S. C. W. EIJFFINGER (2001), "Central Bank Independence. An Update of Theory and Evidence," *Journal of Economic Surveys* 15, 3–40.
- BERNHOLZ, P. (2012), "The Current Account Deficits of the GIPS Countries and Their Target Debts at the ECB," CESifo Forum 13, 33–4.
- BINDSEIL, U. (2009a), "Central Bank Financial Crisis Management from a Risk Management Perspective," in: BINDSEIL, U., F. GONZÁLEZ, and E. TABAKIS (eds.), Risk Management for Central Banks and Other Public Investors, Cambridge University Press: Cambridge, 394– 440.
 - (2009b), "Central Banks and Other Public Institutions as Financial Investors," in: BINDSEIL, U., F. GONZÁLEZ, and E. TABAKIS (eds.), *Risk Management for Central Banks and Other Public Investors*, Cambridge University Press: Cambridge, 3–48.

- BINDSEIL, U. (2013), "Central Bank Collateral, Asset Fire Sales, Regulation and Liquidity," ECB Working Paper 1610, November 2013.
- (2014), Monetary Policy Operations and the Financial System, Oxford University Press: Oxford.
- BINDSEIL, U., P. COUR-THIMANN, and P. J. KÖNIG (2012), "Target2 and Cross-Border Interbank Payments During the Financial Crisis," *CESifo Forum* 13, 83–92.
- BINDSEIL, U., F. GONZÁLEZ, and E. TABAKIS (eds.) (2009), *Risk Management for Central Banks and Other Public Investors*, Cambridge University Press: Cambridge.
- BINDSEIL, U. and J. JABLECKI (2013), "Central Bank Liquidity Provision, Risk-Taking and Economic Efficiency," *ECB Working Paper* 1542, May 2013.
- BINDSEIL, U. and P. J. KÖNIG (2012), "TARGET2 and the European Sovereign Debt Crisis," *Kredit und Kapital* 45, 135–74.
- BINDSEIL, U., A. MANZANARES, and B. WELLER (2004), "The Role of Central Bank Capital Revisited," *ECB Working Paper* 392, September 2004.
- BINDSEIL, U. and F. PAPADIA (2006), "Credit Risk Mitigation in Central Bank Operations and its Effects on Financial Markets. The Case of the Eurosystem," ECB Occasional Paper 49, August 2006.
 - (2009), "Risk Management and Market Impact of Central Bank Credit Operations," in: BINDSEIL, U., F. GONZÁLEZ, and E. TABAKIS (eds.), *Risk Management for Central Banks* and Other Public Investors, Cambridge University Press: Cambridge, 271–302.
- BINDSEIL, U. and A. WINKLER (2013), "Dual Liquidity Crises. A Financial Accounts Framework," *Review of International Economics* 21, 151–63.
- BLANCHARD, O. J. (1990), "Suggestions For a New Set of Fiscal Indicators," OECD Economics Department Working Paper 79, April 1990.
- BLANCHARD, O. J., J.-C. CHOURAQUI, R. HAGEMANN, and N. SARTOR (1990), "The Sustainability of Fiscal Policy. New Answers To an Old Question," OECD Economic Studies 15, 7–35.
- BLANCHARD, O. J. and S. FISCHER (1989), *Lectures on Macroeconomics*, MIT Press: Cambridge, MA.
- BLANCHARD, O. J. and P. WEIL (2001), "Dynamic Efficiency, the Riskless Rate, and Debt Ponzi Games Under Uncertainty," Advances in Macroeconomics 1, 1–23.
- BLINDER, A. S. and R. M. SOLOW (1973), "Does Fiscal Policy Matter?" Journal of Public Economics 2, 319–37.
- BOHN, H. (1991), "Budget Balance through Revenue or Spending Adjustments?" Journal of Monetary Economics 27, 333-59.
- —— (1995), "The Sustainability of Budget Deficits in a Stochastic Economy," Journal of Money, Credit and Banking 27, 257–71.
- (1998), "The Behavior of US Public Debt and Deficits," *Quarterly Journal of Economics* 113, 949–63.
- (2008), "The Sustainability of Fiscal Policy in the United States," in: NECK, R. and J.-E. STURM (eds.), Sustainability of Public Debt, MIT Press: Cambridge, MA, 15–49.
- BOLTON, P., X. FREIXAS, and J. SHAPIRO (2012), "The Credit Ratings Game," Journal of Finance 67, 85–111.
- BOLTON, P. and O. JEANNE (2011), "Sovereign Default Risk and Bank Fragility in Financially Integrated Economies," *IMF Economic Review* 59, 162–94.
- BORDO, M., A. MARKIEWICZ, and L. JONUNG (2011), "A Fiscal Union for the Euro. Some Lessons from History," *NBER Working Paper* 17380, September 2011.
- BRENDEL, M., J. EBERL, and C. WEBER (2015), "Riskante Risikokontrolle," *ifo Schnelldienst* 68, 41–9.
- BRESCIANI-TURRONI, C. (1937), The Economics of Inflation. A Study of Currency Depreciation in Post-War Germany, Allen & Unwin: London.
- BRIXI, H. P. (1998), "Contingent Government Liabilities. A Hidden Risk for Fiscal Stability," World Bank Policy Research Working Paper 1989, October 1998.

—— (2005), "Addressing Contingent Liabilities and Fiscal Risk," in: SHAH, A. (ed.), *Fiscal Management*, World Bank: Washington, 163–90.

- BRIXI, H. P. and A. MODY (2002), "Dealing with Government Fiscal Risk. An Overview," in: BRIXI, H. P. and A. SCHICK (eds.), Government at Risk: Contingent Liabilities and Fiscal Risk, World Bank: Washington, DC, 21-58.
- BRUNNERMEIER, M. K. (2009), "Deciphering the Liquidity and Credit Crunch 2007-2008," Journal of Economic Perspectives 23, 77-100.

(2012), "Macroprudential Regulation. Optimizing the Currency Area," in: EUROPEAN CENTRAL BANK (ed.), *The Great Financial Crisis. Lessons for Financial Stability and Monetary Policy*, European Central Bank: Frankfurt am Main, 29–39.

- BUCK, F. (2013), "Financial Regulation and the Grabbing Hand," *CESifo DICE Report* 11, 3–13.
- BUITER, W. H. (1985), "A Guide to Public Sector Debt and Deficits," *Economic Policy* 1, 13–79.
- (2002), "The Fiscal Theory of the Price Level. A Critique," *Economic Journal* 112, 459–80.
- (2004), "Two Naked Emperors? Concerns About the Stability and Growth Pact and Second Thoughts About Central Bank Independence," *Fiscal Studies* 25, 249–77.
- —— (2005), "New Developments in Monetary Economics. Two Ghosts, Two Eccentricities, a Fallacy, a Mirage and a Mythos," *Economic Journal* 115, 1–31.
- (2007), "Seignorage," NBER Working Paper 12919, February 2007.
- ——— (2008a), "Can Central Banks Go Broke?" CEPR Policy Insight 24, May 2008.
- (2008b), "Monetary Economics and the Political Economy of Central Banking," *mimeo*.
- BUITER, W. H., J. MICHELS, and E. RAHBARI (2011), "ELA. An Emperor without Clothes?" *Citigroup Global Economics View*, January 2011.
- BUITER, W. H. and E. RAHBARI (2012a), "Looking into the Deep Pockets of the ECB," *Citigroup Global Economics View*, February 2012.

— (2012b), "The European Central Bank as Lender of Last Resort for Sovereigns in the European," Journal of Common Market Studies 50, 6-35.

BUITER, W. H. and A. C. SIBERT (2005), "How the Eurosystem's Treatment of Collateral in its Open Market Operations Weakens Fiscal Discipline in the Eurozone (and What To Do About It)," CEPR Discussion Paper 5387, December 2005.

BURNSIDE, C. (2004), "Assessing New Approaches to Fiscal Sustainability Analysis," mimeo.

— (2005a), "Some Tools for Fiscal Sustainability Analysis," in: BURNSIDE, C. (ed.), Fiscal Sustainability in Theory and Practice, World Bank: Washington, DC, 35–79.

- BURNSIDE, C. (2005b), "Theoretical Prerequisites for Fiscal Sustainability Analysis," in: BURN-SIDE, C. (ed.), Fiscal Sustainability in Theory and Practice, World Bank: Washington, DC, 11-34.
- CAGAN, P. D. (1956), "The Monetary Dynamics of Hyperinflation," in: FRIEDMAN, M. (ed.), Studies in the Quantity Theory of Money, University of Chicago Press: Chicago, IL, 25–43.
- CALOMIRIS, C. W. (2009), "A Recipe for Ratings Reform," The Economists' Voice 6, 1-4.
- CALVO, G. A. (1978), "Optimal Seignorage from Monetary Creation. An Analysis in Terms of the Optimum Balance of Payments Deficit Problem," *Journal of Monetary Economics* 4, 503–17.
- CANZONERI, M. B., R. E. CUMBY, and B. T. DIBA (2001), "Is the Price Level Determined by the Needs of Fiscal Solvency?" *American Economic Review* 91, 1221–38.
- CARGILL, T. F. (2005), "Is the Bank of Japan's Financial Structure an Obstacle to Policy?" *IMF Staff Papers* 52, 311–34.
- CARUANA, J. (2012), "Why Central Bank Balance Sheets Matter," *BIS Papers* 66, October 2012.
- CASSARD, M. and D. FOLKERTS-LANDAU (1997), "Risk Management of Sovereign Assets and Liabilities," *IMF Working Paper* 97/166, December 1997.
- CASSOLA, N. and F. KOULISCHER (2014), "The Collateral Channel of Monetary Policy. Evidence from the European Central Bank," *mimeo*.
- CEBOTARI, A., J. M. DAVIS, L. LUSINYAN, A. MATI, P. MAURO, M. PETRIE, and R. VELLOSO (2009), *Fiscal Risks. Sources, Disclosure, and Management*, International Monetary Fund: Washington, DC.
- CHAILLOUX, A., S. GRAY, U. H. KLÜH, S. SHIMIZU, and P. STELLA (2008a), "Central Bank Response to the 2007-08 Financial Market Turbulence. Experiences and Lessons Drawn," *IMF Working Paper* 08/210, September 2008.
- CHAILLOUX, A., S. GRAY, and R. MCCAUGHRIN (2008b), "Central Bank Collateral Frameworks. Principles and Policies," *IMF Working Paper* 08/222, September 2008.
- CHALK, N. A. and R. HEMMING (2000), "Assessing Fiscal Sustainability in Theory and Practice," *IMF Working Paper* 00/81, April 2000.
- CHAMON, M., J. SCHUMACHER, and C. TREBESCH (2014), "Foreign Law Bonds. Can They Reduce Sovereign Borrowing Costs?" mimeo.
- CHAPMAN, J., J. CHIU, and M. MOLICO (2010), "Central Bank Haircut Policy," Bank of Canada Working Paper 23, October 2010.
- CHARI, V. V. and R. JAGANNATHAN (1988), "Banking Panics, Information, and Rational Expectations Equilibrium," *Journal of Finance* 43, 749–61.
- CHEUN, S., I. v. KOEPPEN-MERTES, and B. WELLER (2009), "The Collateral Frameworks of the Eurosystem, the Federal Reserve System and the Bank of England and the Financial Market Turmoil," *ECB Occasional Paper* 107, December 2009.
- CHOUDHRY, M. (2006), An Introduction to Repo Markets, John Wiley & Sons: Chichester.
- CHRIST, C. F. (1968), "A Simple Macroeconomic Model with a Government Budget Restraint," Journal of Political Economy 76, 53–67.
- CLAESSENS, S. M., M. A. KOSE, and M. E. TERRONES (2014), "The Global Financial Crisis. How Similar? How Different? How Costly?" In: CLAESSENS, S. M., M. A. KOSE, L. LAEVEN, and F. VALENCIA (eds.), *Financial Crises. Causes, Consequences, and Policy Responses*, International Monetary Fund: Washington, DC, 209–37.

- CLAEYS, G. (2014), "The (not so) Unconventional Monetary Policy of the European Central Bank since 2008," *European Parliament Policy Department*, June 2014.
- CLAEYS, G., Z. DARVAS, S. MERLER, and G. B. WOLFF (2014), "Addressing Weak Inflation. The European Central Bank's Shopping List," *Bruegel Policy Contribution* 2014/5, May 2014.
- COCHRANE, J. H. (2001), "Long–Term Debt and Optimal Policy in the Fiscal Theory of the Price Level," *Econometrica* 69, 69–116.
- COEURDACIER, N. and H. REY (2013), "Home Bias in Open Economy Financial Macroeconomics," *Journal of Economic Literature* 51, 63–115.
- COEURÉ, B. (2012), "Collateral Scarcity. A Gone or Going Concern?" Speech at the ECB-DNB Joint Central Bank Seminar on Collateral and Liquidity, Amsterdam, 1 October 2012.
- COHEN, D. (2002), "Fiscal Sustainability and a Contingency Trust Fund," in: BRIXI, H. P. and A. SCHICK (eds.), *Government at Risk: Contingent Liabilities and Fiscal Risk*, World Bank: Washington, DC, 143–57.
- COMOTTO, R. (2012), *Haircuts and Initial Margins in the Repo Market*, International Capital Market Association: Zurich.
 - —— (2013), "Shadow Banking. Minimum Haircuts on Collateral," *European Parliament Policy Department*, July 2013.
- COOPER, R. and K. NIKOLOV (2013), "Government Debt and Banking Fragility. The Spreading of Strategic Uncertainty," *NBER Working Paper* 19278, August 2013.
- COUR-THIMANN, P. (2013), "Target Balances and the Crisis in the Euro Area," *CESifo Forum* 14, 1–50.
- COUR-THIMANN, P. and B. WINKLER (2013), "The ECB's Non-Standard Monetary Policy Measures. The Role of Institutional Factors and Financial Structure," *ECB Working Paper* 1528, April 2013.
- CRUCES, J. J. and C. TREBESCH (2013), "Sovereign Defaults. The Price of Haircuts," American Economic Journal: Macroeconomics 5, 85–117.
- CUKIERMAN, A. (2011), "Central Bank Finances and Independence. How Much Capital Should a Central Bank Have?" In: MILTON, S. and P. SINCLAIR (eds.), *The Capital Needs of Central Banks*, Routledge: New York City, NY, 33–46.
- CUKIERMAN, A., S. EDWARDS, and G. TABELLINI (1992), "Seignorage and Political Instability," American Economic Review 82, 537–55.
- CUMMINS, J. D. and O. MAHUL (2009), Catastrophe Risk Financing in Developing Countries. Principles for Public Intervention, World Bank: Washington, DC.
- CURRIE, E. and A. VELANDIA (2002), Risk Management of Contingent Liabilities within a Sovereign Asset-Liability Framework, World Bank: Washington, DC.
- CUTLER, D. M. and R. J. ZECKHAUSER (1999), "Reinsurance for Catastrophes and Cataclysms," in: FROOT, K. A. (ed.), *The Financing of Catastrophe Risk*, University of Chicago Press: Chicago, IL, 233–74.
- DALTON, J. and C. H. DZIOBEK (2005), "Central Bank Losses and Experiences in Selected Countries," *IMF Working Paper* 05/72, April 2005.
- DANG, T. V., G. GORTON, and B. HOLMSTRÖM (2013a), "Haircuts and Repo Chains," *mimeo.* (2013b), "The Information Sensitivity of a Security," *mimeo*.

- DARBYSHIRE, R. (2011), "Financing the Central Bank. Capital Adequacy and Financial Independence. An Accountant's Perspective," in: MILTON, S. and P. SINCLAIR (eds.), *The Capital Needs of Central Banks*, Routledge: New York City, NY, 69–82.
- DAS, U. S., M. G. PAPAIOANNOU, and C. TREBESCH (2012), "Sovereign Debt Restructurings 1950-2010: Literature Survey, Data, and Stylized Facts," *IMF Working Paper* 12/203, August 2012.
- DAVIES, M. and T. NG (2011), "The Rise of Sovereign Credit Risk. Implications for Financial Stability," *BIS Quarterly Review*, September 2011, 59–70.
- DE GRAUWE, P. and Y. JI (2013), "Fiscal Implications of the ECB's Bond Buying Program," Open Economies Review 24, 843–52.
- DEL NEGRO, M. and C. A. SIMS (2014), "When Does a Central Bank's Balance Sheet Require Fiscal Support?" *mimeo*.
- DEUTSCHE BUNDESBANK (2014), "Implications of the Eurosystem's Monetary Operations During the Financial Crisis," *Deutsche Bundesbank Monthly Report*, April 2014.
 - (2015), Allgemeine Geschaeftsbedingungen der Deutschen Bundesbank. Bankrechtliche Regelungen 5, Deutsche Bundesbank: Frankfurt am Main.
- DEWATRIPONT, M. and J. TIROLE (1993), "Efficient Governance Structure. Implications for Banking Regulation," in: MAYER, C. and X. VIVES (eds.), *Capital Markets and Financial Intermediation*, Cambridge University Press: Cambridge, 12–35.
- (1994), The Prudential Regulation of Banks, MIT Press: Cambridge, MA.
- DIAMOND, D. W. and P. H. DYBVIG (1983), "Bank Runs, Deposit Insurance, and Liquidity," Journal of Political Economy 91, 401–19.
- DIAMOND, D. W. and R. G. RAJAN (2011), "Fear of Fire Sales, Illiquidity Seeking, and Credit Freezes," *Quarterly Journal of Economics* 126, 557–91.
- DIAMOND, P. A. (1965), "National Debt in a Neoclassical Growth Model," *American Economic Review* 55, 1126–50.
- DIPPELSMAN, R., C. H. DZIOBEK, and C. G. MANGAS (2012), "What Lies Beneath: The Statistical Definition of Public Sector Debt," *IMF Discussion Note* 12/09, July 2012.
- DOLLS, M., C. FUEST, D. NEUMANN, and A. PEICHL (2014), "Fiscal Integration in the Eurozone. Economic Effects of Two Key Scenarios," *EUROMOD Working Paper 1/14*, January 2014.
- DOMANSKI, D., R. MOESSNER, and W. NELSON (2014), "Central Banks as Lender of Last Resort. Experiences During the 2007-2010 Crisis and Lessons for the Future," *Federal Reserve Board Finance and Economics Discussion Series* 110, May 2014.
- DOMAR, E. D. (1944), "The 'Burden of the Debt' and the National Income," *American Economic Review* 34, 798–827.
- DORNBUSCH, R. (1988), "The European Monetary System, the Dollar and the Yen," in: GI-AVAZZI, F., S. MICOSSI, and M. MILLER (eds.), The European Monetary System, Cambridge University Press: Cambridge, 23–36.
- DRAZEN, A. (1985), "A General Measure of Inflation Tax Revenues," *Economic Letters* 17, 327–30.
- DRECHSLER, I., T. DRECHSEL, D. MARQUES-IBANEZ, and P. SCHNABL (2015), "Who Borrows from the Lender of Last Resort?" *Journal of Finance*, forthcoming.
- DUFFIE, D. (1996), "Special Repo Rates," Journal of Finance 51, 493-526.

- DUNNE, P. G., M. J. FLEMING, and A. ZHOLOS (2011), "Repo Market Microstructure in Unusual Monetary Policy Conditions," *Central Bank of Ireland Research Technical Paper* 08/RT/11, August 2011.
- EASTERLY, W. R., P. MAURO, and K. SCHMIDT-HEBBEL (1995), "Money Demand and Seignorage-Maximizing Inflation," *Journal of Money, Credit and Banking* 27, 583–603.
- EBERL, J. and D. JUS (2012), "The Year of the Cat. Taxing Nuclear Risk with the Help of Capital Markets," *Energy Policy* 51, 364–73.
- EBERL, J. and C. WEBER (2013), "Quality Effects of ECB Collateral Policy," mimeo.
- (2014a), "ECB Collateral Criteria. A Narrative Database 2001–2013," *Ifo Working Paper* 174, February 2014.
- (2014b), "ECB Risk Control. An Analysis of ECB Haircuts 2001–2013," mimeo.
- (2015), "Lessons from the Eurosystem Collateral Pool 2007–2013," mimeo.
- EDWARDS, A. K., L. E. HARRIS, and M. S. PIWOWAR (2007), "Corporate Bond Market Transaction Costs and Transparency," *Journal of Finance* 62, 1421–51.
- ERNHAGEN, T., M. VESTERLUND, and S. VIOTTI (2002), "How Much Equity Does a Central Bank Need?" Sveriges Riksbank Economic Review 2/2002, 5–18.
- ESER, F. and B. SCHWAB (2013), "Assessing Asset Purchases within the ECB's Securities Market Programme," *ECB Working Paper* 1587, September 2013.
- EUROCLEAR (2009), Understanding Repos and the Repo Markets, Euroclear: Brussels.
- EUROPEAN CENTRAL BANK (2000), "Guideline of the European Central Bank of 31 October 2000 on Monetary Policy Instruments and Procedures of the European," Official Journal of the European Communities L 310, 1–82.

(2003), "Guideline of the European Central Bank of 1 December 2003 Amending Guideline ECB/2000/7 on Monetary Policy Instruments and Procedures of the European," *Official Journal of the European Union* L 69, 1–85.

— (2005), "Guideline of the European Central Bank of 3 February 2005 Amending Guideline ECB/2000/7 on Monetary Policy Instruments and Procedures of the European," Official Journal of the European Union L 111, 1-83.

(2006a), "Guideline of the European Central Bank of 31 August 2006 Amending Guideline ECB/2000/7 on Monetary Policy Instruments and Procedures of the European," Official Journal of the European Union L 352, 1–90.

— (2006b), "The Contribution of the ECB and the Eurosystem to European Financial Integration," *ECB Monthly Bulletin*, May 2006, 61-73.

(2006c), "The Single List in the Collateral Framework of the Eurosystem," *ECB Monthly Bulletin*, May 2006, 75–87.

— (2007a), Annual Report 2007, European Central Bank: Frankfurt am Main.

(2007b), "The Collateral Frameworks of the Federal Reserve System, the Bank of Japan and the Eurosystem," *ECB Monthly Bulletin*, October 2007, 85–100.

(2008a), Covered Bonds in the EU Financial System, European Central Bank: Frankfurt am Main.

(2008b), "Guideline of the European Central Bank of 23 October 2008 Amending Guideline ECB/2000/7 on Monetary Policy Instruments and Procedures of the European," *Official Journal of the European Union* L 36, 31–45.

- EUROPEAN CENTRAL BANK (2009), "Decision of the European Central Bank of 2 July 2009 on the Implementation of the Covered Bond Purchase Programme," Official Journal of the European Union L 175, 18–9.
 - (2010a), "Decision of the European Central Bank of 14 May 2010 Establishing a Securities Market Programme," Official Journal of the European Union L 124, 8–9.
 - (2010b), "Guideline of the European Central Bank of 16 September 2010 Amending Guideline ECB/2000/7 on Monetary Policy Instruments and Procedures of the European," *Official Journal of the European Union* L 267, 21–55.
 - (2010c), "The ECB's Monetary Policy Stance During the Financial Crisis," *ECB Monthly Bulletin*, January 2010, 63–71.
 - (2010d), "The ECB's Response to the Financial Crisis," *ECB Monthly Bulletin*, October 2010, 59–74.
 - (2011a), "Decision of the European Central Bank of 25 November 2010 on the Allocation of Monetary Income of the National Central Banks of Member States whose Currency is the Euro," Official Journal of the European Union L 35, 17–25.
 - (2011b), "Decision of the European Central Bank of 3 November 2011 on the Implementation of the Second Covered Bond Purchase Programme," Official Journal of the European Union L 297, 70–1.
 - (2011c), "Ensuring Fiscal Sustainability in the Euro Area," *ECB Monthly Bulletin*, April 2011, 61–77.
 - (2011d), "Guideline of the European Central Bank of 20 September 2011 on Monetary Policy Instruments and Procedures of the Europystem," Official Journal of the European Union L 331, 1–95.
 - (2011e), "The ECB's Non-Standard Measures. Impact and Phasing-Out," *ECB Monthly Bulletin*, July 2011, 55–69.
- (2011f), The Monetary Policy of the ECB, European Central Bank: Frankfurt am Main.
- (2012a), Convergence Report. May 2012, European Central Bank: Frankfurt am Main.
- (2012b), Euro Money Market Study, European Central Bank: Frankfurt am Main.
- (2012c), "Financial Reporting in the Eurosystem," *ECB Monthly Bulletin*, April 2012, 87–98.
- (2012d), "Guideline of the European Central Bank of 26 November 2012 Amending Guideline ECB/2011/14 on Monetary Policy Instruments and Procedures of the European," Official Journal of the European Union L 348, 30–41.
- (2012e), "Protocol (No 4) on the Statute of the European System of Central Banks and of the European Central Bank," *Official Journal of the European Union* C 326, 230–50.
- —— (2013a), Collateral Eligibility Requirements. A Comparative Study across Specific Frameworks, European Central Bank: Frankfurt am Main.
- (2013b), ELA Procedures, European Central Bank: Frankfurt am Main.
- (2013c), "The Eurosystem Collateral Framework Throughout the Crisis," *ECB Monthly Bulletin*, July 2013, 71–86.
- (2014a), Collateral Eligibility and Availability. Follow-up to the Report on 'Collateral Eligibility Requirements' Dated July 2013, European Central Bank: Frankfurt am Main.
- (2014b), "Decision of the European Central Bank of 15 October 2014 on the Implementation of the Third Covered Bond Purchase Programme," Official Journal of the European Union L 335, 22-4.

- EUROPEAN CENTRAL BANK (2014c), Euro Money Market Survey, European Central Bank: Frankfurt am Main.
 - —— (2014d), Euro Repo Market. Improvements for Collateral and Liquidity Management, European Central Bank: Frankfurt am Main.
 - (2015a), "Decision of the European Central Bank of 19 November 2014 on the Implementation of the Asset-Backed Securities Purchase Programme," Official Journal of the European Union L 1, 4–7.
 - (2015b), "Guideline (EU) 2015/510 of the European Central Bank of 19 December 2014 on the Implementation of the Europystem Monetary Policy Framework (ECB/2014/60)," Official Journal of the European Union L 91, 3–135.
- (2015c), The Financial Risk Management of the Eurosystem's Monetary Policy Operations, European Central Bank: Frankfurt am Main.
- EUROPEAN COMMISSION (2000), "Directive 2000/12/EC of the European Parliament and of the Council of 20 March 2000 Relating to the Taking up and Pursuit of the Business of Credit Institutions," Official Journal of the European Communities L 126, 1–59.
 - (2012), "Fiscal Sustainability Report 2012," European Economy 8, 1–204.
- EUROPEAN ECONOMIC COMMUNITY (1983), "Seventh Council Directive 83/349/EEC of 13 June 1983 Based on the Article 54 (3) (g) of the Treaty on Consolidated Accounts," Official Journal of the European Communities L 193, 1–17.
- EUROPEAN PARLIAMENT (2012), "Regulation (EU) of the European Parliament and of the Council of 4 July 2012 on OTC Derivatives, Central Counterparties and Trade Repositories," Official Journal of the European Union L 201, 1–59.
- EUROPEAN REPO COUNCIL (2012), Haircuts and Initial Margins in the Repo Market, International Capital Market Association: Zurich.
- EUROPEAN UNION (2012), "Consolidated Version of the Treaty on the Functioning of the European Union," Official Journal of the European Union C 326, 47–390.
- EWERHART, C., N. CASSOLA, and N. VALLA (2006), "Declining Valuations and Equilibrium Bidding in Central Bank Refinancing Operations," *ECB Working Paper* 668, August 2006.
- EWERHART, C. and J. TAPKING (2008), "Repo Markets, Counterparty Risk, and the 2007/2008 Liquidity Crisis," *ECB Working Paper* 909, June 2008.
- FAMA, E. F. (1977), "Risk-Adjusted Discount Rates and Capital Budgeting Under Uncertainty," Journal of Financial Economics 5, 3–24.
- FARHI, E. and J. TIROLE (2012), "Collective Moral Hazard, Maturity Mismatch, and Systemic Bailouts," *American Economic Review* 102, 60–93.
- (2014), "Deadly Embrace. Sovereign and Financial Balance Sheets Doom Loops," *mimeo*.
- FECHT, F., K. G. NYBORG, J. ROCHOLL, and J. WOSCHITZ (2015), "Collateral, Central Bank Repos, and Systemic Arbitrage," *mimeo*.
- FISCHER, S. (1982), "Seignorage and the Case for a National Money," *Journal of Political Economy* 90, 295–313.
- (1995), "Central-Bank Independence Revisited," American Economic Review 85, 201–6.
- FISCHER, S., R. SAHAY, and C. A. VÉGH (2002), "Modern Hyper-and High Inflations," *Journal* of Economic Literature 40, 837–80.
- FITCH RATINGS (2012), "Repo Emerges from the Shadow," Fitch Ratings Macro Credit Research, 3 February 2012.

- FRENCH, K. R. and J. M. POTERBA (1991), "Investor Diversification and International Equity Markets," *American Economic Review* 81, 222–6.
- FRIEDMAN, M. (ed.) (1956), Studies in the Quantity Theory of Money, University of Chicago Press: Chicago, IL.
 - (1971), "Government Revenue from Inflation," Journal of Political Economy 79, 846–56.
- FRIEDMAN, M. and A. J. SCHWARTZ (1963), A Monetary History of the United States, 1867-1960, Princeton University Press: Princeton, NJ.
- FRY, M. J. (1993), "The Fiscal Abuse of Central Banks," IMF Working Paper 93/58, July 1993.
- FUEST, C. (2013), Stellungnahme gegenüber dem Bundesverfassungsgericht, Zentrum für Europäische Wirtschaftsforschung: Mannheim.
- FUEST, C. and A. PEICHL (2012), "European Fiscal Union. What Is It? Does It Work? And Are There Really 'No Alternatives'?" *CESifo Forum* 13, 3–9.
- FUEST, C. and H.-W. SINN (2015), "Die Risiken der Notkredite," oekonomenstimme.org, 13 November 2015.
- GARBADE, K. D. (2006), "The Evolution of Repo Contracting Conventions in the 1980s," Federal Reserve Bank of New York Economic Policy Review 12, 27–42.
- GENNAIOLI, N., A. MARTIN, and S. ROSSI (2014), "Sovereign Default, Domestic Banks, and Financial Institutions," *Journal of Finance* 69, 819–66.
- GHYSELS, E., J. IDIER, S. MANGANELLI, and O. VERGOTE (2014), "A High Frequency Assessment of the ECB Securites Markets Programme," *ECB Working Paper* 1642, February 2014.
- GIAMMARIOLI, N., C. NICKEL, P. ROTHER, and J.-P. VIDAL (2007), "Assessing Fiscal Soundness. Theory and Practice," *ECB Occasional Paper* 56, March 2007.
- GIANNETTI, M. and L. LAEVEN (2012), "The Flight Home Effect. Evidence from the Syndicated Loan Market During Financial Crises," *Journal of Financial Economics* 104, 23–43.
- GIAVAZZI, F. and A. GIOVANNINI (1989), Limiting Exchange Rate Flexibility. The European Monetary System, MIT Press: Cambridge, MA.
- GLICK, R. and S. LEDUC (2012), "Central Bank Announcements of Asset Purchases and the Impact on Global Financial and Commodity Markets," *Journal of International Money and Finance* 31, 2078–101.
- GOLDSTEIN, I. (2012), "Empirical Literature on Financial Crises. Fundamentals vs. Panic," in: CAPRIO, G. (ed.), *The Evidence and Impact of Financial Globalization*, Academic Press: Amsterdam, 523-34.
- GOLDSTEIN, M. A., E. S. HOTCHKISS, and E. R. SIRRI (2007), "Transparency and Liquidity. A Controlled Experiment on Corporate Bonds," *Review of Financial Studies* 20, 235–73.
- GONZÁLEZ, F. and P. MOLITOR (2009), "Risk Mitigation Measures and Credit Risk Assessment in Central Bank Policy Operations," in: BINDSEIL, U., F. GONZÁLEZ, and E. TABAKIS (eds.), *Risk Management for Central Banks and Other Public Investors*, Cambridge University Press: Cambridge, 303–39.
- GOODFRIEND, M. (2007), "How the World Achieved Consensus on Monetary Policy," *Journal* of Economic Perspectives 21, 47–68.
- GOODHART, C. A. E. (1999), "Myths About the Lender of Last Resort," *International Finance* 2, 339–60.
- GORTON, G. (1988), "Banking Panics and Business Cycles," Oxford Economic Papers 40, 751–81.

- GORTON, G. and A. METRICK (2010), "Haircuts," Federal Reserve Bank of St. Louis Review 92, 507–19.
- (2012), "Securitized Banking and the Run on Repo," *Journal of Financial Economics* 104, 425–51.
- GRILLI, V. U. (1988), "Seignorage in Europe," NBER Working Paper 2778, November 1988.
- GROS, D. (1989), "Seignorage in the EC. The Implications of the EMS and Financial Market Integration," *IMF Working Paper* 89/7, January 1989.
 - (1993), "Seignorage and EMU. The Fiscal Implications of Price Stability and Financial Market Integration," *European Journal of Political Economy* 9, 581–601.
 - (2004), "Financial Aspects of Central Bank Independence and Price Stability. The Case of Turkey," *CEPS EU-Turkey Working Paper* 12, September 2004.
- GROS, D., C. ALCIDI, and A. GIOVANNI (2012), "Central Banks in Times of Crisis. The FED versus the ECB," *European Parliament Policy Department*, June 2012.
- GROSSMAN, H. I. and J. B. VAN HUYCK (1988), "Sovereign Debt as a Contingent Claim. Excusable Default, Repudiation, and Reputation," *American Economic Review* 78, 1088– 97.
- GROSSMAN, S. J. and O. D. HART (1980), "Disclosure Laws and Takeover Bids," *Journal of Finance* 35, 323–34.
- GUILFORD, J. P. (1954), Psychometric Methods, McGraw-Hill: New York City, NY.
- GULATI, M., C. TREBESCH, and J. ZETTELMEYER (2013), "The Greek Debt Restructuring. An Autopsy," *Economic Policy* 28, 513–69.
- HAAS, R. d. and N. VAN HOREN (2012), "Running for the Exit? International Bank Lending During a Financial Crisis," *Review of Financial Studies* 26, 244–85.
- HABER, G. and R. NECK (2008), "The Long Shadow of 'Austrokeynesianism'? Public Debt Sustainability in Austria," in: NECK, R. and J.-E. STURM (eds.), Sustainability of Public Debt, MIT Press: Cambridge, MA, 107–30.
- HAIGHT, F. A. (1967), Handbook of the Poisson Distribution, Wiley: New York.
- HAKKIO, C. S. and M. RUSH (1991), "Is the Budget Deficit Too Large?" *Economic Inquiry* 29, 429–45.
- HALL, R. E. and R. REIS (2015), "Maintaining Central-Bank Solvency under New-Style Central Banking," NBER Working Paper 21173, May 2015.
- HAMILTON, J. D. and M. A. FLAVIN (1986), "On the Limitations of Government Borrowing. A Framework for Empirical Testing," *American Economic Review* 76, 808–19.
- HANSEN, A. H. (1941), Fiscal Policy and Business Cycles, Norton: New York City, NY.
- HANSEN, B. (1958), The Economic Theory of Fiscal Policy, Allen & Unwin: London.
- HAWKINS, J. (2003), "Central Bank Balance Sheets and Fiscal Operations," in: BANK FOR IN-TERNATIONAL SETTLEMENTS (ed.), Fiscal Issues and Central Banking in Emerging Economies, Bank for International Settlements: Basle, 71–83.
- HEINLE, E. and M. KOIVU (2009), "Risk Measurement for a Repo Portfolio. An Application to the Eurosystem's Collateralized Lending Operations," in: BINDSEIL, U., F. GONZÁLEZ, and E. TABAKIS (eds.), Risk Management for Central Banks and Other Public Investors, Cambridge University Press: Cambridge, 359–93.
- HOFFMANN, D. (2015), Die EZB in der Krise. Eine Analyse der wesentlichen Sondermaßnahmen von 2007 bis 2012, Pro Business: Berlin.

- HOFMANN, C. (2011), "Central Bank Collateral and the Lehman Collapse," *Capital Markets Law Journal* 6, 456–69.
- HONOHAN, P. (1996), "Does It Matter how Seignorage Is Measured?" Applied Financial Economics 6, 293–300.
- ILLING, G. and P. KÖNIG (2014), "Die Europäische Zentralbank als Lender of Last Resort," DIW Wochenbericht 24, 541–54.
- INTERNATIONAL CAPITAL MARKET ASSOCIATION (2007), European Repo Market Survey 2007, International Capital Market Association: Zurich.

(2008), European Repo Market Survey 2008, International Capital Market Association: Zurich.

- (2009), European Repo Market Survey 2009, International Capital Market Association: Zurich.
- (2010), European Repo Market Survey 2010, International Capital Market Association: Zurich.
- (2011), European Repo Market Survey 2011, International Capital Market Association: Zurich.
- (2012), European Repo Market Survey 2012, International Capital Market Association: Zurich.
- (2013), European Repo Market Survey 2013, International Capital Market Association: Zurich.
- (2014), European Repo Market Survey 2014, International Capital Market Association: Zurich.
- INTERNATIONAL FEDERATION OF ACCOUNTANTS (2014), Handbook of International Public Sector Accounting Pronouncements, International Federation of Accountants: New York City, NY.
- INTERNATIONAL MONETARY FUND (2002), Assessing Sustainability, International Monetary Fund: Washington, DC.
 - (2003), Sustainability Assessments. Review of Application and Methodological Refinements, International Monetary Fund: Washington, DC.
 - (2005), *Government Guarantees and Fiscal Risk*, International Monetary Fund: Washington, DC.
 - (2013), External Debt Statistics. Guide for Compilers and Users, International Monetary Fund: Washington, DC.
 - (2014a), Global Financial Stability Report. Moving from Liquidity- to Growth-Driven Markets, International Monetary Fund: Washington, DC.
 - (2014b), *Government Finance Statistics Manual 2014*, International Monetary Fund: Washington, DC.
- IZE, A. (2005), "Capitalizing Central Banks. A Net Worth Approach," IMF Staff Papers 52, 289-310.
- (2006), "Spending Seignorage. Do Central Banks Have a Governance Problem?" *IMF Working Paper* 06/58, March 2006.
- IZE, A. and N. OULIDI (2009), "Why Do Central Banks Go Weak?" IMF Working Paper 09/13, January 2009.

- JACKLIN, C. J. and S. BHATTACHARYA (1988), "Distinguishing Panics and Information-Based Bank Runs. Welfare and Policy Implications," *Journal of Political Economy* 96, 568–92.
- JEANNE, O. and L. E. O. SVENSSON (2007), "Credible Commitment to Optimal Escape from a Liquidity Trap. The Role of the Balance Sheet of an Independent Central Bank," *American Economic Review* 97, 474–90.
- JOKIVUOLLE, E. and S. PEURA (2000), "A Model for Estimating Recovery Rates and Collateral Haircuts for Bank Loans," *Bank of Finland Discussion Paper* 2/2000, March 2000.
- JORDAN, T. J. (1994), Seignorage, Defizite, Verschuldung und Europäische Währungsunion, Paul Haupt: Bern.
- KATZ, J. G., E. S. MUNOZ, and C. STEPHANOU (2009), "Credit Rating Agencies. No Easy Regulatory Solutions," World Bank Crisis Response Policy Brief 8, October 2009.
- KELLY, B. T., H. N. LUSTIG, and S. VAN NIEUWERBURGH (2012), "Too-Systemic-To-Fail. What Option Markets Imply About Sector-Wide Government Guarantees," *Fama-Miller Working Paper* 11, March 2012.
- KING, R. G. and C. I. PLOSSER (1985), "Money, Deficits, and Inflation," Carnegie-Rochester Conference Series on Public Policy 22, 147–96.
- KLEIN, M. and M. J. M. NEUMANN (1990), "Seignorage. What Is It and Who Gets It?" Weltwirtschaftliches Archiv 126, 205–21.
- KLUGMAN, S. A., H. H. PANJER, and G. E. WILLMOT (2012), Loss Models. From Data to Decisions, Wiley: Hoboken, NJ.
- KLÜH, U. H. and P. STELLA (2008), "Central Bank Financial Strength and Policy Performance. An Econometric Evaluation," *IMF Working Paper* 08/176, July 2008.
- KOTLIKOFF, L. J. and B. RAFFELHÜSCHEN (1999), "Generational Accounting Around the Globe," *American Economic Review* 89, 161–6.
- KOULISCHER, F. (2015), "Asymmetric Shocks in a Currency Union. The Role of Central Bank Collateral Policy," *mimeo*.
- KOULISCHER, F. and D. STRUYVEN (2014), "Central Bank Liquidity Provision and Collateral Quality," Journal of Banking & Finance 49, 113–30.
- KRISHNAMURTHY, A., S. NAGEL, and D. ORLOV (2014), "Sizing Up Repo," Journal of Finance 69, 2381–417.
- KUNREUTHER, H. (1997), "Rethinking Society's Management of Catastrophic Risks," Geneva Papers on Risk and Insurance 22, 151–76.
- KURTZIG, J. and B. MANDER (2003), "Survey of Central Bank Accounting Practices," in: COURTIS, N. and B. MANDER (eds.), *Accounting Standards for Central Banks*, Central Bank Publications: London, 21–46.
- LAEVEN, L. and F. VALENCIA (2008), "Systemic Banking Crises. A New Database," *IMF Working Paper* 08/224, November 2008.
- (2012), "Systemic Banking Crises Database. An Update," *IMF Working Paper* 12/163, June 2012.
- (2014), "Resolution of Banking Crises. The Good, the Bad, and the Ugly," in: CLAESSENS,
 S. M., M. A. KOSE, L. LAEVEN, and F. VALENCIA (eds.), *Financial Crises. Causes, Consequences, and Policy Responses*, International Monetary Fund: Washington, DC, 397-430.
- LANGE, C. (1995), Seignorage. Eine theoretische und empirische Analyse des staatlichen Geldschöpfungsgewinnes, Duncker & Humblot: Berlin.

- LANGENUS, G. (2006), "Fiscal Sustainability Indicators and Policy Design in the Face of Ageing," National Bank of Belgium Working Paper 102, October 2006.
- LASTRA, R. M. (2011), Cross-Border Bank Insolvency, Oxford University Press: Oxford.
- LATTUGA, E. and M. VALLI (2012), "Digging into ELA," UniCredit Economics & FI/FX Research, 29 May 2012.
- LEEPER, E. M. (1991), "Equilibria under 'Active' and 'Passive' Monetary and Fiscal Policies," Journal of Monetary Economics 27, 129–47.
- LEHMBECKER, P. (2009), The Quality of Eligible Collateral, Central Bank Losses and Monetary Stability. An Empirical Analysis, Peter Lang: Frankfurt am Main.
- LENZA, M., H. PILL, and L. REICHLIN (2010), "Monetary Policy in Exceptional Times," *Economic Policy* 25, 295–339.
- LEONE, A. M. (1994), "Institutional and Operational Aspects of Central Bank Losses," in: BALINO, T. J. and C. COTTARELLI (eds.), *Frameworks for Monetary Stability*, International Monetary Fund: Washington, DC, 738–55.
- LEVELS, A. and J. CAPEL (2012), "Is Collateral Becoming Scarce? Evidence for the Euro Area," Journal of Financial Market Infrastructures 1, 29–53.
- LEVY, A. and S. SCHICH (2010), "The Design of Government Guarantees for Bank Bonds. Lessons from the Recent Financial Crisis," *OECD Journal: Financial Market Trends* 1, 1–32.
- LEVY, A. and A. ZAGHINI (2010), "The Pricing of Government-Guaranteed Bank Bonds," *Banca d'Italia Working Paper* 753, March 2010.
- LEVY, H. and M. LEVY (2014), "The Home Bias is Here to Stay," Journal of Banking & Finance 47, 29–40.
- LEY, E. (2010), "Fiscal (and External) Sustainability," mimeo.
- LÖNNBERG, A. and P. STELLA (2008), "Issues in Central Bank Finance and Independence," *IMF Working Paper* 08/37, February 2008.
- MACAULAY, F. R. (1938), Some Theoretical Problems Suggested by the Movements of Interest Rates, Bond Yields and Stock Prices in the United States since 1856, National Bureau of Economic Research: New York City, NY.
- MACLEOD, H. D. (1856), *The Theory and Practice of Banking*, Longman, Brown, Green, Longmans and Roberts: London.
- MAHARAJ, T., M. PELATA, H. HAWORTH, and C. ARANDA-HASSEL (2012), "Wider ECB Collateral & LTROs," Credit Suisse Fixed Income Research, February 2012.
- MANCINI, L., A. RANALDO, and J. WRAMPELMEYER (2015), "The Euro Interbank Repo Market," *Review of Financial Studies*, forthcoming.
- MANGANELLI, S. (2012), "The Impact of the Securities Market Programme," ECB Research Bulletin 17, Winter 2012.
- MANKIW, N. G. (1987), "The Optimal Collection of Seignorage. Theory and Evidence," Journal of Monetary Economics 20, 327–41.
- MARCO, F. d. and M. MACCHIAVELLI (2014), "The Political Origin of Home Bias. The Case of Europe," *mimeo*.
- MARSHALL, J. (2003), "Fiscal Rule and Central Bank Issues in Chile," in: BANK FOR INTER-NATIONAL SETTLEMENTS (ed.), Fiscal Issues and Central Banking in Emerging Economies, Bank for International Settlements: Basle, 98–106.

- MARTÍNEZ-RESANO, J. R. (2004), "Central Bank Financial Independence," Banco de Espana Occasional Paper 401, January 2004.
- MARTY, A. L. (1967), "Growth and the Welfare Cost of Inflationary Finance," Journal of Political Economy 75, 71–6.
 - (1978), "Inflation, Taxes, and the Public Debt," *Journal of Money, Credit and Banking* 10, 437–52.
- MAS-COLELL, A., M. D. WHINSTON, and J. R. GREEN (1995), *Microeconomic Theory*, Oxford University Press: Oxford.
- MASSON, P. R. (1985), "The Sustainability of Fiscal Deficits," IMF Staff Papers 32, 577-605.
- MCCALLUM, B. T. (1984), "Are Bond-Financed Deficits Inflationary? A Ricardian Analysis," Journal of Political Economy 92, 123–35.
- MEANING, J. and F. ZHU (2011), "The Impact of Recent Central Bank Asset Purchase Programmes," *BIS Quarterly Review*, December 2011, 73–83.
- MENDOZA, E. G. and J. D. OSTRY (2008), "International Evidence on Fiscal Solvency. Is Fiscal Policy 'Responsible'?" *Journal of Monetary Economics* 55, 1081–93.
- MENDOZA, E. G. and P. M. OVIEDO (2006), "Fiscal Solvency and Macroeconomic Uncertainty in Emerging Markets. The Tale of the Tormented Insurer," *NBER Working Paper* 12586, October 2006.
- MERLER, S. and J. PISANI-FERRY (2012), "Who's Afraid of Sovereign Bonds?" Bruegel Policy Contribution 02/2012, February 2012.
- MILES, D. and J. SCHANZ (2014), "The Relevance or Otherwise of the Central Bank's Balance Sheet," *Journal of International Economics* 92, 103–16.
- MILGROM, P. R. (1981), "Good News and Bad News. Representation Theorems and Applications," *Bell Journal of Economics* 12, 380–91.
- MILLER, M. and L. ZHANG (2000), "Creditor Panic, Asset Bubbles and Sharks. Three Views of the Asian Crisis," in: DADUSH, U. B., D. DAS GUPTA, and M. UZAN (eds.), *Private Capital Flows in the Age of Globalization. The Aftermath of the Asian Crisis*, Edward Elgar: Cheltenham, 222–40.
- MOOG, S. and B. RAFFELHÜSCHEN (2012), "Explizite und Implizite Staatsverschuldung in Europa. Eine Tragfähigkeitsanalyse," Zeitschrift für Staats- und Europawissenschaften 10, 277–90.
- MULLER, P., S. DEVNANI, and R. FLYTKJAER (2012), "The Impact of State Guarantees on Banks' Debt Issuing Costs, Lending and Funding Policy," *European Economy Economic Papers* 447, January 2012.
- MUSGRAVE, R. A. (1959), The Theory of Public Finance. A Study in Public Economy, McGraw-Hill: New York City, NY.
- MUSGRAVE, R. A. and P. B. MUSGRAVE (1984), Public Finance in Theory and Practice, McGraw-Hill: New York City, NY.
- NAGEL, J. (2012), "Understanding Central Bank Balance Sheets," International Economy Summer 2012, 14–54.
- NECK, R. and J.-E. STURM (eds.) (2008), Sustainability of Public Debt, MIT Press: Cambridge, MA.
- NEUMANN, M. J. M. (1992), "Seignorage in the United States. How Much Does the US Government Make from Money Production?" Federal Reserve Bank of St. Louis Review 74, 29–40.

- NEUMANN, M. J. M. (1996), "A Comparative Study of Seignorage. Japan and Germany," BOJ Monetary and Economic Studies 14, 104–42.
- (2012), "The Refinancing of Banks Drives Target Debt," CESifo Forum 13, 29–32.
- NGUYEN, T. (2007), Grenzen der Versicherbarkeit von Katastrophenrisiken, Deutscher Universitäts-Verlag: Wiesbaden.
- NYBORG, K. G. (2015), "Central Bank Collateral Frameworks," Swiss Finance Institute Research Paper 15-10, February 2015.
- NYBORG, K. G., U. BINDSEIL, and I. A. STREBULAEV (2002), "Bidding and Performance in Repo Auctions. Evidence from ECB Open Market Operations," *ECB Working Paper* 157, July 2002.
- NYBORG, K. G. and P. ÖSTBERG (2014), "Money and Liquidity in Financial Markets," *Journal* of Financial Economics 112, 30–52.
- NYBORG, K. G. and I. A. STREBULAEV (2001), "Collateral and Short Squeezing of Liquidity in Fixed Rate Tenders," *Journal of International Money and Finance* 20, 769–92.
- O'CONNELL, S. P. and S. A. ZELDES (1988), "Rational Ponzi Games," International Economic Review 29, 431–50.
- OPP, C. C., M. M. OPP, and M. HARRIS (2013), "Rating Agencies in the Face of Regulation," Journal of Financial Economics 108, 46–61.
- ORGANIZATION FOR ECONOMIC CO-OPERATION AND DEVELOPMENT (2014), Government at a Glance 2013, OECD Publishing: Paris.
- PANJER, H. H. and G. E. WILLMOT (1992), *Insurance Risk Models*, Society of Actuaries: Schaumburg.
- PAPADOPOULOS, A. P. and M. G. SIDIROPOULOS (1999), "The Sustainability of Fiscal Policies in the European Union," *International Advances in Economic Research* 5, 289–307.
- PAPI, L. (2011), "Central Bank Capital Adequacy for Central Banks with or without a Monetary Policy," in: MILTON, S. and P. SINCLAIR (eds.), The Capital Needs of Central Banks, Routledge: New York City, NY, 96–106.
- PÂRIS, P. and C. WYPLOSZ (2014), "PADRE. Politically Acceptable Debt Restructuring in the Eurozone," Geneva Reports on the World Economy S3, 1–39.
- PATINKIN, D. (1956), Money, Interest and Prices, Row, Peterson & Co.: Evanston, IL.
- PERSSON, T. and G. TABELLINI (1993), "Designing Institutions for Monetary Stability," Carnegie-Rochester Conference Series on Public Policy 39, 53–84.
- PHELPS, E. S. (1961), "The Golden Rule of Accumulation. A Fable for Growthmen," *American Economic Review* 51, 638–43.
- (1973), "Inflation in the Theory of Public Finance," Swedish Journal of Economics 75, 67–82.
- PISANI-FERRY, J. (2013), "The Known Unknowns and Unknown Unknowns of European Monetary Union," *Journal of International Money and Finance* 34, 6–14.
- PORTES, R., H. REY, and Y. OH (2001), "Information and Capital Flows. The Determinants of Transactions in Financial Assets," *European Economic Review* 45, 783–96.
- POTERBA, J. M. (1997), "Do Budget Rules Work?" In: AUERBACH, A. J. (ed.), Fiscal Policy Lessons from Economic Research, MIT Press: Cambridge, MA, 53-86.
- POTERBA, J. M. and J. J. ROTEMBERG (1990), "Inflation and Taxation with Optimizing Governments," *Journal of Money, Credit and Banking* 22, 1–18.

- POTRAFKE, N. and M. REISCHMANN (2014), "Explosive Target Balances," *Economic Modelling* 42, 439–44.
- (2015), "Fiscal Transfers and Fiscal Sustainability," Journal of Money, Credit and Banking 47, 975–1005.
- PRINGLE, R. (2003a), "Central Bank (In)Solvency," Central Banking Journal 13, 1–2.
 - (2003b), "Why Central Banks Need Capital," Central Banking Journal 14, 76–80.
- REINHART, C. M. and K. S. ROGOFF (2014), "Financial and Sovereign Debt Crises. Some Lessons Learned and Those Forgotten," in: CLAESSENS, S. M., M. A. KOSE, L. LAEVEN, and F. VALENCIA (eds.), *Financial Crises. Causes, Consequences, and Policy Responses*, International Monetary Fund: Washington, DC, 141–56.
- REIS, R. (2013), "The Mystique Surrounding the Central Bank's Balance Sheet, Applied to the European Crisis," *American Economic Review* 103, 135–40.
- ROBICHECK, A. A. and S. C. MYERS (1966), "Conceptual Problems in the Use of Risk-Adjusted Discount Rates," *Journal of Finance* 21, 727–30.
- ROGOFF, K. S. (1985), "The Optimal Degree of Commitment to an Intermediate Monetary Target," *Quarterly Journal of Economics* 100, 1169–89.
- ROSENTHAL, R. and R. L. ROSNOW (1991), Essentials of Behavioral Research. Methods and Data Analysis, McGraw-Hill: New York City, NY.
- Ross, S. M. (2010), Introduction to Probability Models, Academic Press: Amsterdam.
- RUBINSTEIN, M. E. (1973), "A Mean-Variance Synthesis of Corporate Financial Theory," *Jour*nal of Finance 28, 167–81.
- RUCHIN, A. (2011), "Can Securities Lending Transactions Substitute for Repurchase Agreements Transactions?" *Banking Law Journal* 128, 450–80.
- RULE, G. (2012), "Collateral Management in Central Bank Policy Operations," Bank of England CCBS Handbook 31.
- RUSSEL, F. (1973), "Bubble, Bubble. No Toil, No Trouble," American Heritage Magazine 24, 74–86.
- SANGIORGI, F. and C. S. SPATT (2015), "Opacity, Credit Rating Shopping and Bias," mimeo.
- SARGENT, T. J. and N. WALLACE (1981), "Some Unpleasant Monetarist Arithmetic," Federal Reserve Bank of Minneapolis Quarterly Review 5, 1–17.
- SCHLESINGER, H. (2012), "The Balance of Payments Tells Us the Truth," CESifo Forum 13, 11–3.
- SCHOBERT, F. (2008), "Why Do Central Banks Make Losses?" Central Banking Journal 18, 79–83.
- SCHWARZ, C., P. KARAKITSOS, N. MERRIMAN, and W. STUDENER (2014), "Why Accounting Matters. A Central Bank Perspective," *ECB Occasional Paper* 153, May 2014.
- SHAVELL, S. (1986), "The Judgment Proof Problem," International Review of Law and Economics 6, 45-58.
- SHAVIRO, D. (1997), Do Deficits Matter? University of Chicago Press: Chicago, IL.
- SIBERT, A. C. (2009), "What Role Should the ECB Play in Financial Market Supervision," Briefing Paper Prepared for the Committee on Economic and Monetary Affairs of the European Parliament, March 2009.
- SIMS, C. A. (1994), "A Simple Model for Study of the Determination of the Price Level and the Interaction of Monetary and Fiscal Policy," *Economic Theory* 4, 381–99.

- SIMS, C. A. (2004), "Fiscal Aspects of Central Bank Independence," in: SINN, H.-W., M. WIDGRÉN, and M. KÖTHENBÜRGER (eds.), European Monetary Integration, MIT Press: Cambridge, MA, 103–16.
 - (2005), "Limits to Inflation Targeting," in: BERNANKE, B. S. and M. WOODFORD (eds.), *The Inflation-Targeting Debate*, University of Chicago Press: Chicago, IL, 283–309.
- SINCLAIR, P. and S. MILTON (2011), "Central Banks' Capital. An Introduction," in: MILTON, S. and P. SINCLAIR (eds.), *The Capital Needs of Central Banks*, Routledge: New York City, NY, 1–32.
- SINGH, M. (2013), "The Changing Collateral Space," IMF Working Paper 13/25, January 2013.

SINN, H.-W. (1980), Ökonomische Entscheidungen bei Ungewissheit, Mohr Siebeck: Tübingen.

- (1983), Economic Decisions Under Uncertainty, North Holland: Amsterdam.
- (1997), "The Selection Principle and Market Failure in Systems Competition," *Journal* of *Public Economics* 66, 247–74.
- (2000), "Why a Funded Pension System Is Useful and Why It Is not Useful," International Tax and Public Finance 7, 389–410.
- (2008), Risk-Taking, Limited Liability, and the Banking Crisis, Ifo Institute: München.
- (2010a), Casino Capitalism. How the Financial Crisis Came About and What Needs to Be Done Now, Oxford University Press: Oxford.
- (2010b), "Rescuing Europe," CESifo Forum 11, 1–22.
- (2012), *Die Target-Falle*, Carl Hanser: München.
- (2014a), "Responsibilities of States and Central Banks in the Euro Crisis," *CESifo* Forum 15, 3–36.
- (2014b), *The Euro Trap. On Bursting Bubbles, Budgets, and Beliefs*, Oxford University Press: Oxford.
- (2015a), Der Euro. Von der Friedensidee zum Zankapfel, Carl Hanser: München.
- (2015b), "The Greek Tragedy," *CESifo Forum* 16, 5–35.
- SINN, H.-W. and H. FEIST (1997), "Eurowinners and Eurolosers. The Distribution of Seignorage Wealth in EMU," *European Journal of Political Economy* 13, 665–89.
- —— (2000), "Seignorage Wealth in the Eurosystem. Eurowinners and Eurolosers Revisited," *CESifo Working Paper* 353, November 2000.
- SINN, H.-W. and T. WOLLMERSHÄUSER (2012), "Target Loans, Current Account Balances and Capital Flows. The ECB's Rescue Facility," *International Tax and Public Finance* 19, 468– 508.
- SKRETA, V. and L. VELDKAMP (2009), "Ratings Shopping and Asset Complexity. A Theory of Ratings Inflation," *Journal of Monetary Economics* 56, 678–95.
- SPAVENTA, L. (1987), "The Growth of Public Debt. Sustainability, Fiscal Rules, and Monetary Rules," *IMF Staff Papers* 34, 374–99.
- STANDARD AND POOR'S (2012), "Standard and Poor's Ratings Definitions," *Global Credit* Portal Ratings Direct, 22 June 2012.
- STELLA, P. (1997), "Do Central Banks Need Capital?" IMF Working Paper 97/83, July 1997.
- (2003), "Why Central Banks Need Financial Strength," *Central Banking Journal* 14, 23–9.

- STELLA, P. (2005), "Central Bank Financial Strength, Transparency, and Policy Credibility," IMF Staff Papers 52, 335–65.
 - (2008), "Central Bank Financial Strength, Policy Constraints and Inflation," *IMF Working Paper* 08/49, February 2008.
 - (2011), "Central Bank Financial Strength and Macroeconomic Policy Performance," in: MILTON, S. and P. SINCLAIR (eds.), *The Capital Needs of Central Banks*, Routledge: New York City, NY, 47–68.
- STIGLITZ, J. E. and A. WEISS (1981), "Credit Rationing in Markets with Imperfect Information," American Economic Review 71, 393–410.
- STOLPER, A. (2009), "Regulation of Credit Rating Agencies," Journal of Banking and Finance 33, 1266-73.
- STONE, M., K. FUJITA, and K. ISHI (2011), "Should Unconventional Balance Sheet Policies be Added to the Central Bank Toolkit? A Review of the Experience So Far," *IMF Working Paper* 11/145, June 2011.
- SZCZERBOWICZ, U. (2014), "The ECB's Unconventional Monetary Policies. Have They Lowered Market Borrowing Costs for Banks and Governments?" *RIETI Discussion Paper* 14-E-008, January 2014.
- TABAKIS, E. and B. WELLER (2009), "Collateral and Risk Mitigation Frameworks of Central Bank Policy Operations. A Comparison across Central Banks," in: BINDSEIL, U., F. GONZÁLEZ, and E. TABAKIS (eds.), Risk Management for Central Banks and Other Public Investors, Cambridge University Press: Cambridge, 340-58.
- TAMURA, K. and E. TABAKIS (2013), "The Use of Credit Claims as Collateral for Eurosystem Credit Operations," *ECB Occasional Paper* 148, June 2013.
- TANNER, E. (2013), "Fiscal Sustainability. A 21st Century Guide for the Perplexed," IMF Working Paper 13/89, April 2013.
- TESAR, L. L. and I. M. WERNER (1995), "Home Bias and High Turnover," Journal of International Money and Finance 4, 467–92.
- THORNTON, H. (1802), An Enquiry into the Nature and Effects of the Paper Credit of Great Britain, Allen & Unwin: London.
- TREBESCH, C. and J. ZETTELMEYER (2014), "ECB Interventions in Distressed Sovereign Debt Markets. The Case of Greek Bonds," *CESifo Working Paper* 4731, March 2014.
- TREHAN, B. and C. E. WALSH (1991), "Testing Intertemporal Budget Constraints. Theory and Applications to U.S. Federal Budget and Current Account Deficits," *Journal of Money*, *Credit and Banking* 23, 206–23.
- TRICHET, J.-C. (2010), "State of the Union. The Financial Crisis and the ECB's Response between 2007 and 2009," *Journal of Common Market Studies* 48, 7–19.
- TSESMELIDAKIS, Z. and R. C. MERTON (2012), "The Value of Implicit Guarantees," mimeo.
- TUCKER, P. (2009), "The Repertoire of Official Sector Interventions in the Financial System. Last Resort Lending, Market-Making, and Capital," *Remarks at the Bank of Japan International Conference*, 27-28 May 2009.
- UCTUM, M. and M. WICKENS (2000), "Debt and Deficit Ceilings, and Sustainability of Fiscal Policies. An Intertemporal Analysis," Oxford Bulletin of Economics and Statistics 62, 197–222.
- UEDA, K. and B. WEDER DI MAURO (2012), "Quantifying Structural Subsidy Values for Systemically Important Financial Institutions," *IMF Working Paper* 12/128, May 2012.

- UNITED NATIONS (2009), System of National Accounts 2008, United Nations: New York City, NY.
- VAEZ-ZADEH, R. (1991), "Implications and Remedies of Central Bank Losses," in: DOWNES, P. and R. VAEZ-ZADEH (eds.), *The Evolving Role of Central Banks*, International Monetary Fund: Washington, DC, 69–92.
- VAN BEKKUM, S., M. GABARRO, and R. M. IRANI (2015), "Does a Larger Menu Increase Appetite? Collateral Eligibility and Bank Risk-Taking," *mimeo*.
- VAN EWIJK, C. (1992), "The Distribution of Seignorage. A Note on Klein and Neumann," Weltwirtschaftliches Archiv 128, 346-51.
- VAN RIET, A. (2014), "Financial Repression to Ease Fiscal Stress. Turning Back the Clock in the Eurozone?" *mimeo*.
- VERGOTE, O., W. STUDENER, I. EFTHYMIADIS, and N. MERRIMAN (2010), "Main Drivers of the ECB Financial Accounts and ECB Financial Strength Over the First 11 Years," ECB Occasional Paper 111, May 2010.
- VON HAYEK, F. A. (1976), Denationalisation of Money, Institute of Economic Affairs: London.
- VON WEIZSAECKER, C. C. (1962), Wachstum, Zins und optimale Investitionsquote, Kyklos: Basle.
- WALSH, C. E. (1995), "Optimal Contracts for Central Bankers," American Economic Review 85, 150–67.
 - (2010), Monetary Theory and Policy, MIT Press: Cambridge, MA.
- WEBER, C. (2015), "Collateral Criteria as a Monetary Policy Device," Presentation at the CES Doctoral Seminar at Loas, June 2015.
- WEIL, P. (1987), "Permanent Budget Deficits and Inflation," Journal of Monetary Economics 20, 393-410.
- WHELAN, K. (2014a), "TARGET2 and Central Bank Balance Sheets," *Economic Policy* 77, 79–137.
- (2014b), "The ECB's Collateral Policy and Its Future as Lender of Last Resort," *European Parliament Policy Department*, November 2014.
- WHITE, L. J. (2010), "The Credit Rating Agencies," Journal of Economic Perspectives 24, 211–26.
- WHITTAKER, J. (2011), "Eurosystem Debts, Greece, and the Role of Banknotes," Lancaster University Monetary Research, November 2011.
- WILCOX, D. W. (1989), "The Sustainability of Government Deficits. Implications of the Present-Value Borrowing Constraint," Journal of Money, Credit and Banking 21, 291–306.
- WOLFF, G. B. (2014), "Eurosystem Collateral Policy and Framework. Was it Unduly Changed?" Bruegel Policy Contribution 2014/14, November 2014.
- WOODFORD, M. (1995), "Price-Level Determinacy without Control of a Monetary Aggregate," Carnegie-Rochester Conference Series on Public Policy 43, 1–46.
- WORLDBANK AND INTERNATIONAL MONETARY FUND (2014), Revised Guidelines for Public Debt Management, World Bank: Washington, DC.
- XU, D. and P. GHEZZI (2003), "Fundamentals to Spreads. A Fair Spread Model for High Yield Emerging Markets Sovereigns," *Deutsche Bank Global Markets Research*, July 2003.

Curriculum Vitae

seit 04/2011	 Munich Graduate School of Economics an der LMU München Abschluss: Dr. oec. publ. Betreuer: Prof. Dr. Dr. h.c. mult. Hans-Werner Sinn
02/2015 - 06/2015	Massachusetts Institute of Technology (MIT), Cambridge, USAVisiting PhD Student
10/2006 – 02/2011	 LMU München Abschluss: Diplom-Volkswirt Schwerpunkte: "Finanzwissenschaft" und "Empirische Wirtschaftsforschung" Thema der Diplomarbeit: "Der Einfluss von Marktmacht auf die Extraktion von Ressourcen unterschiedlicher Qualität"
02/2009 - 09/2009	 Københavns Universitet, Kopenhagen, Dänemark Studium der Volkswirtschaftslehre Schwerpunkte: "Geldtheorie" und "Spieltheorie"
09/1996 - 06/2005	Gymnasium PuchheimAbschluss: Abitur
24.10.1985	geboren in München